

VU Research Portal

Patient-centered care: facilitating home dialysis and improving health-related quality of life

Bonenkamp, Anna Amarentia

2022

document version Publisher's PDF, also known as Version of record

Link to publication in VU Research Portal

citation for published version (APA)

Bonenkamp, A. A. (2022). Patient-centered care: facilitating home dialysis and improving health-related quality of life. Ridderprint.

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal ?

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

E-mail address: vuresearchportal.ub@vu.nl Patient-centered care: facilitating home dialysis and improving health-related quality of life

Anna A. Bonenkamp

Patient-centered care: facilitating home dialysis and improving health-related quality of life

Anna Amarentia Bonenkamp

Patient-centered care: facilitating home dialysis and improving health-related quality of life

Copyright 2022 © Anna Amarentia Bonenkamp

The Netherlands. All rights reserved. No parts of this thesis may be reproduced, stored in a retrieval system or transmitted in any form or by any means without permission of the author.

The printing of this thesis was kindly supported by Jeroen Bosch Ziekenhuis and Chipsoft.

Provided by thesis specialist Ridderprint, ridderprint.nl Printing: Ridderprint Layout and design: Joppe Klein, persoonlijkproefschrift.nl Cover design: Anna Sieben, Sieben Medical Art

ISBN: 978-94-6458-124-9

VRIJE UNIVERSITEIT

PATIENT-CENTERED CARE: FACILITATING HOME DIALYSIS AND IMPROVING HEALTH-RELATED QUALITY OF LIFE

ACADEMISCH PROEFSCHRIFT

ter verkrijging van de graad Doctor aan de Vrije Universiteit Amsterdam, op gezag van de rector magnificus prof.dr. J.J.G. Geurts, in het openbaar te verdedigen ten overstaan van de promotiecommissie van de Faculteit der Geneeskunde op woensdag 15 juni 2022 om 13.45 uur in een bijeenkomst van de universiteit, De Boelelaan 1105

door

Anna Amarentia Bonenkamp

geboren te Utrecht

Promotoren

Prof. dr. F.J. van Ittersum Prof. dr. F.W. Dekker

Co-promotoren

Dr. B.C. van Jaarsveld Dr. A.C. Abrahams

The work within this thesis (DOMESTICO) was financially supported by the Nierstichting and The Netherlands Organisation for Health Research and Development (ZonMw), in addition to Fresenius Medical Care, Baxter, Dirinco, Vifor Pharma and AstraZeneca.

Beoordelingscommissie

prof. dr. A.M. van Dulmen prof. dr. M.H. Emmelot-Vonk dr. M.P.C. Grooteman prof. dr. M.H. Hemmelder dr. A. Özyilmaz dr. C.B. Terwee



The secret of the care of the patient is in caring for the patient - Dr. Francis W. Peabody

Voor mama

Table of contents

Chapter 1	General introduction	11
Part I	Eligibility for home dialysis in the current dialysis population	
Chapter 2	Trends in home dialysis use differ among age categories in past two decades a Dutch registry study <i>European Journal of Clinical Investigation.</i> 2021 Jul 22e13656.	29
Chapter 3	Comorbidity is not associated with dialysis modality choice <i>Nephrology</i> . 2022. <i>in print</i>	53
Chapter 4	Key elements in selection of pre-dialysis patients for home dialysis <i>Peritoneal Dialysis International.</i> 2021 <i>Sep</i> ;41(5)494-501.	79
Part II	Enhancing technique survival of peritoneal dialysis	
Chapter 5	Differences in hospitalization between peritoneal dialysis and hemodialysis patients European Journal of Clinical Investigation. 2022 Feb 7e13758. (Epub ahead of print)	105
Chapter 6	Modifiable causes of early and late technique failure in peritoneal dialysis <i>Peritoneal Dialysis International.</i> 2022. <i>In print</i>	123
Part III	Shift towards Health-Related Quality of Life	
Chapter 7	Health-Related Quality of Life in Home Dialysis Patients Compared to In-Center Hemodialysis Patients A Systematic Review and Meta-analysis <i>Kidney Medicine 2020 Feb 11;2(2)139-154</i> .	153
Chapter 8	The impact of COVID-19 on the mental health of dialysis patients <i>Journal of Nephrology. 2021 Apr;34(2)337-344.</i>	197
Chapter 9	Health-Related Quality of Life compared between kidney transplantation and nocturnal hemodialysis <i>PLoS One. 2018 Sep 20;13(9)e0204405.</i>	213

Part IV	Discussion and future perspectives	
Chapter 10	Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO) rationale and design Investigating the impact of home dialysis therapies on quality of life clinical outcomes and costs in the Netherlands <i>BMC Nephrology 2019 Sep 18;20(1)361.</i>	235
Chapter 11	Summary and general discussion	253
Appendices		275
	Nederlandse samenvatting	2/3
	Appendix A	283
	Appendix B	285
	DOMESTICO Study Group members	292
	List of publications	291
	Dankwoord	294
	About the author	297



Chapter 1 General introduction

Paragraph 5 of this introduction is in part adapted from: Anna A. Bonenkamp, Maaike K. van Gelder, Alferso C. Abrahams, *et al.* Home haemodialysis in the Netherlands: State of the art. *The Netherlands Journal of Medicine 2018;76(4):144-57.*

General introduction

The number of patients with chronic kidney disease (CKD) and end stage kidney disease (ESKD) is continuing to rise globally.^{1,2} This worldwide growth in the number of patients is caused by both a rise in the prevalence *and* the incidence.² In developed countries, the prevalence is increasing as a result of enhanced dialysis survival^{2, 3}, while in undeveloped countries a remarkably growing number of patients suffer from CKD.² The increase in incidence has two main causes: first, the population is ageing and second, the number of patients suffering from diabetes mellitus and cardiovascular diseases - important risk factors for developing CKD - is growing.^{1, 4} Furthermore, the disease-specific survival of patients with cardiac diseases and diabetes mellitus has increased. Therefore, patients have a higher life-time risk of developing CKD and ESKD in particular.^{5, 6} Overall, the continuing growth in CKD and ESKD patients is a major global problem.



Figure 1. Prevalence of kidney replacement therapy by age categories adapted from RENINE annual report 2020. Available at: https://www.nefrovisie.nl/jaarrapportages/ Accessed: 23-02-2022.

In the Netherlands, the patient population being treated for ESKD has steadily increased over the past 15 years from 11,730 in 2005 to 18,071 in 2020. In the age

category over 65 years, a marked rise in prevalent patients is seen (Figure 1).⁷ This is a result of a high age-specific incidence, i.e. the incidence in a specified age category divided by the general population in that age category. Compared to the age-specific incidence of patients <65 years, the age-specific incidence of elderly patients was 4-5 fold higher in the past two decades.⁸ These findings show that with advanced age, comorbidities such as ESKD accumulate. Not only ESKD, but also other lifestyle diseases accumulate with age. The increase in older patients with diabetes mellitus is particularly striking.⁹ Thus, the current population of patients with ESKD has changed; patients are older, have more comorbidities and will consequently be more frail. This developments, resulting in the growth of a vulnerable patient group, demand adjustment of our healthcare organization.

Home dialysis

ESKD is preferably treated with a kidney transplant, since it is associated with the highest survival.¹⁰ However, as kidney donors are scarce, dialysis therapy remains a cornerstone in the management of ESKD. Dialysis may be performed in a dialysis centre, on average 3 times weekly for a 4 hour session, or at home. Home dialysis provides several advantages, including more independence, no travel time to the hospital and greater scheduling flexibility. Yet, the majority of patients is treated with in-centre haemodialysis (CHD).¹¹ At the end of 2020, there were 4,990 patients on CHD in the Netherlands, accounting for 80% of all dialysis patients.¹²

There are two home-based therapies to choose from, peritoneal dialysis (PD) and home haemodialysis (HHD). In PD, dialysis fluid enters the body through a permanent catheter in the abdominal cavity. Waste products in the peritoneal capillaries of the patient diffuse through the peritoneal membrane to the dialysis fluid. Excess body water is removed by osmotic ultrafiltration. After hours of dwelling time, the fluid filled with waste products and excess body water is drained into a waste bag and the cycle is repeated again. Home haemodialysis is essentially the same therapy as in-centre haemodialysis, yet only the location where the treatment is performed is different. In haemodialysis, the patient's blood exits the body through a vascular access into a dialyzer that consists of a synthetic semi-permeable membrane. In the dialyzer, waste products diffuse through this membrane to the dialysis fluid. In addition, excess body water is removed by hydrostatic ultrafiltration. Home haemodialysis was the first home dialysis therapy to be introduced in 1961 by the Japanese doctor Yukihiko Nosé.¹³ The reason to perform haemodialysis at home was primarily a practical consideration, since a proportion of the patients with ESKD could not be offered CHD due to capacity problems. The HHD incidence reached its global peak in 1970s, but soon decreased due to the introduction of PD, the expansion of CHD facilities, the increase of kidney transplantations and eventually, lack of experience of nephrologists.¹⁴ Nowadays, only a small portion of home dialysis patients are treated with HHD. In the Netherlands, a total of 270 patients is treated with HHD, compared to 1,001 with PD.¹² The low use of HHD compared to PD can be explained by the complexity of the dialysis procedure, because patients are responsible for the dialysis machine set-up, self-cannulation and trouble-shooting. Nevertheless, HHD provides the perfect opportunity for prolonged or more frequent HD sessions. These intensive HD sessions are associated with improved survival, quality of life, blood pressure control and phosphate control.¹⁵⁻¹⁷

Peritoneal dialysis is the leading home-based dialysis therapy worldwide, but its use around the world is quite variable. Different governmental policies combined with financial incentives possibly play a role.¹⁸ In Hong Kong, where a 'PD first' policy is adopted, >70% of dialysis patients are on PD compared to $\leq 20\%$ in Australia, Canada and most European countries.¹⁸ Remarkable is the contrast in recent developments: in Asia and the United States of America an increase in the number of PD patients is seen, whereas the number of PD patients has declined in several European countries, including the Netherlands.¹⁸ For an overview of developments in European countries in recent years, see Appendix A. The use of PD is also quite variable among different regions within countries.^{19, 20} In the Netherlands, the proportion of patients on home dialysis varies between 0 to 40% as shown in Figure 2.⁸ Overall, from the perspective that majority of patients choose home dialysis after extensive treatment education^{20, ²¹, the discrepancies among countries and within countries suggest that there is room for improvement in the uptake of home dialysis as a whole and PD in particular.}



Figure 2. Funnel plot showing centre variation in percentage home dialysis at three months after start dialysis. Home dialysis includes peritoneal dialysis and home haemodialysis. Data is adjusted for age, sex, socioeconomic status, and primary kidney disease categories. Adapted from RENINE annual report 2020. Available at: https://www.nefrovisie.nl/jaarrapportages/Accessed: 23-02-2022.

Multiple reasons for the decline in PD have been suggested. The decreasing number of *incident* PD patients is regarded the main problem ²², and has been attributed to multiple factors including the increasing number of patients that obtain a kidney transplant and the expansion of CHD facilities.^{7,22,23} The latter occurred in the early 2000s as a result of governmental decision to allow dialysis in standalone centres rather than in hospitals only, initiated by an enormous shortage of CHD capacity leaving patients no choice but to opt for PD treatment.^{7,24} Other frequently mentioned reasons are the ageing of the dialysis population and the increasing number of diabetic patients starting dialysis.²² The loss of *prevalent* PD patients due to a high rate of PD technique failure is considered another cause for the low number of PD patients. The reported incidence of PD technique failure within the first year ranges from 12.7% to 26.2%.²⁵ Identifying modifiable causes of technique failure could reverse the decline in the number of PD patients. These developments in both incidence and prevalence create a vicious circle: a decrease in the number of incident PD patients contributes to loss of experience of young nephrologists, which may increase technique failure rate and further prevent new uptake in PD.26,27

Chapter 1

Apart from more independence and flexibility, another important advantage is that home dialysis may be associated with lower costs.²⁸⁻³² Dialysis is a cost-consuming treatment. In the Netherlands, 0.1% of inhabitants have chronic kidney disease, accounting for 0.9% of the total healthcare spending in 2017.³³ Nearly half of these costs are spend on dialysis patients, representing a quarter of the total patients with chronic kidney disease.³⁴ Costs for CHD are the highest of all diseases within internal medicine, estimated at > 300 million. As an extreme example: average annual costs for a CHD patient are \in 50,000, while the costs for over 7000 diabetes mellitus patients with secondary complications are 43 million in total (on average €600 per patient annually).³⁴ Of course, these calculations are based upon healthcare claim costs from diagnosis-related group codes, not total costs per individual patient. For example, a study with Vektis data, which included costs of hospital admissions, medication and indirect non-medical costs such as transportation, showed that the costs for a dialysis patient are in reality between €77,566 and €105,833 per year.²⁸ The lowest costs were found for patients on continuous ambulatory peritoneal dialysis patients, a home dialysis treatment. Also in other previous economic evaluation studies home dialysis, including home hemodialysis, tended to be more cost-effective than CHD.²⁹⁻³² Home dialysis might thus be a sustainable option in a growing population of patients.

Health-Related Quality of Life

Despite major advances in kidney disease and kidney disease-related mortality ³, the survival of patients with ESKD is still considerably lower compared to patients without kidney disease. The unadjusted 5-year survival rate for patients on dialysis is 42.6%.³⁵ For a 45-year old patient starting dialysis, the remaining life expectancy is 10 years. For comparison, a 45-year old patient without kidney disease has an average remaining life expectancy of 35 years.³⁵ According to literature, there are no major differences in survival between home dialysis modalities (both PD and HHD) and CHD.^{22, 36} Since a large European observational cohort study found that survival of patients starting on CHD or PD was similar (adjusted HR 0.91, 95% CI 0.88 – 0.95; PD vs CHD), mortality is no reason to favour CHD to a home dialysis modality.²²

Due to the low survival of dialysis patients, it is important to make life with the burden of ESKD worthwhile. A life with dialysis is characterized by polypharmacy and frequent hospitalizations resulting in morbidity, and often a life-long dialysis dependency that requires a certain structure to everyday life.^{37, 38} Thus, advantages and disadvantages of dialysis treatment modalities should be weighed for each individual

patient. A reason to favour home dialysis therapy is that it can offer more flexibility and autonomy than CHD. These features of home dialysis can likely contribute to less burden and improved quality of life for some patients.

The most well-known definition of quality of life is that from the World Health Organisation: '*individual's perception of their position in life in the context of the culture and value systems in which they live and in relation to their goals, expectations, standards and concerns*'.³⁹ Health-Related Quality of Life (HRQoL) thus not only consists of the physical health status affected by the disease, but rather of every aspect of life that is important to the patient. The key term 'individual perceptions' further stresses that differences are present in each individual's HRQoL. This emphasis on the individual is essential in understanding the concept of HRQoL. Means to objectively measure HRQoL are the use of self-reported questionnaires, HRQoL is thus a patient-reported outcome.

HRQoL is known to be low in patients with ESKD due to the marked burden of kidney disease; patients with ESKD on dialysis have a lower HRQoL compared to patients with chronic conditions such as diabetes mellitus and even malignancies.^{40, 41} In recent years, HRQoL has received quite more attention, along with the interest for patient-centred care and individualized medicine. This has led to a paradigm shift from clinical outcomes - important to compare patient groups - to patient-reported outcomes – more important to the individual patient. In patient focus groups of both CHD and PD patients (including SONG-HD, SONG-PD and ICHOM), outcomes related to HRQoL - life participation, symptom burden and fatigue – were selected as core outcomes along some clinical outcomes – mortality, cardiovascular disease, technique survival.⁴²⁻⁴⁵ These developments underscore the importance of improving HRQoL of dialysis patients.

The most commonly used generic questionnaires to measure HRQoL are the Short Form questionnaires.^{46, 47} The advantage of these questionnaires is that they allow for comparison among different patient groups and the general population. A disadvantage is that they do not specifically map the burden of kidney disease and, therefore, important elements of HRQoL are missed. Therefore, also kidney disease specific HRQoL questionnaires exist, such as the Kidney Disease Quality of Life Instrument.⁴⁸ Disease burden may also be evaluated with a symptom questionnaire, for example with the Dialysis Symptom Index (DSI).⁴⁹. It is conceptually assumed that a high symptom burden causes a decline in functional status and in turn lower HRQoL.⁵⁰ (For an example of a generic and kidney disease specific symptom questionnaire, respectively Short Form 12 and Dialysis Symptom Index, see Appendix B.)

Although validated questionnaires exist and HRQoL is recognized as a core outcome, HRQoL is not frequently measured in daily practice.^{51, 52} Accurate and frequent assessments can help to identify and address a high symptom burden⁵², as well as facilitate patient-physician communication about HRQoL.⁵³ Previous studies showed that due to underrecognition *and* underreport of symptoms, such as depression and itch, patients often remained untreated.^{54, 55} Additionally, early interventions such as re-education, home care worker support, dialysis schedules adjustments, and psychotherapy can help to enhance treatment modality satisfaction and treatment longevity.

DOMESTICO

The abovementioned poor quality of life and marked practice variation in proportion of home dialysis patients among centres were reasons to start the Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO) in 2017. DOMESTICO consists of two multi-centre cohort studies and a implementation project (Figure 3). The retrospective multi-centre cohort study enrolled patients that started dialysis treatment between 1 January 2012 and 31 December 2016 from 37 Dutch dialysis centres, representing two thirds of all dialysis centres in the Netherlands. For each included home dialysis or nocturnal dialysis patient, one CHD patient was randomly selected. The aim of the retrospective DOMESTICO study was to evaluate modifiable factors in technique failure, and to compare outcomes between home dialysis and in-centre dialysis patients. In the prospective study, that started in December 2017, a total of 1600 patients starting dialysis will be followed for at least one year. A total of 56 dialysis centres participate in this study, all but one centres in the Netherlands and 2 from Belgium. The aim of this nationwide study is to compare quality of life, total costs and clinical outcomes between home dialysis and CHD patients. The implementation project aims to provide optimal education for all dialysis modalities, focusing on pre-dialysis education and good practices in dialysis care.

All original data presented in this manuscript originate from the Dutch kidney patient population, predominantly from the DOMESTICO studies. The Dutch patient population is unique in terms of kidney transplantation and dialysis access. The Netherlands consistently ranks amongst the countries with the highest rates of (living) donor kidney transplantations worldwide, has a steady home haemodialysis rate and due to its small surface area, small distances of patient's residence to the hospital.^{2, 11, 56}



Thesis outline

This thesis will contribute to the research surrounding home dialysis and HRQoL. In a growing and ageing patient population, home dialysis should obviously be considered as a sustainable kidney replacement therapy due to its possible cost-effectiveness. Therefore, identifying all patients eligible for home dialysis is crucial. From the same perspective enhancing technique survival should be an important goal. Because the dialysis population is characterized by low survival and high disease burden, HRQoL is the most important outcome, especially in an era with attention for patient-centred care. This thesis thus aims to investigate the following questions in dialysis patients:

- Which patient characteristics identify eligibility for home dialysis in the current dialysis population?
- What are modifiable causes of hospital admissions and technique failure (in PD patients)?
- Does treatment modality influence Health-Related Quality of Life?

The current dialysis population consists of more elderly patients then two decades ago. Therefore, time trends in the use of home dialysis for different age categories over the past 20 years are explored in **chapter 2**. Severe comorbidity is frequently perceived a contra-indication to receive home dialysis. **Chapter 3** examines the association between comorbidity and dialysis modality choice if corrected for age, BMI and differences between centres. Pre-dialysis education and programmes can help to aid patients in dialysis modality choice, and may also help the multidisciplinary team surrounding the patient. **Chapter 4** describes the findings from such a multidisciplinary pre-dialysis programme.

The second part of this thesis then focuses on PD technique survival. Chapter 5 describes the hospitalization rate of PD patients compared to patients that receive HD with a multistate model. In addition, modifiable causes of hospitalization are sought that could contribute to the reduction of hospital admissions. Chapter 6 then focuses on modifiable causes and risk factors of technique failure. Identifying modifiable causes and risk factors may provide important information to enhance technique survival of PD patients.

Part 3 concentrates on HRQoL, an important patient-reported outcome. Chapter 7 systematically reviews the association between home dialysis and HRQoL of dialysis patients worldwide. Chapter 8 examines the effect of a major pandemic, the COVID-19 pandemic, on the mental health of dialysis patients. Chapter 9 describes the HRQoL of patients treated with a specific home based therapy, nocturnal home hemodialysis, in comparison to patients that obtained a kidney transplant.

Finally, the findings and relevance of this thesis are discussed in the **Discussion** and future perspectives. This part also contains the protocol of the prospective DOMESTICO study (chapter 10). The developments presented in this introduction led to the aims of this nationwide study, to investigate the HRQoL and cost-effectiveness of home dialysis in the current ageing dialysis population.

References

- 1. Xie Y, Bowe B, Mokdad AH, Xian H, Yan Y, Li T, *et al.* Analysis of the Global Burden of Disease study highlights the global, regional, and national trends of chronic kidney disease epidemiology from 1990 to 2016. Kidney Int. 2018;94(3):567-81.
- Robinson BM, Akizawa T, Jager KJ, Kerr PG, Saran R, Pisoni RL. Factors affecting outcomes in patients reaching end-stage kidney disease worldwide: differences in access to renal replacement therapy, modality use, and haemodialysis practices. Lancet (London, England). 2016;388(10041):294-306.
- 3. Boenink R, Stel VS, Waldum-Grevbo BE, Collart F, Kerschbaum J, Heaf JG, *et al.* Data from the ERA-EDTA Registry were examined for trends in excess mortality in European adults on kidney replacement therapy. Kidney Int. 2020.
- 4. Brown EA, Johansson L. Dialysis options for end-stage renal disease in older people. Nephron Clin Pract. 2011;119 Suppl 1:c10-3.
- Deeg DJH, van Vliet MJG, Kardaun JWPF, Huisman M. Understanding the mortality decline at older ages: improved life course or improved present period? Annual Rev Gerontol Geriatr. 2013;33:259–91.
- Liyanage T, Ninomiya T, Jha V, Neal B, Patrice HM, Okpechi I, *et al.* Worldwide access to treatment for end-stage kidney disease: a systematic review. The Lancet. 2015;385(9981):1975-82.
- Hemke AC, Dekker FW, Bos WJ, Krediet RT, Heemskerk MB, Hoitsma AJ. [Causes of decreased use of peritoneal dialysis as a kidney replacement therapy in the Netherlands]. Ned Tijdschr Geneeskd. 2012;156(21).
- Hoekstra T, Dekker FW, Cransberg K, Bos WJ, van Buren M, Hemmelder MH. RENINE annual report 2018. Available at: https://www.nefrovisie.nl/jaarrapportages/ Accessed: 04-01-2021. 2018.
- 9. Ceretta ML, Noordzij M, Luxardo R, De Meester J, Diez JMA, Finne P, *et al.* Changes in co-morbidity pattern in patients starting renal replacement therapy in Europe data from the ERA-EDTA Registry. Nephrology Dialysis Transplantation. 2018;33(10):1794-804.
- Wolfe RA, Ashby VB, Milford EL, Ojo AO, Ettenger RE, Agodoa LY, *et al.* Comparison of mortality in all patients on dialysis, patients on dialysis awaiting transplantation, and recipients of a first cadaveric transplant. The New England journal of medicine. 1999;341(23):1725-30.
- ERA-EDTA Registry: ERA-EDTA Registry Annual Report 2018. Amsterdam UMC, location AMC, Department of Medical Informatics, Amsterdam, the Netherlands, 2020. 2018.
- 12. Dutch Renal Registry (RENINE) [Internet]. Available at: https://www.nefrovisie.nl/ nefrodata/ Accessed: 03-07-2021. 2021.
- 13. Nosé Y. Home hemodialysis: a crazy idea in 1963: a memoir. ASAIO journal (American Society for Artificial Internal Organs : 1992). 2000;46(1):13-7.
- 14. Agar JW. International variations and trends in home hemodialysis. Adv Chronic Kidney Dis. 2009;16(3):205-14.
- 15. Mathew A, McLeggon J-A, Mehta N, Leung S, Barta V, McGinn T, *et al.* Mortality and Hospitalizations in Intensive Dialysis: A Systematic Review and Meta-Analysis. Canadian journal of kidney health and disease. 2018;5:2054358117749531-.

- 16. Rocco MV, Lockridge RS, Jr., Beck GJ, Eggers PW, Gassman JJ, Greene T, *et al.* The effects of frequent nocturnal home hemodialysis: the Frequent Hemodialysis Network Nocturnal Trial. Kidney Int. 2011;80(10):1080-91.
- 17. Chertow GM, Levin NW, Beck GJ, Depner TA, Eggers PW, Gassman JJ, *et al.* In-center hemodialysis six times per week versus three times per week. The New England journal of medicine. 2010;363(24):2287-300.
- Li PK-T, Chow KM, Van de Luijtgaarden MWM, Johnson DW, Jager KJ, Mehrotra R, et al. Changes in the worldwide epidemiology of peritoneal dialysis. Nature Reviews Nephrology. 2017;13(2):90-103.
- 19. Bouvier N, Durand PY, Testa A, Albert C, Planquois V, Ryckelynck JP, *et al.* Regional discrepancies in peritoneal dialysis utilization in France: the role of the nephrologist's opinion about peritoneal dialysis. Nephrol Dial Transplant. 2009;24(4):1293-7.
- 20. Oliver MJ, Quinn RR, Richardson EP, Kiss AJ, Lamping DL, Manns BJ. Home care assistance and the utilization of peritoneal dialysis. Kidney Int. 2007;71(7):673-8.
- Shukla AM, Easom A, Singh M, Pandey R, Rotaru D, Wen X, et al. Effects of a Comprehensive Predialysis Education Program on the Home Dialysis Therapies: A Retrospective Cohort Study. Perit Dial Int. 2017;37(5):542-7.
- 22. van de Luijtgaarden MW, Jager KJ, Segelmark M, Pascual J, Collart F, Hemke AC, *et al.* Trends in dialysis modality choice and related patient survival in the ERA-EDTA Registry over a 20-year period. Nephrol Dial Transplant. 2016;31(1):120-8.
- 23. Blake P. Proliferation of Hemodialysis Units and Declining Peritoneal Dialysis Use: An International Trend. American Journal of Kidney Diseases. 2009;54(2):194-6.
- 24. Ministerie van Volksgezondheid, Welzijn en Sport. Staatscourant van het Koninkrijk der Nederlanden, Staatscourant 1999, 126 page 5 (06-07-1999) Available at: https://zoek. officielebekendmakingen.nl/stcrt-1999-126-p5-SC19556.html. Accessed: 30-07-2021.
- 25. Cho Y, See EJ, Htay H, Hawley CM, Johnson DW. Early Peritoneal Dialysis Technique Failure: Review. Perit Dial Int. 2018;38(5):319-27.
- 26. Biesen Wv, Veys N, Lameire N, Vanholder R. Why less success of the peritoneal dialysis programmes in Europe? Nephrology Dialysis Transplantation. 2008;23(5):1478-81.
- Htay H, Cho Y, Pascoe EM, Darssan D, Nadeau-Fredette AC, Hawley C, *et al.* Multicenter Registry Analysis of Center Characteristics Associated with Technique Failure in Patients on Incident Peritoneal Dialysis. Clin J Am Soc Nephrol. 2017;12(7):1090-9.
- 28. Mohnen SM, van Oosten MJM, Los J, Leegte MJH, Jager KJ, Hemmelder MH, *et al.* Healthcare costs of patients on different renal replacement modalities - Analysis of Dutch health insurance claims data. PLoS One. 2019;14(8):e0220800.
- 29. Klarenbach SW, Tonelli M, Chui B, Manns BJ. Economic evaluation of dialysis therapies. Nat Rev Nephrol. 2014;10(11):644-52.
- Klarenbach S, Tonelli M, Pauly R, Walsh M, Culleton B, So H, *et al.* Economic evaluation of frequent home nocturnal hemodialysis based on a randomized controlled trial. J Am Soc Nephrol. 2014;25(3):587-94.
- Li B, Cairns JA, Fotheringham J, Tomson CR, Forsythe JL, Watson C, *et al.* Understanding cost of care for patients on renal replacement therapy: looking beyond fixed tariffs. Nephrol Dial Transplant. 2015;30(10):1726-34.

- 32. Couillerot-Peyrondet AL, Sambuc C, Sainsaulieu Y, Couchoud C, Bongiovanni-Delaroziere I. A comprehensive approach to assess the costs of renal replacement therapy for end-stage renal disease in France: the importance of age, diabetes status, and clinical events. Eur J Health Econ. 2017;18(4):459-69.
- RIVM. Diseases ordered by costs [Ranglijst ziekten op basis van zorgkosten] Bilthoven, The Netherlands: RIVM—National Institut of Public Health and the Environment [internet] https://www.volksgezondheidenzorg.info/ranglijst/ranglijst-aandoeningen-opbasis-van-zorguitgaven Accessed: 15-08-2021.
- The Dutch Healthcare Authority (NZa), Dutch Ministry of Health, Welfare and Sport (VWS). DIS open data. Utrecht, The Netherlands [internet] https://www.opendisdata.nl/ Accessed: 15-08-2021.
- Kramer A, Boenink R, Stel VS, Santiuste de Pablos C, Tomović F, Golan E, *et al.* The ERA-EDTA Registry Annual Report 2018: a summary. Clinical Kidney Journal. 2021;14(1):107-23.
- Marshall MR, Polkinghorne KR, Kerr PG, Hawley CM, Agar JWM, McDonald SP. Intensive Hemodialysis and Mortality Risk in Australian and New Zealand Populations. American Journal of Kidney Diseases. 2016;67(4):617-28.
- Daratha KB, Short RA, Corbett CF, Ring ME, Alicic R, Choka R, *et al.* Risks of subsequent hospitalization and death in patients with kidney disease. Clin J Am Soc Nephrol. 2012;7(3):409-16.
- Burnier M, Pruijm M, Wuerzner G, Santschi V. Drug adherence in chronic kidney diseases and dialysis. Nephrology Dialysis Transplantation. 2014;30(1):39-44.
- Development of the World Health Organization WHOQOL-BREF quality of life assessment. The WHOQOL Group. Psychological medicine. 1998;28(3):551-8.
- 40. Mittal SK, Ahern L, Flaster E, Maesaka JK, Fishbane S. Self-assessed physical and mental function of haemodialysis patients. Nephrol Dial Transplant. 2001;16(7):1387-94.
- 41. van Sandwijk MS, Al Arashi D, van de Hare FM, van der Torren JMR, Kersten MJ, Bijlsma JA, *et al.* Fatigue, anxiety, depression and quality of life in kidney transplant recipients, haemodialysis patients, patients with a haematological malignancy and healthy controls. Nephrol Dial Transplant. 2019;34(5):833-8.
- Manera KE, Johnson DW, Craig JC, Shen JI, Gutman T, Cho Y, *et al.* Establishing a Core Outcome Set for Peritoneal Dialysis: Report of the SONG-PD (Standardized Outcomes in Nephrology-Peritoneal Dialysis) Consensus Workshop. Am J Kidney Dis. 2020;75(3):404-12.
- 43. Evangelidis N, Tong A, Manns B, Hemmelgarn B, Wheeler DC, Tugwell P, *et al.* Developing a Set of Core Outcomes for Trials in Hemodialysis: An International Delphi Survey. American Journal of Kidney Diseases. 2017;70(4):464-75.
- 44. Manns B, Hemmelgarn B, Lillie E, Dip SC, Cyr A, Gladish M, *et al.* Setting research priorities for patients on or nearing dialysis. Clin J Am Soc Nephrol. 2014;9(10):1813-21.
- 45. Verberne WR, Das-Gupta Z, Allegretti AS, Bart HAJ, van Biesen W, García-García G, et al. Development of an International Standard Set of Value-Based Outcome Measures for Patients With Chronic Kidney Disease: A Report of the International Consortium for Health Outcomes Measurement (ICHOM) CKD Working Group. Am J Kidney Dis. 2019;73(3):372-84.

- 46. Wyld M, Morton RL, Hayen A, Howard K, Webster AC. A systematic review and meta-analysis of utility-based quality of life in chronic kidney disease treatments. PLoS medicine. 2012;9(9):e1001307.
- 47. Bonenkamp AA, van Eck van der Sluijs A, Hoekstra T, Verhaar MC, van Ittersum FJ, Abrahams AC, *et al.* Health-Related Quality of Life in Home Dialysis Patients Compared to In-Center Hemodialysis Patients: A Systematic Review and Meta-analysis. Kidney Medicine. 2020;2(2):139-54.
- 48. Hays RD, Kallich JD, Mapes DL, Coons SJ, Carter WB. Development of the kidney disease quality of life (KDQOL) instrument. Qual Life Res. 1994;3(5):329-38.
- 49. Weisbord SD, Fried LF, Arnold RM, Rotondi AJ, Fine MJ, Levenson DJ, *et al.* Development of a symptom assessment instrument for chronic hemodialysis patients: the Dialysis Symptom Index. J Pain Symptom Manage. 2004;27(3):226-40.
- 50. Wilson IB, Cleary PD. Linking Clinical Variables With Health-Related Quality of Life: A Conceptual Model of Patient Outcomes. JAMA. 1995;273(1):59-65.
- Anderson NE, Calvert M, Cockwell P, Dutton M, Kyte D. The Use of Patient-Reported Outcomes in Patients Treated With Maintenance Hemodialysis: A Perspective. Am J Kidney Dis. 2019;74(3):399-406.
- 52. van der Veer SN, Aresi G, Gair R. Incorporating patient-reported symptom assessments into routine care for people with chronic kidney disease. Clin Kidney J. 2017;10(6):783-7.
- 53. van der Willik EM, Hemmelder MH, Bart HAJ, van Ittersum FJ, Hoogendijk-van den Akker JM, Bos WJW, *et al.* Routinely measuring symptom burden and health-related quality of life in dialysis patients: first results from the Dutch registry of patient-reported outcome measures. Clinical Kidney Journal. 2020;14(6):1535-44.
- 54. Rayner HC, Larkina M, Wang M, Graham-Brown M, van der Veer SN, Ecder T, *et al.* International Comparisons of Prevalence, Awareness, and Treatment of Pruritus in People on Hemodialysis. Clin J Am Soc Nephrol. 2017;12(12):2000-7.
- 55. Weisbord SD, Fried LF, Mor MK, Resnick AL, Unruh ML, Palevsky PM, *et al.* Renal provider recognition of symptoms in patients on maintenance hemodialysis. Clin J Am Soc Nephrol. 2007;2(5):960-7.
- 56. Stel VS, Kramar R, Leivestad T, Hoitsma AJ, Metcalfe W, Smits JM, *et al.* Time trend in access to the waiting list and renal transplantation: a comparison of four European countries. Nephrol Dial Transplant. 2012;27(9):3621-31.

General introduction



Part I

Eligibility for home dialysis in the current dialysis population



Chapter 2

Trends in home dialysis use differ among age categories in past two decades: a Dutch registry study

Anna A. Bonenkamp, Tiny Hoekstra, Marc H. Hemmelder, Anita van Eck van der Sluijs , Alferso C. Abrahams, Frans J. van Ittersum, and Brigit C. van Jaarsveld

> European Journal of Clinical Investigation. 2021 Jul 22:e13656. (Epub ahead of print)

Abstract

Background: Although the number of patients with end stage kidney disease is growing, the number of patients who perform dialysis at home has decreased during the past two decades. The aim of this study was to explore time trends in the use of home dialysis in the Netherlands.

Methods: Dialysis episodes of patients who started dialysis treatment were studied using Dutch registry data (RENINE). The uptake of home dialysis between 1997 through 2016 was evaluated in time periods of 5 years. Home dialysis was defined as start with peritoneal dialysis or home haemodialysis, or transfer to either within 2 years of dialysis initiation. All analyses were stratified for age categories. Mixed model logistic regression analysis was used to adjust for clustering at patient level.

Results: A total of 33,340 dialysis episodes in 31,569 patients were evaluated. Mean age at dialysis initiation increased from 62.5 ± 14.0 to 65.5 ± 14.5 years in in-centre haemodialysis patients, whereas it increased from 51.9 ± 15.1 to 62.5 ± 14.6 years in home dialysis patients. In patients < 65 years, the uptake of home dialysis was significantly lower during each 5-year period compared to the previous period, whereas kidney transplantation occurred more often. In patients ≥ 65 years, incidence of home dialysis remained constant, whereas mortality decreased.

Conclusions: In patients < 65 years, the overall use of home dialysis declined consistently over the past 20 years. The age of home dialysis patients increased more rapidly than that of in-centre dialysis patients. These developments have a significant impact on the organization of home dialysis.

Introduction

Globally, the number of patients with chronic kidney disease (CKD) and end stage kidney disease (ESKD) is continuing to rise.^{1, 2} This growth in prevalence of patients who need kidney replacement therapy (kidney transplantation or dialysis) causes a major economic and logistical burden to the healthcare system.^{1, 2} The majority of patients is treated with in-centre haemodialysis (CHD), while the use of dialysis at home is low.³ But home dialysis offers more flexibility and independence, which could improve quality of life.^{4, 5} In addition, home dialysis might be more cost-effective than CHD.⁶

Another important development is global ageing, also resulting in the ageing of the dialysis population. A further contribution to this is that older patients are not often eligible for kidney transplantation. The ageing of the dialysis population might be a reason for the low use of home dialysis modalities.⁷ In the past, home dialysis generally was performed by young, employed patients. However, nowadays young patients are frequently transplanted with kidneys from living donors.⁸

Consequently, in order to increase the use of home dialysis, it would be helpful to gain better understanding of the impact of age on the home dialysis use, for example to reduce the economic burden of a growing patient population. The aim of this study is to explore time trends in the use of home dialysis in the Netherlands. This country had a pronounced decline in home dialysis patients during the last two decades, and it consistently ranks amongst the countries with the highest rates of kidney transplantations worldwide.^{3, 9, 10} Therefore, we studied the uptake of home dialysis between 1997-2016 in patients commencing dialysis treatment, stratified for age categories.

Methods

Study design

Anonymized registry data from the Dutch Renal Registry (RENINE) were used for this multicentre cohort study. RENINE collects treatment data of dialysis patients in all Dutch dialysis units; > 95% of all Dutch dialysis patients are registered in RENINE.¹¹ Kidney replacement therapies are registered as CHD, peritoneal dialysis (PD), home haemodialysis (HD) or kidney transplantation. Modality and centre transfers are updated regularly. For this analysis, age at start of dialysis treatment, sex, dates of modality transfers, and information on recovery of kidney function, kidney transplantation, and death were provided. All patients provided informed consent for registration of the data and usage of data for conducting scientific research. Reporting of the study conforms to broad EQUATOR guidelines.^{12, 13}

Study population

Dialysis episodes of patients who started *maintenance* dialysis treatment between 1-1-1997 through 31-12-2016 in the Netherlands were included, including dialysis episodes of patients who previously underwent kidney transplantation. Each dialysis episode was followed for 2 years, the last day of follow-up was 31-12-2018. A patient may have had multiple dialysis episodes during the study period and may thus be included more than once. Dialysis episodes instead of individual patients were chosen because we considered that a dialysis modality choice is made in each new dialysis episode, including in episodes of patients with a dialysis history. Dialysis episodes shorter than 90 days were excluded. In addition, dialysis episodes of patients < 20 years of age were excluded, since paediatric care is different from adult patient policy and this patient population is small.

Study outcomes

Primary outcome was start of home dialysis, i.e. PD and home HD. Both home dialysis at the beginning of the dialysis episode as well as a transfer to PD or home HD within 2 years of dialysis initiation were included. Subsequent switches after the start of home dialysis were ignored. A complete list of registry codes used to define study outcomes is provided in Appendix S1.

A relatively long transfer period of 2 years was chosen to also include *home HD* patients; in this registry study, the median time to HHD was 16 months [IQR 9 - 28] while the median time of transfer to PD was 4 months [IQR 2 - 12]. As in literature a shorter transfer period is more common¹⁴, start of home dialysis within 12 months of dialysis initiation was also evaluated as a sensitivity analysis.

As secondary outcome, start of PD and start of home HD were analysed separately.

Statistical analysis

The age of incident patients was reported as mean with standard deviation (SD) and sex of incident patients as proportions.

Logistic regression was used to assess the uptake of home dialysis between 1997 through 2016. Calendar time at dialysis initiation was equally divided into 5-years periods: 1997-2001, 2002-2006, 2007-2011, and 2012-2016. The period 2002 to 2006 was set as reference category. During this period the incidence of CHD in the Netherlands peaked after opening of standalone dialysis centres following a governmental decision to allow dialysis treatment in satellite and independent centres.^{15,} ¹⁶ Follow-up time for each episode was maximum 2 years and censoring occurred at recovery of kidney function, kidney transplantation, or death (for corresponding codes, see Appendix S1). A logistic mixed model analysis was performed to adjust for clustering of dialysis vintage, and transplantation history. Due to the interaction of age with the different time periods, analyses were stratified for the following age categories: 20-44 years, 45-64 years, 65-74 years, or ≥ 75 years.¹⁷

A competing risk model was used to estimate the cumulative incidence function (CIF) for start of home dialysis in incident patients with recovery of kidney function, kidney transplantations, and all-cause mortality as competing events.¹⁸ The 2-year cumulative incidence is the proportion of the study population, that is incident dialysis patients, who develop the outcome of interest during this time before the occurrence of a competing event. Subsequently, CIFs were estimated for kidney transplantations and all-cause mortality. In these analyses, the other three outcomes were treated as competing events. The three curves were plotted simultaneously. The curve for all-cause mortality was plotted as 1 minus CIF.

To further explore the robustness of results, three sensitivity analyses were conducted as follows: (i) home dialysis was defined as start with home dialysis, or transfer to home dialysis within the *first* year after start dialysis - instead of within 2 years; (ii) only the first dialysis episode of patients were analysed, analysing patients instead of dialysis episodes and using logistic regression instead of mixed model logistic regression analysis; and (iii) only episodes of patients who were still treated with dialysis after two years were analysed. All incident dialysis episodes followed by recovery of kidney function (n=771), kidney transplant (n=4118), or death (n=7786) within 2 years were excluded, irrespective of dialysis treatment modality.
Overall, a p-value of <0.05 was considered statistically significant. All analyses were performed using SPSS Statistics 25 (IBM) or STATA 14.

Results

A total of 33,340 chronic dialysis episodes between 1997 and 2016 fulfilled our inclusion criteria; these episodes belonged to 31,569 adult patients (Figure 1). Table 1 shows the characteristics of dialysis episodes and incident patients included in the study. Both the total number of dialysis episodes as the total number of incident patients increased from 1997 to 2016, whereas the total number of home dialysis episodes decreased (from 3,037 to 2,390). The total number of home HD was low, yet increased (from 67 to 253). The increase in the total number of incident patients was attributable to the increase in elderly patients: the number of patients aged ≥ 65 years increased from 2,921 to 4,889, whereas the number of patients aged 20-44 years decreased from 1,133 to 747.



Figure 1. Flowchart of the study

* Main analysis

** Second sensitivity analysis

	1997-2001	2002-2006	2007-2011	2012-2016
Dialysis episodes				
Total number of dialysis episodes	7230	8107	9004	8999
Total number of home dialysis episodes *	3037	2668	2535	2390
Total number of PD episodes *	2980	2580	2412	2155
Total number of home HD episodes *#	67	98	131	253
Incident patients				
Total number of incident patients	6496	7329	8047	8020
Aged 20-44 years	1133	1025	863	747
Aged 45-64 years	2442	2480	2536	2384
Aged 65-74 years	1881	2132	2236	2439
Aged \geq 75 years	1040	1692	2412	2450
Mean age at start dialysis (years ±SD)	59.6 ± 15.0	62.5 ± 14.8	64.9 ± 14.5	65.6 ± 14.1
Male (%)	3909 (60)	4487 (61)	5002 (62)	5009 (62)

Table 1. Characteristics of dialysis episodes and incident patients of the study population, by time period

*within 2 years of dialysis initiation

46 home haemodialysis episodes, were preceded by PD treatment

In Figure 2, the mean age in years at the start of a dialysis episode between 1997 and 2016 is shown. The age of home dialysis patients increased from 51.9 ± 15.1 to 62.5 ± 14.6 years during this period, while the age of CHD patients increased from 62.5 ± 14.0 to 65.5 ± 14.5 years.



Figure 2. Mean age at start of a dialysis episode between 1997 and 2016. Mean age is presented with confidence intervals. Each dot represents a year from 1997 to 2016

Time trends in uptake of home dialysis

Table 2 shows the results of the logistic regression assessing the uptake of home dialysis in each time period for the four age categories, using 2002-2006 as the reference period since governmental policies introduced around this period incentivized the growth of dialysis centres.^{15, 16} Table 2A shows the uptake of home dialysis within 2 years of dialysis initiation, and Table 2B shows the start of home dialysis within one year of dialysis initiation. During 1997-2001, for all age categories the uptake of home dialysis was significantly higher compared to the reference period (adjusted odds ratios (OR) ranging from 1.30 to 2.17, Table 2A). In the youngest two age categories, that is dialysis episodes of patients <65 years, the uptake of home dialysis was significantly lower in time periods 2007-2011 and 2012-2016 than for the period 2002-2006 (adjusted ORs ranging from 0.36 to 0.63). Each time period of 5 years was associated with a significantly lower uptake of home dialysis compared to the previous period in these age categories.

In the 65- to 74-year category, adjusted ORs in the time periods 2007-2011 and 2012-2016 were not significantly different from the reference period. In patients aged \geq 75 years, the two most recent time periods were associated with a higher uptake of home dialysis (adjusted ORs 1.21 and 1.52 resp.) compared to the reference period. As findings were similar for the analyses with, respectively, a 2-year and a 1-year transfer period, all further analyses were performed with a transfer period of 2 years to allow for the longer transfer time of home HD.

		Time period						
		1997-2001 OR (95	5% CI)	2002-2006 ^a	2007-2011 OR (95	% CI)	2012-2016 OR (95°	% CI)
		V	B		Α	B	A	B
Age	unadjusted	2.24 (1.70 – 2.94)	2.44 (1.82 – 3.27)	1.0	0.52 (0.40 – 0.69)	0.49 (0.37 – 0.67)	0.34 (0.25 – 0.46)	0.32 (0.23 – 0.44)
20-44	$adjusted^b$	2.17 (1.66 – 2.84)	2.35 (1.77 – 3.13)	1.0	0.54 (0.42-0.72)	$0.53 \ (0.40 - 0.70)$	0.36 (0.27 – 0.49)	0.35 (0.26-0.48)
Age	unadjusted	1.61 (1.35 – 1.92)	1.63 (1.37 – 1.95)	1.0	0.62 (0.52 – 0.74)	$0.60\ (0.50-0.71)$	0.44 (0.36 – 0.53)	0.41 (0.34 - 0.50)
45-64	$adjusted^b$	1.60 (1.34 – 1.91)	1.62 (1.35 – 1.95)	1.0	0.63 (0.53 – 0.75)	0.60 (0.50 - 0.72)	0.43 (0.36 – 0.53)	0.41 (0.34 - 0.51)
Age	unadjusted	1.30 (1.14 – 1.49)	1.28 (1.12 – 1.47)	1.0	0.95 (0.84 - 1.09)	0.94 (0.82 - 1.08)	0.98 (0.86 – 1.11)	0.93 (0.82 - 1.06)
65-74	$adjusted^b$	1.30 (1.14 – 1.49)	1.28 (1.12 – 1.47)	1.0	0.95 (0.84 - 1.09)	0.94 (0.83 – 1.07)	0.98 (0.86 – 1.12)	0.94 (0.82 – 1.07)
Age above	unadjusted	1.35 (1.10 – 1.66)	1.37 (1.11 – 1.68)	1.0	1.22 (1.03 - 1.45)	1.21 (1.02 – 1.44)	1.54 (1.30 - 1.81)	1.50 (1.27 – 1.78)
75	$adjusted^b$	$1.35\ (1.10-1.67)$	1.36 (1.11 – 1.68)	1.0	1.21 (1.02 - 1.44)	1.20 (1.01 – 1.43)	1.52 (1.29 - 1.80)	1.49 (1.25 – 1.76)
^a time period	1 2002-2006	was regarded as re	eference period					

Table 2. Uptake of home dialysis in 2 years of dialysis initiation (A) and in the first year of dialysis initiation (B) (n=33,340), by time period and age category

^a time period 2002-2006 was regarded as reference period ^b adjusted for sex, dialysis vintage, and transplantation history

Time trends in uptake of PD or home HD

The uptake of PD was quite similar to the overall uptake of home dialysis (Table 3). However in the 65-74 years category, the last time period was associated with a borderline significant lower uptake of PD (adjusted OR 0.89, 95% CI 0.78 - 1.01).

		Time period			
		1997-2001 OR (95% CI)	2002- 2006 ^a	2007-2011 OR (95% CI)	2012-2016 OR (95% CI)
Age 20-44	unadjusted	2.41 (1.81 – 3.22)	1.0	0.46 (0.34 – 0.62)	0.26 (0.19 – 0.37)
	adjusted ^b	2.33 (1.75 – 3.11)	1.0	0.49 (0.36– 0.66)	0.28 (0.20 – 0.39)
Age 45-64	unadjusted	1.68 (1.40 – 2.01)	1.0	0.60 (0.50 – 0.72)	0.38 (0.31 – 0.46)
	adjusted ^b	1.67 (1.39 – 2.00)	1.0	0.60 (0.50 – 0.72)	0.37 (0.30 – 0.46)
Age 65-74	unadjusted	1.31 (1.15 – 1.50)	1.0	0.93 (0.81 – 1.06)	0.88 (0.78 – 1.01)
	adjusted ^b	1.31 (1.15 – 1.50)	1.0	0.93 (0.82 – 1.07)	0.89 (0.78 – 1.01)
Age above 75	unadjusted	1.36 (1.10 – 1.67)	1.0	1.22 (1.03 - 1.45)	1.40 (1.18 - 1.66)
	adjusted ^b	1.36 (1.10 – 1.67)	1.0	1.21 (1.02 - 1.44)	1.39 (1.17 - 1.64)

Table 3. Uptake of peritoneal dialysis within 2 years of dialysis initiation (n=33,340), by time period and age category

^a time period 2002-2006 was regarded as reference period

^b adjusted for sex, dialysis vintage, and transplantation history

After correction for sex, age, dialysis vintage, and transplantation history, the home HD use increased for each time period (Supplementary Table S1). The last time period had an adjusted OR of 3.57 (2.59 - 4.92). As the number of home HD episodes was too low, no stratification for age categories was performed.

Time trends in incidence of home dialysis

Figure 3 shows the results of a competing risk approach modelling the cumulative incidences for start of home dialysis and those for kidney transplantation and death following CHD within 2 years after dialysis initiation, categorized by time period and age group. Figure 3A shows that the 2-year incidence of home dialysis for patients aged 20-44 years decreased in subsequent time periods from 58% to 34%. Figure 3B shows that the 2-year incidence for patients aged 45-64 years also decreased from 45% to 29%. In patients aged 65-74 years, the 2-year incidence of home dialysis was 29% in time period 1997-2001 and remained 24% during the other time periods (Figure 3C). In patients aged \geq 75 years, the 2-year incidence of home dialysis was low: 17% in the first, 14% in the second, 16% in the third and 19% in the last time period (Figure 3D).

In the youngest age groups the 2-years incidence of kidney transplantation whilst on CHD increased considerably, from 7% to 30% in patients aged 20-44 years and from 5% to 16% in patients aged 45-64 years. In patients aged 65-74 years the incidence of kidney transplantation was 8% during the last time period, and in patients aged > 75 year, this incidence was almost nihil. In the time period 1997-2001, the 2-year incidence of home dialysis and kidney transplantations combined was 65% for patients aged 20-44 years, which was comparable with the combined incidence in the time period 2012-2016 in this age category. In patients aged 45-64 years, the combined 2-year incidence was 50% in 1997-2001 and 45% in 2012-2016. In patients aged 65-74 years, the 2-year incidences were 30% and 32% respectively.

The 2-year incidence of mortality on CHD decreased over the time periods for all age categories (Figure 3A-D). This phenomenon was most pronounced in older patients: the incidence of mortality decreased from 27% to 16% in patients aged 65-74 years and from 38% to 27% in patients aged \geq 75 years. In addition, the proportion of patients that stayed on CHD increased from 43% to 52% in patients aged 65-74 years and from 44% to 53% in patients aged \geq 75 years.



Figure 3. Cumulative 2 year incidences of home dialysis, kidney transplantation, CHD and



death in incident patients. A, 20-44 years; B, 45-64 years; C, 65-74 years; D, ≥ 75 years

Sensitivity analyses

Supplementary Table S2 shows the results of the sensitivity analysis of only first dialysis episodes between 1997-2016. A total of 29,892 patients were analysed. The uptake of home dialysis was still significantly lower in the last two time periods for patients < 65 years old, yet ORs tended to be higher if compared to the original analysis. The OR for home dialysis uptake in period 2012-2016 was 0.54 (95% CI 0.45-0.66) in patients aged 20-44 years and 0.59 (95% CI 0.52-0.66) in patients aged 45-64 years. In the third sensitivity analysis, only episodes of patients that were still on dialysis two years after dialysis initiation were evaluated, excluding episodes that ended with recovery of kidney function, kidney transplantation or death. A home dialysis, that is start or transfer to home dialysis within 2 years of dialysis initiation. The results of this analysis were similar to the results from the original analysis, except that in dialysis episodes of patients ≥ 75 years of age the uptake of home dialysis during the first time period was no longer significantly higher compared to the reference period (Supplementary Table S3).

Discussion

In this large cohort of Dutch patients, the home dialysis use in patients aged < 65 years declined over the time periods from 1997 to 2016. In these younger patients a considerable increase in the number of kidney transplants was seen. In contrast, the older population showed a constant home dialysis use over time for the patients 65-74 years of age and a significant increase for the patients above 75 years of age. As a result, the home dialysis population aged remarkably. In both elderly age groups, kidney transplantation was negligible, but a clear decrease in mortality was found in elderly patients starting dialysis. Most of the elderly patients remained on CHD over time. The predominantly used home-dialysis treatment in this cohort was PD. Although numbers are low, over time the home HD use increased.

Multiple factors possibly influenced the changes in the uptake of home dialysis. In the first time period, 1997-2001, the uptake of PD was quite high, in part explained by a shortage of CHD facilities. After a change in legislation regarding initiating a dialysis centre by the Dutch government in 1999, this capacity problem was resolved. Consequently, many new dialysis centres appeared and an increase in patients starting

CHD was observed in 2002.^{15, 16} Apparently, such a policy change can have a major influence on the choice of dialysis modality within a population, as has also been reported in North America.^{19, 20} In contrast, in Australia and China governmental initiatives to promote home dialysis have resulted in a stabilization or even an increase in the prevalence of home dialysis patients.^{14, 21} In such large countries with extensive rural areas and great distance to the nearest dialysis centre home dialysis may be a favoured treatment.²¹ Indeed, in China 20% of the total dialysis population is treated with PD, while in Australia 25% is treated with home dialysis.^{21,22} The small country of Hong Kong has even the highest percentage of home dialysis throughout the world, as 76% of dialysis patients are treated with PD, due to a three-decade PDfirst policy adopted due to its cost-effectiveness.²³ Another country in which policy changes had a marked effect is the USA. In 2018, this country had a total of 12% of dialysis patients on a home-based therapy compared to 9% prior to differences in reimbursement.²⁴ Main reasons for changing the reimbursement were rising healthcare costs and improving healthcare efficiency.^{21, 25} In Europe, the proportion varies from 7% in Greece to 30% in Scandinavian countries such as Finland.³ In the latter a home first policy was adopted, partly due to a capacity problem but more importantly to provide individualized dialysis treatment which may be best achieved at home.²⁶ Overall, practices in these countries suggest that governmental policies to promote home dialysis are important and can have a large impact on uptake of home dialysis. In the present analysis, the initial decrease of home dialysis came to a halt in the time periods following 2002-2006 in the elderly patient groups, possibly due to dedication and initiatives of both nephrologists and nurses who stimulated home dialysis in these patients.

The average home dialysis patient aged significantly over a period of 20 years. First, this can be explained by the aging of the total dialysis population since more elderly patients started dialysis. Second, this can be explained by less younger patients starting home dialysis, since these patients are more often transplanted. This suggestion is supported by the fact that the CHD population has aged less than the home dialysis population and that we observed a clear increase in kidney transplantations after CHD initiation in patients under <75 years of age. This trend is in agreement with European data.²⁷. One could claim that 'the home dialysis patient of 20 years ago obtains a kidney transplant at present'. Indeed, in the younger population the combined 2-year incidence of transplantation and home dialysis remained more or less the same over a period of 20 years.

In addition, a decrease in the 2-year incidence in mortality on CHD was found from 1997 to 2016, most pronounced in patients above 65 years of age. This is consistent with a recent study of the ERA-EDTA, contributing a 10-year reduction in mortality not solely to a better survival in the general population, but also to improvements in dialysis care.²⁸ As a consequence of the decreased mortality rates, more elderly patients are on long-term maintenance dialysis. The majority of these patients is treated with CHD; the 2-year incidence of home dialysis in most elderly patients (\geq 75 years) increased only slightly from 17% to 19%. The low incidence of home dialysis in the elderly patients might be explained by the notion that elderly patients are too frail to be treated with home dialysis.⁷ However, the greater proportion of patients staying on CHD over time, could also suggest that more elderly patients would be able to start home dialysis if sufficiently assisted. Nevertheless, the aging of the dialysis population will have implications for the organization of predialysis education and home dialysis, as older patients may require additional support.

Over the past 15 years, several international initiatives were introduced to promote home dialysis, especially in elderly patients.²⁹ These initiatives include training of community-based home care workers to perform dialysis tasks at the patient's home, prolonging training time for the elderly patient and updating educating programs to enhance informed decision making.³⁰⁻³² Although we observed a 50% higher uptake of home dialysis in patients above 75 years of age, the overall use in these elderly patients remained low: the proportion increased from 14% in the reference period, that is 2002-2006, to 19% in the most recent time period. It should also be noted that the 2-year incidence for home dialysis was 17% in the time period prior to the governmental legislation. Thus, the abovementioned initiatives possibly helped to revive home dialysis after the governmental decision. In other countries a higher proportion of elderly patients is treated with home dialysis. Especially in Australia and New Zealand this proportion is quite high, 24% and 47% respectively, suggesting that it is possible for many elderly patients to perform home dialysis.³³ Incorporating more initiatives to promote home dialysis may allow more elderly patients to start home dialysis in the future.

The growing number of, especially elderly, dialysis patients puts pressure on healthcare expenses worldwide, since dialysis is an expensive treatment.^{1, 2} Home dialysis might be a mean of relieving this financial burden, since especially continuous ambulatory PD is supposed to be more cost-effective.⁶ Moreover, elderly patients may as well

benefit from home dialysis: they might obtain better quality of life and might be more satisfied with assisted PD than with CHD.^{4, 34} However, home dialysis in elderly patients emphasizes the need for adaptation in organization of home dialysis care, yet total expenses, including those for home care workers, remains unknown. The results presented in this study have implications for further research, and underscore the need of cost-effectiveness studies in elderly patients.

The results of our study remained robust in three different sensitivity analyses, a strength of this study. Other strengths of this study include its large sample size and the inclusion of dialysis episodes of nearly all chronic dialysis patients in the Netherlands over 20 years. This enabled us to explore in detail the various shifts in kidney replacement therapy and in competing events, that is kidney transplantation and mortality, over time. However, registry data are also a limitation to this study. Not all potentially relevant confounders are registered in the registry; we were for example unable to explore the effect of a pre-dialysis education program.³⁵ Other patient-specific characteristics that are known to influence dialysis modality choice, such as comorbidities and acute start of dialysis, could also have changed the main results since these demographic characteristics have supposedly changed over time in the home dialysis population.^{14, 30, 36} We evaluated shifts in kidney replacement therapy after dialysis initiation; the effect of pre-emptive kidney transplants is not evaluated in the present analysis.¹¹ Furthermore, not necessarily a limitation but noteworthy nevertheless, we presented the 2-year incidences of kidney transplantation and mortality for incident patients initiating CHD, not the kidney transplantation and mortality incidences for patients that initiated treatment with home dialysis.

Conclusions

From 1997 to 2016, the home dialysis use in patients aged <65 years declined sharply. This decrease can in part be explained by an increase in kidney transplantations. In incident patients above 65 years of age, the uptake of home dialysis remained stable, possibly explained by initiatives to promote home dialysis in the elderly. This study demonstrated that the home dialysis population has aged considerably, which was more pronounced than the ageing of the dialysis population in general.

Within a growing population with ESKD, sufficient resources to facilitate home dialysis must be offered to support this older patient population in their dialysis modality of choice.

References

- Couser WG, Remuzzi G, Mendis S, Tonelli M. The contribution of chronic kidney disease to the global burden of major noncommunicable diseases. Kidney Int. 2011;80(12):1258-70.
- 2. Xie Y, Bowe B, Mokdad AH, Xian H, Yan Y, Li T, *et al.* Analysis of the Global Burden of Disease study highlights the global, regional, and national trends of chronic kidney disease epidemiology from 1990 to 2016. Kidney Int. 2018;94(3):567-81.
- 3. ERA-EDTA Registry: ERA-EDTA Registry Annual Report 2018. Amsterdam UMC, location AMC, Department of Medical Informatics, Amsterdam, the Netherlands, 2020.
- 4. Bonenkamp AA, van Eck van der Sluijs A, Hoekstra T, Verhaar MC, van Ittersum FJ, Abrahams AC, *et al.* Health-Related Quality of Life in Home Dialysis Patients Compared to In-Center Hemodialysis Patients: A Systematic Review and Meta-analysis. Kidney Medicine. 2020;2(2):139-54.
- Dahlerus C, Quinn M, Messersmith E, Lachance L, Subramanian L, Perry E, *et al.* Patient perspectives on the choice of dialysis modality: results from the Empowering Patients on Choices for Renal Replacement Therapy (EPOCH-RRT) Study. Am J Kidney Dis. 2016;68(6):901-10.
- 6. Mohnen SM, van Oosten MJM, Los J, Leegte MJH, Jager KJ, Hemmelder MH, *et al.* Healthcare costs of patients on different renal replacement modalities Analysis of Dutch health insurance claims data. PLoS One. 2019;14(8):e0220800.
- 7. van de Luijtgaarden MW, Noordzij M, Stel VS, Ravani P, Jarraya F, Collart F, *et al.* Effects of comorbid and demographic factors on dialysis modality choice and related patient survival in Europe. Nephrol Dial Transplant. 2011;26(9):2940-7.
- 8. Cecka JM. Kidney Transplantation from Living Unrelated Donors. Annual Review of Medicine. 2000;51(1):393-406.
- Robinson BM, Akizawa T, Jager KJ, Kerr PG, Saran R, Pisoni RL. Factors affecting outcomes in patients reaching end-stage kidney disease worldwide: differences in access to renal replacement therapy, modality use, and haemodialysis practices. Lancet (London, England). 2016;388(10041):294-306.
- 10. ERA-EDTA Registry: ERA-EDTA Registry 2003 Annual Report. Academic Medical Center, Amsterdam, The Netherlands, May 2005.
- 11. Hoekstra T, Dekker FW, Cransberg K, Bos WJ, van Buren M, Hemmelder MH. RENINE annual report 2018. Available at: https://www.nefrovisie.nl/jaarrapportages/ Accessed: 04-01-2021. 2018.
- 12. Simera I, Moher D, Hoey J, Schulz KF, Altman DG. A catalogue of reporting guidelines for health research. European journal of clinical investigation. 2010;40(1):35-53.

- 13. Benchimol EI, Smeeth L, Guttmann A, Harron K, Moher D, Petersen I, *et al.* The REporting of studies Conducted using Observational Routinely-collected health Data (RECORD) statement. PLoS medicine. 2015;12(10):e1001885.
- Ethier I, Cho Y, Hawley C, Pascoe EM, Roberts MA, Semple D, *et al.* Effect of patient- and center-level characteristics on uptake of home dialysis in Australia and New Zealand: a multicenter registry analysis. Nephrol Dial Transplant. 2020.
- Hemke AC, Dekker FW, Bos WJ, Krediet RT, Heemskerk MB, Hoitsma AJ. [Causes of decreased use of peritoneal dialysis as a kidney replacement therapy in the Netherlands]. Ned Tijdschr Geneeskd. 2012;156(21).
- Ministerie van Volksgezondheid, Welzijn en Sport. Staatscourant van het Koninkrijk der Nederlanden, Staatscourant 1999, 126 page 5 (06-07-1999) retrieved from: https://zoek. officielebekendmakingen.nl/stcrt-1999-126-p5-SC19556.html.
- 17. Kramer AP, M.; Noordzij, M.; Stel V.A.; Andrusev, A.M.; *et al.* The European Renal Association European Dialysis and Transplant Association (ERA-EDTA) Registry Annual Report 2016: a summary. Clinical Kidney Journal. 2019.
- Fine JP, Gray RJ. A Proportional Hazards Model for the Subdistribution of a Competing Risk. Journal of the American Statistical Association. 1999;94(446):496-509.
- 19. Walker DR, Inglese GW, Sloand JA, Just PM. Dialysis facility and patient characteristics associated with utilization of home dialysis. Clin J Am Soc Nephrol. 2010;5(9):1649-54.
- 20. Blake P. Why is the proportion of patients doing peritoneal dialysis declining in North America? Perit Dial Int. 2001.
- Li PK-T, Chow KM, Van de Luijtgaarden MWM, Johnson DW, Jager KJ, Mehrotra R, et al. Changes in the worldwide epidemiology of peritoneal dialysis. Nature Reviews Nephrology. 2017;13(2):90-103.
- 22. ANZDATA Registry. 43rd Report, Chapter 2: Prevalence of Renal Replacement Therapy for End Stage Kidney Disease. Australia and New Zealand Dialysis and Transplant Registry, Adelaide, Australia. 2020. Available at: http://www.anzdata.org.au Accessed: 13-07-2021. 2020.
- 23. Leung CB, Cheung WL, Li PK. Renal registry in Hong Kong-the first 20 years. Kidney international supplements. 2015;5(1):33-8.
- Johansen KL, Chertow GM, Foley RN, Gilbertson DT, Herzog CA, Ishani A, *et al.* US Renal Data System 2020 Annual Data Report: Epidemiology of Kidney Disease in the United States. American Journal of Kidney Diseases. 2021;77(4, Supplement 1):A7-A8.
- Sedor JR, Watnick S, Patel UD, Cheung A, Harmon W, Himmelfarb J, *et al.* ASN End-Stage Renal Disease Task Force: perspective on prospective payments for renal dialysis facilities. J Am Soc Nephrol. 2010;21(8):1235-7.
- Honkanen EOaR, V.M. . What happened in Finland to increase home hemodialysis? Hemodialysis International 2008;12.
- 27. van de Luijtgaarden MW, Jager KJ, Segelmark M, Pascual J, Collart F, Hemke AC, *et al.* Trends in dialysis modality choice and related patient survival in the ERA-EDTA Registry over a 20-year period. Nephrol Dial Transplant. 2016;31(1):120-8.
- 28. Boenink R, Stel VS, Waldum-Grevbo BE, Collart F, Kerschbaum J, Heaf JG, *et al.* Data from the ERA-EDTA Registry were examined for trends in excess mortality in European adults on kidney replacement therapy. Kidney Int. 2020.

- 29. Segall L, Nistor I, Van Biesen W, Brown EA, Heaf JG, Lindley E, *et al.* Dialysis modality choice in elderly patients with end-stage renal disease: a narrative review of the available evidence. Nephrol Dial Transplant. 2017;32(1):41-9.
- 30. Oliver MJ, Quinn RR, Richardson EP, Kiss AJ, Lamping DL, Manns BJ. Home care assistance and the utilization of peritoneal dialysis. Kidney Int. 2007;71(7):673-8.
- 31. Hurst H, Figueiredo AE. The Needs of Older Patients for Peritoneal Dialysis: Training and Support at Home. Perit Dial Int. 2015;35(6):625-9.
- 32. Maaroufi A, Fafin C, Mougel S, Favre G, Seitz-Polski B, Jeribi A, *et al.* Patients' preferences regarding choice of end-stage renal disease treatment options. Am J Nephrol. 2013;37(4):359-69.
- 33. Brown EA, Johansson L. Dialysis options for end-stage renal disease in older people. Nephron Clin Pract. 2011;119 Suppl 1:c10-3.
- Iyasere OU, Brown EA, Johansson L, Huson L, Smee J, Maxwell AP, *et al.* Quality of Life and Physical Function in Older Patients on Dialysis: A Comparison of Assisted Peritoneal Dialysis with Hemodialysis. Clin J Am Soc Nephrol. 2016;11(3):423-30.
- 35. Castledine CI, Gilg JA, Rogers C, Ben-Shlomo Y, Caskey FJ. Renal centre characteristics and physician practice patterns associated with home dialysis use. Nephrol Dial Transplant. 2013;28(8):2169-80.
- Rioux JP, Cheema H, Bargman JM, Watson D, Chan CT. Effect of an in-hospital chronic kidney disease education program among patients with unplanned urgent-start dialysis. Clin J Am Soc Nephrol. 2011;6(4):799-804.

	Time period			
	1997-2001 OR (95% CI)	2002- 2006 ª	2007-2011 OR (95% CI)	2012-2016 OR (95% CI)
unadjusted	0.72 (0.50 - 1.02)	1.0	1.26 (0.93 – 1.71)	2.90 (2.17 – 3.88)
adjusted ^ь	0.63 (0.43 – 0.93)	1.0	1.36 (0.99 – 1.87)	3.57 (2.59 – 4.92)

Supplementary material

Supplementary Table S1. Uptake of home haemodialysis within the first 2 years of dialysis initiation (n=33,340), by time period

^a time period 2002-2006 was regarded as reference period

^b adjusted for sex, age, dialysis vintage, and transplantation history

Supplementary Table S2. Sensitivity analysis: uptake of home dialysis in the first 2 years of
dialysis initiation in patients with a first dialysis episode in 1997-2016 (n=29,892), by time
period and age category

		Time period			
		1997-2001 OR (95% CI)	2002- 2006 ª	2007-2011 OR (95% CI)	2012-2016 OR (95% CI)
Age 20-44	unadjusted	1.63 (1.37 – 1.93)	1.0	0.71 (0.59 – 0.85)	0.54 (0.45 – 0.66)
	adjusted ^b	1.63 (1.37 – 1.93)	1.0	0.70 (0.59 – 0.84)	0.54 (0.45 – 0.66)
Age 45-64	unadjusted	1.30 (1.17 – 1.46)	1.0	0.80 (0.70 - 0.88)	0.59 (0.52 – 0.66)
	adjusted ^b	1.30 (1.17 – 1.46)	1.0	0.79 (0.70 – 0.88)	0.59 (0.52 – 0.66)
Age 65-74	unadjusted	1.30 (1.13 – 1.49)	1.0	0.96 (0.84 – 1.10)	0.99 (0.87 – 1.13)
	adjusted ^b	1.30 (1.14 – 1.50)	1.0	0.96 (0.84 – 1.10)	0.99 (0.86 – 1.13)
Age above 75	unadjusted	1.35 (1.10 – 1.66)	1.0	1.24 (1.05 – 1.48)	1.55 (1.31 – 1.84)
	adjusted ^b	1.36 (1.10 – 1.67)	1.0	1.24 (1.04 – 1.48)	1.54 (1.30 – 1.82)

^a time period 2002-2006 was regarded as reference period

 $^{\it b}$ adjusted for sex and transplantation history

		Time period			
		1997-2001 OR (95% CI)	2002- 2006 ª	2007-2011 OR (95% CI)	2012-2016 OR (95% CI)
Age 20-44	unadjusted	2.05 (1.48 – 2.86)	1.0	0.45 (0.31 – 0.64)	0.37 (0.25 - 0.55)
	adjusted ^b	1.96 (1.43 – 2.68)	1.0	0.49 (0.35 – 0.68)	0.42 (0.29 – 0.61)
Age 45-64	unadjusted	1.72 (1.37 – 2.16)	1.0	0.59 (0.47 – 0.74)	0.40 (0.30 - 0.52)
	adjusted ^b	1.71 (1.36 – 2.15)	1.0	0.60 (0.48 – 0.75)	0.41 (0.31 – 0.52)
Age 65-74	unadjusted	1.42 (1.20 – 1.69)	1.0	0.95 (0.81 - 1.13)	0.88 (0.75 – 1.04)
	adjusted ^b	1.42 (1.20 - 1.68)	1.0	0.95 (0.81 – 1.12)	0.88 (0.75 – 1.04)
Age above 75	unadjusted	1.09 (0.82 - 1.45)	1.0	1.16 (0.93 – 1.44)	1.40 (1.13 – 1.73)
	adjusted ^b	1.09 (0.82 – 1.45)	1.0	1.16 (0.93 – 1.45)	1.40 (1.13 – 1.73)

Supplementary Table S3. Sensitivity analysis: uptake of home dialysis within the first 2 years of dialysis initiation in dialysis episodes with a dialysis duration of at least 2 years (n=20,665), by time period and age category

^a time period 2002-2006 was regarded as reference period

^b adjusted for sex, dialysis vintage, and transplantation history

VALUE = therap	Original code in registry	Defined as
	in-centre HD Including active in-centre HD and nocturnal in-centre HD	CHD
	Home HD	Home HD
	PD Including CAPD and APD	PD
	Kidney transplantation Including deceased donor kidney transplantation and living donor kidney transplantation	Kidney transplantation
	Lost to follow-up	Lost to follow-up
In Re	Informed consent withdrawn	Lost to follow-up
	Recovery of kidney function	Recovery of kidney function
	Conservative treatment In use from 2015	Lost to follow-up
	Stop dialysis treatment and start palliative care	Death
	Death	Death

Appendix S1. Complete list of codes used to classify study outcomes and events during follow-up



Chapter 3

Comorbidity is not associated with dialysis modality choice in patients with end-stage kidney disease

Anna A. Bonenkamp, Sanne Vonk, Alferso C. Abrahams, Yolande M. Vermeeren, Anita van Eck van der Sluijs, Tiny Hoekstra, Frans J. van Ittersum and Brigit C. van Jaarsveld on behalf of the DOMESTICO study group

Nephrology. 2022. in print

Abstract

Background: Over the past years the proportion of home dialysis patients has decreased in the Netherlands. In addition, the home dialysis use varies significantly among centres. It is unclear whether this is the result of differences in comorbidity, other case mix factors or differently perceived barriers for home modalities by dialysis centres. Our aim was to investigate the association between comorbidity and dialysis modality choice.

Methods: The multi-centre DOMESTICO cohort study collected comorbidity data of patients who started dialysis in 35 Dutch centres from 2012 to 2016. Comorbidity was assessed by the Charlson comorbidity index. Home dialysis was defined as any peritoneal dialysis or home haemodialysis treatment during follow-up. Logistic regression analysis was used to assess the association between comorbidity and dialysis modality, with a mixed model approach to adjust for clustering of patients within dialysis centres. Other case mix factors, including age and body mass index, were included as confounders in the model.

Results: A total of 1358 patients were included, of whom 628 were treated with home dialysis. In crude mixed model analyses, the probability of receiving home dialysis was lower when comorbidity score was higher: having a high comorbidity score resulted in an odds ratio of 0.74 (95% CI 0.54-1.00, p-value 0.05) when compared to patients without comorbidities. After adjustments for age, sex, ethnic background, body mass index and dialysis vintage, there was no association between comorbidity and home dialysis.

Conclusion: Comorbidity was not significantly associated with home dialysis choice, after adjustment for several confounding factors including age and body mass index. Future studies should aim at unravelling the centre-specific characteristics that probably play a role in dialysis modality choice.

Introduction

The proportion of home dialysis patients has declined in several European countries, including the United Kingdom and the Netherlands.^{1, 2} In the Netherlands, the proportion of prevalent home dialysis patients almost halved over 15 years: from 30% in 2003 to 18% in 2018.³

This decrease in home dialysis is often explained by the increasing number of patients with diabetes mellitus and cardiovascular disease.^{1, 4-6} Patients have to be able to perform dialysis at home and as a result a high degree of comorbidity may be seen as a barrier to home dialysis. Indeed, peritoneal dialysis (PD) patients in older cohorts had fewer comorbidities than in-centre haemodialysis (CHD) patients.⁷⁻¹² Another perceived barrier is advanced age of patients with kidney failure, caused by ageing of the general population and by more kidney transplantations in younger patients.¹ In a registry study among different European countries, it was found that elderly patients and patients with various comorbidities were less likely to receive PD.⁵

However, in the proportion of patients treated with home dialysis, a large variation exists among countries and even among dialysis centres within a country.^{3, 5, 13} In the Netherlands, with a nation-wide home dialysis prevalence around 20%, the proportion of home dialysis varies considerably from 0% to even 40%.³ This variation could be explained by different characteristics of dialysis patients among centres, most importantly regarding comorbidity and age. However, this variation could also indicate different selection criteria for home dialysis among physicians. It remains unclear what the impact is of comorbidity on final dialysis modality choice.

The aim of this study is to investigate the association between comorbidity and type of dialysis treatment – home dialysis versus in-centre dialysis - in patients with end-stage kidney disease initiating dialysis between 2012-2017, accounting for centres' practice patterns.

Methods

Study design and patient population

The Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO) is a multi-centre retrospective cohort study investigating characteristics and outcomes of home and nocturnal dialysis patients, in comparison with in-centre dialysis patients. Eligible patients were adults who started maintenance dialysis treatment between 1 January 2012 and 1 January 2017, including those with graft failure. Patients who stopped dialysis or died within 30 days after dialysis initiation were excluded. In DOMESTICO, all patients who were treated with home dialysis (or nocturnal dialysis) during the study period were selected and CHD patients were randomly selected in a systematic manner. Patients were followed until kidney transplantation, wish to stop dialysis, death or study end on 1 January 2017. Local medical ethics committees of all participating dialysis centres approved the study.

Determinants

Comorbidity was assessed with Deyo's Charlson comorbidity index (CCI).¹⁴ The adaption of Deyo *et al.*, in which lymphoma and leukemia are scored under the condition 'malignancy', is most frequently used.^{14, 15} The CCI was calculated from the presence of a total of 17 conditions with several assigned weights ranging from 1 to 6 (Supplementary Table S1).^{14, 16} The total score in dialysis patients ranges from 2 to 29, as ESKD results in a CCI score of 2 points. The score was divided into three groups according to literature: a score of 2 reflecting no comorbidity (only ESKD), a score of 3-4 reflecting intermediate comorbidity, and a score of 5 or more points reflecting high comorbidity.¹⁶

In addition, the association of various single comorbidities with dialysis modality was evaluated. These comorbidities were: diabetes mellitus, ischemic heart disease, heart failure, cerebrovascular disease, any malignancy and chronic lung disease.

Data collection

All comorbidities were collected at dialysis initiation from patients' medical charts. Also age, sex, body mass index (BMI), ethnic background, cause of kidney failure, presence and duration of previous dialysis (i.e. dialysis vintage), and presence of previous transplantation were identified from patients' charts. BMI was divided into three groups according to the WHO classification: BMI < 25 kg/m², BMI 25 - 30

kg/m² (overweight), and BMI \geq 30 kg/m² (obese). A high home dialysis volume was considered a marker for a successful home dialysis programme. Home dialysis centre size was thus defined based on the mean annual number of prevalent home dialysis patients according to registry data and subsequently dichotomized into <30 and \geq 30 home dialysis patients.¹⁷

Outcome

In the present study, dialysis modality was defined as CHD (including nocturnal in-centre haemodialysis) or home dialysis, the latter including both PD and home haemodialysis (home HD). All patients who started with home dialysis or were ever treated with home dialysis during the follow-up were defined as home dialysis patients to reflect dialysis modality choice. If a patient was treated with both PD and home HD during the study period, the first episode of home treatment determined the category of home dialysis treatment.

Statistical analyses

All normally distributed continuous variables were reported as means with standard deviation (SD), non-normally continuous variables as median with interquartile range (IQR), and categorical variables as proportions. For examining differences between patients groups, t-tests, Mann-Whitney, and Chi-square tests were used where appropriate.

To assess the association between comorbidity and dialysis modality, logistic mixed model analysis was performed with CCI or single comorbidities as determinant. The assumption of linearity was validated and if violated, the CCI score was presented as categories. A mixed model - also known as multilevel model or hierarchical model - was chosen to account for the dependency of patients within a centre. This correction was performed by means of applying a random intercept for dialysis centre. Individual patients (level 1) were thus clustered within dialysis centres (level 2). The addition of a random slope was also tested, to allow for the association between comorbidity and dialysis modality to be different among dialysis centres. All analyses were corrected for age, sex, BMI, ethnic background, and dialysis vintage at study start. To investigate possible interaction of dialysis choice, interactions for home dialysis centre size, age, and BMI were investigated. In addition, important confounders were evaluated as individual risk factors as well. BMI was missing in 17% of the cases, therefore

weight and length were imputed with standard multiple imputation techniques using 10 repetitions and predictive mean matching (SPSS).¹⁸

Three sensitivity analyses were conducted, (i) using the Davies comorbidity score instead of the CCI (ii) including only patients with home dialysis as initial therapy and (iii) defining home dialysis as PD only, excluding all home HD patients.^{19, 20} The latter was performed because the association between comorbidity and home dialysis could be different for the two individual types of home dialysis. Finally, the two types of home dialysis were analysed separately using a multinomial logistic regression, in which outcomes were CHD, PD, and home HD.

A p-value of <0.05 was considered statistically significant. All analyses were performed using SPSS Statistics version 26 (IBM Corp) or STATA 14 (StataCorp LP).

Results

A total of 1358 patients were included in this study, of whom 46% was treated with home dialysis during the study period: 41% was treated with PD (n=564) and 5% with home HD (n=64). Most home dialysis patients (72%) started home dialysis as initial therapy. Median follow-up time, i.e. inclusion in the study to end of the study (kidney transplantation, death, stop of dialysis or January 1st 2017), was 1.7 years (IQR 0.8 - 2.9). Baseline characteristics of the patients are described in Table 1.

	All patients n=1358	Home dialysis n=628	In-center haemodialysis n=730
Male sex, n (%)	832 (61)	390 (62)	442 (61)
Age (yr), mean ± SD	62.4 ± 15.7	61.6 ± 15.6	63.1 ± 15.8
Body mass index (kg/m ²), mean \pm SD	26.8 ± 5.6	26.4 ± 5.1	27.2 ± 6.0
Ethnic background, n (%)			
Caucasian	805 (59)	403 (64)	402 (55)
Moroccan/Turkish	73 (5)	13 (2)	60 (8)
Asian	71 (5)	35 (6)	36 (5)
Afro-American	60 (4)	21 (3)	39 (5)
Unknown	330 (24)	146 (23)	184 (25)

Table 1. Characteristics of the 1358 included dialysis patients, divided by dialysis modality

	All patients	Home dialysis	In-center haemodialysis
	n=1358	n=628	n=730
ERA-EDTA code, n (%)			
Glomerulonephritis/pyelonephritis	261 (19)	125 (20)	136 (19)
Cystic kidney disease	78 (6)	39 (6)	39 (5)
Renovascular kidney disease	355 (26)	164 (26)	191 (26)
Diabetes mellitus	243 (18)	102 (16)	141 (19)
Other/unknown	421 (31)	198 (32)	223 (31)
Previous dialysis, n (%)	276 (20)	121 (19)	155 (21)
Dialysis vintage (mo), median [IQR] [#]	29.4 [11.0 - 57.7]	17.7 [2.2 – 45.4]	38.4 [15.4 - 62.8]
Previous renal transplant, n (%)	241 (18)	92 (15)	149 (20)
Charlson comorbidity index, n (%)			
2 (no comorbidity)	409 (30)	202 (32)	207 (28)
3-4 (intermediate comorbidity score)	553 (41)	257 (41)	296 (41)
\geq 5 (high comorbidity score)	396 (29)	169 (27)	227 (31)
Davies comorbidity score, n (%)			
0 (no comorbidity)	398 (29)	194 (31)	204 (28)
1-2 (intermediate risk)	722 (53)	330 (53)	392 (54)
\geq 3 (high comorbidity score)	238 (18)	104 (17)	134 (18)
Diabetes Mellitus, n (%)	465 (34)	193 (31)	272 (37)
Ischaemic heart disease, n (%)	378 (28)	182 (29)	196 (27)
Heart failure, n (%)	149 (11)	83 (13)	66 (9)
Cerebrovascular disease, n (%)	187 (14)	84 (13)	103 (14)
Any malignancy, n (%)	192 (14)	81 (13)	111 (15)
Chronic lung disease, n (%)	159 (12)	67 (11)	92 (13)

 Table 1. Characteristics of the 1358 included dialysis patients, divided by dialysis modality (continued)

[#] Dialysis vintage presented for patients with previous dialysis only.

The prevalence of comorbidity was: diabetes mellitus 34%, ischaemic heart disease 28%, heart failure 11%, cerebrovascular disease 14% and any malignancy 14%. Mean age at dialysis initiation was slightly higher in CHD patients compared to home dialysis patients (63.1 ± 15.8 vs. 61.6 ± 15.6 years, resp.). Patients receiving home dialysis were more likely to be Caucasian, had a shorter dialysis vintage at dialysis initiation, and less often a previous renal transplant. In Table 2, clinical characteristics of patients from small and large home dialysis centres are shown. No differences in

CCI, age and BMI were found between patients from small and large home dialysis centres.

	Patients from centres with < 30 home dialysis patients N=535	Patients from centres with \geq 30 home dialysis patients N=823	p-value
CCI of CHD patients, n (%)			
2 (no comorbidity)	94 (27)	113 (29)	0.45
3-4 (intermediate comorbidity score)	147 (43)	149 (38)	
\geq 5 (high comorbidity score)	101 (30)	126 (32)	
CCI of home dialysis patients, n (%)			
2 (no comorbidity)	69 (36)	133 (31)	0.24
3-4 (intermediate comorbidity score)	80 (41)	177 (41)	
\geq 5 (high comorbidity score)	44 (23)	125 (29)	
Mean age (± SD)	62.4 ± 15.0	62.4 ± 16.1	0.98
Mean BMI (± SD)	26.9 ± 5.7	26.7 ± 5.5	0.50

Table 2. Characteristics of patients from small and large home dialysis centres

CCI, Charlson comorbidity index; CHD, in-centre haemodialysis; BMI, body mass index.

Association between comorbidity and dialysis modality

Table 3 shows the association between comorbidity and home dialysis as dialysis modality choice. CCI was analysed in categories, since the linearity assumption was violated. Intermediate comorbidity, i.e. 3-4 points, was not associated with home dialysis as modality choice (unadjusted OR 0.97, 95% CI 0.73 – 1.28). A high comorbidity score, i.e. a score of ≥ 5 points, was associated with a lower probability of receiving home dialysis (unadjusted OR 0.74, 95% CI 0.54 – 1.00, p-value 0.05). After adjustments for age, sex, BMI, ethnic background, and dialysis vintage, a higher comorbidity score was no longer associated with home dialysis (adjusted OR 0.88, 95% CI 0.63 – 1.23). Age and BMI were the most important confounders in the model, they induced the greatest change in the regression coefficient respectively 28% and 30%. The other confounders induced changes of less than 10%. Adding a random slope to the model with CCI as a continuous variable did not change our results, indicating that dialysis centre did not influence the association between CCI and

dialysis modality choice. This suggests that comorbidity was not weighted differently among centres.

Patients with heart failure (n=149) were more likely to receive home dialysis, even after adjustments for confounders (adjusted OR 1.60, 95% CI 1.09 – 2.37). Diabetic patients were less likely to receive home dialysis in the unadjusted analysis, but after correction for confounders this association lost significance (adjusted OR 0.83, 95% CI 0.64 – 1.08). Patients with ischaemic heart disease, cerebrovascular disease, malignancies or chronic lung disease were as likely to receive home dialysis as CHD. The Davies comorbidity score had also no association with home dialysis choice (Supplementary Table S2). Comparable results to the original analysis were also found in a sensitivity analysis that included only patients with home dialysis as initial therapy.

•	y S1S
:	dial
	aemo
-	re h
	n-cent
•	t0 1
-	npared
	, co
	y SIS.
-	dial
	ome o
	t wit!
	treatmen
-	and
-	dity
-	comorbi
Ċ	ot
•	tion
•	ocia
	ASSC
	3.
	ف
-	g
I	(0

	Logistic mixed model regression analysis*					
	Odds ratio [95% CI] Crude	P-value	Odds ratio [95% CI] Adjusted'	P-value	Odds ratio [95% CI] Adjusted#	P-value
Charlson comorbidity index						
CCI 2	REF	0.82	REF	0.73	REF	0.53
CCI 3-4	0.97 [0.73 - 1.28]	0.05	1.05 [0.78 - 1.41]	0.30	1.10[0.82 - 1.49]	0.44
$CCI \ge 5$	0.74 [0.54 - 1.00]		$0.84 \ [0.61 - 1.17]$		$0.88 \ [0.63 - 1.23]$	
At least 1 comorbidity	0.86 [0.67 - 1.12]	0.26	0.97 [0.74 - 1.27]	0.81	1.01 [0.77 - 1.33]	0.93
Diabetes Mellitus †	0.75 [0.59 - 0.97]	0.03	$0.78 \ [0.61 - 1.01]$	0.06	$0.83 \ [0.64 - 1.08]$	0.17
Ischaemic heart disease	1.08 [0.83 - 1.40]	0.57	$1.17 \left[0.89 - 1.55 \right]$	0.26	$1.23 \ [0.93 - 1.63]$	0.15
Heart failure	1.47 [1.01 - 2.14]	0.05	1.54 [1.05 - 2.25]	0.03	1.60 [1.09 - 2.37]	0.02
Cerebrovascular disease	$0.79 \ [0.57 - 1.11]$	0.18	$0.83 \ [0.59 - 1.17]$	0.28	0.81 [0.57 - 1.15]	0.24
Any malignancy	0.91 [0.65 - 1.28]	0.58	$0.92 \ [0.65 - 1.30]$	0.65	$0.89 \ [0.63 - 1.26]$	0.50
Chronic lung disease	0.83 [0.58 - 1.21]	0.34	$0.88 \left[0.61 - 1.29 \right]$	0.52	$0.88 \ [0.60 - 1.29]$	0.50
* I ogistic mixed model regressic	n analvsis with dialvsis ce	ntre as rand	om intercent with individ	dual natients	as first level	

* Logistic mixed model regression analysis with dialysis centre as random intercept, with individual patients as first level. † Adjusted for age, sex, and BMI

Adjusted for age, sex, BMI, ethnic background, and dialysis vintage

† Adjusted for age, sex, ethnic background, and dialysis vintage. CCI, Charlson comorbidity index; BMI, body mass index.

Interaction of dialysis centre, age and BMI on the association between comorbidity and dialysis modality

Home dialysis centre size and age were no interactions in the association between comorbidity and home dialysis choice (Supplementary Tables S3 and S4). However, obese patients (BMI \ge 30 kg/m²) with an intermediate or high comorbidity score were significantly less likely to receive home dialysis compared to obese patients without comorbidities, adjusted OR 0.40 (95% CI 0.18 – 0.86, p-value 0.02) for intermediate comorbidity score and adjusted OR 0.43 (95% CI 0.20 – 0.93) for high comorbidity score (Table 4). Patients with a BMI <25 kg/m² with an intermediate comorbidity score were significantly more likely to receive home dialysis compared to patients with a BMI <25 kg/m² without comorbidities (adjusted OR 1.59, 95% CI 1.01 – 2.49, p-value 0.04).

	I ogistic mired model			
	regression analysis*	L		
	Odds ratio [95% CI] Crude	P-value	Odds ratio [95% CI] Adjusted [#]	P-value
Patients with BMI <25 kg/n $n = 493^{\dagger}$	1 ²			
Charlson comorbidity index CCI 2 CCI 3-4 CCI ≥ 5	REF 1.37 [0.89 – 2.11] 1.03 [0.62 – 1.69]	0.15 0.92	REF 1.59 [1.01 – 2.49] 1.22 [0.72 – 2.07]	0.04 0.46
Overweight patients (BMI 2 $n = 379 \ddagger$	$25 - 30 \ kg/m^2$			
Charlson comorbidity index CCI 2 CCI 3-4 CCI ≥ 5	REF 1.00 [0.61 – 1.64] 0.71 [0.41 – 1.24]	0.99 0.22	REF 1.10 [0.66 – 1.83] 0.76 [0.43 – 1.35]	0.72 0.36
Obese patients (BMI \ge 30 kg $n = 257$ †	$g/m^2)$			
Charlson comorbidity index CCI 2	REF	0.02	REF	0.02
$\frac{CCI > 4}{CCI \ge 5}$	0.40 [0.19 - 0.86] 0.42 [0.20 - 0.88]	0.02	0.40 [0.18 - 0.86] 0.43 [0.20 - 0.93]	0.02

 Table 4. Interaction of BMI in the association between CCI and treatment with home dialysis, compared with in-centre haemodialysis

* Logistic mixed model analysis with dialysis centre as random intercept, with individual patients as first level.

Adjusted for age, sex, ethnic background, and dialysis vintage

† A total of 1358 patients were analysed: for 229 patients imputed data for BMI were used. BMI was divided according to the WHO classification: BMI <25 kg/m², BMI 25 - 30 kg/m² (overweight), and BMI \geq 30 kg/m² (obese).

BMI, body mass index; CCI, Charlson comorbidity index.

Association between age or BMI and dialysis modality

Older age, analysed as an individual risk factor, was associated with a lower probability of receiving home dialysis both in unadjusted and adjusted analyses (Supplementary Table S5). Elderly patients (≥ 65 years of age) were less likely to receive home dialysis compared to patients younger than 65 years of age, adjusted OR 0.67 (95% CI 0.53 – 0.86, p-value 0.002). Also BMI, analysed as an individual risk factor, was associated with dialysis modality choice (Supplementary Table S6). Obese patients were less likely to receive home dialysis compared to patients with a BMI <25 kg/m2, adjusted OR 0.68 (95% CI 0.48 – 0.96, p-value 0.03)

Association between comorbidity and PD or HHD

A sensitivity analysis comparing PD with CHD, revealed similar results as the original analysis (Supplementary Table S7). Finally, the two types of home dialysis were analysed separately using a multinomial logistic regression, with CHD as reference treatment (Supplementary Table S8). A high comorbidity score of \geq 5 was significantly associated with a lower probability of receiving peritoneal dialysis compared to CHD, with a crude OR of 0.74 (95% CI 0.55 - 0.98, p-value 0.04). After adjusting for confounders, the association lost significance (OR 0.83, 95% CI 0.61 – 1.14). The adjusted OR for a high comorbidity score and receiving home HD was 1.42 (95% CI 0.69 – 2.93).

Discussion

In this study, ESKD patients with a high comorbidity score measured by CCI were less likely to receive home dialysis as compared to CHD. However, when adjusted for confounders including age and BMI, we found no association between comorbidity and dialysis modality choice. In addition, no association was found with diabetes mellitus, ischaemic heart disease, malignancy and cerebrovascular disease. Patients with heart failure were more likely to receive home dialysis, while obese patients with comorbidities were more likely to receive CHD.

The association between comorbidities and PD as home dialysis modality has been investigated in different populations, including in the USA and Europe.^{5, 7-9, 11, 12, 21, 22} Similar results were found in an older European cohort from 1998-2006, in which a high comorbidity score was highly associated with receiving CHD in unadjusted analyses yet almost lost significance in analyses adjusted only for age and sex.⁵ However in their study, patients with malignancy and cerebrovascular disease were less likely to receive PD while patients with diabetes mellitus were more likely to receive PD (adjusted OR 1.09 (1.00 – 1.20)). In contrast, in another study, French patients with diabetes mellitus were more likely to receive CHD and patients with heart failure were more likely to receive PD, both similar to our results.⁸ In studies from the USA, both heart failure and higher comorbidity scores were associated with a lower probability of receiving PD.^{7, 9, 11} These studies however originate from before 2000, when the use of PD was historically low in the USA making comparisons with the current population difficult.^{23, 24} Finally, in a study from Australia and New

Zealand, several comorbidities including diabetes mellitus were associated with a lower probability of receiving home dialysis.²² Overall, these discrepancies among countries indicate that wide variation in selection of home dialysis exists and that comorbidity alone is not a justified contraindication for home dialysis.

In our study, both age and BMI were important confounders in the association between comorbidity and dialysis modality. Thus far, only few other studies corrected for both factors.^{9, 21} In the French study of Couchoud *et al.*, only patients aged \geq 75 years and single comorbidities were evaluated.²¹ They found a positive association between heart failure and home dialysis (adjusted OR 1.8, 95% CI 1.5 – 2.3). The study of Stack *et al.* from USA, also evaluated single comorbidities only and was conducted prior to 2000.⁹ Although few studies correct for age and BMI, increasing age is associated with a lower probability of receiving home dialysis in recent studies⁵, ^{22, 25}, as is obesity.²² The often-reported association between comorbidity and dialysis modality may be largely explained by the confounding effect of age. The same may be true for BMI, as many conditions including cardiovascular disease are initiated by an unhealthy lifestyle.

Age should not be a barrier to receive home dialysis. Although elderly patients frequently have functional limitations and cognitive impairment that may limit the possibilities for self-care²⁶, this does not necessarily rule out a home-based treatment.²⁷ Assisted PD is an important and emerging treatment option for older dialysis patients with similar outcomes to CHD, such as mortality, hospitalisation rates, and health-related quality of life.²⁸⁻³⁰ Moreover, PD provides ultrafiltration more slowly and is not associated with intradialytic hypotension frequently occurring in CHD, which is especially important in frail elderly patients.³¹ Because of the considerable growth in the number of elderly dialysis patients, it is essential to consider home dialysis treatment as a feasible option for elderly patients.

Obese patients were less likely to receive home dialysis treatment in several studies.^{8, 9, 22} It is possible that in obese patients CHD is preferred, due to the survival advantage known as the 'obesity paradox' in obese CHD patients that lacks in PD.³² Another explanation may be that obesity (BMI >30 kg/m²) in PD patients is associated with higher risk of leakage and PD-associated infections.³³⁻³⁶ The latter could be related to the common co-existence of Diabetes Mellitus and lower socioeconomic status in obese patients, or it might be due to obese abominable folds.^{34, 35} But, using extended

catheters or even pre-sternal catheters reduced this risk of infections in several studies.³⁷ Many nephrologists may consider obesity a contraindication for treatment with PD as PD can induce weight gain, but this issue is controversial.³¹ Overall, obesity may not be considered an absolute contraindication for performing PD.

In keeping with findings of previous studies, our study identified that heart failure is associated with a higher probability of receiving home dialysis.^{8, 21} PD is indeed suggested as ultrafiltration treatment in patients with diuretic-resistant heart failure.³⁸ In this seriously ill-group, percutaneous PD catheter insertion under local anaesthesia may be performed by interventional radiologists to avoid general anaesthesia.^{37, 39} Since PD lacks the intradialytic hypotension known in CHD, it is a suitable treatment option in all patients with heart failure.³⁸

Comorbidity alone does not explain the variation in percentage of home dialysis among centres. The present study results suggests that other factors in modality selection are weighted differently among centres. These factors likely include age and BMI, but since these factors were not different between centres with a high or low volume of home dialysis patients – considering a high volume a proxy for a successful home dialysis programme - other factors must also define dialysis modality choice. Indeed, in a French study analysing differences between centres in the use of PD, there was variation in PD use among regions but also huge variation in the evaluation of different patient characteristics.8 The authors thus suggested that other regional practice patterns, such as the organisation of a home dialysis programme, play a role in modality selection. Ethier et al., reporting on the ANZDATA registry and using a mixed model, stated that variation in the use of home dialysis among centres was associated more with centre factors, such as centre size and proportion of patients with a vascular access at dialysis initiation, than patient characteristics.²² Also, logistic and financial factors form barriers for home dialysis and can be weighed differently by individual dialysis centres.⁴⁰⁻⁴³ Further studies are needed to explore these centrespecific factors that might also influence dialysis modality selection.

The strengths of this study include the extensive statistical analyses and the definition of both determinant and outcome. The latter was defined as a start with or transfer to home dialysis during follow-up, reflecting dialysis modality choice. The determinant comorbidity was defined both in validated scores and in single comorbidities providing insight in the association from several points of view, especially heart failure was positively associated with home dialysis. With mixed models, we corrected for centre differences in patient selection which has not often been performed in studies.^{7, 22} However, we had a relatively small sample compared to others.^{5, 8, 9, 21} Although CHD patients were randomly selected, the DOMESTICO study was not designed for the present research question and the population used might not represent a true reflection of modality selection. For this research question, it might have been better to match patients according to their total duration of follow-up. Due to the retrospective design of the study, we were unable to investigate causes of low use of home dialysis, but mere associations instead. Finally, CCI and Davies are developed for mortality predictions and not for dialysis modality choice. These scores might not adequately reflect the impact of comorbidity on dialysis modality choice, especially since the various single comorbidities had associations in different directions.

Notwithstanding these limitations, in this study comorbidity was not significantly associated with home dialysis choice if corrected for age, BMI and centre. Only obese patients with comorbidities were significantly less likely to receive home dialysis. Other factors than comorbidity possibly also influence dialysis modality choice. Differences in prevalence of obesity and age distribution, but probably also centre-specific factors may be related to the variation in the proportion of patients treated with home dialysis among centres. We suggest that future studies should focus on the centre-specific factors that determine dialysis modality selection.

References

- 1. van de Luijtgaarden MW, Jager KJ, Segelmark M, Pascual J, Collart F, Hemke AC, *et al.* Trends in dialysis modality choice and related patient survival in the ERA-EDTA Registry over a 20-year period. Nephrol Dial Transplant. 2016;31(1):120-8.
- ERA-EDTA Registry: ERA-EDTA Registry Annual Report 2018. Amsterdam UMC, location AMC, Department of Medical Informatics, Amsterdam, the Netherlands, 2020. 2018.
- 3. Hoekstra T, Dekker FW, Cransberg K, Bos WJ, van Buren M, Hemmelder MH. RENINE annual report 2018. Available at: https://www.nefrovisie.nl/jaarrapportages/ Accessed: 04-01-2021. 2018.
- 4. Xie Y, Bowe B, Mokdad AH, Xian H, Yan Y, Li T, *et al.* Analysis of the Global Burden of Disease study highlights the global, regional, and national trends of chronic kidney disease epidemiology from 1990 to 2016. Kidney Int. 2018;94(3):567-81.
- van de Luijtgaarden MW, Noordzij M, Stel VS, Ravani P, Jarraya F, Collart F, *et al.* Effects of comorbid and demographic factors on dialysis modality choice and related patient survival in Europe. Nephrol Dial Transplant. 2011;26(9):2940-7.
- Ceretta ML, Noordzij M, Luxardo R, De Meester J, Diez JMA, Finne P, *et al.* Changes in co-morbidity pattern in patients starting renal replacement therapy in Europe data from the ERA-EDTA Registry. Nephrology Dialysis Transplantation. 2018;33(10):1794-804.
- Miskulin DC, Meyer KB, Athienites NV, Martin AA, Terrin N, Marsh JV, et al. Comorbidity and other factors associated with modality selection in incident dialysis patients: the CHOICE Study. Choices for Healthy Outcomes in Caring for End-Stage Renal Disease. Am J Kidney Dis. 2002;39(2):324-36.
- Couchoud C, Savoye E, Frimat L, Ryckelynck JP, Chalem Y, Verger C. Variability in case mix and peritoneal dialysis selection in fifty-nine French districts. Perit Dial Int. 2008;28(5):509-17.
- 9. Stack AG. Determinants of modality selection among incident US dialysis patients: results from a national study. J Am Soc Nephrol. 2002;13(5):1279-87.
- Chanouzas D, Ng KP, Fallouh B, Baharani J. What influences patient choice of treatment modality at the pre-dialysis stage? Nephrology Dialysis Transplantation. 2011;27(4):1542-7.
- Vonesh EF, Snyder JONJ, Foley RN, Collins AJ. The differential impact of risk factors on mortality in hemodialysis and peritoneal dialysis. Kidney International. 2004;66(6):2389-401.
- Jager KJ, Korevaar JC, Dekker FW, Krediet RT, Boeschoten EW, Netherlands Cooperative Study on the Adequacy of Dialysis Study G. The effect of contraindications and patient preference on dialysis modality selection in ESRD patients in The Netherlands. Am J Kidney Dis. 2004;43(5):891-9.
- 13. Mendelssohn DC, Mujais SK, Soroka SD, Brouillette J, Takano T, Barre PE, *et al.* A prospective evaluation of renal replacement therapy modality eligibility. Nephrol Dial Transplant. 2009;24(2):555-61.
- Deyo RAC, D.C.; Ciol, M.A. Adapting a clinical comorbidity index for use with ICD-9-CM administrative databases. J Clin Epidemiol. 1992;45(6):613-9.
- 15. Sharabiani MTA, Aylin P, Bottle A. Systematic Review of Comorbidity Indices for Administrative Data. Medical Care. 2012;50(12).
- Charlson MEP, P; Ales, K.L.; MacKenzie, C.R. A new method of classifying prognostic comorbidity in longitduinal studies: development and validation. J Chron Dis. 1987;40(5):373-83.
- 17. Hoekstra T, van Ittersum FJ, Rabelink AJ, Berger SP, Hemmelder MH. Analyse van kwaliteitsindicatoren Chronische Nierschade [internet]. [cited 2 november 2020] available from: https://www.nefrovisie.nl/richtlijnen-indicatoren/toelichting/ 2013-2016.
- 18. Blazek K, van Zwieten A, Saglimbene V, Teixeira-Pinto A. A practical guide to multiple imputation of missing data in nephrology. Kidney International. 2021;99(1):68-74.
- 19. Davies SJ, Phillips L, Naish PF, Russell GI. Quantifying comorbidity in peritoneal dialysis patients and its relationship to other predictors of survival. Nephrol Dial Transplant. 2002;17(6):1085-92.
- 20. Davies SJ, Russell L, Bryan J, Phillips L, Russell GI. Comorbidity, urea kinetics, and appetite in continuous ambulatory peritoneal dialysis patients: their interrelationship and prediction of survival. Am J Kidney Dis. 1995;26(2):353-61.
- 21. Couchoud C, Moranne O, Frimat L, Labeeuw M, Allot V, Stengel B. Associations between comorbidities, treatment choice and outcome in the elderly with end-stage renal disease. Nephrol Dial Transplant. 2007;22(11):3246-54.
- 22. Ethier I, Cho Y, Hawley C, Pascoe EM, Roberts MA, Semple D, *et al.* Effect of patient- and center-level characteristics on uptake of home dialysis in Australia and New Zealand: a multicenter registry analysis. Nephrol Dial Transplant. 2020.
- 23. Li PK-T, Chow KM, Van de Luijtgaarden MWM, Johnson DW, Jager KJ, Mehrotra R, *et al.* Changes in the worldwide epidemiology of peritoneal dialysis. Nature Reviews Nephrology. 2017;13(2):90-103.
- 24. Blake P. Why is the proportion of patients doing peritoneal dialysis declining in North America? Perit Dial Int. 2001.
- 25. Castledine CI, Gilg JA, Rogers C, Ben-Shlomo Y, Caskey FJ. Renal centre characteristics and physician practice patterns associated with home dialysis use. Nephrol Dial Transplant. 2013;28(8):2169-80.
- Goto NA, van Loon IN, Morpey MI, Verhaar MC, Willems HC, Emmelot-Vonk MH, et al. Geriatric Assessment in Elderly Patients with End-Stage Kidney Disease. Nephron. 2019;141(1):41-8.
- 27. Brown EA, Finkelstein FO, Iyasere OU, Kliger AS. Peritoneal or hemodialysis for the frail elderly patient, the choice of 2 evils? Kidney Int. 2017;91(2):294-303.
- Iyasere O, Brown E, Gordon F, Collinson H, Fielding R, Fluck R, *et al.* Longitudinal Trends in Quality of Life and Physical Function in Frail Older Dialysis Patients: A Comparison of Assisted Peritoneal Dialysis and In-Center Hemodialysis. Perit Dial Int. 2019;39(2):112-8.
- 29. Brown EA, Wilkie M. Assisted Peritoneal Dialysis as an Alternative to In-Center Hemodialysis. Clin J Am Soc Nephrol. 2016;11(9):1522-4.
- Béchade C, Lobbedez T, Ivarsen P, Povlsen JV. Assisted Peritoneal Dialysis for Older People with End-Stage Renal Disease: The French and Danish Experience. Perit Dial Int. 2015;35(6):663-6.
- 31. Eroglu E, Heimburger O, Lindholm B. Peritoneal dialysis patient selection from a comorbidity perspective. Semin Dial. 2020.

- 32. Ladhani M, Craig JC, Irving M, Clayton PA, Wong G. Obesity and the risk of cardiovascular and all-cause mortality in chronic kidney disease: a systematic review and meta-analysis. Nephrol Dial Transplant. 2017;32(3):439-49.
- Leblanc M, Ouimet D, Pichette V. Dialysate leaks in peritoneal dialysis. Semin Dial. 2001;14(1):50-4.
- Nessim SJ, Komenda P, Rigatto C, Verrelli M, Sood MM. Frequency and Microbiology of Peritonitis and Exit-Site Infection among Obese Peritoneal Dialysis Patients. Peritoneal Dialysis International. 2013;33(2):167-74.
- 35. Jegatheesan D, Johnson DW, Cho Y, Pascoe EM, Darssan D, Htay H, *et al.* The Relationship between Body Mass Index and Organism-Specific Peritonitis. Peritoneal Dialysis International. 2018;38(3):206-14.
- Obi Y, Streja E, Mehrotra R, Rivara MB, Rhee CM, Soohoo M, *et al.* Impact of Obesity on Modality Longevity, Residual Kidney Function, Peritonitis, and Survival Among Incident Peritoneal Dialysis Patients. Am J Kidney Dis. 2018;71(6):802-13.
- Crabtree JH, Shrestha BM, Chow KM, Figueiredo AE, Povlsen JV, Wilkie M, et al. Creating and Maintaining Optimal Peritoneal Dialysis Access in the Adult Patient: 2019 Update. Perit Dial Int. 2019;39(5):414-36.
- Puttagunta H, Holt SG. Peritoneal Dialysis for Heart Failure. Perit Dial Int. 2015;35(6):645-9.
- Abdel-Aal AK, Dybbro P, Hathaway P, Guest S, Neuwirth M, Krishnamurthy V. Best practices consensus protocol for peritoneal dialysis catheter placement by interventional radiologists. Perit Dial Int. 2014;34(5):481-93.
- Robinski M, Mau W, Wienke A, Girndt M. The Choice of Renal Replacement Therapy (CORETH) project: dialysis patients' psychosocial characteristics and treatment satisfaction. Nephrol Dial Transplant. 2017;32(2):315-24.
- de Jong RW, Stel VS, Heaf JG, Murphy M, Massy ZA, Jager KJ. Non-medical barriers reported by nephrologists when providing renal replacement therapy or comprehensive conservative management to end-stage kidney disease patients: a systematic review. Nephrol Dial Transplant. 2021;36(5):848-62.
- 42. van de Luijtgaarden MWM, Jager KJ, Stel VS, Kramer A, Cusumano A, Elliott RF, *et al.* Global differences in dialysis modality mix: the role of patient characteristics, macroeconomics and renal service indicators. Nephrology Dialysis Transplantation. 2013;28(5):1264-75.
- 43. Hahn Lundström U, Abrahams AC, Allen J, Altabas K, Béchade C, Burkhalter F, *et al.* Barriers and opportunities to increase PD incidence and prevalence: Lessons from a European Survey. Peritoneal Dialysis International. 2021:08968608211034988.

Supplementary material

Supplementary Table S1. Scoring of the Charlson comorbidity, as proposed by Deyo et al.¹

Conditions	Points for Charlson comorbidity index
Myocardial infarction	1
Heart failure	1
Peripheral vascular disease	1
Cerebrovascular disease	1
Dementia	1
Chronic lung disease	1
Rheumatologic disease	1
Peptic ulcer disease	1
Mild liver disease	1
Diabetes mellitus (without chronic complications)	1
Diabetes mellitus with chronic complications	2
Hemiplegia	2
Any malignancy including lymphoma and leukemia	2
End stage kidney disease	2
Moderate to severe liver disease	3
Metastatic solid tumor	6
AIDS	6

	Logistic mixed model regression analysis*			
	Odds ratio [95% CI] Crude	P-value	Odds ratio [95% CI] Adjusted [#]	P-value
Davies comorbidity score				
Davies 0	REF		REF	
Davies 1-2	0.90 [0.69 – 1.18]	0.45	1.04 [0.78 – 1.39]	0.80
Davies ≥ 3	0.78 [0.55 – 1.12]	0.19	0.96 [0.65 - 1.42]	0.83

Supplementary Table S2. Association of the Davies comorbidity score and treatment with home dialysis, compared with in-centre haemodialysis

* Logistic mixed model regression analysis with dialysis centre as random intercept, with individual patients as first level.

Adjusted for age, sex, BMI, ethnic background, and dialysis vintage

BMI, body mass index.

The Davies comorbidity score was calculated from the presence of the following conditions, with a score ranging from 0 to 7: malignancy, ischaemic heart disease, peripheral vascular disease, left ventricular dysfunction, diabetes mellitus, systemic collagen vascular disease, and other significant pathology / life-threatening disease.^{2,3} The score was divided into three groups according to literature: 0 for patient with absent conditions, 1-2 conditions for intermediate comorbidity, and 3 or more conditions reflecting high comorbidity.²

1. Deyo RAC, D.C.; Ciol, M.A. . Adapting a clinical comorbidity index for use with ICD-9-CM administrative databases. J Clin Epidemiol. 1992;45(6):613-9.

2. Davies SJ, Russell L, Bryan J, Phillips L, Russell GI. Comorbidity, urea kinetics, and appetite in continuous ambulatory peritoneal dialysis patients: their interrelationship and prediction of survival. Am J Kidney Dis. 1995;26(2):353-61.

3. Davies SJ, Phillips L, Naish PF, Russell GI. Quantifying comorbidity in peritoneal dialysis patients and its relationship to other predictors of survival. Nephrol Dial Transplant. 2002;17(6):1085-92.

· · ·		-		
	Logistic mixed model regression analysis*			
	Odds ratio [95% CI] Crude	P-value	Odds ratio [95% CI] Adjusted [#]	P-value
Small centre size < 30 patient n = 535	ts			
Charlson comorbidity index CCI 2 CCI 3-4 CCI ≥ 5	REF 0.89 [0.56 – 1.42] 0.63 [0.37 – 1.07]	0.63 0.09	REF 0.98 [0.61 – 1.59] 0.78 [0.45 – 1.36]	0.95 0.40
Large centre size ≥ 30 patien $n = 823$	ts			
Charlson comorbidity index CCI 2 CCI 3-4	REF 1.01 [0.71 – 1.44]	0.60	REF 1.17 [0.81 – 1.70]	0.40
00123	0.79 [0.54 – 1.16]	0.23	0.93 [0.62 – 1.39]	0./2

Supplementary Table S3 Interaction of centre size in the association between CCI and treatment with home dialysis, compared with in-centre haemodialysis

* Logistic mixed model analysis with dialysis centre as random intercept, with individual patients as first level.

Adjusted for age, sex, BMI, ethnic background, and dialysis vintage

CCI, Charlson comorbidity index; BMI, body mass index.

	Logistic mixed model regression analysis*			
	Odds ratio [95% CI] Crude	P-value	Odds ratio [95% CI] Adjusted [#]	P-value
Younger patients <65 years on $n = 641$	f age			
Charlson comorbidity index CCI 2 CCI 3-4 CCI ≥ 5	REF 0.98 [0.67 – 1.45] 0.70 [0.44 – 1.11]	0.95 0.13	REF 1.08 [0.72 – 1.60] 0.72 [0.45 – 1.18]	0.71 0.19
Older patients ≥ 65 years of a $n = 717$	age			
Charlson comorbidity index CCI 2 CCI 3-4	REF 1.07 [0.69 – 1.65]	0.77	REF 1.08 [0.70 – 1.68]	0.73
$CCI \ge 5$	0.87 [0.56 – 1.37]	0.56	0.95 [0.60 – 1.50]	0.82

Supplementary Table S4 Interaction of age in the association between CCI and treatment with home dialysis, compared with in-centre haemodialysis

* Logistic mixed model analysis with dialysis centre as random intercept, with individual patients as first level.

Adjusted for age, sex, BMI, ethnic background, and dialysis vintage

CCI, Charlson comorbidity index; BMI, body mass index.

Supplementary Table S5 Association of AGE and treatment with home dialysis, compared with in-centre haemodialysis

	Logistic mixed model regression analysis*			
	Odds ratio [95% CI] Crude	P-value	Odds ratio [95% CI] Adjusted#	P-value
Age				
<65 years of age	REF		REF	
\geq 65 years of age	0.76 [0.60 – 0.97]	0.02	0.67 [0.53 – 0.86]	0.002

* Logistic mixed model regression analysis with dialysis centre as random intercept, with individual patients as first level.

Adjusted for sex and ethnic background

Supplementary Table S6 Association of BMI and treatment with home dialysis, compared with in-centre haemodialysis

	Logistic mixed model regression analysis*			
	Odds ratio [95% CI] Crude	P-value	Odds ratio [95% CI] Adjusted [#]	P-value
BMI < 25 kg/m ²	REF		REF	
BMI 25 - 30 kg/m ²	0.81 [0.61 – 1.09]	0.17	0.80 [0.59 – 1.08]	0.14
$BMI \ge 30 \text{ kg/m}^2$	0.63 [0.45 - 0.88]	0.006	0.68 [0.48 – 0.96]	0.03

* Logistic mixed model regression analysis with dialysis centre as random intercept, with individual patients as first level.

Adjusted for age, sex, CCI, ethnic background, and dialysis vintage BMI, body mass index; CCI, Charlson comorbidity index.

Supplementary Table S7. Association of comorbidity and treatment with peritoneal dialysis (n=564), compared to in-centre haemodialysis (n=730)

	Logistic mixed model regression analysis*			
	Odds ratio [95% CI] Crude	P-value	Odds ratio [95% CI] Adjusted [#]	P-value
Charlson comorbidity index	:			
CCI 2	REF		REF	
CCI 3 - 4	0.97 [0.72 – 1.29]	0.82	1.10 [0.81 – 1.51]	0.55
$CCI \ge 5$	0.70 [0.51 – 0.97]	0.03	0.84 [0.59 – 1.20]	0.35
At least 1 comorbidity	0.85 [0.65 – 1.11]	0.22	1.00 [0.75 – 1.34]	0.98
Diabetes Mellitus †	0.77 [0.59 – 0.99]	0.04	0.86 [0.66 – 1.12]	0.27
Ischaemic heart disease	1.05 [0.80 – 1.37]	0.73	1.20 [0.89 – 1.61]	0.23
Heart failure	1.44 [0.98 – 2.11]	0.07	1.59 [1.06 – 2.38]	0.03
Cerebrovascular disease	0.80 [0.57 – 1.14]	0.22	0.83 [0.58 - 1.20]	0.32
Any malignancy	0.82 [0.58 - 1.18]	0.29	0.80 [0.55 – 1.15]	0.23
Chronic lung disease	0.84 [0.57 – 1.22]	0.36	0.86 [0.58 - 1.28]	0.45

* Logistic mixed model regression analysis with dialysis centre as random intercept, with individual patients as first level.

Adjusted for age, sex, BMI, ethnic background, and dialysis vintage

[†] Adjusted for age, sex, ethnic background, and dialysis vintage

CCI, Charlson comorbidity index; BMI, body mass index.

	Multinominal regress	Multinominal regression analysis			
Dialysis modality	Odds ratio [95% CI] Crude	P-value	Odds ratio [95% CI] Adjusted*	P-value	
In-centre haemodialysis	REF		REF		
Peritoneal dialysis					
CCI 3 - 4	0.89 [0.68 – 1.16]	0.39	0.96 [0.73 – 1.27]	0.79	
$CCI \ge 5$	0.74 [0.55 – 0.98]	0.04	0.83 [0.61 – 1.14]	0.26	
Home haemodialysis					
CCI 3 - 4	0.88 [0.47 – 1.65]	0.70	1.14 [0.58 – 2.25]	0.70	
$CCI \ge 5$	1.01 [0.53 – 1.93]	0.98	1.42 [0.69 – 2.93]	0.35	

Supplemental Table S8. Multinomial regression analysis of the association between Charlson comorbidity index and three dialysis modalities

* Adjusted for age, sex, BMI, ethnic background, and dialysis vintage.

CCI, Charlson comorbidity index; BMI, body mass index.



Chapter 4

Key elements in selection of pre-dialysis patients for home dialysis

Anna A. Bonenkamp, Tom D. Y. Reijnders, Anita van Eck der Sluijs, E. Christiaan Hagen, Alferso C. Abrahams, Frans J. van Ittersum and Brigit C. van Jaarsveld.

Peritoneal Dialysis International. 2021 Sep;41(5):494-501.

Abstract

Background Most pre-dialysis patients are medically eligible for home dialysis, and home dialysis has several advantages over incentre dialysis. However, accurately selecting patients for home dialysis appears to be difficult, since uptake of home dialysis remains low. The aim of this study was to investigate which medical or psychosocial elements contribute most to the selection of patients eligible for home dialysis.

Methods All patients from a Dutch teaching hospital, who received treatment modality education and subsequently started dialysis treatment, were included. The pre-dialysis programme consisted of questionnaires for the patient, nephrologist and social worker, followed by an assessment of eligibility for home dialysis by a multidisciplinary team. Clinimetric assessment and logistic regression were used to identify domains and questions associated with home dialysis treatment.

Results A total of 135 patients were included, of whom 40 were treated with home dialysis and 95 with incentre haemodialysis. The key elements associated with long-term home dialysis treatment were part of the domains 'suitability of the housing', 'self-care', 'social support' and 'patient capacity', with adjusted odds ratios ranging from 0.13 for negative to 18.3 for positive associations.

Conclusion The assessment of contraindications by a nephrologist followed by the assessment of possibilities by a social worker or dialysis nurse who investigates four key elements, ideally during a home visit, and subsequent detailed education offered by specialized nurses is an optimal way to select patients for home dialysis.

Introduction

Home dialysis, that is, peritoneal dialysis (PD) or home haemodialysis (HD), offers more flexibility and independence than conventional incentre haemodialysis (CHD), whereas patient survival is comparable or better.¹⁻³ Therefore, it is not surprising that extensive pre-dialysis programmes lead to a preference for home dialysis in 70% of pre-dialysis patients.⁴ Nevertheless, the percentage of patients treated with home dialysis is only about 9-11% throughout the world.^{5,6} An important barrier to uptake of home dialysis is limited pre-dialysis care.⁷⁻¹¹

However, it remains uncertain which elements of pre-dialysis programmes influence a patient's treatment decision. Identifying key elements linked to long-term home dialysis treatment could help various centres to assess eligibility for home dialysis in more patients and to present home dialysis as a viable option among other kidney replacement therapies (KRT). Elements with a negative correlation for home dialysis can be addressed during treatment modality education. Therefore, the aim of this study was to assess which elements of a multidisciplinary structured pre-dialysis programme¹² contribute most to adequate selection of patients eligible for treatment with home dialysis.

Methods

Study population and design

All patients who had completed the pre-dialysis programme, were assessed for eligibility for home dialysis by both the nephrologist and the social worker and started dialysis between June 2013 and August 2018 in a large, non-academic teaching hospital in the Netherlands (Meander Medical Centre, Amersfoort, the Netherlands) were eligible for inclusion into this retrospective study. From 2013, this teaching hospital adopted a 'home first' policy, that resulted in adjusting the pre-dialysis programme. Prior to the implementation of this programme, the home dialysis rate was 18% as was noted in the article about the implementation.¹² All eligible patients had an eGFR of ≤ 15 ml/min/1.73m² or Chronic Kidney Disease stage 4 with rapid deterioration of kidney function prior to referral to the pre-dialysis programme. The programme, that is, eligibility assessment for home dialysis and treatment modality education, was also offered to patients who had an unplanned start of dialysis, that is, acutely started

patients or so-called 'crash landers'. The study was approved by the Medical Research Ethics Committee of the VU University Medical Centre, Amsterdam.

Structured pre-dialysis programme

The structured pre-dialysis programme started with three questionnaires (provided as Supplementary material in Appendix A): for the patient (containing 31 questions), the social worker (20 questions), and the nephrologist (30 questions) of this patient (Figure 1).¹² The questionnaires were developed by Medworg project 'Gezonde Nieren (Healthy Kidneys)', aiming to collect as much relevant information as possible regarding patients and their possibilities for home dialysis. The patient's questionnaire consisted of questions about physical performance, daily activities, and the patient's social support system. The social worker's questionnaire consisted of questions about hygiene, availability of space in the patient's housing, and the capability of the patient and his family to perform dialysis at home. This questionnaire was ideally completed after the social worker performed a home visit. The nephrologist's questionnaire consisted of relative and absolute contra-indications to home dialysis and CHD, e.g. questions about non-compliance, multiple abdominal surgery, and morbid obesity for peritoneal dialysis and impossibility for a vascular access and severe heart failure for (home) HD. Both the social worker and the nephrologist assessed eligibility of all patients for home dialysis, based on the questionnaires. These eligibility assessments were not binding but a mere recommendation, that is, a patient who was judged by the nephrologist as not eligible for home dialysis might initiate home dialysis. The time necessary to fill in a questionnaire was expected to be about 25, 35, and 10 minutes for a patient, social worker and nephrologist respectively.



Figure 1. Overview of the structured pre-dialysis programme NQ, nephrologist's questionnaire; SWQ, social worker's questionnaire; PTQ, patient's questionnaire; MDT, multidisciplinary team meeting; KRT, kidney replacement therapy (including conservative care).

Results of the three questionnaires and home visit were discussed during a Multi-Disciplinary Team (MDT) meeting (Figure 1), in which nephrologists, (pre-)dialysis nurses, and social workers were present. After this meeting the patient received education on all medically feasible treatment modalities by pre-dialysis nurses, including kidney transplantation and conservative care. Depending on the preferences of the patient and the health-care professionals, additional education was provided by transplant nurses and nurses specialized in home dialysis. After completing the education, the definitive choice for KRT was made by the patient, in consultation with his nephrologist. All health-care professionals involved in the pre-dialysis programme were thoroughly informed about the potential benefits of home dialysis at the start of the pre-dialysis programme.

Data collection

Baseline demographic data, eGFR according to chronic kidney disease epidemiology collaboration (CKD-EPI; ml/min/1,73m²) and comorbidities were collected from patients' charts. Comorbidities were scored according to the Charlson comorbidity index.^{13, 14} Social situation and education level were retrieved from the questionnaires. Higher level of education was classified as university or college attendance. Treatment modality was assessed from patients' charts.

Definition of outcome

Treatment modality was defined as the modality, that is, CHD or home dialysis, used at 180 days after dialysis initiation to reflect long-term use. Both (assisted) PD and home HD were considered as home dialysis. The time point of 180 days was chosen to ensure that patients who were eligible and willing to perform PD or home HD, but started for any reason with CHD, were identified as home dialysis patients. For example, patients presenting with acute kidney injury often start with CHD before switching to PD, and treatment with home HD is always preceded by CHD.

Statistical analysis

Continuous variables were reported as means with standard deviation (SD) or as medians with interquartile range (IQR), where appropriate. Categorical variables were presented as proportions. In general, a p-value of <0.05 was considered statistically significant.

Clinimetric properties of all questionnaires were evaluated by three independent researchers: two nephrologist-epidemiologists (FJvI. and BCvJ) and an investigator (AAB). The questions were analysed according to having a formative or reflective nature, and subsequently grouped within separate domains. Logistic regression analysis was performed to investigate the association between long-term home dialysis treatment and the questions within each domain. All questions within a domain with p-value <0.20 in univariable analysis were added to a multivariable model, to correct for correlation. The multivariable model was additionally adjusted for age and Charlson comorbidity index, where appropriate. All analyses were performed using SPSS Statistics 25 (Armonk New York: IBM Corp) or STATA® 14 (Texas: StataCorp LP).

Results

Patient characteristics

A total of 362 patients started the pre-dialysis programme (Figure 2). During the study period, 43 patients died or showed recovery of kidney function. A total of 66 out of 319 patients (21%) opted for conservative treatment. For 72 patients the questionnaires were incomplete. Furthermore, 41 patients had not commenced KRT by the end of

the study period and 71 patients obtained a pre-emptive kidney transplant (n=28, with a kidney transplant rate of 8%) or preferred conservative care (n=43, Figure 2).



Figure 2. Flow chart of pre-dialysis program and subsequent renal replacement therapy CHD, incentre haemodialysis.

The remaining 135 patients were included in the analysis, 85% of whom received a home visit. In 15% of patients, the social worker did not succeed in performing a home visit due to time constraints or because the patient did not consent to a home visit. Table 1 presents the characteristics of all included patients. More CHD patients lived alone and had congestive heart failure compared to home dialysis patients.

	All Patients	Home Dialysis CHD	
	n = 135	n = 40	n = 95
Demographics			
Male sex	85 (63)	22 (55)	63 (62)
Age (years)	66.8 ± 13.8	65.4 ± 14.6	67.5 ± 13.5
Living alone	61 (45)	14 (35)	47 (49)
Higher education	23 (24)	12 (30)	21 (22)
Employment	26 (19)	9 (23)	17 (18)
eGFR at start education*	12.4 ± 5.9	12.2 ± 3.6	12.5 ± 6.7
eGFR at start KRT*	8.2 ± 2.8	8.7 ± 3.0	8.0 ± 2.7
Comorbidities			
Charlson CI	4 [2-5]	4 [3-5]	4 [2-5]
Age-adjusted Charlson CI	6 [4-8]	6 [5-7]	6 [4-8]
Diabetes mellitus	54 (40)	16 (40)	38 (40)
Ischaemic heart disease	33 (24)	11 (28)	22 (23)
Congestive heart failure	12 (9)	1 (3)	11 (12)

Table 1. Characteristi	cs of the	included	patients
------------------------	-----------	----------	----------

Data are shown as n (%), mean ± standard deviation or median with interquartile range [IQR]. CHD, incentre haemodialysis; KRT, kidney replacement therapy; Charlson CI, Charlson Comorbidity Index; IQR, interquartile range.

* eGFR according to CKD-EPI creatinine equation in ml/min/1,73m²

Preference of patients, health-care professionals and treatment decision

Of the 135 included patients, initial preferences at the start of the programme were: 58 patients preferred CHD (43%), 47 PD (35%), 11 home HD (8%), 5 preemptive kidney transplantation (4%), 3 conservative care (2%) and 11 did not have a preference (8%). The social workers considered the overall burden of home dialysis too high in 56 patients. The nephrologists considered previous abdominal surgery (8%), severe obesity (8%), large cystic kidneys (2%) and other reasons (3%) absolute contraindications for PD and no possibility for a vascular access (1%) a contraindication for home HD. These other reasons were intellectual disability, manic-depressive illness, or complete lack of self-sufficiency. For the final decision in the eligibility assessment in 69% of patients. The nephrologist found that 8 patients were possibly ineligible for home dialysis, whereas the social worker found that home dialysis could be an eligible option. In 34 patients, the nephrologist found that home dialysis could be an eligible option, whereas the social worker found these patients ineligible for home dialysis (Supplementary Table S1).

Dialysis treatment was initiated at a median of 5 months [IQR 0-11] after starting the programme, at a mean eGFR of 8.2 ml/min/1,73 m². At 180 days after start of dialysis, 95 patients were treated with CHD (70%), 34 with PD (25%), and 6 patients were treated with home HD (4%). The number of patients treated with home dialysis was comparable between acutely and non-acutely started patients (8 of 35 acutely (23%) vs. 32 of 100 non-acutely started patients (32%), p = 0.31). The rate of home dialysis in the total dialysis population, that is, including patients with incomplete questionnaires, was 29% (52/177), see Supplementary Figure S1.

Characteristics of Questionnaires

Assessment of face validity of the questionnaires, indicated that there was some overlap between the questionnaires regarding medical and housing parameters. Evaluation of the measurement model also indicated that the questions were predominantly formative (as opposed to reflective), meaning that the measured variables are considered to be the cause – and not a reflection - of the latent variable.¹⁵ For example, the question about having enough space for the storage of supplies is a formative question, whereas a question on the consequences of home dialysis, e.g. having more time for education or work, would be a reflective question. In this context this implies that the questions for patient and social worker can be considered formative for, or having a causative relation to, final eligibility for home dialysis. By clinimetric assessment, the questions of the questionnaires for the patient and social worker were classified into the seven domains: work, mental health, patient capacity, physical health, social support, self-care, and suitability of the housing (Figure 3). The questions from the nephrologists' questionnaire addressed assumed contra-indications for home dialysis; these were not considered as a domain but as a professional practice pattern.



Figure 3. Domains within the patient's and social worker's questionnaires, with examples of the most discriminating questions.

Questionnaires: questions associated with home dialysis

The results of the associations between the different questions and long-term home dialysis treatment are depicted in Table 2. All questions that were associated with home dialysis in univariable analysis are shown. From the domain 'suitability of the housing' the general question 'Is the housing suitable for home dialysis' and 'Is there enough space available for dialysis supplies' were associated with home dialysis treatment, with an adjusted odds ratio (OR) of 9.34 (95% confidence interval (CI) 3.01 - 28.96) and 3.27 (95% CI 0.80 - 13.38, P 0.10) respectively. Not having an active lifestyle – domain 'self-care' - was associated with CHD treatment (adjusted OR 0.13, 95% CI 0.04 - 0.42). The question 'Does the patient have a strong social support system' from the corresponding domain was associated with home dialysis

(adjusted OR 4.86, 95% CI 1.87 – 12.60). The question 'Is the patient able to bear the extra workload of home dialysis' was also strongly associated with home dialysis (adjusted OR 18.60, 95% CI 3.11 - 111.21). Other questions did not show relevant associations with home dialysis after adjustment. Thus, the questions indicative of suitable housing, self-care, social support, and-patient capacity were most strongly associated with long-term home dialysis treatment.

Domains	OR (95% CI) crude	p-value	OR (95% CI) adjusted*	p-value
Suitability of the housing				
Owner occupied home	2.06 (0.97 – 4.38)	0.06	1.07 (0.44 – 2.64)	0.88
Does the property have stairs?	1.78 (0.83 – 3.78)	0.14	1.42 (0.59 – 3.42)	0.44
Enough space available for dialysis machine	2.98 (1.37 - 6.49)	0.01	0.34 (0.08 – 1.51)	0.16
Enough space available for dialysis supplies	4.74 (2.03 – 11.04)	<0.001	3.27 (0.80 - 13.38)	0.10
The housing is suitable for home dialysis	9.33 (4.02 – 21.68)	<0.001	9.34 (3.01 – 28.96)	<0.001
Self-care				
Each hour of care by home care agency or caregiver	0.77 (0.61 – 0.97)	0.03	0.88 (0.71 – 1.08)	0.22
Each point on Katz scale#	2.31 (0.79 – 6.78)	0.13	1.14 (0.37 – 3.54)	0.82
The patient doesn't have an active lifestyle	0.10 (0.03 – 0.30)	<0.001	0.13 (0.04 – 0.42)	0.001
Social support				
Is your partner, with whom you live together, in good health?	2.28 (1.06 – 4.89)	0.04	1.76 (0.71 – 4.38)	0.23
Do you have people in the household to help you?	2.62 (0.99 - 6.90)	0.05	2.54 (0.71 – 9.04)	0.15
Does the patient have a strong social support system?	4.62 (1.86 – 11.46)	0.001	4.86 (1.87 – 12.60)	0.001
Physical health				
Do you have trouble breathing?	0.32 (0.09 – 1.17)	0.09	0.31 (0.08 – 1.12)	0.07
Do you have any other diseases?	2.11 (0.74 – 6.04)	0.16	2.66 (0.89 - 7.98)	0.08

Table 2. Association of questions with home dialysis at 180 days
--

Domains	OR (95% CI) crude	p-value	OR (95% CI) adjusted*	p-value
Patient capacity				
Is the patient's understanding of their illness good?	1.38 (1.00 – 1.91)	0.05	0.99 (0.67 – 1.48)	0.96
Is the patient's mental health eligble for home dialysis?	6.87 (1.97 – 23.97)	0.002	0.71 (0.11 – 4.81)	0.73
Are there sufficient financial resources for home dialysis?	2.50 (0.69 – 9.11)	0.17	1.98 (0.47 - 8.39)	0.35
Is the patient able to bear the extra workload of home dialysis?	15.56 (4.49 – 54.01)	<0.001	18.60 (3.11 – 111.21)	0.001
Mental health				
How would you rate your quality of life on a scale from 0 to 10?	1.31 (0.92 – 1.85)	0.12	1.31 (0.92 – 1.85)	0.12

Table 2. Association of questions with home dialysis at 180 days. (continued)

OR, Odds ratio; CI, confidence interval.

*Multivariable models were adjusted for other questions within the same domain; questions from the domains social support and physical health were additionally adjusted for age and Charlson comorbidity index.

[#] This scale elaborates the independency in activities of daily life and ranges from 0 to 6, in which higher scores reflect a more independent patient (Ref. Katz S, JAMA 1963; 185:914-9).

Eligibility assessment by nephrologists and social workers and long-term dialysis treatment

The nephrologists classified 83 patients (61% of all dialysis patients) eligible for home dialysis of whom 37 patients actually were on home dialysis at 180 days, resulting in a positive predictive value (PPV) of 45%. In comparison, the social worker classified 57 patients (42% of all dialysis patients) eligible for home dialysis of whom 35 patients performed home dialysis at 180 days, resulting in a PPV of 61%. Both the nephrologist and the social worker regarded few true home dialysis patients initially ineligible for home dialysis (Supplementary Table S2 & S3).

Discussion

In this study on a pre-dialysis programme, we disentangled the value of different questions and characteristics that are commonly addressed during preparation for

dialysis care. We identified and quantified the value of questions that best predicted uptake of home dialysis following a eligibility assessment. We present 4 key questions on suitable housing, self-care, social support and patient capacity for optimal selection of patients for home dialysis. These elements should also be addressed in subsequent education, especially if the lack of these elements prevents home dialysis to be seen as treatment option.

This study arose from the observation of a significant increase in the proportion of home dialysis patients in a centre that adopted a structured pre-dialysis programme.¹² Within this programme, we sought those elements that had the highest association with-long-term home dialysis treatment. We discovered that a selected set of questions, in combination with information gathered during a home visit, is very efficient for selecting patients for home dialysis during pre-dialysis education. A barrier in the uptake of home dialysis is the feeling of lack of family support.^{16, 17} In our analysis, the simple question 'Does the patient have a strong social support system?' appears to be a good selection question for home dialysis.

Offering adequate treatment modality education is an important process involving multidisciplinary input. It is of utmost importance that patients are provided information on all forms of KRT, including home dialysis, and choose the treatment that suits them best in a process of shared decision making. In clinical practice, negative associations with home dialysis unintendedly expressed by nephrologists or dialysis nurses may guide the patient's decision and form barriers to home dialysis.¹⁸ By identifying elements of a patient's social and physical condition that most clearly distinguish long-term home dialysis treatment, our approach has the potential to increase the efficiency of the pre-dialysis decision process while ensuring a shared decision.

In our study, 80% of patients were medically eligible for home dialysis, compared to 76 – 87% in other studies.^{7,9} The nephrologists in the teaching hospital of this study considered some conditions, e.g. large polycystic kidneys and previous abdominal surgery absolute contra-indications for PD, yet many studies showed that in similar patients PD can be performed with necessary precautions.^{19, 20} In a previous study, it was mentioned that significant variation in eligibility assessments among centres existed.⁷ This practice variation in medical eligibility for home dialysis urges the need for more guidelines on contra-indications for home dialysis, especially in respect of the

increasing number of elderly patients with chronic kidney disease.² Elderly patients are often frail and more frequently have multiple comorbidities and thus might be considered ineligible for home dialysis treatment.²¹ However, PD might be an excellent therapy option in elderly patients with for example hemodynamic instability.²⁰ Frail patients might need the assistance of caregivers or homecare workers, but with options for assisted PD home dialysis is also a feasible option for such patients.²²

In the presented programme, three phases can be distinguished: the collection of information about the patient by nephrologist and the social worker and the general education session. Several studies suggest that a multi-step pre-dialysis programme is associated with a higher percentage of home dialysis patients. Shukla et al. 2017 found that a group education session followed by an individual session led to a steep increase in the number of patients starting home dialysis (38%).⁴ Manns et al. 2005 randomized patients between standard education and an educational intervention including a group education session, combined with standard education. ²³ They reported that patients in the intervention group opted for home dialysis and selfcare haemodialysis significantly more often. Velasco et al. 2015 reported that the multicentre implementation of an education programme, consisting of an education session at home and multiple reflective sessions, resulted in a PD incidence of 48%, as opposed to the national-PD incidence of 15% in Spain.²⁴ Of interest, this was the only study of these 3 articles on multi-step pre-dialysis programmes, that reported the incidence rate of patients choosing conservative care. They reported a rate of 5%, while the incidence was 21% in our study. This underscores that our pre-dialysis programme provides optimal informed decision making on all KRT programs and conservative care.



Figure 4. Proposed sequence in a structured pre-dialysis programme

To provide a practical workflow in pre-dialysis care, based on the experience collected in this study, one could adopt the following sequence of three-phase predialysis programme (Figure 4). In this scenario, the professional knowledge of the nephrologist in assessing which patient *cannot* perform home dialysis for medical reasons is combined with the specific expertise of the social worker or dialysis nurse in determining which patient *can* perform home dialysis. We considered that a home visit was an important addition to this program to satisfactorily assess the suitability of housing for home dialysis. But a home visit might also help to inform and reassure the social system surrounding the patient. In a study evaluating treatment modality education at home, family members that were present demonstrated improved understanding of dialysis and experienced fewer concerns and fears.²⁵

A limitation of this study is that data were collected during the implementation of a new structured pre-dialysis programme in a single centre and that our study analysed the elements of this programme retrospectively. Therefore, we were unable to examine the influence of different professionals on treatment decision nor the added value of a home visit instead of office consultations, whether verbal and non-verbal communication played a role in the treatment decision and whether other related factors correlating with home dialysis not measured in the questionnaire affected the assessment of eligibility (residual confounding). The questionnaires used were not validated for construct validity or reliability. Also, we did not investigate whether it makes a difference which health care professional assesses the four key questions. However, as some questions are best answered during a home visit, we think that the home visit rather than the social worker should play a central role in a pre-dialysis programme. A home visit could not be performed in every patient, but as the home visit was performed in 85% of patients we believe that our conclusion about this part of pre-dialysis care is sufficiently well-founded.

Since certain questions involve a direct judgement by the social worker and are thus dependent on his expertise, a next step would be the validation of the key questions with assessments by other health professionals including dialysis nurses in external cohorts. Future studies might also specifically evaluate the effect of age and frailty on the treatment decision, as the number of elderly patients with chronic kidney disease increases.

The strength of this study includes the long follow-up period. Home dialysis was intentionally defined as a home modality at 180 days after start of dialysis, to enable inclusion of late home dialysis starters. This definition is likely a reflection of the long-term dialysis modality. At 180 days after dialysis initiation as compared to 90 days after dialysis initiation, we were able to select 2 extra home HD patients. In addition, another strength of this study includes using clinimetrics to reduce a large number of questions to 4 key elements that can be easily assessed during a pre-dialysis programme.

In conclusion, if there are no contraindications for home dialysis and a patient prefers this treatment, then a selection process including 4 key questions on suitable housing, self-care, social support, and patient capacity, if possible addressed during a home visit, is an optimal way to assess a patient's eligibility for home dialysis. This strategy helps to do justice to the wish of many patients to be treated, or in fact treat themselves, with a dialysis modality at home.

References

- Dahlerus C, Quinn M, Messersmith E, Lachance L, Subramanian L, Perry E, *et al.* Patient Perspectives on the Choice of Dialysis Modality: Results From the Empowering Patients on Choices for Renal Replacement Therapy (EPOCH-RRT) Study. American Journal of Kidney Diseases. 2016;68(6):901-10.
- 2. van de Luijtgaarden MW, Jager KJ, Segelmark M, Pascual J, Collart F, Hemke AC, *et al.* Trends in dialysis modality choice and related patient survival in the ERA-EDTA Registry over a 20-year period. Nephrol Dial Transplant. 2016;31(1):120-8.
- 3. Walker RC, Hanson CS, Palmer SC, Howard K, Morton RL, Marshall MR, *et al.* Patient and caregiver perspectives on home hemodialysis: a systematic review. Am J Kidney Dis. 2015;65(3):451-63.
- Shukla AM, Easom A, Singh M, Pandey R, Rotaru D, Wen X, et al. Effects of a Comprehensive Predialysis Education Program on the Home Dialysis Therapies: A Retrospective Cohort Study. Perit Dial Int. 2017;37(5):542-7.
- Saran R, Robinson B, Abbott KC, Bragg-Gresham J, Chen X, Gipson D, *et al.* US Renal Data System 2019 Annual Data Report: Epidemiology of Kidney Disease in the United States. Am J Kidney Dis. 2020;75(1 Suppl 1):A6-A7.
- 6. Kramer AP, M.; Noordzij, M.; Stel V.A.; Andrusev, A.M.; *et al.* The European Renal Association European Dialysis and Transplant Association (ERA-EDTA) Registry Annual Report 2016: a summary. Clinical Kidney Journal. 2019.
- Mendelssohn DC, Mujais SK, Soroka SD, Brouillette J, Takano T, Barre PE, et al. A prospective evaluation of renal replacement therapy modality eligibility. Nephrol Dial Transplant. 2009;24(2):555-61.
- 8. McLaughlin K, Manns B, Mortis G, Hons R, Taub K. Why patients with ESRD do not select self-care dialysis as a treatment option. Am J Kidney Dis. 2003;41(2):380-5.
- 9. Mehrotra R, Marsh D, Vonesh E, Peters V, Nissenson A. Patient education and access of ESRD patients to renal replacement therapies beyond in-center hemodialysis. Kidney International. 2005;68(1):378-90.
- Smart NA, Dieberg G, Ladhani M, Titus T. Early referral to specialist nephrology services for preventing the progression to end-stage kidney disease. Cochrane Database Syst Rev. 2014(6):CD007333.
- 11. Walker RC, Howard K, Morton RL, Palmer SC, Marshall MR, Tong A. Patient and caregiver values, beliefs and experiences when considering home dialysis as a treatment option: a semi-structured interview study. Nephrol Dial Transplant. 2016;31(1):133-41.
- de Maar JS, de Groot MA, Luik PT, Mui KW, Hagen EC. GUIDE, a structured predialysis programme that increases the use of home dialysis. Clin Kidney J. 2016;9(6):826-32.
- Deyo RAC, D.C.; Ciol, M.A. Adapting a clinical comorbidity index for use with ICD-9-CM administrative databases. J Clin Epidemiol. 1992;45(6):613-9.
- Charlson MEP, P; Ales, K.L.; MacKenzie, C.R. A new method of classifying prognostic comorbidity in longitduinal studies: development and validation. J Chron Dis. 1987;40(5):373-83.
- 15. de Vet HCW, Terwee CB, Mokkink LB, Knol DL. Measurement in Medicine: A Practical Guide: Cambridge University Press; 2011.

- 16. Zhang AH, Bargman JM, Lok CE, Porter E, Mendez M, Oreopoulos DG, *et al.* Dialysis modality choices among chronic kidney disease patients: identifying the gaps to support patients on home-based therapies. Int Urol Nephrol. 2010;42(3):759-64.
- 17. Morton RL, Tong A, Howard K, Snelling P, Webster AC. The views of patients and carers in treatment decision making for chronic kidney disease: systematic review and thematic synthesis of qualitative studies. BMJ. 2010;340:c112.
- 18. Frongillo M, Feibelmann S, Belkora J, Lee C, Sepucha K. Is there shared decision making when the provider makes a recommendation? Patient Educ Couns. 2013;90(1):69-73.
- 19. Haggerty S, Roth S, Walsh D, Stefanidis D, Price R, Fanelli RD, *et al.* Guidelines for laparoscopic peritoneal dialysis access surgery. Surgical endoscopy. 2014;28(11):3016-45.
- 20. Eroglu E, Heimburger O, Lindholm B. Peritoneal dialysis patient selection from a comorbidity perspective. Semin Dial. 2020.
- Goto NA, van Loon IN, Morpey MI, Verhaar MC, Willems HC, Emmelot-Vonk MH, et al. Geriatric Assessment in Elderly Patients with End-Stage Kidney Disease. Nephron. 2019;141(1):41-8.
- 22. Oliver MJ, Quinn RR, Richardson EP, Kiss AJ, Lamping DL, Manns BJ. Home care assistance and the utilization of peritoneal dialysis. Kidney Int. 2007;71(7):673-8.
- 23. Manns BJ, Taub K, Vanderstraeten C, Jones H, Mills C, Visser M, *et al.* The impact of education on chronic kidney disease patients' plans to initiate dialysis with self-care dialysis: a randomized trial. Kidney Int. 2005;68(4):1777-83.
- 24. Prieto-Velasco M, Quiros P, Remon C, Spanish Group for the Implementation of a Shared Decision Making Process for RRTCwPDAT. The Concordance between Patients' Renal Replacement Therapy Choice and Definitive Modality: Is It a Utopia? PLoS One. 2015;10(10):e0138811.
- 25. Ismail SY, Luchtenburg AE, Timman R, Zuidema WC, Boonstra C, Weimar W, *et al.* Home-based family intervention increases knowledge, communication and living donation rates: a randomized controlled trial. Am J Transplant. 2014;14(8):1862-9.

Supplemental material

Appendix A1. Questionnaire Patient

Questions	Answer options
Are you in a relationship?	Y/N
Do you live together with your partner?	Y/N
Is your partner in good health, or is he/she limited in the activities that he/she can do?	Good health/limited
Do you have children?	Y/N
How many children do you have	number
Do you live together with your children?	Y/N
How many children still live at home?	number
Do you have siblings?	Y/N
Do you receive home care service?	Y/N
How many hours a week do you receive home care service?	number
Do you receive informal care from a caregiver?	Y/N
How many hours a week do you receive informal care from a caregiver?	number
Do you have people in the household to help you?	Y/N
Are you a homeowner?	Y/N
What type of home do you live in?	house, apartment/flat, retirement condo, other
If you live in an apartment, is there an elevator in the building?	Y/N
Does your property have stairs?	Y/N
What is your level of education?	primary education, secondary education, further education, higher education
Are you currently employed?	Y/N
How many hours do you work a week?	number
How would you rate your quality of life?	Scale 1-10
How would you rate your current health?	Scale 1-10
Do you have any other conditions/diseases besides your kidney failure?	Y/N
Do you feel like you have sufficient energy to do the things you want to do throughout the day?	Y/N
Do you have trouble breathing?	Y/N

Appendix A1.	Question	naire Patient	(continued)
--------------	----------	---------------	-------------

Questions	Answer options
The following questions belong to the KATZ-scale	Scale 0-6
BATHING	Independence/dependence
DRESSING	Independence/dependence
TOILETING	Independence/dependence
TRANSFERRING	Independence/dependence
CONTINENCE	Independence/dependence
FEEDING	Independence/dependence

Appendix A2. Questionnaire social worker

Questions	Answer options
Does the residence have enough space available for a dialysis machine/ equipment? (2 square meters)	Y/N
Does the residence have enough space available for the necessary (dialysis) supplies/materials? (1 cubic meter)	Y/N
Does the residence contain a residual current device?	Y/N
Are water supply and drainage systems present on the floor where dialysis may take place?	Y/N
Are the hygienic conditions in the residence sufficient for home dialysis?	Y/N
Have prior adaptations been made to the residence to aid in activities of daily living?	Y/N
Does the patient have an internet connection?	Y/N
Is the patient's residential housing suitable for home dialysis?	Y/N
What means of transport does the patient use for traveling?	Patient drives car, public transport, patient transport
Does the patient have an active lifestyle (do his/her daily activities make successful home dialysis likely)?	Y/N
Is the patient's understanding of their illness good?	Y/N
Does the patient have auditory or visual impairments that would hinder home dialysis?	Y/N
Does the patient's current physical health state allow for home dialysis?	Y/N
Have any major life events occurred in the patient's life in the past year?	Y/N
Does the patient's current mental health state allow for home dialysis?	Y/N
Are there sufficient financial resources (monetary or material) for home dialysis?	Y/N
Is the patient able to carry the burden of home dialysis?	Y/N
Can the patient's partner bear the extra burden of home dialysis?	Y/N

Questions	Answer options
Does the patient have a strong social support system, i.e. is the patient's social network supportive for home dialysis?	Y/N
Is the patient able to bear the extra workload of home dialysis, i.e. do the patient's personal and social circumstances allow for home dialysis?	Y/N

Appendix A2. Questionnaire social worker (continued)

Appendix A3. Questionnaire nephrologist

Questions	Answer options
Does the patient want to be eligible for kidney replacement therapy?	Y/N
Is renal transplantation an option for this patient?	Y/N
Did the patient undergo abdominal surgery?	Y/N
Does the type of abdominal surgery constitute an absolute contraindication for peritoneal dialysis?	Y/N
Does the patient have an abdominal wall defect?	Y/N
Is it possible to correct the abdominal wall defect?	Y/N
Is the patient obese (BMI >30 kg/m2)?	Y/N
Is the patient morbidly obese (BMI >35 kg/m2)?	Y/N
Does the patient's weight constitute an absolute contraindication for peritoneal dialysis?	Y/N
Does the patient have severe Chronic Obstructive Pulmonary Disease (GOLD 3/4)?	Y/N
Does the patient's COPD (GOLD 3/4) constitute an absolute contraindication for peritoneal dialysis?	Y/N
Is there an active inflammatory process in the abdomen?	Y/N
Does this inflammatory process constitute an absolute contraindication for peritoneal dialysis?	Y/N
Does the patient have polycystic kidney disease?	Y/N
Are the polycystic kidneys large?	Y/N
Do the kidney cysts constitute an absolute contraindication for peritoneal dialysis?	Y/N
Are there other relative contraindications for peritoneal dialysis?	Y/N
Are there other absolute contraindications for peritoneal dialysis?	Y/N
Does the patient have heart failure?	Y/N
What is New York Heart Association (NYHA) Functional Classification of the patient's heart failure?	Scale 1 - 4
Is the patient's heart failure a reason to recommend peritoneal dialysis?	Y/N
Is it possible to place a AV fistula or graft?	Y/N

Appendix A3. Questionnaire nephrologist (continued)

Questions	Answer options
Is vascular access possible?	Y/N
Are there other relative contraindications for haemodialysis?	Y/N
Are there other absolute contraindications for haemodialysis?	Y/N
Does the patient exhibit non-cooperative behaviour that precludes home dialysis?	Y/N
Are there other relative contraindications for home dialysis?	Y/N
Are there other absolute contraindications for home dialysis?	Y/N
Are there other relative contraindications for in-centre haemodialysis?	Y/N
Are there other absolute contraindications for in-centre haemodialysis?	Y/N



Supplementary Figure 1. Flow-chart of the total number of patients that started dialysis following the structured pre-dialysis program KRT, kidney replacement therapy

	Social worker: eligible for home dialysis	Social worker: ineligible for home dialysis	TOTAL
Nephrologist: eligible for home dialysis	49	34	83
Nephrologist: ineligible for home dialysis	8	44	52
TOTAL	57	78	135

Supplemental Table 1. Final eligibility assessment of both nephrologist and social worker

Supplemental Table 2. Eligibility assessment by the nephrologist compared to treatment at 180 days

	Treatment Home	Treatment CHD	TOTAL
Nephrologist: eligible for home dialysis	37	46	83
Nephrologist: ineligible for home dialysis	3	49	52
TOTAL	40	95	135

CHD, in-centre haemodialysis.

Supplemental Table 3. Eligibility assessment by the social worker compared to treatment at 180 days

	Treatment Home	Treatment CHD	TOTAL
Social worker: eligible for home dialysis	35	22	57
Social worker: ineligible for home dialysis	5	73	78
TOTAL	40	95	135

CHD, in-centre haemodialysis.



Part II

Enhancing technique survival of home dialysis



Chapter 5

Differences in hospitalization between peritoneal dialysis and haemodialysis patients

Anita van Eck van der Sluijs, Anna A. Bonenkamp, Vera A. van Wallene, Tiny Hoekstra, Birgit I. Lissenberg-Witte, Friedo W. Dekker, Frans J. van Ittersum, Marianne C. Verhaar, Brigit C. van Jaarsveld and Alferso C. Abrahams on behalf of the DOMESTICO study group

> European Journal of Clinical Investigation. 2022 Feb 7:e13758. (Epub ahead of print)
Abstract

Background Dialysis is associated with frequent hospitalizations. Studies comparing hospitalizations between peritoneal dialysis (PD) and hemodialysis (HD) report conflicting results and mostly analyze data of patients that remain on their initial dialysis modality. This cohort study compares hospitalizations between PD and HD patients taking into account transitions between modalities.

Methods The retrospective Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes collected hospitalization data of patients who started dialysis between 2012 and 2017. Primary outcome was hospitalization rate, analyzed with a multi-state model that attributed each hospitalization to the current dialysis modality. Secondary outcomes were risk for first hospitalization, number of hospitalizations, number of hospital days per patient-year, and causes of hospitalizations.

Results In total, 695 patients (252 PD and 443 HD at 3 months) treated in 31 Dutch hospitals were included. The crude hospitalization rate for PD was 2.3 (\pm 5.0) and for HD 1.4 (\pm 3.2) hospitalizations per patient-year. The adjusted hazard ratio for hospitalization rate was 1.1 (95%CI 1.02-1.3) for PD compared to HD. The risk for first hospitalization was 1.3 times (95%CI 1.1-1.6) higher for PD compared to HD during the first year after dialysis initiation. The number of hospitalizations and number of hospital days per patient-year were significantly higher for PD. The most common causes of PD and HD hospitalizations were peritonitis (23%) and vascular access-related problems (33%), respectively.

Conclusion PD was associated with higher hospitalization rate, higher risk for first hospitalization, and higher number of hospitalizations compared to HD.

Introduction

Dialysis treatment for end-stage kidney disease (ESKD) is associated with high morbidity, frequently resulting in hospitalization.¹⁻⁴ The hospitalization rate of dialysis patients varies between 1.2 - 1.7 per patient-year, compared to 0.8 per patient-year for patients with a kidney transplant.^{2, 5} Dialysis patients also have a higher risk of readmission, with a hazard ratio of 1.8 for readmission within one year compared to a control group of patients without kidney disease.^{2, 6} Infections and cardiovascular diseases are the leading causes for hospitalization in dialysis patients.^{2, 7, 8}

Hospitalization is an indirect measure of morbidity in dialysis patients, as well as a risk factor for mortality.^{6, 9} Also, hospitalization negatively affects the quality of life and increases the costs of dialysis.^{7, 10, 11} Hospitalization costs are one of the most expensive elements of dialysis treatment.¹⁰⁻¹² Therefore, prevention of hospitalization of dialysis patients is of utmost importance.

Differences in hospitalization between peritoneal dialysis (PD) and haemodialysis (HD) patients have been the subject of previous studies. However, there are several problems with these studies. First, they report conflicting results with studies describing an equal number and duration of hospital admissions for PD patients compared to HD patients¹³⁻¹⁶, while other studies conclude that PD patients are more likely to be hospitalized.^{3, 5, 17-21} Second, most studies do not take into account the time on dialysis, which also seems to affect hospitalization rates. The hospitalization rate for HD patients is highest during their first year of dialysis with a decrease thereafter, while PD patients experience an increase in hospitalization rate as their dialysis duration progresses, according to the 2018 report from the United States Renal Data System (USRDS).² Finally, and most importantly, most studies only analyse data from patients who remain on their initial dialysis modality or do not take transitions between dialysis modalities into account.^{3, 13-15, 18, 19, 21} However, a transition from one dialysis modality to another, for example from PD to HD, occurs frequently in daily practice. Analysing only the data of patients who continue their original dialysis modality introduces selection bias in the results reported. Therefore, the aim of this study was to compare hospitalizations between incident PD and HD patients taking into account transitions between dialysis modalities and time on dialysis.

Methods

Study population

The Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO) is a multi-centre cohort study among dialysis patients in the Netherlands. For this analysis, retrospectively collected hospitalization data from a cohort of patients from 31 hospitals were used. Eligible patients were adults (\geq 18 years) who started dialysis treatment (i.e. PD or HD) between January 1, 2012 and January 1, 2017 with a minimum dialysis treatment duration of 3 months. Patients were allowed to have had previous kidney replacement therapy in the form of (dialysis followed by) kidney transplantation. Follow-up of patients was conducted until after kidney transplantation, a patient's wish to stop dialysis, death, or the end of the study period on January 1, 2017. The study was approved by local medical ethics committees of the participating dialysis centres.

Baseline characteristics

Baseline characteristics were collected at dialysis initiation. For the baseline data, patients were grouped according to their dialysis modality (i.e. PD or HD) at 3 months after dialysis initiation. Primary kidney disease was classified according to the European Renal Association – European Dialysis and Transplant Association (ERA-EDTA) codes and categorized into: glomerulonephritis/pyelonephritis, cystic kidney disease, renovascular kidney disease, diabetes mellitus, and other/unknown.²² Comorbidities were classified according to both the Charlson Comorbidity Index (CCI) and the Davies score.^{23, 24} Kidney replacement therapy vintage and dialysis vintage were presented as the months that patients received kidney replacement therapy (i.e. kidney transplantation and dialysis combined) or dialysis alone in the past. Residual glomerular filtration rate was calculated as the creatinine clearance (ml/min), using creatinine measurements in blood and 24 hours urine collections. Patients were indicated as acute starters if they had never been under outpatient monitoring by a nephrologist prior to initiation of dialysis.

Hospitalization

Hospitalization was defined as a hospital admission with a minimum duration of 24 hours. The start and end dates of each hospitalization were recorded along with the reason using ICD-10 codes.²⁵ The primary outcome was hospitalization rate, which was defined as the number of hospitalizations per patient-year. Patient-years were

defined as the number of years a patient performed a dialysis modality within the study period.

Secondary outcomes were risk for first hospitalization, total number of hospitalizations per patient, number of hospital days per patient-year, and causes of hospitalization. Causes of hospitalization were grouped into the following categories: access-related (including vascular access infection, fistula operation, and PD catheter leakage, exchange or removal), peritonitis, fluid overload, cardiac disease (including myocardial ischemia or infarction, cardiac arrest or arrhythmia, cardiac failure, and hemorrhagic pericarditis), vascular disease (including pulmonary embolus, stroke, cerebrovascular hemorrhage, ruptured vascular aneurysm, mesenteric infarction, and peripheral vascular disease), non-dialysis related infection, gastrointestinal disease (excluding PD peritonitis), malignancy, transplantation, and other/unknown.

Statistical analysis

Baseline characteristics were presented as mean with standard deviation (SD), median with interquartile range (IQR) or as number with percentages. Groups were compared with a Chi-square test, an independent samples t-test or Mann–Whitney U test, where appropriate.

Since patients can transition between dialysis modalities over time (i.e. PD patients transition to HD or HD patients transition to PD), all analyses were performed with models that allow for such transitions. Hospitalization rate was analysed with a multi-state model with recurrent events, which attributed every hospitalization to the dialysis modality the patient performed at the time of admission. Patients who died were censored. The results of this model are presented with hazard ratios (HR).

The risk for first hospitalization was analysed with a Cox regression model with dialysis modality as time varying covariate. The proportional hazards assumption was tested and if it was violated, data were presented for two different time periods. Number of hospitalizations and number of hospital days per patient-year were analysed with negative binomial regression. The last two outcomes were analysed in a multilevel model, in which dialysis modality was the first level and the patient the second level. This analysis thus corrected for the dependency of both dialysis modalities within the same patient.

All analyses were adjusted for potential confounders. In the first model, adjustments were made for age and sex, in a second model data were also adjusted for CCI, dialysis vintage, and acute start of dialysis. Statistical analyses were conducted with IBM SPSS Statistics version 25 and R version 3.6.1.

Results

Baseline characteristics

The study cohort consisted of 695 dialysis patients, of whom 252 (36%) were receiving PD and 443 (64%) HD at 3 months after dialysis initiation. Baseline characteristics are presented in Table 1. Mean age was 63.0 (\pm 15.3) years for both groups, and the majority of patients were male. The comorbidity scores were similar between PD and HD patients. PD patients had a dialysis vintage of 16 months [IQR 9 – 41], whereas HD patients had a significantly longer dialysis vintage of 39 months [IQR 19 – 64]. PD patients less often had a previous kidney transplant compared to HD patients, 10% and 25% respectively (p<0.001). Only 4% of the PD patients had an acute start of dialysis, whereas 20% of HD patients did (p<0.001). Just over half of the patients performed PD themselves; the rest were assisted by a nurse or other caregiver at home.

Variable	Full sample n=695	PD n=252	HD n=443
Age (yr), mean ± SD	63.0 ± 15.3	63.1 ± 14.9	62.9 ± 15.6
Sex (male), n (%)	418 (60)	160 (64)	258 (58)
Ethnic background, n (%)			
Caucasian	395 (57)	149 (59)	246 (56)
Other	123 (18)	30 (12)	93 (21)
Unknown	177 (25)	73 (29)	104 (23)
Primary kidney disease, n (%)			
Glomerulonephritis/pyelonephritis	141 (20)	39 (16)	102 (23)
Cystic kidney disease	38 (6)	19 (8)	19 (4)
Renovascular kidney disease	193 (28)	71 (28)	122 (28)
Diabetes mellitus	119 (17)	49 (19)	70 (16)
Other/unknown	204 (29)	74 (29)	130 (29)
BMI (kg/m ²), mean \pm SD	26.8 ± 5.5	26.6 ± 4.7	26.9 ± 6.0

Table 1. Baseline characteristics according to dialysis modality at 3 months.

Variable	Full sample	PD n=2.52	HD n=443
Smoking, n (%)	n-075	n - 2 5 2	H =115
Yes	117 (17)	42 (17)	75 (17)
Quit	172 (25)	67 (27)	105 (24)
Unknown	103 (15)	36 (14)	67 (15)
CCI score, n (%) ^a	× /	~ /	~ /
2	208 (30)	84 (33)	124 (28)
3 – 4	281 (41)	97 (39)	184 (42)
≥ 5	204 (29)	71 (28)	133 (30)
Davies score, n (%)			
0	182 (26)	77 (31)	105 (24)
1 – 2	370 (53)	125 (50)	245 (56)
≥ 3	141 (20)	50 (20)	91 (21)
KRT vintage (months), median [IQR] ^b	150 [64-212]	138 [44-181]	154 [69-230]
Dialysis vintage (months), median [IQR] ^c	35 [15-58]	16 [9-41]	39 [19-64]
Previous transplant, n (%)	138 (20)	26 (10)	112 (25)
Residual GFR (ml/min), median [IQR]	7.8 [4.6-11.6]	9.5 [6.7-12.9]	6.6 [3.3-10.4]
Residual diuresis (ml/day), mean ± SD	1459 ± 841	1708 ± 743	1317 ± 862
Acute start of dialysis, n (%)	98 (14)	11 (4)	87 (20)

Table 1. Baseline characteristics according to dialysis modality at 3 months. (continued)

PD= peritoneal dialysis; HD= haemodialysis; SD=standard deviation; CCI= Charlson comorbidity index; KRT= kidney replacement therapy; IQR=interquartile range; GFR= glomerular filtration rate.

* Groups were compared with a Chi-square test, an independent samples t-test or Mann–Whitney U test, where appropriate.

a. By definition, dialysis patients have a minimum CCI score of 2.

b. KRT vintage was only calculated for the 159 patients (23%) who received previous kidney replacement therapy: 33 PD patients (13%) and 126 HD patients (28%)

c. Previous dialysis treatment was only calculated for the 148 patients (21%) who received dialysis before inclusion: 30 PD patients (12%) and 118 HD patients (27%)

Dialysis treatment and follow-up

The median dialysis duration for the entire study cohort was 22.0 months [IQR 11.1 – 36.4]. PD patients had a shorter dialysis duration [19.1 months, IQR 10.4 – 30.5] than HD patients [23.6 months, IQR 11.7 – 38.6] (p=0.001). Patients transitioned more often from PD to HD (33%), than from HD to PD (11%) (p<0.001).

Hospitalization rate

A total of 521 hospitalizations took place during PD, while 959 hospitalizations took place during HD. The crude hospitalization rate for PD was 2.3 (\pm 5.0) hospitalizations per patient-year and for HD 1.4 (\pm 3.2) hospitalizations per patient-year. Using a multi-state model, the adjusted HR for hospitalization rate was 1.1 (95% confidence interval (CI) 1.02 – 1.3) for PD compared to HD patients (Table 2).

Table 2.	Comparison	of hospitalization	rate	(hospitalizations	per	patient-year)	and	risk	for
first hosp	oitalization.								

Dialysis modality	Crude HR (95	5% CI)	Adjusto HR (95	ed* 5% CI)	Adjuste HR (95	ed** 5% CI)
Hospitalizations per	patient-y	year				
PD vs HD	1.1	(1.03 - 1.3)	1.1	(1.02 - 1.3)	1.1	(1.02 - 1.3)
Risk for first hospital	ization d	luring first year	r after di	alysis initiation	ı	
PD vs HD	1.3	(1.1 - 1.6)	1.3	(1.1 - 1.6)	1.3	(1.1 - 1.6)
Risk for first hospital	lization 2	≥ 1 year after di	ialysis in	itiation		
PD vs HD	1.8	(1.4 - 2.5)	1.8	(1.4 - 2.5)	1.9	(1.4 - 2.5)

HR= hazard ratio; PD= peritoneal dialysis; HD= haemodialysis. The hospitalization rate was calculated with a multi-state model with recurrent events, which attributed every hospitalization to the dialysis modality the patient performed at the time of admission. The risk for first hospitalization was analysed with a Cox regression model with dialysis modality as time varying covariate.

* Adjusted for age and sex

** Adjusted for age, sex, Charlson Comorbidity Index, dialysis vintage, and acute start of dialysis

Risk for first hospitalization, number of hospitalizations, and number of hospital days per patient-year

Figure 1 shows the estimated cumulative incidence curves for the first hospitalization for PD and HD patients according to the Cox regression model. The model was adjusted for age, sex, CCI, dialysis vintage, and acute start of dialysis.



Figure 1. Risk for first hospitalization for PD and HD patients.

Estimated cumulative incidence curves for first hospitalization for PD and HD patients derived from a multi-state Cox regression model. Model is adjusted for age, sex, Charlson Comorbidity Index, dialysis vintage, and acute start of dialysis. PD= peritoneal dialysis; HD= haemodialysis.

Because the proportional hazards assumption was violated, HRs for risk for first hospitalization were calculated separately for the first year after dialysis initiation and for the period thereafter, conditional on having survived the first year. The adjusted HR for risk for first hospitalization during the first year was 1.3 (95% CI 1.1 – 1.6) for PD versus HD. For the period thereafter, the adjusted HR was 1.9 (95% CI 1.4 – 2.5) (Table 2).

The number of PD hospitalizations, corrected for the total PD duration, was significantly higher than the number of HD hospitalizations, corrected for the total HD duration (crude incidence rate ratio of PD relative to HD 1.3; 95% CI 1.1 – 1.6). Additional adjustments for age, sex, CCI, dialysis vintage, and acute start of dialysis resulted in a further increase in incidence rate ratio to 1.7 (95% CI 1.2 – 2.3) (Table 3).

The crude median number of hospital days per patient-year was 4.2 for PD patients [IQR 0 – 15.3] and 0.8 for HD patients [IQR 0 – 10.8]. The adjusted incidence rate ratio for number of hospital days per patient-year was 1.5 (95% CI 1.2 – 2.1) for PD compared to HD (Table 3).

 Table 3. Comparison of number of hospitalizations and number of hospital days per patientyear.

Dialysis modality	Crude IRR (95	% CI)	Adjuste IRR (95	d* 5% CI)	Adjuste IRR (95	d** 5% CI)
Number of hospitaliz	ations					
PD vs HD	1.3	(1.1-1.6)	1.7	(1.3-2.3)	1.7	(1.2-2.3)
Number of hospital d	ays per p	atient-year				
PD vs HD	1.6	(1.2-2.1)	1.6	(1.2-2.1)	1.5	(1.2-2.1)

IRR= incidence rate ratio of PD relative to HD; PD= peritoneal dialysis; HD= haemodialysis. * Adjusted for age and sex

** Adjusted for age, sex, Charlson Comorbidity Index, dialysis vintage, and acute start of dialysis

Causes

Causes of hospitalizations are presented in Table 4. The main cause for hospitalizations during PD treatment was peritonitis (23%), while the second most common cause were non-dialysis related infections (15%). The main cause for hospitalization during HD treatment was a vascular access-related reason (33%), such as a fistula operation or a dialysis access infection. The second most common cause for hospitalization during HD treatment were non-dialysis related infections (18%). For both PD and HD, hospitalizations for fluid overload were rare (2 - 3%).

Causes	PD n=521	HD n=959
Access-related ^a	69 (13)	317 (33)
Peritonitis	117 (23)	N/A
Fluid overload	14 (3)	22 (2)
Cardiac disease ^b	57 (11)	87 (9)
Vascular disease ^c	28 (5)	50 (5)
Infection ^d	79 (15)	170 (18)
Gastrointestinal disease	46 (9)	94 (10)
Malignancy	9 (2)	25 (3)
Transplantation	13 (2)	25 (2)
Other / unknown	89 (17)	169 (18)

Table 4. Causes of hospitalizations.

Data are presented as n (%). PD= peritoneal dialysis; HD= haemodialysis; N/A= not applicable. Access-related includes vascular access infection, fistula operation, PD catheter leakage/ exchange/removal.

Cardiac disease includes myocardial ischaemia/infarction, cardiac arrest/arrhythmia, cardiac failure, haemorrhagic pericarditis.

Vascular disease includes pulmonary embolus, stroke, cerebrovascular haemorrhage, ruptured vascular aneurysm, mesenteric infarction, peripheral arterial disease.

a. Non-dialysis related infections.

Discussion

In this retrospective cohort study among 695 dialysis patients, PD treatment was associated with a higher hospitalization rate, a higher risk for first hospitalization, a higher number of hospitalizations and a higher number of hospital days per patientyear compared to HD treatment, when hospitalizations were attributed to the dialysis modality the patient was receiving upon admission. In addition, PD hospitalizations were mainly caused by peritonitis, while vascular access-related reasons were the main causes for HD hospitalizations.

A higher PD hospitalization rate compared to HD is found in several other studies. Banshodani *et al.* retrospectively showed that emergency hospitalization rates for cardiovascular diseases and infectious diseases were significantly higher for 130 PD patients compared to 130 HD patients, with HRs of 2.70 (95% CI 1.53 – 4.77) and 4.16 (95% CI 2.59 – 6.68), respectively.^{3,21} Lafrance *et al.* also retrospectively showed

that infection-related hospitalization rates were significantly higher for PD patients compared to HD patients (HR 1.52, 95% CI 1.38 – 1.68).¹⁸ Besides the fact that Banshodani *et al.* had a smaller study population than our study and Lafrance *et al.* investigated younger patients (HD 58.5 ± 16.4 years and PD 58.8 ± 14.5 years) during the period 2001 to 2007, both studies did not take transitions in dialysis modality into account. Banshodani *et al.* censored all patients who changed dialysis modality and Lafrance *et al.* attributed all hospitalizations of patients according to their dialysis modality at 90 days.^{3, 18, 21} These studies defined patients according to a single dialysis modality, which does not do justice to daily practice at all.

That it is important to take transitions from and to different dialysis modalities into account is also shown in a study by Murphy et al.¹⁷ In their prospective Canadian cohort, they showed that PD patients had a lower hospitalization rate (defined as the total number of hospitalization days relative to the survival of the patient) compared to HD patients (rate ratio 0.85, 95% CI 0.82 - 0.87) when hospitalizations were attributed to the dialysis modality at baseline, while they had a higher hospitalization rate (rate ratio 1.31, 95% CI 1.27 - 1.34) when hospitalizations were attributed to the dialysis modality at 3 months.¹⁷ In addition, Murphy et al. performed an analysis in which hospitalizations were attributed to the dialysis modality the patient was receiving upon admission, which showed that PD treatment was associated with a higher hospitalization rate than HD treatment, with a rate ratio of 1.10 (95% CI 1.07 - 1.13).¹⁷ This study advocated the use of treatment-received analyses in comparing hospitalization rates, which we did, instead of intention-to-treat analyses. However, our study defined hospitalization rate as the number of hospitalizations per patientyear, which is much more commonly used in studies, also investigated the risk for first hospitalization, and described a more recent study population.

In two Canadian cohorts, Quinn *et al.* and Oliver *et al.* used the number of hospitalization days per patient year for calculating their hospitalization rates. In their analyses with dialysis as time-varying covariate, they showed equal hospitalization rates for PD compared to HD (Quinn *et al.*: rate ratio 1.28, 95% CI 0.63 – 2.61. Oliver *et al.*: rate ratio 0.93, 95% CI 0.51 – 1.71).^{8, 16} However, besides the fact that they used a different measure for hospitalization rate, which makes comparison with our study difficult, they did not investigate the risk for first hospitalization, and Oliver *et al.* only investigated patients on assisted PD. Several other studies showed that hospitalization rates of PD and HD patients are equal.^{13-15, 19, 26} However, these studies

performed an intention-to-treat analysis by attributing hospitalizations of patients to their initial dialysis modality, which is not a valid analysis for the present research question, as argued above.

In our study, the main cause of PD hospitalizations was peritonitis, while HD hospitalizations were mainly vascular access-related. Also in a Japanese survey among 89,748 patients, these were most common causes for PD and HD hospitalizations.²⁰ Several other studies have identified infections and specifically peritonitis as an important cause for PD hospitalizations.^{16, 18, 21, 27}

Apparently, PD patients have a higher risk for hospitalization than HD patients. This could be attributed to the dialysis modality per se, or could be the result of circumstantial factors. A possible explanation could be that the threshold for hospitalization is lower for PD than for HD patients. HD patients frequently visit the hospital for dialysis, in most cases at least three times a week for four hours. If, for example, they develop an infection, assessment and (start of) antibiotic treatment can easily be performed during the dialysis session in hospital. Moreover, the effect of the antibiotic treatment can be evaluated during the next scheduled dialysis session and adapted based on culture results. On the other hand, PD patients are treated at home and visit the hospital much less frequently. If they develop an infection, they must visit the hospital for evaluation. In addition, they have to attend the hospital again for evaluation of the treatment effect. It is conceivable that this need for frequent hospital visits could lead to a lower threshold for hospitalization in PD patients. Finally, we cannot exclude residual confounding as possible or additional explanation for finding a higher hospitalization risk in PD compared to HD.

To our knowledge, this is the first European study to describe several important hospitalization outcomes of PD and HD, taking into account transitions between dialysis modalities and thus properly showing the risk for hospitalization of the different dialysis modalities. Almost one-fifth of our population changed dialysis modality, underscoring that a model allowing this is superior to models evaluating hospitalizations on an intention-to-treat basis. Besides the fact that we used a multistate model in a relatively large cohort of patients, we also describe a recent dialysis population, which is relevant because the composition of the dialysis population has changed in previous years, for example with respect to age.^{28, 29} However, our study has some limitations. First, all types of admissions with a minimum duration of 24

hours were analyzed, possibly including admissions for PD training and vascular access procedures. Consequently, both PD and HD admissions might be overrated. Second, no center correction has been conducted, while the decision to admit a patient might differ between centers. Third, it should be noted that a very small number of hemodialysis patients were treated with home hemodialysis (n=45) and hospitalizations during this treatment (n=57) were counted among HD hospitalizations, which may have affected the results. Finally, the model we used, which allows transitions between dialysis modalities over time, was not compatible with competing risk regression models, whereas death should be considered a competing event. However, in our population, only 17 patients died without being hospitalised, while 140 patients died during or after at least one hospitalisation. Thus, we do believe that accounting for competing risks would not have altered our results.

In conclusion, our study shows that, when hospitalizations are attributed to the type of dialysis treatment upon admission, PD is associated with a higher hospitalization rate, a higher risk for first hospitalization, a higher number of hospitalizations and a higher number of hospital days per patient-year compared to HD. Since the PD hospitalizations were mainly caused by peritonitis, more attention to infection prevention is necessary for reducing the number of hospitalizations in the future.

References

- 1. Vanholder R and Van Biesen W, Incidence of infectious morbidity and mortality in dialysis patients. Blood Purif, 2002. 20(5): p. 477-80.
- United States Renal Data System, 2018 USRDS Annual Data Report: Epidemiology of kidney disease in the United States. National Institutes of Health, National Institute of Diabetes and Digestive and Kidney Diseases, Bethesda, MD, 2018.
- 3. Banshodani M, Kawanishi H, Moriishi M, Shintaku S, and Tsuchiya S, Association between Dialysis Modality and Cardiovascular Diseases: A Comparison between Peritoneal Dialysis and Hemodialysis. Blood Purif, 2020. **49**(3): p. 302-309.
- 4. Himmelfarb J, Vanholder R, Mehrotra R, and Tonelli M, The current and future landscape of dialysis. Nat Rev Nephrol, 2020. 16(10): p. 573-585.
- 5. Molnar AO, Moist L, Klarenbach S, et al., Hospitalizations in Dialysis Patients in Canada: A National Cohort Study. Can J Kidney Health Dis, 2018. 5: p. 2054358118780372.
- 6. Daratha KB, Short RA, Corbett CF, et al., Risks of subsequent hospitalization and death in patients with kidney disease. Clin J Am Soc Nephrol, 2012. 7(3): p. 409-16.
- Zhang AH, Cheng LT, Zhu N, Sun LH, and Wang T, Comparison of quality of life and causes of hospitalization between hemodialysis and peritoneal dialysis patients in China. Health Qual Life Outcomes, 2007. 5: p. 49.
- Quinn RR, Ravani P, Zhang X, et al., Impact of modality choice on rates of hospitalization in patients eligible for both peritoneal dialysis and hemodialysis. Perit Dial Int, 2014. 34(1): p. 41-8.
- 9. Kim H, An JN, Kim DK, et al., Elderly Peritoneal Dialysis Compared with Elderly Hemodialysis Patients and Younger Peritoneal Dialysis Patients: Competing Risk Analysis of a Korean Prospective Cohort Study. PLoS One, 2015. 10(6): p. e0131393.
- Manns BJ, Mendelssohn DC, and Taub KJ, The economics of end-stage renal disease care in Canada: incentives and impact on delivery of care. Int J Health Care Finance Econ, 2007. 7(2-3): p. 149-69.
- Mohnen SM, van Oosten MJM, Los J, et al., Healthcare costs of patients on different renal replacement modalities - Analysis of Dutch health insurance claims data. PLoS One, 2019. 14(8): p. e0220800.
- Lee H, Manns B, Taub K, et al., Cost analysis of ongoing care of patients with end-stage renal disease: the impact of dialysis modality and dialysis access. Am J Kidney Dis, 2002. 40(3): p. 611-22.
- 13. Harris SA, Lamping DL, Brown EA, Constantinovici N, and North Thames Dialysis Study G, Clinical outcomes and quality of life in elderly patients on peritoneal dialysis versus hemodialysis. Perit Dial Int, 2002. 22(4): p. 463-70.
- Yang JY, Chen L, Chao CT, et al., Comparative Study of Outcomes among Patients with Polycystic Kidney Disease on Hemodialysis and Peritoneal Dialysis. Sci Rep, 2015. 5: p. 12816.
- 15. Laurin LP, Harrak H, Elftouh N, et al., Outcomes of Infection-Related Hospitalization according to Dialysis Modality. Clin J Am Soc Nephrol, 2015. 10(5): p. 817-24.
- Oliver MJ, Al-Jaishi AA, Dixon SN, et al., Hospitalization Rates for Patients on Assisted Peritoneal Dialysis Compared with In-Center Hemodialysis. Clin J Am Soc Nephrol, 2016. 11(9): p. 1606-14.

- 17. Murphy SW, Foley RN, Barrett BJ, et al., Comparative hospitalization of hemodialysis and peritoneal dialysis patients in Canada. Kidney Int, 2000. 57(6): p. 2557-63.
- 18. Lafrance JP, Rahme E, Iqbal S, et al., Association of dialysis modality with risk for infection-related hospitalization: a propensity score-matched cohort analysis. Clin J Am Soc Nephrol, 2012. 7(10): p. 1598-605.
- 19. Perl J, McArthur E, Bell C, et al., Dialysis Modality and Readmission Following Hospital Discharge: A Population-Based Cohort Study. Am J Kidney Dis, 2017. **70**(1): p. 11-20.
- 20. Nitta K, Masakane I, Hanafusa N, et al., Annual dialysis data report 2017, JSDT Renal Data Registry. Renal Replacement Therapy, 2019. 5(1).
- 21. Banshodani M, Kawanishi H, Moriishi M, Shintaku S, and Tsuchiya S, Association between Dialysis Modality and Infectious Diseases: Peritoneal Dialysis versus Hemodialysis. Blood Purif, 2020: p. 1-10.
- 22. van Dijk PC, Jager KJ, de Charro F, et al., Renal replacement therapy in Europe: the results of a collaborative effort by the ERA-EDTA registry and six national or regional registries. Nephrol Dial Transplant, 2001. 16: p. 1120-1129.
- 23. Charlson ME, Pompei P, Ales KL, and MacKenzie CR, A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. J Chronic Dis, 1987. 40(5): p. 373-83.
- 24. Davies SJ, Russell L, Bryan J, Phillips L, and Russell GI, Comorbidity, Urea Kinetics, and Appetite in Continuous Ambulatory Peritoneal-Dialysis Patients Their Interrelationship and Prediction of Survival. American Journal of Kidney Diseases, 1995. **26**(2): p. 353-361.
- WHO, International Statistical Classification of Diseases and Related Health Problems, 10th Revision. Available at https://class.whofic.nl/browser.aspx?scheme=ICD10-nl.cla. 2007.
- 26. Williams VR, Quinn R, Callery S, Kiss A, and Oliver MJ, The impact of treatment modality on infection-related hospitalization rates in peritoneal dialysis and hemodialysis patients. Perit Dial Int, 2011. **31**(4): p. 440-9.
- 27. Jeon Y, Kim HD, Hong YA, et al., Clinical outcomes of infection-related hospitalization in incident peritoneal dialysis patients. Kidney Res Clin Pract, 2020. **39**(4): p. 460-468.
- 28. van de Luijtgaarden MWM, Jager KJ, Segelmark M, et al., Trends in dialysis modality choice and related patient survival in the ERA-EDTA Registry over a 20-year period. Nephrol Dial Transplant, 2016. **31**: p. 120-128.
- 29. Bonenkamp AA, Hoekstra T, Hemmelder MH, et al., Trends in home dialysis use differ among age categories in past two decades: A Dutch registry study. Eur J Clin Invest, 2021: p. e13656.

Hospitalization of dialysis patients



Chapter 6

Technique failure in peritoneal dialysis: modifiable causes and patient-specific risk factors

Anna A. Bonenkamp, Anita van Eck van der Sluijs, Friedo W. Dekker,Dirk G. Struijk, Carola W.H. de Fijter, Yolande M. Vermeeren, FransJ. van Ittersum, Marianne C. Verhaar, Brigit C. van Jaarsveld, andAlferso C. Abrahams on behalf of the DOMESTICO study group

Peritoneal Dialysis International. 2022. In print

Abstract

Background Technique survival is a core outcome for peritoneal dialysis (PD), according to SONG-PD. This study aimed to identify modifiable causes and risk factors of technique failure in a large Dutch cohort using standardized definitions.

Methods Patients who participated in the retrospective DOMESTICO cohort study and started PD between 2012 and 2016 were included, and followed until January 1st 2017. The primary outcome was technique failure, defined as transfer to in-center hemodialysis for \geq 30 days or death. Death-censored technique failure was analyzed as secondary outcome. Cox regression models and competing risk models were used to assess the association between potential risk factors and technique failure.

Results A total of 695 patients were included, of whom 318 experienced technique failure during follow-up. Technique failure rate in the first year was 29%, while the death-censored technique failure rate was 23%. Infections were the most common modifiable cause for technique failure, accounting for 20% of all causes during the entire follow-up. Leakage and catheter problems were important causes within the first six months of PD treatment (both accounting for 15%). APD use was associated with a lower risk of technique failure (HR 0.66, 95% CI 0.53 – 0.83).

Conclusion Infections, leakage, and catheter problems were important modifiable causes for technique failure. As the first-year death-censored technique failure rate remains high, future studies should focus on infection prevention and catheter access to improve technique survival.

Introduction

Peritoneal dialysis (PD) is an established treatment for kidney failure, offering patients more flexibility and independence compared to in-center hemodialysis.^{1, 2} Improving the technique survival of PD, i.e. preventing technique failure, remains a challenge despite advances in technique survival over the past decades.³⁻⁵ In fact, technique survival was chosen as one of the five core outcomes for PD according to the Standardized Outcomes in Nephrology-Peritoneal Dialysis (SONG-PD) study.⁶

Identifying modifiable causes and risk factors of technique failure could contribute to develop strategies to improve PD technique survival. Previous research has identified causes and risk factors of technique failure during the first months of PD treatment.⁷⁻⁹ Although technique failure after the first months of PD treatment is also relevant for the loss of prevalent PD patients, few studies have explored the various causes over an extended period of PD treatment.¹⁰⁻¹²

Moreover, comparing previous research on technique failure is hampered by the lack of standard definitions.⁸ Technique failure is defined differently in almost every other study, especially in handling death as a cause of technique failure. Lan *et al.* therefore advocated the use of a standardized definition of technique failure, including both transfer to in-center hemodialysis (CHD) and death.¹³ Few studies to date have used this standardized definition.^{3,7}

In addition, the characteristics of PD patients have changed over time and studies on technique failure in the current PD population are scarce. Therefore, this study aims to investigate the causes, risk factors, and center variation of PD technique failure in a recent Dutch cohort, all according to the standardized definitions.

Methods

Study design and research population

Patients were enrolled from the retrospective Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO), a multi-center cohort study in the Netherlands. In this study, 33 centers included PD patients, representing nearly two thirds of all dialysis centers in the Netherlands. Eligible patients were adults who started PD between 1 January 2012 and 1 January 2017, and had a minimum PD treatment duration of 14 days. Patients who were previously treated with dialysis or kidney transplantation were also included. Patients who stopped dialysis or died within 30 days after dialysis initiation were excluded. Patients were followed until kidney transplantation, wish to stop dialysis, death or end of study period on 1 January 2017. Local medical ethics committees of all participating dialysis centers approved the study. Reporting of the study conforms to broad STROBE guidelines.¹⁴

Definition of PD technique failure

The primary outcome of this study was PD technique failure, defined as a transfer to CHD for \geq 30 days, death on PD or death within 30 days after transfer to CHD, in accordance with the previously proposed standardized definition.¹³ In patients with multiple episodes of technique failure, only the first episode of technique failure was analyzed. The following causes for technique failure were collected from the electronic patient charts: PD-related infections consisting of PD peritonitis and exit-site infections, catheter-related problems, clearance or ultrafiltration (UF) problems, peritoneal leakage, psychosocial problems, risk for or diagnosis of encapsulating peritoneal sclerosis (EPS), another reason, stop dialysis, and death.¹⁵

In addition, patients were stratified into an early and a late technique failure group. Early technique failure was defined as technique failure during the first 6 months after start of PD, and late technique failure was defined as technique failure that occurred more than 6 months after start of PD.^{8, 9, 16}

Secondary outcomes were death-censored technique failure, death and permanent technique failure, the latter was defined as a transfer to CHD for \geq 180 days, death on PD or death within 180 days after transfer to CHD.¹³

Covariates

Demographic, clinical, and dialysis-related data at dialysis initiation were collected from electronic patient charts. These included age, sex, ethnic background, employment status, smoking, body mass index (BMI), primary kidney disease, comorbid conditions, dialysis vintage, and kidney transplant history. PD modality, i.e. continuous ambulatory PD (CAPD) or automated PD (APD), was defined as the modality the patient used most of the time during follow-up. BMI was divided into three groups according to the WHO classification: BMI <25 kg/m², BMI 25 - 30 kg/m²

(overweight), and BMI \geq 30 (obese). Comorbid conditions were scored into three groups according to the Charlson Comorbidity Index (CCI): low (2 points, since patients with kidney failure by definition already have 2 points), intermediate (3-4 points), and severe comorbidity (\geq 5 points).¹⁷ Causes of death, coded according to the ERA-EDTA coding system, were retrieved from the Dutch renal registry (RENINE).¹⁸ For each participating center PD volume was calculated from data provided by RENINE, as mean annual number of prevalent patients, and divided into tertiles.¹⁹ Variation in practice patterns were collected with an additional questionnaire that was send to the local investigators of the participating centers.

Statistical analysis

Baseline characteristics were expressed as number with percentages for categorical variables and as mean with standard deviation (SD) or median with interquartile range (IQR) for continuous variables. Incidence of all-cause technique failure was presented as a Kaplan Meier curve. Cumulative incidence curves of cause-specific technique failure were calculated using a competing risk model.²⁰ Causes of early and late technique failure were shown as percentages.

To investigate the association between possible risk factors and technique failure, a cox regression model was conducted. This model was censored for kidney transplantation. BMI and PD modality were selected as potentially modifiable patient-specific risk factors according to literature.^{3, 7, 9, 12, 16} Each potentially modifiable risk factor was adjusted for plausible predetermined confounders (age, sex, employment status, BMI, CCI, and center PD volume). The proportional hazard assumption was verified in the unadjusted models on the basis of Schoenfeld residuals and Kaplan Meier graphs. Several sensitivity analyses were conducted. First, a competing risk model was used to investigate the association between possible risk factors and technique failure in the presence of a competing event.²⁰ In such a model, a participant with the competing event (i.e. kidney transplantation) remains in the analysis. This model was also used to investigate the association between possible risk factors and death-censored technique failure, in which both kidney transplantation and death were competing events. Second, hypothesizing that PD modality at PD cessation might be different from PD modality used most of the time and be related to technique failure, in patients with technique failure the PD modality at PD cessation was used.

Finally, a funnel plot was constructed to evaluate the early technique failure rate of the participating centers, adjusted for age and sex. This is a graphical method to evaluate center performance with a reference standard, i.e. the overall early technique failure rate, and an indication of precision through control limits based on sample sizes.^{21,22} The early technique failure rate was chosen, because especially early failure is associated with catheter-related problems and thus possible modifiable causes.⁸

Missing confounders (maximum of 25% missing for BMI and CCI) were imputed using standard multiple imputation techniques in SPSS (10 repetitions and predictive mean matching). All analyses were performed using SPSS Statistics version 26 (IBM) or STATA 14 (StataCorp LP, College Station, TX). A p-value of < 0.05 was considered statistically significant.

Results

A total of 708 adult patients started PD treatment between 2012 and 2016 in the participating centers, of whom 13 patients were excluded since they had a total PD duration of less than 14 days. The study population thus consisted of 695 patients (See Flow diagram, Figure 1).



Figure 1. Flow chart of the patients included in the study

Baseline characteristics are presented in Table 1. Mean age at dialysis initiation was 62.9 ± 15.1 years and 27% of patients had a high CCI score indicating severe comorbidity. A history of previous dialysis was present in 15% of patients. APD was the predominantly used PD modality in 29% of patients with early technique failure and 53% of patients with late technique failure, reflecting common practice in the Netherlands to start PD therapy with CAPD. The median PD follow-up time for all patients was 13 months [IQR 6 – 22.2 months], with a minimum of 0 and a maximum of 59 months.

	All patients	Patients with technique failure	Patients without technique failure	p-value
	n=695	n=318	n=377	
Age (yr), mean ± SD	62.9±15.1	64.8±14.8	61.4±15.1	0.003
Sex (male), n (%)	447 (64)	210 (66)	237 (63)	NS
Ethnic background, n (%)				NS
Caucasian	422 (61)	191 (60)	231 (61)	
Moroccan/Turkish	22 (3)	11 (4)	11 (3)	
Asian	39 (6)	15 (5)	24 (6)	
Afro-American	23 (3)	9 (3)	14 (4)	
Other/unknown	189 (27)	92 (29)	97 (26)	
Primary kidney disease, n (%)				NS
Glomerulonephritis	81 (12)	32 (10)	49 (13)	
Polycystic kidney disease	37 (5)	11 (4)	26 (7)	
Renovascular kidney disease	210 (30)	112 (35)	98 (26)	
Diabetes mellitus	123 (18)	58 (18)	65 (17)	
Other	183 (26)	84 (27)	99 (26)	
Unknown	61 (9)	21 (7)	40 (11)	
Employment status, n (%)	167 (28)	61 (22)	106 (32)	0.006
Current smoker, n (%)	111 (16)	52 (17)	59 (16)	NS
BMI (kg/m ²), mean \pm SD	26.4±5.0	26.9±5.1	26.1±4.9	0.05
BMI, n (%)				NS
< 25 kg/m2	239 (46)	98 (42)	141 (49)	
25 – 30 kg/m2	177 (34)	85 (36)	92 (32)	
≥ 30 kg/m2	107 (20)	51 (22)	56 (19)	

Table 1. Baseline characteristics of 695 patients treated with peritoneal dialysis

	All patients	Patients with technique failure n=318	Patients without technique failure n=377	p-value
Charlson comorbidity index, n (?	%)			
2 (low)*	168 (32)	58 (25)	110 (38)	0.001
3-4 (intermediate)	212 (41)	95 (41)	117 (40)	
≥ 5 (severe)	139 (27)	77 (33)	62 (21)	
Diabetes mellitus, n (%)	164 (32)	81 (35)	83 (29)	NS
Ischemic heart disease, n (%)	146 (28)	80 (35)	66 (23)	0.002
Heart failure, n (%)	69 (13)	38 (17)	31 (11)	NS
Vascular disease, n (%)	130 (23)	65 (26)	65 (21)	NS
History of dialysis at dialysis initiation, n (%)	103 (15)	39 (12)	64 (17)	NS
Dialysis vintage (months), median [IQR]	12 [1–36]	12 [4-37]	11 [1-33]	NS
History of kidney transplant at dialysis initiation, n (%)	73 (11)	29 (9)	44 (12)	NS
Kidney transplant (months), median [IQR]	120 [64-171]	99 [64-171]	135 [63-173]	NS
APD, n (%)	350 (50)	146 (46)	204 (54)	0.03

Table 1. Baseline characteristics of 695 patients treated with peritoneal dialysis (continued)

BMI, body mass index; APD, automated peritoneal dialysis; SD, standard deviation; IQR, interquartile range. Groups are defined according to the 30-day definition of technique failure. * kidney failure alone represents a Charlson Comorbidity Index of 2 points

Incidence of technique failure

A total of 318 patients developed technique failure during the study, of whom 22 patients experienced a recurrent episode of technique failure. The PD patients experienced a mean of 0.36 episodes of technique failure per person–year of follow-up. The 1- and 2-year technique failure rates were 29% and 52% respectively (Figure 2A). The median time to technique failure was 1.85 years. Patients with technique failure were older, had higher comorbidity scores, were more likely to have ischemic heart disease, and were more frequently treated with CAPD (Table 1). A total of 202 patients developed death-censored technique failure during the study (0.24 episodes of death-censored technique failure per person-year). The 1- and 2-year death-censored technique failure and 35% respectively (Figure 2B). The median time to death-censored technique failure was 3.58 years.



Figure 2. Technique failure, as a composite outcome (with transfer to CHD or death) (A) and as death-censored technique failure (B). Technique failure was defined as a transfer to CHD for \geq 30 days, death on PD or death within 30 days after transfer to CHD. First day of receiving CHD was the date assigned as technique failure.

Causes of technique failure

Figure 3 shows that death was the most common cause of technique failure, followed by PD-related infections (20%). The other causes of technique failure occurred in about 10% or less than 10% of the patients who experienced technique failure. The predominant causes for death were cardiovascular disease (28%), infections other than PD peritonitis (15%) and malignancies (13%). None of the deaths were attributable to a PD peritonitis.



Figure 3. Cumulative incidence of different causes for technique failure Shows the occurrence of different causes for technique failure over time in a population of patients with technique failure (n=318, 100%). UF, ultrafiltration.

Figure 4 shows the different causes of early (i.e. during the first 6 months after start of PD) and late (i.e. more than 6 months after start of PD) technique failure. A total of 99 patients developed early technique failure, and 219 patients developed late technique failure. Catheter-related problems were the cause of early technique failure in 15% of patients, whereas this was the cause of late technique failure in only 5% of patients. Similarly, PD fluid leakage was the cause in 15% and 5%, respectively. Infections and clearance problems were a major cause of both early and late technique failure; infections were in 20% of patients the cause of technique failure and clearance problems in 11–12% of patients. EPS was a cause of technique failure in less than 1% of patients. The group of 'other reasons' included (temporary) discontinuations of PD due to major (abdominal) surgery with hospitalization.



Figure 4. Comparison of causes of early and late technique failure Early PD technique failure is defined as occurrence of technique failure in the first 6 months after start of PD (n=99). Late PD technique failure is defined as occurrence of technique failure more than 6 months after start of PD (n=219). EPS, encapsulating peritoneal sclerosis.

Risk factors

The patient-specific risk factors sex, age, employment status and BMI were not associated with technique failure (Table 2). APD compared to CAPD was associated with a reduced risk of technique failure (adjusted hazard ratio (HR) 0.66 (95% Confidence Interval (CI) 0.53 - 0.83). The patient-specific risk factors for death-censored technique failure were similar to those for technique failure including death in the definition (Supplementary Table S1); only APD was associated with a reduced risk of death-censored technique failure (adjusted HR 0.60, 95% CI 0.46 - 0.80). In addition, APD use was not associated with death as a separate outcome while age was associated with death (Supplementary Table S2).

Table 2. Patient-specific	risk factors associated with t	technique fail	ure in a Cox regression mod	del		
Risk factors	Crude HR (95% CI)	p-value	Adjusted model 1 HR (95% CI)	p-value	Adjusted model 2 HR (95% CI)	p-value
Male sex	1.15(0.91 - 1.45)	0.24				
Age (10-year)	1.05(0.97 - 1.13)	0.25				
Employed	$0.80\ (0.60-1.07)$	0.13				
CCI						
low	Reference					
intermediate	1.41 (1.02 - 1.96)	0.04				
severe	1.81(1.29 - 2.55)	0.001				
PD volume						
< 15 patients	Reference					
15-25 patients	$1.05\ (0.68 - 1.63)$	0.83				
>25 patients	$0.81 \ (0.53 - 1.24)$	0.33				
BMI						
< 25 kg/m ²	Reference		Reference			
$25 - 30 \text{ kg/m}^2$	1.21 (0.91 - 1.62)	0.20	1.17 (0.87 - 1.58)	0.31		
≥ 30 kg/m²	$1.21 \ (0.86 - 1.69)$	0.28	1.23 (0.88 - 1.71)	0.22		
APD (vs CAPD)	0.66(0.53 - 0.83)	<0.001	0.67 (0.54 - 0.84)	<0.001	0.66(0.53 - 0.83)	<0.001
Model 1 is adjusted for a Model 2 is adjusted for a	tge and sex age, sex, employment status,	BMI, CCI, ai	nd center PD volume			

In this cox regression model both pre-selected potentially modifiable risk factors, BMI and PD modality, and all determinants used for adjustments are shown. HR, hazard ratio; BMI, body mass index; APD, automated peritoneal dialysis; CAPD, continuous ambulatory peritoneal dialysis; CCI, Charlson Comorbidity index. The sensitivity analysis in which the association between patient-specific risk factors and technique failure was investigated with a competing risk model, showed similar results for these associations as the original analyses (Supplementary Table S3). In a sensitivity analysis using PD modality at PD cessation, similar results were found (for APD compared to CAPD, adjusted HR 0.60 (95% CI 0.47 – 0.75)).

Center variation in technique failure

All centers used icodextrin and antibiotic prophylaxis during PD catheter insertion (Supplementary Table S4). Most centers used neutral pH low glucose degradation products (GDP) solutions (91%) and exit site antibiotic prophylaxis (79%). The initial antibiotic regimen for peritonitis varied across centers and antifungal prophylaxis during antibiotic therapy was provided only in 6% of centers.



Figure 5. Funnel plot of early technique failure in incident study patients Each circle represents the early technique failure rate for a participating center (n=31). Rates are adjusted for age and sex. The overall early technique failure rate is used as a reference (blue). The 90%, 95%, and 98% control limits are provided as dotted lines. Using the 95% control limit, one center with 29 incidents patients during the study period had a significantly higher early technique failure rate and performed worse than expected.

The center variation in technique failure rate is shown in Figure 5. The overall early technique failure rate, shown as the reference standard, was 16%, which is the total number of patients with early technique failure divided by the total number of PD patients from all centers that were not lost to follow-up at 6 months (due to transplantation or study end, n = 73). Most centers had an early technique failure rate around the overall rate of 16%. Four centers had a higher rate, of which only one center was outside the 95% control limits of the reference standard.

Permanent technique failure

A total of 254 patients developed permanent technique failure during the study: i.e. at 180 days after transfer to CHD they had not returned to PD (0.26 episodes of permanent technique failure per person-year). The 1- and 2-year permanent technique failure rate was 22% and 43% respectively (Supplementary Figure S1). The median time to permanent technique failure was 2.7 years. The most common cause of permanent technique failure was death, followed by infections. A total of 72 patients developed early permanent technique failure and 182 patients developed late permanent technique failure. Again, early technique failure was associated with catheter-related problems and leakage, while infection and clearance problems were important causes for both early and late technique failure (Supplementary Figure S2 and Supplementary Table S5).

Discussion

In this cohort of 695 Dutch patients who were treated with PD between 2012 and 2017, the technique failure rate within the first year of PD treatment was 29%. Death was the most common cause of technique failure. Death-censored technique failure rate at 1 year was 23%. In 20% of patients with technique failure, infections were a possible modifiable cause. In addition, early technique failure was frequently caused by catheter-related problems and leakage (both accounting for 15%). We found that APD use had a protective effect on technique failure.

Only few studies to date have used the standardized technique failure definition as proposed by L*an et al.*^{3,7,13} See *et al.*, reporting on Australian patients that started PD between 2000 and 2014, also used the standardized 30-day definition and found a first year technique failure rate of 26%.⁷ In an older study by Descoeudres *et al.*, not using

the standard definition but a similar definition of technique failure including death by any cause, the technique failure rate at 1 year was 25%.²³ The technique failure rate in our study is thus comparable to other studies that included death as a cause for technique failure. Death was the most common cause for technique failure during the entire follow-up, as would be expected in a study on dialysis patients since mortality rates of both PD and CHD patients are high.²⁴ Yet the death-censored technique failure rate was still high. This, in addition to the decline of the number of PD patients in the Netherlands, underscores the need to find modifiable causes for technique failure.

In recent decades, significant advances in PD treatment have declined the overall rate of technique failure.³⁻⁵ Boyer et al. state that this is, in addition to improved patient survival, attributable to less infection-related technique failure.⁵ Nevertheless, infections were still an important cause of technique failure - both in early and late technique failure - indicating that prevention of infections is pivotal in technique survival. Recommendations for the prevention of peritonitis from the ISPD, including exit-site prophylaxis and antibiotic prophylaxis during PD catheter insertion, were generally well followed by participating centers especially if compared to international data from PDOPPS.²⁵⁻²⁷ In a recent study by PDOPPS, antibiotic prophylaxis during PD catheter insertion was indeed associated with a lower peritonitis risk.²⁸ On the other hand, most centers in the Netherlands did not use antifungal prophylaxis during antibiotic therapy although prophylaxis was associated with a significant risk reduction of fungal peritonitis in a systematic review.²⁹ According to the results of PDOPPS antifungal prophylaxis was also variably used across countries, the lowest in Japan (8% of facilities) and the highest in Australia (89%).²⁷ So a greater reduction in infections may be possible if all centers would adhere to current guidelines.

The ISPD guidelines refrain from recommending a specific antibiotic regimen for peritonitis based on a Cochrane systematic review due to lack of superiority.^{25, 30} As a result, the initial antibiotic regimen varied across centers. Of note, one third of all centers used a combination with glycopeptides, possibly based on a systematic review in which glycopeptides were proven most effective in combination with ceftazidim.³¹ Also in PDOPPS a variable use of vancomycin across countries has been reported.²⁷ However, because evidence for antibiotic regimens including glycopeptides remain weak³⁰, future clinical trials may evaluate good practices from single centers. Examples are temporary discontinuation of PD without removing the catheter (peritoneal rest) combined with intravenous meropenem and meropenem intracatheter as lock (Mero-

PerRest protocol) in case of enteric peritonitis and the treatment with amphotericin B catheter lock for salvage of the PD catheter in case of *Candida* peritonitis.^{32, 33}

Catheter-related problems have been identified as an important cause of early technique failure in previous studies.^{10, 23} In this study, we identified leakage as another important cause of early technique failure. This underscores the need for a multidisciplinary team with sufficient experience in catheter care and insertion.³⁴ In a study from Australia and New Zealand, small center volume - possibly indicative of low center experience - was associated with technique failure due to mechanical complications.³ A striking variation in PD catheter survival among different centers in the UK suggests differences in access protocols.¹⁵ Still, previous studies have not yielded results that could lead to recommendations for the preferred use of a catheter delivery technique or specific PD catheter type.^{34, 35} The workgroup PD catheter access of PDOPPS hypothesize that standardized protocols for catheter insertion will be associated with a reduction of technique failure, the results of this working group are thus eagerly awaited.¹⁵

A possible other reduction in technique failure might be the increased interest in assisted PD due to the ageing dialysis population.³⁶ Within this demographic shift, assistance during PD treatment is a mean to provide home dialysis to elderly patients that may be unable to perform PD themselves due to frailty or physical impairments. In a recent study, family-assisted PD was associated with lower risk on catheterrelated technique failure.³⁷ The authors hypothesized that involving family members in dialysis treatment may lead to better adherence to diet restrictions resulting in less constipation. Of note, in this study also a lower risk on technique failure due to clearance problems was found in both family assisted and nurse-assisted PD. The nurse or family member supervising the treatment likely ameliorates the patient's adherence to dialysis prescriptions.³⁷ Clearance problems, in our report the main cause of death-censored technique failure following infections, may thus also be perceived as a modifiable cause for technique failure. These aforementioned modifiable causes - infections, leakage, catheter-related problems and clearance problems - accounted for 48% of technique failure within our cohort, hence, quality improvements aimed at these causes can have a major impact on technique survival.

APD use had a protective effect on death-censored technique failure in our analysis, even after adjustments for age and comorbidity. In recent literature conflicting results have been presented: APD use was associated with an adjusted lower technique

failure rate and higher patient survival in one study³⁸, while in other studies APD use was associated with a higher risk of technique failure.^{3, 7} There may be a link with infections, since CAPD use was associated with a higher rate of peritonitis in recent studies.^{28, 39} Also in the only two randomized controlled trials to date - although originating from <2000 - higher peritonitis rates with CAPD use were found.^{40,41} This association with peritonitis might be due to better adaptation of therapy to patient needs, as the authors of a recent study suggest³⁹, or to fewer connections between catheter and dialysis bags when using APD instead of CAPD and thus less risk of breaching hygiene measures. Although the suggestion of fewer connections resulting in less infections is disputed²⁵, new devices that assists the patient are hypothesized to reduce infection risk.⁴² APD might also be used more often by patients themselves than for assisted PD³⁷, which could explain the protective effect since self-care may be associated with a lower peritonitis rate.^{43,44} However, the association between APD use and technique failure may also reflect long-term PD treatment, as patients with early technique failure may not be able to transfer to APD (in other words: confounding by indication). In the Netherlands, most patients start PD treatment with CAPD to familiarize themselves with performing exchanges by hand prior to a transfer to APD. The reason for the protective effect of APD is thus uncertain, therefore the choice for APD or CAPD should ideally be based on patient preference.²⁵

In a previous study from the Netherlands by Huisman *et al.*, smaller centers with on average less than 20 PD patients had a significantly higher risk of technique failure than larger centers.⁴⁵ The association between center volume and technique failure however likely reflects center experience.¹⁶ Indeed, others confirmed that in larger centers technique failure due to modifiable causes, i.e. infections, catheter - and ultrafiltration problems, were less common.⁴⁶ Guillouët *et al.* found that center volume and patients characteristics alone could not fully explain the center effect on technique failure. They suggested that factors of center experience such as patient education and nephrologist's views on home dialysis play an important role in technique failure.¹⁶ Contributing to this, we showed that the early technique failure rate – often caused by infections, leakage and catheter-related problems – was similar across all centers and was not related to the number of incident study patients. This probably indicates that it is not the center volume itself that matters, but the experience within a center and having a dedicated team.

In this study, technique failure consisted of a composite outcome of death and transfer to CHD, in accordance with the standardized definition.¹³ Death is an objective measure but transfer to CHD is subjective; often a choice is made by the nephrologist to discontinue treatment and this decision will be weighed differently by each nephrologist. A considerable proportion of the causes of technique failure may have been modifiable, i.e. infections, leakage and catheter problems, since practice variation exists in peritonitis rate and in the treatment of infections and access.^{27, 39} Because the definition of technique failure partly consists of the decision to discontinue PD, studies on infection prevention and catheter access such as the PDOPPS will help to increase technique survival.¹⁵

Strengths of this study include the use of the standardized definitions of technique failure, including the death-censored and permanent definition, the analysis of causes of both early and late technique failure, the use of a patient cohort reflecting current practice patterns and extensive adjustments for confounders. In addition, most studies were conducted on registry data whereas our cohort study enabled to identify the causes of technique failure in more detail. Yet, the study sample of this analysis was relatively small and the study was conducted in a single country. The study duration of this study was a respectable 5 years, yet the median follow-up duration was 13 months. As a result, the proportion of technique failure after 1 year should be interpreted with caution.

In conclusion, in this multi-center Dutch study of PD patients PD-related infections, leakage and catheter problems were important modifiable causes for technique failure. As almost a quarter of patients experience death-censored technique failure within the first year, future studies should emphasize on prevention of infections and PD catheter access problems to improve technique survival.

References

- 1. Francois K, Bargman JM. Evaluating the benefits of home-based peritoneal dialysis. Int J Nephrol Renovasc Dis. 2014;7:447-55.
- Dahlerus C, Quinn M, Messersmith E, Lachance L, Subramanian L, Perry E, *et al.* Patient perspectives on the choice of dialysis modality: results from the Empowering Patients on Choices for Renal Replacement Therapy (EPOCH-RRT) Study. Am J Kidney Dis. 2016;68(6):901-10.
- Htay H, Cho Y, Pascoe EM, Darssan D, Nadeau-Fredette AC, Hawley C, et al. Multicenter Registry Analysis of Center Characteristics Associated with Technique Failure in Patients on Incident Peritoneal Dialysis. Clin J Am Soc Nephrol. 2017;12(7):1090-9.
- Sukul N, Mukhopadhyay P, Schaubel DE, Pearson J, Turenne M, Saran R, *et al.* Peritoneal Dialysis and Mortality, Kidney Transplant, and Transition to Hemodialysis: Trends From 1996-2015 in the United States. Kidney Medicine. 2020;2(5):610-9.e1.
- 5. Boyer A, Lanot A, Lambie M, Verger C, Guillouet S, Lobbedez T, *et al.* Trends in Peritoneal Dialysis Technique Survival, Death, and Transfer to Hemodialysis: A Decade of Data from the RDPLF. Am J Nephrol. 2021:1-10.
- Manera KE, Johnson DW, Craig JC, Shen JI, Gutman T, Cho Y, *et al.* Establishing a Core Outcome Set for Peritoneal Dialysis: Report of the SONG-PD (Standardized Outcomes in Nephrology-Peritoneal Dialysis) Consensus Workshop. Am J Kidney Dis. 2020;75(3):404-12.
- See EJ, Johnson DW, Hawley CM, Pascoe EM, Badve SV, Boudville N, *et al.* Risk Predictors and Causes of Technique Failure Within the First Year of Peritoneal Dialysis: An Australia and New Zealand Dialysis and Transplant Registry (ANZDATA) Study. Am J Kidney Dis. 2018;72(2):188-97.
- 8. Cho Y, See EJ, Htay H, Hawley CM, Johnson DW. Early Peritoneal Dialysis Technique Failure: Review. Perit Dial Int. 2018;38(5):319-27.
- Béchade C, Guittet L, Evans D, Verger C, Ryckelynck JP, Lobbedez T. Early failure in patients starting peritoneal dialysis: a competing risks approach. Nephrol Dial Transplant. 2014;29(11):2127-35.
- 10. Kolesnyk I, Dekker FW, Boeschoten EW, Krediet RT. Time-dependent reasons for peritoneal dialysis technique failure and mortality. Perit Dial Int. 2010;30(2):170-7.
- 11. Guo A, Mujais S. Patient and technique survival on peritoneal dialysis in the United States: evaluation in large incident cohorts. Kidney Int Suppl. 2003(88):S3-12.
- 12. Jaar BG, Plantinga LC, Crews DC, Fink NE, Hebah N, Coresh J, *et al.* Timing, causes, predictors and prognosis of switching from peritoneal dialysis to hemodialysis: a prospective study. BMC Nephrology. 2009;10(1):3.
- Lan PG, Clayton PA, Johnson DW, McDonald SP, Borlace M, Badve SV, *et al.* Duration of Hemodialysis Following Peritoneal Dialysis Cessation in Australia and New Zealand: Proposal for a Standardized Definition of Technique Failure. Perit Dial Int. 2016;36(6):623-30.
- von Elm E, Altman DG, Egger M, Pocock SJ, Gøtzsche PC, Vandenbroucke JP. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. J Clin Epidemiol. 2008;61(4):344-9.
- Perl J, Davies SJ, Lambie M, Pisoni RL, McCullough K, Johnson DW, *et al.* The Peritoneal Dialysis Outcomes and Practice Patterns Study (PDOPPS): Unifying Efforts to Inform Practice and Improve Global Outcomes in Peritoneal Dialysis. Perit Dial Int. 2016;36(3):297-307.
- Guillouët S, Veniez G, Verger C, Béchade C, Ficheux M, Uteza J, *et al.* Estimation of the Center Effect on Early Peritoneal Dialysis Failure: A Multilevel Modelling Approach. Peritoneal dialysis international : journal of the International Society for Peritoneal Dialysis. 2016;36(5):519-25.
- Charlson MEP, P; Ales, K.L.; MacKenzie, C.R. A new method of classifying prognostic comorbidity in longitduinal studies: development and validation. J Chron Dis. 1987;40(5):373-83.
- Hoekstra T, Dekker FW, Cransberg K, Bos WJ, van Buren M, Hemmelder MH. RENINE annual report 2018. Available at: https://www.nefrovisie.nl/jaarrapportages/ Accessed: 04-01-2021. 2018.
- 19. Hoekstra T, van Ittersum FJ, Rabelink AJ, Berger SP, Hemmelder MH. Analyse van kwaliteitsindicatoren Chronische Nierschade [internet]. [cited 2 november 2020] available from: https://www.nefrovisie.nl/richtlijnen-indicatoren/toelichting/ 2013-2016.
- 20. Fine JP, Gray RJ. A Proportional Hazards Model for the Subdistribution of a Competing Risk. Journal of the American Statistical Association. 1999;94(446):496-509.
- 21. van der Willik EM, van Zwet EW, Hoekstra T, van Ittersum FJ, Hemmelder MH, Zoccali C, *et al.* Funnel plots of patient-reported outcomes (PROs) to evaluate healthcare quality: basic principles, pitfalls and considerations. Nephrology (Carlton). 2020.
- 22. Spiegelhalter DJ. Funnel plots for comparing institutional performance. Statistics in medicine. 2005;24(8):1185-202.
- 23. Descoeudres B, Koller MT, Garzoni D, Wolff T, Steiger J, Schaub S, *et al.* Contribution of early failure to outcome on peritoneal dialysis. Perit Dial Int. 2008;28(3):259-67.
- 24. van de Luijtgaarden MW, Jager KJ, Segelmark M, Pascual J, Collart F, Hemke AC, *et al.* Trends in dialysis modality choice and related patient survival in the ERA-EDTA Registry over a 20-year period. Nephrol Dial Transplant. 2016;31(1):120-8.
- 25. Li PK, Szeto CC, Piraino B, de Arteaga J, Fan S, Figueiredo AE, *et al.* ISPD Peritonitis Recommendations: 2016 Update on Prevention and Treatment. Perit Dial Int. 2016;36(5):481-508.
- 26. Szeto CC, Li PK, Johnson DW, Bernardini J, Dong J, Figueiredo AE, *et al.* ISPD Catheter-Related Infection Recommendations: 2017 Update. Perit Dial Int. 2017;37(2):141-54.
- 27. Boudville N, Johnson DW, Zhao J, Bieber BA, Pisoni RL, Piraino B, *et al.* Regional variation in the treatment and prevention of peritoneal dialysis-related infections in the Peritoneal Dialysis Outcomes and Practice Patterns Study. Nephrol Dial Transplant. 2019;34(12):2118-26.
- 28. Perl J, Fuller DS, Bieber BA, Boudville N, Kanjanabuch T, Ito Y, *et al.* Peritoneal Dialysis-Related Infection Rates and Outcomes: Results From the Peritoneal Dialysis Outcomes and Practice Patterns Study (PDOPPS). Am J Kidney Dis. 2020;76(1):42-53.
- 29. Campbell D, Mudge DW, Craig JC, Johnson DW, Tong A, Strippoli GF. Antimicrobial agents for preventing peritonitis in peritoneal dialysis patients. Cochrane Database Syst Rev. 2017;4(4):Cd004679.

- Ballinger AE, Palmer SC, Wiggins KJ, Craig JC, Johnson DW, Cross NB, *et al.* Treatment for peritoneal dialysis-associated peritonitis. Cochrane Database of Systematic Reviews. 2014(4).
- Barretti P, Doles JVP, Pinotti DG, El Dib R. Efficacy of antibiotic therapy for peritoneal dialysis-associated peritonitis: a proportional meta-analysis. BMC Infectious Diseases. 2014;14(1):445.
- 32. van der Sluijs AVE, Eekelschot KZ, Frakking FN, Haas PA, Boer WH, Abrahams AC. Salvage of the peritoneal dialysis catheter in Candida peritonitis using amphotericin B catheter lock. Perit Dial Int. 2021;41(1):110-4.
- 33. Abrahams AC, Ruger W, Ter Wee PM, van Ittersum FJ, Boer WH. Improved Outcome of Enteric Peritonitis in Peritoneal Dialysis Patients Aged 50 Years and Older with Temporary Discontinuation of Peritoneal Dialysis and Intravenous Meropenem. Perit Dial Int. 2017.
- Crabtree JH, Shrestha BM, Chow KM, Figueiredo AE, Povlsen JV, Wilkie M, et al. Creating and Maintaining Optimal Peritoneal Dialysis Access in the Adult Patient: 2019 Update. Perit Dial Int. 2019;39(5):414-36.
- Sun ML, Zhang Y, Wang B, Ma TA, Jiang H, Hu SL, *et al.* Randomized controlled trials for comparison of laparoscopic versus conventional open catheter placement in peritoneal dialysis patients: a meta-analysis. BMC Nephrol. 2020;21(1):60.
- Giuliani A, Karopadi AN, Prieto-Velasco M, Manani SM, Crepaldi C, Ronco C. Worldwide Experiences with Assisted Peritoneal Dialysis. Perit Dial Int. 2017;37(5):503-8.
- Lanot A, Bechade C, Boyer A, Ficheux M, Lobbedez T. Assisted peritoneal dialysis and transfer to haemodialysis: a cause-specific analysis with data from the RDPLF. Nephrol Dial Transplant. 2021;36(2):330-9.
- Wang IK, Yu TM, Yen TH, Lin SY, Chang CL, Lai PC, *et al.* Comparison of patient survival and technique survival between continuous ambulatory peritoneal dialysis and automated peritoneal dialysis. Perit Dial Int. 2020;40(6):563-72.
- Nadeau-Fredette AC, Johnson DW, Hawley CM, Pascoe EM, Cho Y, Clayton PA, et al. Center-Specific Factors Associated with Peritonitis Risk-A Multi-Center Registry Analysis. Perit Dial Int. 2016;36(5):509-18.
- 40. de Fijter CW, Oe PL, Nauta JJ, van der Meulen J, ter Wee PM, Snoek FJ, *et al.* A prospective, randomized study comparing the peritonitis incidence of CAPD and Y-connector (CAPD-Y) with continuous cyclic peritoneal dialysis (CCPD). Advances in peritoneal dialysis Conference on Peritoneal Dialysis. 1991;7:186-9.
- 41. Bro S, Bjorner JB, Tofte-Jensen P, Klem S, Almtoft B, Danielsen H, *et al.* A prospective, randomized multicenter study comparing APD and CAPD treatment. Perit Dial Int. 1999;19(6):526-33.
- 42. Hess S, Dubach M, Meboldt M, Foggensteiner L. Evaluating Patient Safety And Ease Of Use Of A Novel Connection-Assist Device For Peritoneal Dialysis. Patient Prefer Adherence. 2019;13:1785-90.
- 43. Ng JK, Chan GC, Chow KM, Fung W, Pang WF, Law MC, *et al.* Helper-assisted continuous ambulatory peritoneal dialysis: Does the choice of helper matter? Perit Dial Int. 2020;40(1):34-40.
- Béchade C, Lobbedez T, Ivarsen P, Povlsen JV. Assisted Peritoneal Dialysis for Older People with End-Stage Renal Disease: The French and Danish Experience. Perit Dial Int. 2015;35(6):663-6.

- 45. Huisman RM, Nieuwenhuizen MG, Th de Charro F. Patient-related and centre-related factors influencing technique survival of peritoneal dialysis in The Netherlands. Nephrol Dial Transplant. 2002;17(9):1655-60.
- 46. Mujais S, Story K. Peritoneal dialysis in the US: Evaluation of outcomes in contemporary cohorts. Kidney International. 2006;70:S21-S6.

Risk factors	Crude	p-value	Adjusted model 1	p-value	Adjusted model 2	p-value
	SHR (95% CI)		SHR (95% CI)		SHR (95% CI)	
Male sex	1.14(0.86 - 1.53)	0.36				
Age (10 year)	0.94 (0.86 - 1.03)	0.16				
Employed	1.07 (0.78 - 1.47)	0.69				
CCI						
low	Reference					
intermediate	1.15(0.79 - 1.68)	0.47				
severe	1.13(0.74 - 1.74)	0.57				
BMI						
$< 25 \ {\rm kg/m^2}$	Reference		Reference			
25 – 30 kg/m	1.14(0.80 - 1.62)	0.47	1.16(0.82 - 1.63)	0.41		
≥ 30 kg/m²	0.88 (0.57 - 1.37)	0.58	$1.01 \ (0.67 - 1.52)$	0.97		
APD (vs CAPD)	0.64 (0.49 - 0.84)	0.001	0.62 (0.47 - 0.81)	0.001	0.60(0.46 - 0.80)	<0.001
Model 1 is adjust Model 2 is adjust SHR, subdistribu	ed for age and sex ed for age, sex, employment sta ttion hazard ratio, BMI, body n	cus, BMI, CCI, ass index; API	and center PD volume), automated peritoneal d	ialvsis: CAPD.	continuous ambulatory per	itoneal dialysis;

Supplementary Table S1. Patient-specific risk factors associated with death-censored technique failure in a competing risk analysis

Supplemental material

CCI, Charlson Comorbidity index.

145

)			
Risk factors	Crude HR (95% CI)	p-value	Adjusted model 1 HR (95% CI)	p-value	Adjusted model 2 HR (95% CI)	p-value
Male sex	1.14 (0.78 - 1.67)	0.51				
Age (10 year)	1.40(1.20 - 1.63)	<0.001				
Employed	0.34 (0.18 - 0.66)	0.001				
CCI						
low	Reference					
intermediate	2.41 (1.24 – 4.67)	0.009				
severe	4.27 (2.22 - 8.19)	<0.001				
BMI						
$< 25 \ { m kg/m^2}$	Reference		Reference			
$25 - 30 \text{ kg/m}^2$	1.33 (0.79 – 2.23)	0.28	$1.16\ (0.68 - 1.97)$	0.59		
$\geq 30 \text{ kg/m}^2$	1.92(1.12 - 3.30)	0.02	1.62 (0.97 - 2.71)	0.07		
APD (vs CAPD)	$0.74\ (0.52 - 1.07)$	0.11	$0.83 \ (0.57 - 1.20)$	0.32	$0.87\ (0.59 - 1.27)$	0.46
Model 1 is adjusted for	tge and sex		and conton DD infanto			

Model 1 is adjusted for age and sex Model 2 is adjusted for age, sex, employment status, BMI, CCI, and center PD volume HR, hazard ratio; BMI, body mass index; APD, automated peritoneal dialysis; CAPD, continuous ambulatory peritoneal dialysis; CCI, Charlson

Supplementary Table S3	. Patient-specific risk factors	s associated w	ith technique failure in a co	mpeting risk	analysis	
Risk factors	Crude SHR (95% CI)	p-value	Adjusted model 1 SHR (95% CI)	p-value	Adjusted model 2 SHR (95% CI)	p-value
Male sex	1.16 (0.93 - 1.46)	0.20				
Age (10 year)	1.11 (1.03 - 1.20)	0.009				
Employed	$0.67\ (0.51-0.89)$	0.006				
CCI						
low	Reference					
intermediate	1.54 (1.11 - 2.15)	0.01				
severe	2.10 (1.49 – 2.95)	<0.001				
BMI						
$< 25 \ { m kg/m^2}$	Reference		Reference			
$25 - 30 \text{ kg/m}^2$	$1.20\ (0.90 - 1.61)$	0.21	$1.14 \ (0.84 - 1.54)$	0.40		
≥ 30 kg/m²	1.23(0.89 - 1.71)	0.20	$1.22\ (0.89 - 1.68)$	0.21		
APD (vs CAPD)	0.63 (0.51 - 0.79)	<0.001	0.65 (0.52 - 0.81)	<0.001	$0.65\ (0.52 - 0.81)$	<0.001
Model 1 is adjusted for ,	ige and sex					

acting sick and more christing failure in a 111 wint for ú. 4 Datio Tabla C2 -

Model 2 is adjusted for age, sex, employment status, BMI, CCI, and center PD volume

SHR, subdistribution hazard ratio; BMI, body mass index; APD, automated peritoneal dialysis; CAPD, continuous ambulatory peritoneal dialysis; CCI, Charlson Comorbidity index.

Supplementa	rv Table	S4. Cente	er charact	teristics
o ap prementen		0 001100	/ endered	ceriocieo

	N (%)
Center type	
Academic	8 (24)
non-academic	25 (76)
PD solution	
Conventional	3 (9)
neutral pH low GDP	30 (91)
Icodextrin	33 (100)
Exit site antibiotic prophylaxis	26 (79)
Antibiotic prophylaxis during PD catheter insertion	33 (100)
Antifungal prophylaxis during antibiotic therapy	2 (6)
PD volume	
< 15 patients	6 (18)
15-25 patients	12 (36)
\geq 26 patients	15 (45)
HHD volume	
< 5 patients	12 (36)
5-10 patients	13 (39)
≥ 11 patients	8 (24)

PD, peritoneal dialysis; GDP, glucose degradation products; HHD, home hemodialysis.



Supplementary Figure S1. Permanent technique failure

Permanent technique failure was defined as a transfer to CHD for \geq 180 days, death on PD or death within 180 days after transfer to CHD.(1)



Supplementary Figure S2. Comparison of causes of early and late permanent technique failure Early PD technique failure is defined as occurrence of permanent technique failure in the first 6 months after start of PD (n=72). Late technique failure is defined as occurrence of permanent technique failure more than 6 months after start of PD (n=182).

	30 day definition = technique failure n = 318	180-day definition = permanent technique failure n = 254
Infections	64 (20%)	41 (16%)
Catheter problems	25 (8%)	13 (5%)
Clearance	37 (12%)	31 (12%)
Leakage	25 (8%)	17 (7%)
Psychosocial	21 (7%)	21 (8%)
EPS	2 (1%)	1 (0%)
Another reason	32 (10%)	14 (6%)
Stop dialysis	17 (5%)	17 (7%)
Death	92 (29%)	95 (37%)
Unknown	3 (1%)	4 (2%)

Supplementary Table S5. Causes of technique failure by definition

Technique failure according to the 30-day definition was defined as a transfer to CHD for \geq 30 days, death on PD or death within 30 days after transfer to CHD.

Permanent technique failure according to the 180-day definition was defined as a transfer to CHD for \geq 180 days, death on PD or death within 180 days after transfer to CHD.

All were in accordance with the standardized definition as proposed by Lan et al.1

1. Lan PG, Clayton PA, Johnson DW, McDonald SP, Borlace M, Badve SV, *et al.* Duration of Hemodialysis Following Peritoneal Dialysis Cessation in Australia and New Zealand: Proposal for a Standardized Definition of Technique Failure. Perit Dial Int. 2016;36(6):623-30.



Part III

Shift towards Health-Related Quality of Life



Chapter 7

Health-Related Quality of Life in home dialysis patients compared to in-center hemodialysis patients: a systematic review and meta-analysis

Anna A. Bonenkamp*, Anita van Eck van der Sluijs*, Tiny Hoekstra, Marianne C.
 Verhaar, Frans J. van Ittersum, Alferso C. Abrahams and Brigit C. van Jaarsveld.
 ** These authors contributed equally to this work.*

Kidney Medicine 2020 Feb 11;2(2):139-154.

Abstract

Rationale & Objective Dialysis patients judge Health-Related Quality of Life (HRQoL) as an essential outcome. Remarkably, little is known about HRQoL differences between home dialysis and in-center hemodialysis (HD) patients across the world.

Study design Systematic review and meta-analysis.

Setting & Study populations Search strategies were performed on the Cochrane Library, Pubmed, and Embase databases between 2007 and 2019. Home dialysis was defined as both peritoneal dialysis and home hemodialysis.

Selection criteria for studies Randomized controlled trials and observational studies that compared HRQoL in home dialysis patients versus in-center HD patients.

Data extraction The data extracted by two authors included: HRQOL scores of different questionnaires, dialysis modality, and subcontinent.

Analytical approach Data was pooled using a random-effects model and results were expressed as standardized mean difference (SMD) with 95% confidence intervals. Heterogeneity was explored using subgroup analyses.

Results Forty-six articles reporting on 41 study populations were identified. Most studies were cross-sectional in design (90%), conducted on PD patients (95%), and used the 12-item or 36-item Short-Form Health Survey questionnaires (83%). More than half of the studies showed moderate or high risk of bias. Pooled analysis of 4,158 home dialysis patients and 7,854 in-center HD patients showed marginally better physical HRQoL score in home dialysis patients compared to in-center HD patients (SMD 0.14, 95%CI 0.04 to 0.24), although heterogeneity was high (I²>80%). In a subgroup analysis, Western European home dialysis patients had higher physical HRQoL score (SMD 0.39, 95%CI 0.17 to 0.61) compared to in-center HD, while home dialysis patients from Latin America had a lower physical score (SMD -0.20, 95%CI -0.28 to -0.12) compared to in-center HD. Mental HRQoL showed no difference in all analyses.

Limitations No randomized controlled trials were found and high heterogeneity among studies existed.

Conclusions Although pooled data showed a marginally better physical HRQoL for home dialysis patients, the quality of design of the included studies was poor. Large prospective studies with adequate adjustments for confounders are necessary to establish whether home dialysis results in better HRQoL. PROSPERO 95985

Summary

Health-Related Quality of Life (HRQoL) is an essential outcome for dialysis patients. However, little is known about differences across the world between HRQoL of home dialysis patients versus in-center hemodialysis patients. A systematic review was conducted, which yielded 46 articles. Subsequently, a meta-analysis showed that home dialysis patients have marginally better physical HRQoL compared to in-center hemodialysis patients. In a subgroup analysis, Western European home dialysis patients had a higher physical score, while home dialysis patients from Latin America had a lower physical score. Mental HRQoL showed no difference. However, the quality of design of the included studies was poor, so large prospective studies are necessary to establish whether home dialysis results in better HRQoL.

Introduction

End Stage Kidney Disease (ESKD) is associated with poor survival. Patients starting on dialysis have a median five-year survival rate of only 45%.¹ Observational studies comparing patients performing home dialysis, mostly peritoneal dialysis (PD), with incenter hemodialysis (ICHD) show comparable survival between groups.²⁻⁴ Therefore, these survival studies will not help patients in choosing a dialysis modality.

Counterintuitive to what some clinicians assume, patients with ESRD consider quality of life far more important than survival.⁵⁻¹⁰ Many patients experience dialysis as a heavy burden; they even have poorer Health-Related Quality of Life (HRQoL) than patients with diabetes or malignancies.^{11,12} Patients also indicate HRQoL aspects as important research topics.^{13,14} This has affected the research performed in the medical field during the last decade, with focus shifting from clinical outcomes to patient reported outcomes.^{15,16} Indeed, the number of articles reporting HRQoL in dialysis patients has multiplied over the last 10 years.

Reducing the impact of ESKD and its treatment on daily life could potentially improve HRQoL. Performing dialysis at home, instead of being treated with ICHD, has the advantage of more independence and flexibility during the day.¹⁷⁻²⁰ Moreover, due to the possibility of self-care and fewer hospital visits with home based therapies, patients are able to return to work and to engage in daily social activities.^{18,21-23} Home hemodialysis (HHD) enables intensifying the dialysis regime, allowing a reduction in medication burden.²⁴ All the aforementioned factors could contribute to an improvement of HRQoL.

Many cross-sectional and some cohort studies from different regions across the world have reported on HRQoL of home dialysis patients in comparison to ICHD patients. Interpretation of these studies are hampered by a large variety in type of questionnaire used and applied study design.²⁵⁻²⁷ In addition, as these studies are conducted in different countries, disparity exists in study populations since the percentage of patients on home dialysis varies across the world. This difference in practice patterns, together with a difference in local cultures, are suggested to influence HRQoL.²⁸ Investigators of 'The Dialysis Outcomes and Practice Patterns Study' found different HRQoL scores between ICHD patients across Japan, Europe, and the United States, after adjustment for several confounders including comorbidities.²⁸ Due to inequalities

among studies, it is difficult to determine whether home dialysis patients have a better HRQoL. Differences in HRQoL of home dialysis patients and ICHD patients should be interpreted in relation to the country of residence.

Hence, a systematic review and meta-analysis was conducted to summarize and evaluate the available studies on HRQoL between home dialysis and ICHD patients, with a special focus on differences across the world.

Methods

Search strategy and selection criteria

The Cochrane Library, Pubmed, and Embase databases were searched for relevant articles using all synonyms and abbreviations of the terms 'dialysis' and 'quality of life' (Table S1). The search was limited to publications during the last 10 years, since the perception of quality of life in patients treated with dialysis changed over time, for example by improved metabolic control over the years.²⁹ After removing the duplicates, two authors (AB and AE) independently performed screening of titles and abstracts according to predetermined in- and exclusion criteria. All articles comparing the HRQoL of adult (i.e. \geq 18 yr.) home dialysis patients with HRQoL of ICHD patients were included. Articles other than randomized controlled trials (RCTs) and observational studies were excluded, such as validation and reliability studies on quality of life questionnaires. In addition, articles in a language other than English were excluded.

The remaining articles were read full text by two authors (AB and AE) and screened for additional references. All articles assessing HRQoL, by applying worldwide most commonly used questionnaires,³⁰ were included (see Table S2). The full text articles were also checked for outdated patient data (data collected before 2007), which was reason for exclusion, and missing HRQoL scores. When no quantitative scores were reported for home dialysis and ICHD patients, the authors were emailed. If they provided the quantitative data, the article was subsequently included in the critical appraisal. Final inclusion was based on consensus between the two authors (AB and AE). In case they failed to reach consensus, a third author (TH) was asked for an opinion that was decisive. The selection process is summarized in Figure 1.

Data extraction

Data extraction was performed and checked by two authors (AB and AE). The included studies were structured according to the dialysis modality, country and subcontinent of conductance, number of participants with characteristics (age, dialysis vintage, and gender), and type of HRQoL questionnaire used. From all studies, HRQoL scores were extracted and evaluated. If no standard deviation (SD) was reported, it was calculated (e.g. from interquartile range (IQR), confidence interval (CI), or standard error) or substituted from another study with similar characteristics.³¹ Subcontinents were classified according to the regional boards of the International Society of Nephrology.³²

For the meta-analysis, the Physical Component Summary (PCS) was used as score for the physical domain and the Mental Component Summary (MCS) for the mental domain. If the summary scores of the Short Form (SF) were not available, the physical functioning score or the mental health score were used respectively. If the World Health Organization Quality of Life (WHOQOL)-BREF was assessed, the physical health score was used for the physical domain and the psychological health score for the mental domain. If the EuroQol-5D (EQ-5D) was reported, the visual analogue scale (VAS) was used for the analysis.

Risk of Bias assessment

After full text screening, articles eligible for critical appraisal were independently appraised by two authors (AB and AE) using criteria based on the Critical Appraisal Skills Programme Cohort Study checklist and the Newcastle-Ottawa Scale.^{33,34} The following criteria were assessed: study design, patient selection, comparability of patients between groups, accurate measurement of outcome, correction for confounding, duration of follow-up, selective reporting, and conflict of interest (details are provided in Table S3). They were scored as + (low risk of bias), - (high risk of bias) or unclear (?) based on consensus between the two authors (AB and AE). In case of disagreement a third opinion (BJ) was decisive. After completing the critical appraisal, the corresponding authors of the articles were contacted if any uncertainty remained (i.e. criteria scored as unclear). Any given comment was taken into account for the final critical appraisal.

Analytical approach

With the extracted HRQoL scores a meta-analysis was performed. Heterogeneity, both in clinical characteristics (e.g. variability in patients) and methodological aspects

(i.e. design, and risk of bias), was explored by visual inspection and quantified by an I² above 75%.³⁵ Significant heterogeneity was expected, due to the use of different types of HRQoL questionnaires and differences between countries regarding practice patterns and accessibility for home dialysis leading to differences between patient populations.²⁸ Therefore, standardized mean difference (SMD) of HRQoL scores and a random effects model were used.

The following subgroup analyses were performed: different subcontinents and subgroups of studies according to overall risk of bias (as scored by authors: low, moderate, or high). When appropriate, type of home dialysis (PD or HHD) was compared to ICHD. Additional analyses were conducted for the following subgroups: type of questionnaire used, different age categories (<45 years, 45 – 60 years, >60 years), and dialysis vintage (<36 months vs. \geq 36 months). Finally, a sensitivity analysis was conducted that excluded articles for which the SD was calculated or substituted. All analyses were performed with Stata/SE 14.1 for Windows.

Protocol and registration

This systematic review was registered in PROSPERO, the International prospective register of systematic reviews. The study protocol can be retrieved from the PROSPERO website (https://www.crd.york.ac.uk/prospero/) using the registration number 95985.

Results

Study selection

The initial literature search was performed on the 21st of November 2017 and was last updated in January 2019. The final search yielded 1,647 articles, after removal of duplicates. Subsequently, articles were excluded based on title and abstract, according to previously determined in- and exclusion criteria. Systematic reviews that were among these articles were checked for references before they were excluded.^{21,25,26,30,36-46} This resulted in one article; however its data collection was performed before 2007 and therefore was excluded.⁴⁷

The full-text of the remaining 80 articles were retrieved and assessed for eligibility. A total of 35 articles were excluded due to the following reasons: comparison group other than ICHD,⁴⁸⁻⁵⁰ groups were not separately presented,⁵¹⁻⁵⁵ unspecified HRQoL

questionnaire, ⁵⁶⁻⁵⁹ HRQoL data exclusively presented in graphs, ⁶⁰⁻⁶² unclear calculation of HRQoL scores, ^{63,64} and outdated population data (data collected before 2007). ⁶⁵⁻⁸⁰ The studies of Garg¹⁷ (FHN trials) and Jardine⁸¹ (ACTIVE dialysis trial) were excluded since they focused on frequent hemodialysis which was not exclusively performed at home. The remaining 45 articles were screened for additional references, resulting in 1 article that was evaluated and included (Figure 1). ⁸²





HRQoL, Health-Related Quality of Life.

* Exclusion criteria: Articles describing data older than 10 years, case-reports, congress abstracts, editorials, language other than English, letters, opinion papers, reviews, validation and reliability studies on quality of life questionnaires.

A total of 46 articles was eligible for critical appraisal.⁸²⁻¹²⁷ The following articles presented overlapping patient data and were appraised as one: Bujang and Liu,^{91,92} Chkhotua and Maglakelidze,^{94,95} Griva and Yang,^{103,104} 2 articles by Kontodimopoulos,^{111,112} and 2 articles by Theofilou,^{120,121} leaving 41 studies for analysis.

Study characteristics

The characteristics of the included studies are described in Table 1. Most of the studies (32%) were conducted in Western Europe, followed by Asia (27%). From the 41 studies included, only 3 compared the HRQoL of HHD patients with ICHD patients,^{82,123,124} while the remaining focused on the comparison PD versus ICHD. The predominantly used questionnaire was the SF, either as a separate questionnaire or as part of the KDQOL (83%).

Mean age of the home dialysis population was 55.9 ± 13.8 years, while the ICHD patients where slightly younger (mean age 54.8 ± 14.1 years). There was a difference in dialysis vintage between both groups with a median of 34.1 months for home dialysis patients (IQR 22.8 - 43.4) and 56.9 months for ICHD patients (IQR 31.0-77.2). The majority (55%) of the total dialysis population was male. One study was conducted in females only.⁸⁷ Half of the home dialysis population was male (range 27-90%) compared to 57% of the ICHD population (range 44%-85%). In the included studies there were no RCTs of ICHD versus home dialysis. Furthermore, most studies had a cross-sectional design, comparing prevalent patients on ICHD with prevalent home dialysis patients.

It should be noted that 4 studies were observational cohort studies with a longitudinal follow-up. Da Silva-Gane *et al.* assessed HRQoL of dialysis patients every 3 months until 12 months after dialysis initiation.⁹⁷ Baseline PCS scores were lower in ICHD patients. However, after a median follow-up period of 14.7 months HRQoL between dialysis modalities was equal. As the follow-up results of PD and ICHD patients are not shown in the article, in the following meta-analysis only baseline data of this study could be used.

ble 1. Study cha	racteristic	s of 41 studies							
hor, year	Home dialysis modality	Country, Subcontinent*	No. of patients (home/ ICHD)	Age, years (SD) (home/ ICHD)	Dialysis vintage, months (SD) (home/ICHD)	HRQoL questionnaire	Physical score, mean (SD) (home/ICHD)	Mental score, mean (SD) (home/ICHD)	Study conclu- sion
Wakeel, 2012	PD	Saudi Arabia, Middle East	100 / 100	51.0 (13.5) / 47.5 (13.8)	34.1 (26.9) / 77.2 (75.5)	KDQOL	47.7 (23.6) / 53.1 (32.0)	61.9 (13.5) / 50.5 (14.8)	Favors PD
vares, 2012	DD	Brazil, Latin America	788 / 1,621	55.6 (15.3) / 48.9 (14.5)	39.7 (42.5) / 53.9 (55.1)	SF	41.0 (9.4) / 43.0 (9.6)	44.7 (8.0) / 44.6 (7.6)	Favors ICHD
apour, 2016	DD	Iran, Middle East	46/46	51.0 (12.5) / 47.8 (10.6)	18.8 (13.7) / 24.4 (14.8)	SF	60.5 (10.4) / 56.2 (10.3)	<i>55.7</i> (7.1) / <i>55.</i> 1 (6.2)	Favors PD
ırata, 2015	DD	Portugal, Western Europe	31/94	N/A	N/A	WHOQOL- Bref	61.7 (12.7) / 43.7 (13.9)	56.1 (11.4) / 46.0 (12.2)	Favors PD
ısok, 2009	DD	Turkey, Eastern Europe	21/24	45.2 (8.9) / 43.1 (12.4)	N/A	SF	43.2 (9.8) / 47.4 (10.2)	44.5 (10.9) / 50.2 (12.6)	N/A
ıykan, 2012	DD	Turkey, Eastern Europe	41 / 42	40.6 (11.9) / 49.1 (12.0)	N/A	SF	53.2 (7.6) / 47.0 (9.2)	45.2 (6.7) / 42.2 (6.7)	N/A
rowiak, 2009	DD	Poland, Eastern Europe	50 / 50	58.9 (13.2) / 59.6 (13.4)	N/A	EQ-5D VAS**	<i>55.</i> 3 (21.7)/ <i>5</i> 3.2 (16.2)	55.3 (21.7)/ 53.2 (16.2)	Equal
own , 2010	DD	UK, Western Europe	70 / 70	73.1 (5.5) / 73.4 (5.1)	30.5 (28.3) / 31.4 (26.5)	SF	36.0 (12.1) / 34.3 (9.7)	55.0 (8.4) / 51.3 (12.9)	Favors PD
ijang , 2015 and u , 2014	DD	Malaysia, Asia	539 / 793	52.8 (15.4) / 55.5 (15.3)	45.6 (37.2) / 91.2 (74.4)	WHOQOL- BREF	55.5 (15.5) / 56.6 (16.1)	60.2 (16.0) / 59.6 (17.3)	Favors PD
len, 2017	DD	China, Asia	103 / 2 <i>5</i> 3	63.1 (12.7) / 56.6 (12.1)	N/A	KDQOL	40.3 (12.0) / 37.4 (12.6)	50.3 (10.0) / 51.0 (10.3)	Favors PD

-

_

Author, year	Home dialysis modality	Country, Subcontinent*	No. of patients (home/ ICHD)	Age, years (SD) (home/ ICHD)	Dialysis vintage, months (SD) (home/ICHD)	HRQoL questionnaire	Physical score, mean (SD) (home/ICHD)	Mental score, mean (SD) (home/ICHD)	Study conclu- sion
Chkhotua, 2011 and Maglakelidze 2011	PD	Georgia, Eastern Europe	43 / 120	N/A	N/A	SF	55.7 (52.2) / 56.9 (53.4)	47.5 (47.9) / 49.9 (51.4)	Equal
Czyzewski, 2014	PD	Poland, Eastern Europe	30 / 40	N/A	39.6778.0	KDQOL	37.5 (10.6) / 34.7 (7.4)	49.9 (7.0) / 43.7 (11.1)	Equal
Da Silva-Gane, 2012	PD	UK, Western Europe	44/80	48.0 (15.6) / 60.6 (14.9)	N/A	SF	30.1 (6.5) / 25.2 (8.8)	45.9 (10.6) / 47.6 (10.7)	Favors PD
De Fijter, 2018	PD	The Netherlands, Western Europe	33 / 42	66.0 (14.0) / 66.0 (11.0)	16/27	KDQOL	43.0 (20.0) / 35.0 (21.0)	56.0 (24.0) / 49.0 (20.0)	Favors PD
Fructuoso, 2011	PD	Portugal, Western Europe	14/37	38.9 (13.3) / 67.3 (14.9)	22.8 (15.6) / 73.2 (78.0)	KDQOL	44.9 (5.6) / 35.9 (9.0)	46.2 (10.2) / 42.6 (12.6)	Favors PD
Garcia-Llana, 2013	PD	Spain, Western Europe	31/30	47.9 (15.9) / 60.6 (16.7)	31.4 (28.6) / 56.9 (81.7)	SF	39.4 (8.7) / 34.3 (8.7)	49.8 (11.5) / 47.1 (10.7)	Favors PD
Ginieri- Coccossis, 2008	PD	Greece, Western Europe	48/41	64.1 (10.4) / 65.3 (8.4)	43.4 (24.0) / 49.8 (30.8)	WHOQOL- Bref	13.5 (2.8) / 12.4 (3.8)	13.2 (3.2) / 12.9 (3.5)	Favors PD
Goncalves, 2015	DD	Brazil, Latin America	116 / 222	58 (13.9) / 54.4 (15.2)	N/A	KDQOL	45.8 / 52.8	44.3 / 56.6	Favors ICHD
Griva, 2014 and Yang, 2015	PD	Singapore, Asia	266 <i>1</i> 236	59.3 (12.5) / 54.4 (10.6)	42.6 (39.4) / 76.4 (66.5)	KDQOL	37.1 (9.7) / 38.9 (9.6)	46.6 (11.2) / 46.3 (10.4)	Favors ICHD
Günalay, 2018	PD	Turkey, Eastern Europe	10 / 50	52.4 (15.1) / 50.0 (18.9)	38.5 (14.2) / 53.5 (48.3)	EQ-5D VAS**	58.1 (13.1) / 66.7 (22.3)	58.1 (13.1) / 66.7 (22.3)	Equal

Author, year	Home dialysis modality	Country, Subcontinent*	No. of patients (home/ ICHD)	Age, years (SD) (home/ ICHD)	Dialysis vintage, months (SD) (home/ICHD)	HRQoL questionnaire	Physical score, mean (SD) (home/ICHD)	Mental score, mean (SD) (home/ICHD)	Study conclu- sion
Ibrahim, 2011	DD	Malaysia, Asia	91 / 183	N/ A	N/A	SF	74.6 / 68.4	77.1 / 70.9	Favors PD
Ikonomou, 2015	PD	Greece, Western Europe	39/90	58.0 (16.0) / 57.9 (13.8)	N/A	SF	42.4 (10.0) / 40.7 (11.3)	52.3 (9.1) / 49.3 (10.3)	Equal
Iyasere, 2016	PD	UK, Western Europe	129 / 122	76.0/75.0	22.0 / 27.5	SF	33.0/ 31.7	49.3 / 50.8	Equal
Kang , 2017	PD	Korea, Asia	366 / 1,250	54.1 (11.9) / 56.4 (13.2)	63.6 (46.8) / 61.2 (55.2)	KDQOL	58.5 (23.0) / 61.9 (21.2)	55.5 (24.9) / 59.8 (21.2)	Favors ICHD
Kim , 2013	PD	Korea, Asia	65 / 172	N/A	N/A	KDQOL	38.7 (9.0) / 39.3 (9.7)	44.8 (6.4) / 44.6 (7.0)	Favors PD
Kontodimo- poulos, 2008 and 2009	PD	Greece, Western Europe	65 / 642	58.7 (12.9) / 58.1 (14.9)	63.6 (67.2) / 74.4 (68.4)	SF	49.2 (30.7) / 49.2 (30.6)	53.0 (26.1) / 55.1 (22.7)	Equal
Nakayama, 2015	PD	Japan, Asia	102/77	62.5 (12.0) / 63.5 (12.4)	N/A	SF	25.4 (25.3) / 32.1 (20.6)	45.6 (12.1) / 46.1 (10.5)	N/A
Neumann, 2018	Q	Germany, Western Europe	153 / 200	59.0 (15.4) / 59.8 (16.0)	N/A	SF	Baseline 38.3 (9.8) / 39.9 (10.8) 12 months 35.4 (11.6) / 37.9 (11.5)	Baseline 52.1 (9.4) / 52.1 (10.0) 12. months 46.1 (11.6)	Equal

Table 1. Study characteristics of 41 studies (continued)

Author, year	Home dialysis modality	Country, Subcontinent*	No. of patients (home/ ICHD)	Age, years (SD) (home/ ICHD)	Dialysis vintage, months (SD) (home/ICHD)	HRQoL questionnaire	Physical score, mean (SD) (home/ICHD)	Mental score, mean (SD) (home/ICHD)	Study conclu- sion
Okpechi, 2013	PD	South Africa, Africa	26 / 56	36.0 (6.1) / 38.6 (10.5)	14.5 (11.6) / 49.8 (71.5)	KDQOL	67.5 (27.5) / 65.4 (53.1)	75.0 (23.5) / 74.6 (21.0)	Equal
Ören, 2013	PD	Turkey, Eastern Europe	12 <i>5 </i> 175	46.4 (14.6) / 47.6 (15.3)	45.4 (34.8) / 94.4 (60.0)	SF	58.4 (25.9) / 48.6 (26.5)	63.3 (18.9) / 57.0 (19.8)	Favors PD
Painter, 2012	ПНН	USA, North America	10/13	42.6 (12.4) / 45.5 (10.4)	33.8 (44.3) / 28.5 (21.2)	KDQOL	Baseline 45.3 (11.3) / 48.8 (10.0) 6 months 49.6 (9.1) /	Baseline 48.1 (14.6) / 51.1 (9.1) 6 months 48.9 (12.6) /	Favors HHD
Ramos, 2015	DD	Brazil, Latin America	60 / 257	56.5 (15.3) / 57.9 (15.9)	N/A	SF	48.4 (7.4) 51.3 (27.8) / 53.5 (29.7)	51.7 (9.6) 71.7 (20.4) / 68.7 (22.6)	Equal
Ruiz de Alegría - Fernández de Retana, 2013	Q	Spain, Western Europe	45 / 53	50.8 (13.3) / 52.3 (13.1)	N/A	SF	3 months 42.6 (8.9) / 40.8 (8.9) 6 months 40.6 (9.8) / 42.2 (9.7) 12 months 43.9 (9.8) / 39 9 (9.7)	3 months 50.5 (13.0) / 46.3 (13.4) 6 months 50.3 (11.6) / 49.3 (11.6) / 12 months 50.5 (11.6) / 49.6 (11.6) /	N/A
Tannor, 2017	PD	South Africa, Africa	48 / 58	36.1 (10.7) / 42.8 (9.8)	26.4 / 72.0	KDQOL	55.5 (21.7) / 54.7 (19.4)	62.7 (19.7) / 68.6 (17.9)	Equal

Table 1. Study characteristics of 41 studies (continued)

1

HRQoL of home dialysis patients worldwide

Author, year	Home dialysis modality	Country, Subcontinent*	No. of patients (home/ ICHD)	Age, years (SD) (home/ ICHD)	Dialysis vintage, months (SD) (home/ICHD)	HRQoL questionnaire	Physical score, mean (SD) (home/ICHD)	Mental score, mean (SD) (home/ICHD)	Study conclu- sion
Theofilou, 2011 and 2013	PD	Greece, Western Europe	60 / 84	64.3 (12.5) / 58.1 (16.1)	38.4 (24.0) / 87.6 (85.2)	WHOQOL- Bref	13.7 (3.0) / 12.7 (3.7)	13.4 (3.1) / 13.3 (3.7)	Favors PD
Turkmen, 2012	PD	Turkey, Eastern Europe	64/90	52.4 (15.3) / 55.0 (15.7)	19.8 (14.3) / 22.7 (13.1)	SF	47.6 (18.5) / 59.4 (20.7)	41.7 (17.2) / 63.9 (20.6)	Favors ICHD
Watanabe, 2014	ДНН	Japan, Asia	46/34	54.0 (8.3) / 57.1 (7.6)	76.8 (68.4) / 88.8 (99.6)	KDQOL	48.7 (9.2) / 37.1 (12.9)	51.2 (8.9) / 49.6 (6.2)	Favors HHD
Wright ^a , 2015	ДНН	USA, North America	22/29	N/A	N/A	KDQOL	40.4 (12.7) / 42.8 (9.8)	50.6 (9.4) / 50.4 (10.0)	Equal
Wright ^{b,} 2015	PD	USA, North America	26/29	N/A	N/A	KDQOL	43.2 (8.8) / 42.8 (9.8)	51.1 (8.2) / 50.4 (10.0)	Equal
Wu , 2013	PD	China, Asia	93/97	54.5 (15.5) / 58.3 (17.5)	25.5/31.0	SF	34.0 (11.9) / 30.5 (14.5)	41.3 (10.0) / 38.5 (12.0)	Equal
Ying, 2014	PD	Malaysia, Asia	73 / 147	N/A	N/A	SF	60.2 (21.9) / 49.6 (20.2)	67.1 (19.4) / 58.0 (20.3)	Favors PD
Yongsiri, 2014	PD	Thailand, Asia	26/34	53.0 (14.4) / 61.1 (15.5)	N/A	WHOQOL- BREF	3.0 (0.9) / 2.9 (0.8)	3.7 (0.7) / 3.7 (0.6)	Equal

Table 1. Study characteristics of 41 studies (continued)

Study conclu- sion	N/A
Mental score, mean (SD) (home/ICHD)	N/A
Physical score, mean (SD) (home/ICHD)	N/A
HRQoL questionnaire	N/A
Dialysis vintage, months (SD) (home/ICHD)	34.1 [#] (22.8- 43.4) 56.9 [#] (31.0- 77.2)
Age, years (SD) (home/ ICHD)	55.9 (13.8) 54.8 (14.1)
No. of patients (home/ ICHD)	4,158 7,854
Country, Subcontinent*	N/A
Home dialysis modality	N/A
Author, year	TOTAL

~ ~
ontinued
C
studies (
-
4
Ŧ
0
characteristics
dy 6
Stu
÷
able
Ĥ

PD, peritoneal dialysis; HHD, home hemodialysis; UK, United Kingdom; USA, United States of America; ICHD, in-center hemodialysis; SD, standard deviation; N/A, not applicable or not available; HRQoL, Health-Related Quality of Life; KDQOL, Kidney Disease Quality Of Life instrument; SF, Short Form; EQ-5D VAS, EuroQol-5D Visual Analogue Scale; WHOQOL-BREF, World Health Organization Quality of Life-BREF. ^{a, b} Wright et al. included 3 patient populations: HHD, PD, and ICHD

* The regional boards of the International Society of Nephrology were used for the classification of countries into subcontinents

** EQ-5D VAS score was used as a surrogate for both physical score and mental score

Median with interquartile range

The study by Neumann *et al.* investigated the change in social networks and social support, and their association with HRQoL, of dialysis patients over a 12 month period.¹¹⁴ The PCS and MCS scores of PD and ICHD patients decreased equally during follow-up. The follow-up HRQoL scores at 12 months were used in this meta-analysis. The study by Painter *et al.* examined exercise capacity after modality switch from ICHD to HHD, yet also assessed HRQoL.⁸² Modality switch was associated with a significant improvement in physical HRQoL scores after 6 months. The follow-up HRQoL scores at 6 months were used in this meta-analysis. The study by Ruiz de Alegría - Fernández de Retana *et al.* related coping mechanisms to HRQoL.¹¹⁸ SF-36 questionnaires were collected at 3, 6, and 12 months after dialysis initiation. Separate HRQoL scores for PD and ICHD were obtained from the author. These unpublished data showed improvement in MCS for ICHD patients, but PCS remained the same in both groups. HRQoL scores 12 months after initiation of dialysis treatment were used in this meta-analysis.

Risk of Bias assessment

The results of the critical appraisal are presented in Table S4. Seventeen of the 41 studies were assessed as having an overall low risk of bias. There was a general lack of adequate presentation of patient characteristics, with 6 studies presenting baseline data without separation by dialysis modality^{86,106,110,126} or no baseline data at all.^{95,96} Few studies adequately adjusted HRQoL scores for confounding between groups.^{84,97,98,108,109,114} Apart from adjustment for confounders, also a stratified analysis was considered as a low risk of bias. HRQoL, as a patient reported outcome measure, should be self-reported or assessed by a trained research-assistant.¹²⁸ For 8 studies it was unknown whether the professional performing the interview was trained to assess HRQoL, leading to potential bias in outcome assessment.^{89,99,102,106,115,116,120,126}

Meta-analysis

The included studies for the meta-analysis compared HRQoL for a total of 4,158 home dialysis patients with 7,854 ICHD patients. The study by Wright *et al.* compared two home dialysis populations (HHD and PD) with ICHD patients and is presented twice in the meta-analysis.¹²⁴

Although heterogeneity was high, HRQoL on the physical domain was marginally better in home dialysis patients compared to ICHD patients with a SMD of 0.14

(95% CI 0.04 to 0.24). The HRQoL on the mental domain was equal between the two groups (SMD 0.06, 95% CI -0.03 to 0.15).

A comparison among subcontinents showed that patients on home dialysis in Western Europe had higher physical HRQoL scores compared to ICHD patients (SMD 0.39, 95% CI 0.17 to 0.61), whereas patients on home dialysis from Latin America had lower physical HRQoL scores (SMD -0.20, 95% CI -0.28 to -0.12) (Figure 2A). The HRQoL on the mental domain showed no difference among the subcontinents (Figure 2B).

If studies were divided according to overall level of bias, an increased risk of bias was associated with an increase in SMD in physical HRQoL (high risk of bias: SMD 0.26, 95% CI -0.01 to 0.52) (Figure 3A). For the mental domain, there was no difference among the different levels of bias (Figure 3B). The subgroup analysis regarding type of home dialysis (PD or HHD) provided no additional insights, recognizing that only 3 studies focused on HHD (data not shown). Heterogeneity remained after all subgroup analyses. Additional analyses regarding type of questionnaire used, different age categories, and dialysis vintage did not alter results nor influenced heterogeneity (Figures S1A, S1B, S2A, and S2B).

The SD for the HRQoL scores in 5 studies had to be calculated, if sufficient data were available,^{95,98,108} or substituted.^{102,106} Also, WHOQOL-BREF scores in 2 studies were transformed into a 100-scale.^{101,121} To further explore the robustness of data, sensitivity analysis was performed which did not change the aforementioned results.



Figure 2A. Meta-analysis of Physical Health-Related Quality of Life among subcontinents. SMD, standardized mean difference; 95% CI, 95% confidence interval; HRQoL, Health-Related Quality of Life.

Study ID	SMD (95% CI)	% Weight
North America I Painter Wright a Subtotal (I-squared = 0.0%, p = 0.799)	-0.25 (-1.08, 0.57) 0.02 (-0.53, 0.57) 0.08 (-0.45, 0.61) -0.00 (-0.35, 0.34)	0.95 1.61 1.69 4.24
Latin America Alvares Goncalves Ramos Subtotal (I–squared = 92.9%, p = 0.000)	0.01 (-0.08, 0.09) -0.63 (-0.86, -0.40) 0.14 (-0.15, 0.42) -0.16 (-0.58, 0.26)	3.54 2.99 2.75 9.28
Western Europe Barata Brown Da Silva–Gane Fjiter Fructuoso Garcia–Llana Ginieri–Coccossis Ikonomou Iyasere Kontodimopoulos Neumann Ruiz de Alegria – Fernandez de Retana Theofilou Subtotal (I–squared = 52.1%, p = 0.015)	$\begin{array}{c} 0.84 \ (0.42, 1.26) \\ 0.34 \ (0.01, 0.67) \\ -0.16 \ (-0.53, 0.21) \\ 0.32 \ (-0.14, 0.78) \\ 0.30 \ (-0.32, 0.92) \\ 0.24 \ (-0.28, 0.55) \\ 0.30 \ (-0.28, 0.55) \\ 0.30 \ (-0.28, 0.55) \\ 0.30 \ (-0.28, 0.55) \\ -0.02 \ (-0.35, 0.16) \\ -0.03 \ (-0.24, 0.18) \\ 0.08 \ (-0.32, 0.48) \\ 0.05 \ (-0.28, 0.38) \\ 0.13 \ (-0.27) \end{array}$	2.11 2.50 2.33 1.95 1.41 1.78 2.12 2.29 2.91 2.87 3.08 2.20 2.51 30.07
Eastern Europe	-0.48 (-1.07, 0.12) 0.45 (0.01, 0.88) 0.11 (-0.28, 0.50) -0.05 (-0.40, 0.30) 0.65 (0.16, 1.13) -0.41 (-1.09, 0.28) -1.15 (-1.50, -0.81) 0.32 (0.09, 0.55) -0.06 (-0.49, 0.37)	1.48 2.04 2.23 2.43 1.85 1.24 2.44 2.99 16.69
Africa Okpechi Tannor Subtotal (I-squared = 14.6%, p = 0.279)	0.02 (-0.45, 0.48) -0.31 (-0.70, 0.07) -0.17 (-0.50, 0.15)	1.92 2.26 4.18
Middle East Al Wakeel Atapour Subtotal (I–squared = 87.3%, p = 0.005)	0.80 (0.52, 1.09) 0.09 (0.32, 0.50) 0.46 (0.24, 1.16)	2.71 2.16 4.87
Asia Bujang and Liu Chen Griva and Yang Ibrahim Kang Kim Nakayama Watanabe Wu Yung Yongsiri Subtotal (I-squared = 68.0%, p = 0.001)	$\begin{array}{c} 0.04 \ (-0.07, 0.15) \\ -0.07 \ (-0.30, 0.16) \\ 0.02 \ (-0.15, 0.20) \\ 0.31 \ (0.06, 0.57) \\ -0.19 \ (-0.31, -0.08) \\ 0.03 \ (-0.26, 0.31) \\ -0.04 \ (-0.34, 0.25) \\ 0.25 \ (-2.46, 0.65) \\ 0.25 \ (-2.46, 0.65) \\ 0.25 \ (-2.56, 0.43) \\ 0.07 \ (-0.05, 0.19) \end{array}$	3.47 3.00 3.23 2.88 3.45 2.73 2.68 2.01 2.73 2.73 2.73 2.73 1.75 30.66
Overall (I-squared = 78.1%, p = 0.000)	0.06 (-0.03, 0.15)	100.00
NOTE: Weights are from random effects analysis	1 1	
-1.5 -15 0 .5 Better HRQoL in-center hemodialysis Better HRQoL home	1 1.5 dialysis	

Figure 2B. Meta-analysis of Mental Health-Related Quality of Life among subcontinents. SMD, standardized mean difference; 95% CI, 95% confidence interval; HRQoL, Health-Related Quality of Life.

Study								SMD (95% CI)	% Weight
Low									
Al Wakeel				€—i -				-0.19 (-0.47, 0.09)	2.74
Alvares				H II.				-0.20 (-0.29, -0.12)	3.37
Atapour				- H+-	+	_		0.42 (0.01, 0.83)	2.19
Brown								0.16 (-0.18, 0.49)	2.51
Bujang and Liu				- H -T				-0.07 (-0.18, 0.04)	3.32
Chen				- 14 4				0.23 (0.00, 0.46)	2.93
Da Silva–Gane				- I G				0.61 (0.23, 0.98)	2.33
Filter						_		0.39 (-0.07, 0.85)	2.01
Griva and Yang				- I -				-0.19(-0.36 -0.01)	3.12
lkonomou				11 I I I I I I I I I I I I I I I I I I				0.16 (-0.22, 0.52)	2 22
lyacara				1.1				0.10 (-0.22, 0.33)	2.55
Kentedimeneules				10.0	10			0.13 (=0.12, 0.36)	2.00
Nelsuame					- C			0.00 (-0.26, 0.26)	2.03
Nakayama				a - 1				-0.29 (-0.58, 0.01)	2.00
Neumann				- L.				-0.22(-0.43, -0.01)	3.00
Painter				101				0.15 (-0.68, 0.97)	1.04
Ramos			_	-	1000			-0.07 (-0.36, 0.21)	2.72
Ruiz de Alegria – Fernandez de Retana						-		0.41 (0.01, 0.82)	2.23
Subtotal (I-squared = 71.4%, p = 0.000)				9				0.03 (-0.08, 0.14)	44.18
Moderate				- i					
Barata				1			+>	1.32 (0.88, 1.76)	2.09
Basok			+					-0.42 (-1.01, 0.17)	1.58
Baykan								0.73 (0.29, 1.18)	2.07
Borowiak			_					0.11 (-0.28, 0.50)	2.27
Chkhotua and Maglakelidze					-			-0.02 (-0.37, 0.33)	2.45
Czvzewski					•	-		0.31 (-0.16, 0.79)	1.95
Garcia–I Jana				<u> </u>				0.59 (0.08, 1.10)	1.82
Ginieri-Coccossis				<u> </u>				0.48 (0.06, 0.90)	2 15
Kang				A				-0.16(-0.27 -0.04)	3 30
Kim				1.1				-0.06 (-0.25, 0.32)	2 70
Tapper				The second	1.0			-0.00 (-0.55, 0.22)	2.70
Tudes en								0.04 (=0.34, 0.42)	2.51
Turkmen								-0.60 (-0.92, -0.27)	2.55
Watanabe				- L	_			1.06 (0.59, 1.54)	1.96
Wright a								-0.22 (-0.77, 0.34)	1.69
Wright b				- 10 A M				0.04 (-0.49, 0.57)	1.77
Wu				- 125	•			0.26 (-0.02, 0.55)	2.70
Yongsiri			_					0.12 (-0.39, 0.63)	1.83
Subtotal (I-squared = 84.3%, p = 0.000)					>			0.20 (-0.02, 0.42)	37.17
High				1					
Fructuoso			-	. 13				1.09 (0.44, 1.74)	1.42
Goncalves				HI).		00000		-0.26 (-0.49, -0.04)	2.94
Günalay								-0.41 (-1.09, 0.28)	1.34
Ibrahim				- 1	•			0.30 (0.04, 0.55)	2.84
Okpechi			_	- 10				0.05 (-0.42, 0.51)	1.99
Theofilou				- E-				0.46 (0.13.0.80)	2.50
Ying				1 2 -		-		0.51 (0.23, 0.79)	2.71
Ören								0 37 (0 14 0 60)	2.02
Subtotal (I-squared = 80.8%, p = 0.000)				V	>			0.26 (-0.01, 0.52)	18.65
Overall (I-squared = 81.3%, p = 0.000)				6				0.14 (0.04, 0.24)	100.00
NOTE: Weights are from random effects an	alvsis			Ť				and formal out of	100100
inone magnits are nonnandon ellects al	1419313	1			1				
	-1.5	-1	5	0	.5	1	1.5		

Figure 3A. Meta-analysis of Physical Health-Related Quality of Life among level of bias. SMD, standardized mean difference; 95% CI, 95% confidence interval; HRQoL, Health-Related Quality of Life.

ID						SMD (95% CI)	% Weigh
Low							
Al Wakeel				. —		0.80 (0.52, 1.09)	2.71
Alvares				-		0.01 (-0.08, 0.09)	3.54
Atapour				•		0.09 (-0.32, 0.50)	2.16
Brown						0.34 (0.01, 0.67)	2.50
Bujang and Liu				-		0.04 (-0.07.0.15)	3.47
Chen				<u> </u>		-0.07(-0.30,0.16)	3.00
Da Silva-Gane						-0.16(-0.53,0.21)	2 33
Filter				1 IN		0.32 (_0.14.0.79)	1.05
Critica and Vana						0.02 (0.15, 0.20)	2 22
lkonomou						0.02 (-0.02 0.69)	3.23
Ikonomou			1.1			0.30 (-0.08, 0.68)	2.29
lyasere				1		-0.12(-0.36, 0.13)	2.91
Kontodimopoulos				1		-0.09 (-0.35, 0.16)	2.87
Nakayama				1		-0.04 (-0.34, 0.25)	2.68
Neumann				-		-0.03 (-0.24, 0.18)	3.08
Painter						-0.25 (-1.08, 0.57)	0.95
Ramos						0.14 (-0.15, 0.42)	2.75
Ruiz de Alegria – Fernandez de Retana				•		0.08 (-0.32, 0.48)	2.20
Subtotal (I-squared = 60.3%, p = 0.001)			- 1			0.07 (-0.02, 0.17)	44.62
Moderate				i			
Barata				·	+	0.84 (0.42, 1.26)	2.11
Basok				<u> </u>		-0.48 (-1.07, 0.12)	1.48
Baykan					_	0.45 (0.01, 0.88)	2.04
Borowiak				•		0.11 (-0.28, 0.50)	2.23
Chkhotua and Maglakelidze						-0.05 (-0.40, 0.30)	2.43
Czvzewski						0.65 (0.16, 1.13)	1.85
Garcia–Llana			· · · ·			0.24 (-0.26.0.75)	1.78
Ginieri-Coccossis						0.13 (-0.28, 0.55)	2 12
Kang						-0.19(-0.31 -0.08)	3.45
Kim				<u>.</u>		0.03 (-0.36 0.31)	2 72
Tannar		_	101			0.03 (-0.20, 0.31)	2.75
Turden en						-0.31(-0.70, 0.07)	2.20
Turkmen						-1.15 (-1.50, -0.81)	2.44
Watanabe				- M		0.20 (-0.24, 0.65)	2.01
Wright a						0.02 (-0.53, 0.57)	1.61
Wright b						0.08 (-0.45, 0.61)	1.69
Wu						0.25 (-0.03, 0.54)	2.73
Yongsiri						-0.08 (-0.59, 0.43)	1.75
Subtotal (I-squared = 81.8%, p = 0.000)				\geq		0.04 (-0.17, 0.24)	36.69
High							
Fructuoso						0.30 (-0.32, 0.92)	1.41
Goncalves			_			-0.63 (-0.86, -0.40)	2.99
Günalay				<u> </u>		-0.41 (-1.09, 0.28)	1.24
Ibrahim						0.31 (0.06, 0.57)	2.88
Okpechi						0.02 (-0.45, 0.48)	1.92
Theofilou				•		0.05 (-0.28, 0.38)	2.51
Ying						0.45 (0.17, 0.74)	2.73
Ören						0.32 (0.09.0.55)	2.99
Subtotal (I-squared = 86.9%, p = 0.000)			\sim			0.06 (-0.26, 0.38)	18.69
Overall (I–squared = 78.1%, p = 0.000)				5		0.06 (-0.03, 0.15)	100.0
NOTE: Weights are from random effects a	nalysis			T			
	Т	1		·		1	
	-1.5	-1	-5 0	5	1	15	

Figure 3B. Meta-analysis of Mental Health-Related Quality of Life among level of bias. SMD, standardized mean difference; 95% CI, 95% confidence interval; HRQoL, Health-Related Quality of Life.

Discussion

This meta-analysis shows a better physical HRQoL for patients treated with a form of home dialysis compared to ICHD patients, while the mental HRQoL is comparable between these two patient groups. However, higher physical HRQoL scores in home dialysis patients were only found in Western Europe. Home dialysis patients from Latin America were found to have poorer physical HRQoL compared to ICHD patients. No studies were conducted in Oceania or Russia and only a few in Africa and the Middle East, hampering the comparison regarding HRQoL in the dialysis population worldwide. Furthermore, it should be noted that included studies were generally low in quality and showed high heterogeneity. Therefore, the conclusion regarding better HRQoL of home dialysis patients compared to ICHD patients lacks the necessary robustness.

The finding that home dialysis patients from Western Europe had better physical HRQoL compared to ICHD patients, could be explained by the fact that PD patients from some of the Western European studies were younger due to practice patterns, suggestive for confounding by indication.^{97,99,100} Although most studies performed statistical adjustments of their analyses, important residual confounding between these patient groups might still be present. In contrast to West-European home dialysis patients, those from Latin America were found to have a poorer physical HRQoL. However, these results could also be subject to confounding by indication since in Brazil, the country where these studies were conducted, it is common practice to perform PD only if patients are not eligible for ICHD.⁸⁴ Brazilian ICHD patients in general.^{84,102} This was emphasized by Ramos *et al.* as in this study, PD and ICHD patients were more comparable and physical HRQoL scores were found to be equal.¹¹⁷

The differences in HRQoL of dialysis patients across the world could also be explained by differences in access to dialysis. Liyanage *et al.* modelled inaccessibility among countries and estimated that at least 47% and at most 73% of the world population has no access to renal replacement therapy (RRT). In Latin America, up to 52% of ESKD patients have no access to dialysis, while Africa and Asia have the highest inaccessibility rates, 83% and 91% respectively.¹²⁹ In South-Africa more than half of the patients in need for RRT cannot be treated.^{130,131} Due to limited resources, prolonged maintenance dialysis is not applied and only patients suitable for transplantation are eligible for RRT. As a result, the elderly or unemployed and patients with diabetes or drug abuse are hardly accepted for dialysis treatment.^{130,131} In India, less than 10% of patients start RRT and yet more than two-thirds cease dialysis treatment due to financial problems, often within 3 months. Most dialysis facilities belong to private hospitals and although PD has gained popularity, due to financial restrictions both home dialysis and ICHD are reserved for the rich minority.¹³² In most countries of North and South Asia dialysis care is publicly funded, as is most common in the rest of the world, whereas only 31% of countries in Southeast Asia provide free publicly funded dialysis care.¹³³ Particularly patients from low income countries worldwide depend on private funding.^{133,134} In high income countries inaccessibility is very low, with a maximum of 30%, in comparison to 98% in low income countries.^{129,135} Due to these accessibility issues, the dialysis patients from high income countries (e.g. Western Europe) substantially differ from patients worldwide, which could influence HRQoL scores importantly.

This meta-analysis also underscores the effect of bias in HRQoL. A high risk of bias was associated with better HRQoL in favor of home dialysis if compared to studies with low risk of bias. Remarkably, in all studies with a high risk of bias HRQoL questionnaires were not completed by patients themselves, yet were administered by researchers for whom it was unclear whether they had been trained. In the manual of the Short Form it is stated that the questionnaire should be completed by the patient alone, prior to any contact with the clinician, to avoid influencing the patient and reduce risk of socially desirable answers.¹²⁸ Hood *et al.* has found that assessment by an interviewer is a potential risk of significant bias.¹³⁶ The aforementioned conclusion is confirmed by the results of this meta-analysis.

No RCTs with randomization between home and in-center dialysis were found in the literature search, as previous experiences have learned that a patient's choice between home dialysis and ICHD is too fundamental to let it be determined by fate.^{20,137} In this meta-analysis most studies had a cross-sectional design and did not adjust for confounding even though populations were not comparable at baseline. However, patients performing home dialysis are principally different from ICHD patients. Therefore, in cross-sectional studies the observed associations are less likely to be causative. Korevaar *et al.* showed that patients starting home dialysis had higher HRQoL scores than ICHD patients even in adjusted analysis¹³⁸, while Manns *et al.* reported that *choosing* home dialysis improved HRQoL even prior to initiation of

home dialysis.¹³⁹ The prospective studies in this meta-analysis had a follow-up period of 6 to 12 months. However, it might take longer for patients to return to social activities and work, two factors suggested to be of major influence on HRQoL.^{18,21-23} Therefore, prospective studies with at least one year of follow-up will be necessary to provide a valid assessment of HRQoL of home dialysis patients.

Unfortunately, few studies reported on disease specific domains, whereas dialysis modality possibly has a greater impact on specific complaints or domains than on generic physical and mental HRQoL scores.^{140,141} Future studies should also incorporate disease specific domains as outcome measure.

The most important limitation of this meta-analysis is the high heterogeneity among studies. High heterogeneity remained despite several subgroup analyses, emphasizing the clinical and methodological diversity among studies. Yet, this systematic review and meta-analysis provides a detailed overview of current literature on HRQoL of home dialysis patients across the world, while previous reviews were unable to provide such a detailed insight.²⁵⁻²⁷ Another limitation was that only three studies focused on HHD, illustrating the knowledge gap regarding this modality.

In conclusion, although pooled data in this meta-analysis shows a marginally better physical HRQoL for home dialysis patients; the quality of design of the included studies is poor and large heterogeneity among studies exist. Therefore, no definitive conclusions on HRQoL of patients treated with home dialysis can be drawn. Large prospective studies with adequate follow-up and adjustments for confounders are necessary to evaluate HRQoL of home dialysis patients.

References

- 1. Kramer A, Pippias M, Noordzij M, et al. The European Renal Association European Dialysis and Transplant Association (ERA-EDTA) Registry Annual Report 2015: a summary. Clin Kidney J. 2018;11(1):108-122.
- van de Luijtgaarden MWM, Jager KJ, Segelmark M, et al. Trends in dialysis modality choice and related patient survival in the ERA-EDTA Registry over a 20-year period. Nephrol Dial Transplant. 2016;31:120-128.
- 3. Merchant AA, Quinn RR, Perl J. Dialysis modality and survival: does the controversy live on? Curr Opin Nephrol Hypertens. 2015;24(3):276-283.
- 4. Marshall MR, Walker RC, Polkinghorne KR, et al. Survival on Home Dialysis in New Zealand. Plos One. 2014;9(5):1-11.
- Verberne WR, Das-Gupta Z, Allegretti AS, et al. Development of an International Standard Set of Value-Based Outcome Measures for Patients With Chronic Kidney Disease: A Report of the International Consortium for Health Outcomes Measurement (ICHOM) CKD Working Group. Am J Kidney Dis. 2019;73(3):372-384.
- 6. Lee MB, Bargman JM. Survival by Dialysis Modality Who Cares? CJASN. 2016.
- 7. Finkelstein FO. Performance measures in dialysis facilities: what is the goal? Clin J Am Soc Nephrol. 2015;10(1):156-158.
- 8. Nissenson AR. Improving outcomes for ESRD patients: shifting the quality paradigm. Clin J Am Soc Nephrol. 2014;9(2):430-434.
- 9. Morton RL, Snelling P, Webster AC, et al. Factors influencing patient choice of dialysis versus conservative care to treat end-stage kidney disease. CMAJ. 2012;184(5):E277-E283.
- 10. Manera KE, Tong A, Craig JC, et al. An international Delphi survey helped develop consensus-based core outcome domains for trials in peritoneal dialysis. Kidney Int. 2019.
- 11. Mittal SK, Ahern L, Flaster E, et al. Self-assessed physical and mental function of haemodialysis patients. Nephrol Dial Transplant. 2001;16(7):1387-1394.
- 12. van Sandwijk MS, Al Arashi D, van de Hare FM, et al. Fatigue, anxiety, depression and quality of life in kidney transplant recipients, haemodialysis patients, patients with a haematological malignancy and healthy controls. Nephrol Dial Transplant. 2019;34(5):833-838.
- Tong A, Sainsbury P, Carter SM, et al. Patients' priorities for health research: focus group study of patients with chronic kidney disease. Nephrol Dial Transplant. 2008;23(10):3206-3214.
- 14. Manns B, Hemmelgarn B, Lillie E, et al. Setting research priorities for patients on or nearing dialysis. Clin J Am Soc Nephrol. 2014;9(10):1813-1821.
- Perl J, Dember LM, Bargman JM, et al. The Use of a Multidimensional Measure of Dialysis Adequacy-Moving beyond Small Solute Kinetics. Clin J Am Soc Nephrol. 2017;12(5):839-847.
- 16. Black N, Burke L, Forrest CB, et al. Patient-reported outcomes: pathways to better health, better services, and better societies. Qual Life Res. 2016;25(5):1103-1112.
- 17. Garg AX, Suri RS, Eggers P, et al. Patients receiving frequent hemodialysis have better health-related quality of life compared to patients receiving conventional hemodialysis. Kidney Int. 2017;91(3):746-754.
- 18. Xi W, Singh PM, Harwood L, et al. Patient experiences and preferences on short daily and nocturnal home hemodialysis. Hemodial Int. 2013;17(2):201-207.
- Cases A, Dempster M, Davies M, et al. The experience of individuals with renal failure participating in home haemodialysis: an interpretative phenomenological analysis. J Health Psychol. 2011;16(6):884-894.
- 20. Suri RS, Garg AX, Chertow GM, et al. Frequent Hemodialysis Network (FHN) randomized trials: Study design. Kidney International. 2007;71(4):349-359.
- 21. Miller BW, Himmele R, Sawin DA, et al. Choosing Home Hemodialysis: A Critical Review of Patient Outcomes. Blood Purif. 2018;45(1-3):224-229.
- 22. Ageborg M, Allenius BL, Cederfjall C. Quality of life, self-care ability, and sense of coherence in hemodialysis patients: a comparative study. Hemodial Int. 2005;9:S8-14.
- 23. Piccoli GB, Bechis F, Iacuzzo C, et al. Why Our Patients Like Daily Hemodialysis. Hemodial Int. 2000;4:47-50.
- 24. Abdel-Kader K, Unruh ML. Benefits of short daily home hemodialysis in the FREEDOM Study: is it about person, place, time, or treatment? Kidney Int. 2012;82(5):511-513.
- 25. Zazzeroni L, Pasquinelli G, Nanni E, et al. Comparison of Quality of Life in Patients Undergoing Hemodialysis and Peritoneal Dialysis: a Systematic Review and Meta-Analysis. Kidney Blood Press Res. 2017;42(4):717-727.
- 26. Ho YF, Li IC. The influence of different dialysis modalities on the quality of life of patients with end-stage renal disease: A systematic literature review. Psychol Health. 2016;31(12):1435-1465.
- 27. Liem YS, Bosch JL, Arends LR, et al. Quality of life assessed with the Medical Outcomes Study Short Form 36-Item Health Survey of patients on renal replacement therapy: a systematic review and meta-analysis. Value Health. 2007;10(5):390-397.
- Fukuhara S, Lopes AA, Bragg-Gresham JL, et al. Health-related quality of life among dialysis patients on three continents: the Dialysis Outcomes and Practice Patterns Study. Kidney Int. 2003;64(5):1903-1910.
- 29. Mazairac AH, De Wit GA, Penne EL, et al. Changes in quality of life over time—Dutch haemodialysis patients and general population compared. Nephrol Dial Transplant. 2011;26:1984-1989.
- 30. Wyld M, Morton RL, Hayen A, et al. A Systematic Review and Meta-Analysis of Utility-Based Quality of Life in Chronic Kidney Disease Treatments. PLOS. 2012;9(9):1-10.
- 31. Furukawa TA, Barbui C, Cipriani A, et al. Imputing missing standard deviations in metaanalyses can provide accurate results. J Clin Epidemiol. 2006;59(1):7-10.
- 32. International Society of Nephrology. Available at: https://www.theisn.org/about-isn/ regions Accessed: 28-5-2019.
- 33. Critical Appraisal Skills Programme. (2017). CASP (Cohort Study) Checklist. [online] Available at: https://casp-uk.net/wp-content/uploads/2018/01/CASP-Cohort-Study-Checklist_2018.pdf. Accessed: 14-2-2018.
- 34. Wells GA, Shea B, O'Connell D, et al. The Newcastle-Ottawa quality assessment scale (Cohort studies). Available at: http://www.ohri.ca/programs/clinical_epidemiology/ nosgen.pdf Accessed: 14-2-2018.
- 35. Higgins JPT, Green S. Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration. 2011.

- Walker RC, Howard K, Morton RL. Home hemodialysis: A comprehensive review of patient-centered and economic considerations. ClinicoEconomics and Outcomes Research. 2017;9:149-161.
- Homaie RE, Mostafavi H, Delavari S, et al. Health-related Quality of Life in Patients on Hemodialysis and Peritoneal Dialysis: a Meta-Analysis of Iranian Studies. Iran J Kidney Dis. 2015;9(5):386-393.
- Ishani A, Slinin Y, Greer N, et al. Comparative Effectiveness of Home-Based Kidney Dialysis Versus In-Center or Other Outpatient Kidney Dialysis Locations – A Systematic Review. Washington (DC): Department of Veterans Affairs (US), 2015.
- Palmer SC, Palmer AR, Craig JC, et al. Home versus in-centre haemodialysis for end-stage kidney disease. Cochrane Database Syst Rev. 2014(11):CD009535.
- 40. Avramovic M, Stefanovic V. Health-related quality of life in different stages of renal failure. Artif Organs. 2012;36(7):581-589.
- 41. Boateng EA, East L. The impact of dialysis modality on quality of life: a systematic review. J Ren Care. 2011;37(4):190-200.
- Liem YS, Bosch JL, Hunink MG. Preference-based quality of life of patients on renal replacement therapy: a systematic review and meta-analysis. Value Health. 2008;11(4):733-741.
- 43. Aguiar R, Pei M, Qureshi AR, et al. Health-related quality of life in peritoneal dialysis patients: A narrative review. Semin Dial. 2019;32(5):452-462.
- 44. Li H, Xie L, Yang J, et al. Symptom burden amongst patients suffering from end-stage renal disease and receiving dialysis: A literature review. Int J Nurs Sci. 2018;5(4):427-431.
- 45. Queeley GL, Campbell ES. Comparing treatment modalities for end-stage renal disease: A meta-analysis. Value in Health. 2014;17(3):A290.
- Valentijn PP, Pereira FA, Ruospo M, et al. Person-Centered Integrated Care for Chronic Kidney Disease: A Systematic Review and Meta-Analysis of Randomized Controlled Trials. Clin J Am Soc Nephrol. 2018;13(3):375-386.
- 47. Arenas VG, Barros LFNM, Lemos FB, et al. Quality of Life: comparison between patients on automated peritoneal dialysis and patients on hemodialysis. Acta Paul Enferm. 2009;22:535-539.
- 48. Zhang J, Huang C, Li Y, et al. Health-related quality of life in dialysis patients with constipation: a cross-sectional study. Patient Prefer Adherence. 2013;7:589-594.
- 49. Osthus TB, Preljevic V, Sandvik L, et al. Renal transplant acceptance status, health-related quality of life and depression in dialysis patients. J Ren Care. 2012;38(2):98-106.
- Fong E, Bargman JM, Chan CT. Cross-sectional comparison of quality of life and illness intrusiveness in patients who are treated with nocturnal home hemodialysis versus peritoneal dialysis. Clinical Journal of the American Society of Nephrology. 2007;2(6):1195-1200.
- Dodson S, Osicka T, Huang L, et al. Multifaceted Assessment of Health Literacy in People Receiving Dialysis: Associations With Psychological Stress and Quality of Life. J Health Commun. 2016;21(sup2):91-98.
- 52. Murali R, Sathyanarayana D, Muthusethupathy M. Assessment of quality of life in chronic kidney disease patients using the kidney disease quality of life-short formtm questionnaire in indian population: A community based study. Asian Journal of Pharmaceutical and Clinical Research. 2015;8(1):271-274.

- 53. Davison SN, Jhangri GS. Existential and religious dimensions of spirituality and their relationship with health-related quality of life in chronic kidney disease. Clinical Journal of the American Society of Nephrology. 2010;5(11):1969-1976.
- 54. Chiu Y-W, Teitelbaum I, Misra M, et al. Pill burden, adherence, hyperphosphatemia, and quality of life in maintenance dialysis patients. Clinical Journal of the American Society of Nephrology. 2009;4(6):1089-1096.
- 55. Bohlke M, Nunes DL, Marini SS, et al. Predictors of quality of life among patients on dialysis in southern Brazil. Sao Paulo Med J. 2008;126(5):252-256.
- Laudanski K, Nowak Z, Niemczyk S. Age-related differences in the quality of life in endstage renal disease in patients enrolled in hemodialysis or continuous peritoneal dialysis. Med Sci Monit. 2013;19:378-385.
- 57. Theofilou P. Quality of life and mental health in hemodialysis and peritoneal dialysis patients: the role of health beliefs. Int Urol Nephrol. 2012;44(1):245-253.
- 58. Panagopoulou A, Hardalias A, Berati S, et al. Psychosocial issues and quality of life in patients on renal replacement therapy. Saudi J Kidney Dis Transpl. 2009;20(2):212-218.
- 59. Dąbrowska-Bender M, Dykowska G, Żuk W, et al. The impact on quality of life of dialysis patients with renal insufficiency. Patient Preference and Adherence. 2018;12:577-583.
- 60. Martinez-Sanchis S, Bernal MC, Montagud JV, et al. Quality of life and stressors in patients with chronic kidney disease depending on treatment. Span J Psychol. 2015;18:E25.
- 61. Peng YS, Chiang CK, Hung KY, et al. Comparison of self-reported health-related quality of life between Taiwan hemodialysis and peritoneal dialysis patients: a multi-center collaborative study. Qual Life Res. 2011;20(3):399-405.
- 62. Conde SA, Fernandes N, Santos FR, et al. Cognitive decline, depression and quality of life in patients at different stages of chronic kidney disease. J Bras Nefrol. 2010;32(3):242-248.
- 63. Aghakhani N, Nia HS, Zadeh SS, et al. Quality of life during hemodialysis and study dialysis treatment in patients referred to teaching hospitals in Urmia-Iran in 2007. Caspian Journal of Internal Medicine. 2011;2(1):183-188.
- 64. de Abreu MM, Walker DR, Sesso RC, et al. Health-related quality of life of patients recieving hemodialysis and peritoneal dialysis in Sao Paulo, Brazil: a longitudinal study. Value Health. 2011;14(5 Suppl 1):S119-S121.
- 65. Makkar V, Kumar M, Mahajan R, et al. Comparison of Outcomes and Quality of Life between Hemodialysis and Peritoneal Dialysis Patients in Indian ESRD Population. J Clin Diagn Res. 2015;9(3):OC28-OC31.
- 66. Osthus TB, Preljevic VT, Sandvik L, et al. Mortality and health-related quality of life in prevalent dialysis patients: Comparison between 12-items and 36-items short-form health survey. Health Qual Life Outcomes. 2012;10:46.
- 67. Izbirak G, Akan H, Mistik S, et al. Comparison of health-related quality of life of patients on hemodialysis and continuous ambulatory peritoneal dialysis. Turkiye Klinikleri Journal of Medical Sciences. 2010;30(5):1595-1602.
- 68. Thong MS, van DS, Noordzij M, et al. Symptom clusters in incident dialysis patients: associations with clinical variables and quality of life. Nephrol Dial Transplant. 2009;24(1):225-230.
- 69. Noshad H, Sadreddini S, Nezami N, et al. Comparison of outcome and quality of life: haemodialysis versus peritoneal dialysis patients. Singapore Med J. 2009;50(2):185-192.

- Hallinen T, Soini EJ, Martikainen JA, et al. Costs and quality of life effects of the first year of renal replacement therapy in one Finnish treatment centre. J Med Econ. 2009;12(2):136-140.
- 71. Abdel-Kader K, Myaskovsky L, Karpov I, et al. Individual quality of life in chronic kidney disease: influence of age and dialysis modality. Clin J Am Soc Nephrol. 2009;4(4):711-718.
- 72. Timmers L, Thong M, Dekker FW, et al. Illness perceptions in dialysis patients and their association with quality of life. Psychol Health. 2008;23(6):679-690.
- Shrestha S, Ghotekar LR, Sharma SK, et al. Assessment of quality of life in patients of end stage renal disease on different modalities of treatment. JNMA J Nepal Med Assoc. 2008;47(169):1-6.
- 74. Mau LW, Chiu HC, Chang PY, et al. Health-related quality of life in Taiwanese dialysis patients: effects of dialysis modality. Kaohsiung J Med Sci. 2008;24(9):453-460.
- Malmstrom RK, Roine RP, Heikkila A, et al. Cost analysis and health-related quality of life of home and self-care satellite haemodialysis. Nephrol Dial Transplant. 2008;23(6):1990-1996.
- 76. Sayin A, Mutluay R, Sindel S. Quality of life in hemodialysis, peritoneal dialysis, and transplantation patients. Transplant Proc. 2007;39(10):3047-3053.
- 77. Molsted S, Prescott L, Heaf J, et al. Assessment and clinical aspects of health-related quality of life in dialysis patients and patients with chronic kidney disease. Nephron Clinical Practice. 2007;106(1):c24-c33.
- 78. Lausevic M, Nesic V, Stojanovic M, et al. Health-related quality of life in patients on peritoneal dialysis in Serbia: comparison with hemodialysis. Artif Organs. 2007;31(12):901-910.
- 79. Kutner NG, Zhang R, Huang Y, et al. Association of sleep difficulty with Kidney Disease Quality of Life cognitive function score reported by patients who recently started dialysis. Clinical Journal of the American Society of Nephrology. 2007;2(2):284-289.
- Kalender B, Ozdemir AC, Dervisoglu E, et al. Quality of life in chronic kidney disease: Effects of treatment modality, depression, malnutrition and inflammation. International Journal of Clinical Practice. 2007;61(4):569-576.
- Jardine MJ, Gray NA, De Zoysa J, et al. Design and participant baseline characteristics of 'A Clinical Trial of IntensiVE Dialysis': The ACTIVE Dialysis Study. Nephrology. 2015;20:257-265.
- 82. Painter P, Krasnoff JB, Kuskowski M, et al. Effects of modality change on health-related quality of life. Hemodial Int. 2012;16(3):377-386.
- 83. Al Wakeel J, Al Harbi A, Bayoumi M, et al. Quality of life in hemodialysis and peritoneal dialysis patients in Saudi Arabia. Ann Saudi Med. 2012;32(6):570-574.
- 84. Alvares J, Cesar CC, Acurcio FA, et al. Quality of life of patients in renal replacement therapy in Brazil: comparison of treatment modalities. Qual Life Res. 2012;21(6):983-991.
- 85. Atapour A, Nasr S, Boroujeni AM, et al. A comparison of the quality of life of the patients undergoing hemodialysis versus peritoneal dialysis and its correlation to the quality of dialysis. Saudi J Kidney Dis Transpl. 2016;27(2):270-280.
- 86. Barata NE. Dyadic Relationship and Quality of Life Patients with Chronic Kidney Disease. J Bras Nefrol. 2015;37(3):315-322.
- Basok EK, Atsu N, Rifaioglu MM, et al. Assessment of female sexual function and quality of life in predialysis, peritoneal dialysis, hemodialysis, and renal transplant patients. Int Urol Nephrol. 2009;41(3):473-481.

- 88. Baykan H, Yargic I. Depression, anxiety disorders, quality of life and stress coping strategies in hemodialysis and continuous ambulatory peritoneal dialysis patients. Klinik Psikofarmakoloji Bulteni. 2012;22(2):167-176.
- 89. Borowiak E, Braksator E, Nowicki M, et al. Quality of life of chronic hemodialysis and peritoneal dialysis patients. Clinical and Experimental Medical Letters. 2009;50(1):37-42.
- 90. Brown EA, Johansson L, Farrington K, et al. Broadening Options for Long-term Dialysis in the Elderly (BOLDE): differences in quality of life on peritoneal dialysis compared to haemodialysis for older patients. Nephrol Dial Transplant. 2010;25(11):3755-3763.
- 91. Bujang MA, Musa R, Liu WJ, et al. Depression, anxiety and stress among patients with dialysis and the association with quality of life. Asian J Psychiatr. 2015;18:49-52.
- 92. Liu WJ, Musa R, Chew TF, et al. Quality of life in dialysis: A Malaysian perspective. Hemodial Int. 2014;18(2):495-506.
- 93. Chen JY, Wan EYF, Choi EPH, et al. The Health-Related Quality of Life of Chinese Patients on Hemodialysis and Peritoneal Dialysis. Patient. 2017:1-10.
- 94. Chkhotua A, Pantsulaia T, Managadze L. The quality of life analysis in renal transplant recipients and dialysis patients. Georgian Med News. 2011;11(200):10-17.
- 95. Maglakelidze N, Pantsulaia T, Tchokhonelidze I, et al. Assessment of health-related quality of life in renal transplant recipients and dialysis patients. Transplant Proc. 2011;43(1):376-379.
- 96. Czyzewski L, Sanko-Resmer J, Wyzgal J, et al. Assessment of health-related quality of life of patients after kidney transplantation in comparison with hemodialysis and peritoneal dialysis. Ann Transplant. 2014;19:576-585.
- 97. Da Silva-Gane M, Wellsted D, Greenshields H, et al. Quality of life and survival in patients with advanced kidney failure managed conservatively or by dialysis. Clin J Am Soc Nephrol. 2012;7(12):2002-2009.
- 98. de Fijter CWH, Diepen AT, Amiri F, et al. Patient-reported outcomes (PROs) argue against the limited use of peritoneal dialysis in end-stage renal disease. Clinical Nephrology. 2018;90(2):94-101.
- 99. Fructuoso M, Castro R, Oliveira L, et al. Quality of life in chronic kidney disease. Nefrologia. 2011;31(1):91-96.
- 100. Garcia-Llana H, Remor E, Selgas R. Adherence to treatment, emotional state and quality of life in patients with end-stage renal disease undergoing dialysis. Psicothema. 2013;25(1):79-86.
- 101. Ginieri-Coccossis M, Theofilou P, Synodinou C, et al. Quality of life, mental health and health beliefs in haemodialysis and peritoneal dialysis patients: investigating differences in early and later years of current treatment. BMC Nephrol. 2008;9:14.
- 102. Goncalves FA, Dalosso IF, Borba JM, et al. Quality of life in chronic renal patients on hemodialysis or peritoneal dialysis: a comparative study in a referral service of Curitiba PR. Jornal brasileiro de nefrologia. 2015;37(4):467-474.
- Griva K, Kang AW, Yu ZL, et al. Quality of life and emotional distress between patients on peritoneal dialysis versus community-based hemodialysis. Qual Life Res. 2014;23(1):57-66.
- 104. Yang F, Griva K, Lau T, et al. Health-related quality of life of Asian patients with endstage renal disease (ESRD) in Singapore. Qual Life Res. 2015;24(9):2163-2171.

- 105. Günalay S, Oztürk YK, Akar H, et al. The relationship between malnutrition and quality of life in haemodialysis and peritoneal dialysis patients. Revista da Associacao Medica Brasileira. 2018;64(9):845-852.
- 106. Ibrahim N, Chiew-Tong NK, Desa A. Symptoms and health-related quality of life in patients with heamodialysis and continuous ambulatory peritoneal dialysis. Research Journal of Medical Sciences. 2011;5(5):252-256.
- 107. Ikonomou M, Skapinakis P, Balafa O, et al. The impact of socioeconomic factors on quality of life of patients with chronic kidney disease in Greece. J Ren Care. 2015;41(4):239-246.
- 108. Iyasere OU, Brown EA, Johansson L, et al. Quality of Life and Physical Function in Older Patients on Dialysis: A Comparison of Assisted Peritoneal Dialysis with Hemodialysis. Clin J Am Soc Nephrol. 2016;11(3):423-430.
- 109. Kang SH, Do JY, Lee SY, et al. Effect of dialysis modality on frailty phenotype, disability, and health-related quality of life in maintenance dialysis patients. PLoS One. 2017;12(5):e0176814.
- 110. Kim JY, Kim B, Park KS, et al. Health-related quality of life with KDQOL-36 and its association with self-efficacy and treatment satisfaction in Korean dialysis patients. Qual Life Res. 2013;22(4):753-758.
- 111. Kontodimopoulos N, Niakas D. An estimate of lifelong costs and QALYs in renal replacement therapy based on patients' life expectancy. Health Policy. 2008;86(1):85-96.
- 112. Kontodimopoulos N, Pappa E, Niakas D. Gender- and age-related benefit of renal replacement therapy on health-related quality of life. Scand J Caring Sci. 2009;23(4):721-729.
- 113. Nakayama M, Ishida M, Ogihara M, et al. Social functioning and socioeconomic changes after introduction of regular dialysis treatment and impact of dialysis modality: a multicentre survey of Japanese patients. Nephrology (Carlton). 2015;20(8):523-530.
- 114. Neumann D, Lamprecht J, Robinski M, et al. Social relationships and their impact on health-related outcomes in peritoneal versus haemodialysis patients: A prospective cohort study. Nephrology Dialysis Transplantation. 2018;33(7):1235-1244.
- 115. Okpechi IG, Nthite T, Swanepoel CR. Health-related quality of life in patients on hemodialysis and peritoneal dialysis. Saudi J Kidney Dis Transpl. 2013;24(3):519-526.
- 116. Ören B, Enc N. Quality of life in chronic haemodialysis and peritoneal dialysis patients in Turkey and related factors. International journal of nursing practice. 2013;19(6):547-556.
- 117. Ramos EC, Santos I, Zanini R, et al. Quality of life of chronic renal patients in peritoneal dialysis and hemodialysis. Jornal brasileiro de nefrologia. 2015;37(3):297-305.
- 118. Ruiz de Alegría-Fernández de Retana B, Basabe-Barañano N, Saracho-Rotaeche R. Coping mechanisms as a predictor for quality of life in patients on dialysis: a longitudinal and multi-centre study. Nefrologia. 2013;33(3):342-354.
- 119. Tannor EK, Archer E, Kapembwa K, et al. Quality of life in patients on chronic dialysis in South Africa: a comparative mixed methods study. BMC Nephrol. 2017;18(1):4.
- 120. Theofilou P. Quality of life in patients undergoing hemodialysis or peritoneal dialysis treatment. J Clin Med Res. 2011;3(3):132-138.
- 121. Theofilou P. Association of insomnia symptoms with kidney disease quality of life reported by patients on maintenance dialysis. Psychol Health Med. 2013;18(1):70-78.
- Turkmen K, Yazici R, Solak Y, et al. Health-related quality of life, sleep quality, and depression in peritoneal dialysis and hemodialysis patients. Hemodial Int. 2012;16(2):198-206.

- 123. Watanabe Y, Ohno Y, Inoue T, et al. Home hemodialysis and conventional in-center hemodialysis in Japan: a comparison of health-related quality of life. Hemodial Int. 2014;18 Suppl 1:S32-S38.
- 124. Wright LS, Wilson L. Quality of Life and Self-Efficacy in Three Dialysis Modalities: Incenter Hemodialysis, Home Hemodialysis, and Home Peritoneal Dialysis. Nephrol Nurs J. 2015;42(5):463-476.
- 125. Wu F, Cui L, Gao X, et al. Quality of life in peritoneal and hemodialysis patients in China. Ren Fail. 2013;35(4):456-459.
- 126. Ying SC, Krishnan M. Interpretation of quality of life outcomes amongst end stage renal disease patients in selected hospitals of Malaysia. International Journal of Pharmaceutical Sciences and Research. 2014;5(1):60-69.
- 127. Yongsiri S, Thammakumpee J, Prongnamchai S, et al. The association between bioimpedance analysis and quality of life in pre-dialysis stage 5 chronic kidney disease, hemodialysis and peritoneal dialysis patients. J Med Assoc Thai. 2014;97(3):293-299.
- 128. Ware JE. SF-36 Health Survey. Manual and Interpretation Guide. 1997.
- 129. Liyanage T, Ninomiya T, Jha V, et al. Worldwide access to treatment for end-stage kidney disease: a systematic review. The Lancet. 2015;385(9981):1975-1982.
- 130. Moosa MR, Maree JD, Chirehwa MT, et al. Use of the 'Accountability for Reasonableness' Approach to Improve Fairness in Accessing Dialysis in a Middle-Income Country. PLoS One. 2016;11(10):e0164201.
- 131. Kilonzo KG, Jones ESW, Okpechi IG, et al. Disparities in dialysis allocation: An audit from the new South Africa. PLoS One. 2017;12(4):e0176041.
- 132. Sakhuja V, Sud K. End-stage renal disease in India and Pakistan: burden of disease and management issues. Kidney Int Suppl. 2003(83):S115-118.
- 133. Bello AK, Alrukhaimi M, Ashuntantang GE, et al. Global overview of health systems oversight and financing for kidney care. Kidney Int Suppl (2011). 2018;8(2):41-51.
- 134. van der Tol A, Lameire N, Morton RL, et al. An International Analysis of Dialysis Services Reimbursement. Clin J Am Soc Nephrol. 2019;14(1):84-93.
- 135. Robinson BM, Akizawa T, Jager KJ, et al. Factors affecting outcomes in patients reaching end-stage kidney disease worldwide: differences in access to renal replacement therapy, modality use, and haemodialysis practices. The Lancet. 2016;388(10041):294-306.
- 136. Hood K, Robling M, Ingledew D, et al. Mode of data elicitation, acquisition and response to surveys: a systematic review. Health Technology Assessment. 2012;16(27).
- 137. Korevaar JC, Feith GW, Dekker FW, et al. Effect of starting with hemodialysis compared with peritoneal dialysis in patients new on dialysis treatment: A randomized controlled trial. Kidney Int. 2003;64:2222-2228.
- 138. Korevaar JC, Jansen MAM, Merkus MP, et al. Quality of life in predialysis end-stage renal disease patients at the initiation of dialysis therapy. Peritoneal Dialysis International. 2000;20:69-75.
- 139. Manns BJ, Walsh MW, Culleton BF, et al. Nocturnal hemodialysis does not improve overall measures of quality of life compared to conventional hemodialysis. Kidney Int. 2009;75(5):542-549.
- 140. Black N. Patient reported outcome measures could help transform healthcare. BMJ. 2013;346:f167.

141. van der Willik EM, Meuleman Y, Prantl K, et al. Patient-reported outcome measures: selection of a valid questionnaire for routine symptom assessment in patients with advanced chronic kidney disease - a four-phase mixed methods study. BMC Nephrol. 2019;20(1):344.

Supplementary material

Supplementary table S1. Search strings for Cochrane, Embase, and Pubmed databases

Database	Search
Cochrane	(((hemodialys*:ab,ti,kw OR haemodialys*:ab,ti,kw OR "hemo- dialys*":ab,ti,kw OR "haemo-dialys*":ab,ti,kw OR "renal dialys*":ab,ti,kw OR "dialysis modalit*":ab,ti,kw OR "artificial kidney*":ab,ti,kw) AND (home:ab,ti,kw OR homebased:ab,ti,kw)) OR "peritoneal dialys*":ab,ti,kw OR "peritoneum dialys*":ab,ti,kw) AND ("patient reported outcome":ab,ti,kw or "life qualit*":ab,ti,kw or "quality of life":ab,ti,kw or qol:ab,ti,kw or hrql:ab,ti,kw or frqol:ab,ti,kw or "SF 36":ab,ti,kw or SF36:ab,ti,kw or "SF 12":ab,ti,kw or SF12:ab,ti,kw or "short form 36":ab,ti,kw or "short form 12":ab,ti,kw or "EQ 5D*":ab,ti,kw or QALY:ab,ti,kw or QALE:ab,ti,kw) <i>Search dates from 1 January 2007 until 1 January 2019</i>
Embase	('peritoneal dialysis'/exp OR 'home dialysis'/exp OR ('hemodialysis'/de OR 'artificial kidney'/exp OR hemodialys*:ab,ti OR haemodialys*:ab,ti OR 'hemo- dialys*':ab,ti OR 'haemo-dialys*':ab,ti OR 'renal dialys*':ab,ti OR (dialysis NEAR/3 modalit*):ab,ti OR 'artificial kidney*':ab,ti AND (home:ab,ti OR homebased:ab,ti)) OR 'peritoneal dialys*':ab,ti OR (peritoneum NEAR/3 dialys*):ab,ti) AND ('patient-reported outcome'/exp OR 'quality of life'/exp OR 'patient reported outcome':ab,ti OR life AND qualit*:ab,ti OR 'quality of life':ab,ti OR qol:ab,ti OR hrql:ab,ti OR hrqol:ab,ti OR 'sf 36':ab,ti OR sf36:ab,ti OR 'sf 12':ab,ti OR sf12:ab,ti OR 'short form 36':ab,ti OR 'short form 12':ab,ti OR qalys:ab,ti OR qale:ab,ti) <i>Search dates from 1 January 2007 until 1 January 2019</i>
Pubmed	("Peritoneal Dialysis" [Mesh] OR "Hemodialysis, Home" [Mesh] OR (("Renal Dialysis" [Mesh:noexp] OR "Kidneys, Artificial" [Mesh] OR hemodialys* [tiab] OR haemodialys* [tiab] OR hemo-dialys* [tiab] OR haemo-dialys* [tiab] OR renal dialys* [tiab] OR dialysis modalit* [tiab] OR artificial kidney* [tiab]) AND (home[tiab] OR homebased [tiab])) OR peritoneal dialys* [tiab] OR peritoneum dialys* [tiab]) AND ("Patient Reported Outcome Measures" [Mesh] OR "Quality of Life" [Mesh] OR "Quality-Adjusted Life Years" [Mesh] OR "patient reported outcome" [tiab] OR life qualit* [tiab] OR "quality of life" [tiab] OR qol[tiab] OR hrql[tiab] OR hrqol[tiab] OR SF 36 [tiab] OR SF36 [tiab] OR SF 12 [tiab] OR SF12 [tiab] OR short form 36 [tiab] OR short form 12 [tiab] OR EQ 5D* [tiab] OR QALY [tiab] OR QUALY [tiab] OR QALYS [tiab] OR QALE [tiab]) Search dates from 21 November 2007 until 1 January 2019

Questionnaire	Content
Short Form (SF)	The long version of the SF (SF-36) consists of eight domains: Physical functioning, Role-physical, Bodily pain, General health, Vitality, Social function, Role-emotional, and Mental health. ¹ These domains are summarized in the Physical Component Summary (PCS) and Mental Component Summary (MCS). The shorter version of the SF (SF-12) only reports the PCS and MCS. ² The SF questionnaires are the most widely used. ³
Kidney Disease Quality Of Life Instrument (KDQOL)	The long version of the KDQOL (KDQOL-SF) consist of the SF-36 questionnaire and the following kidney disease specific domains: Symptoms, Effects of kidney disease, Burden of kidney disease, Work status, Cognitive function, Quality of social interaction, Sexual function, Sleep, Social support, Dialysis staff encouragement, and Patient satisfaction. ⁴ The short version of the KDQOL (KDQOL-36) consists of the SF-12 and the first three kidney disease specific domains (Symptoms, Effects of kidney disease, and Burden of kidney disease).
EuroQol-5D (EQ-5D)	The EuroQol-5D (EQ-5D) is a short questionnaire that can be used to calculate quality adjusted life years (QALYs) and reports on the following domains: mobility, self-care, usual activities, pain/ discomfort, and anxiety/depression. The EQ-5D is widely used in cost-effectiveness research. ⁵
World Health Organization Quality of Life (WHOQOL-BREF)	The World Health Organization Quality of Life (WHOQOL) has developed the WHOQOL-BREF questionnaire which measures four domains (physical health, psychological, social relationships, and environment) and an overall assessment of quality of life and general health. ⁶

Supplementary table S2. HRQoL questionnaires

1. Ware JE, Snow KK, Kosinski M, Gandek B. SF-36 Health Survey: Manual and Interpretation Guide. Boston, MA: The Health Institute, New England Medical Center; 1993.

2. Ware JE, Kosinski MM, Keller SD. A 12-Item Short-Form Health Survey: Construction of Scales and Preliminary Tests of Reliability and Validity. Medical Care. 1996;34:220-233.

3. Wyld M, Morton RL, Hayen A, *et al*. A Systematic Review and Meta-Analysis of Utility-Based Quality of Life in Chronic Kidney Disease Treatments. PLOS. 2012;9(9):1-10.

4. Hays RD, Kallich JD, Mapes DL, Coons SJ, Carter WB. Development of the Kidney Disease Quality of Life (KDQOL) instrument. Qual Life Res. 1994; 3:329–338.

5. Versteegh MM, Vermeulen KM, Evers SMAA, *et al.* Dutch Tariff for the Five-Level Version of EQ-5D. Value in Health. 2016;19:343-352.

6. WHOQOL Group. Development of the World Health Organization WHOQOL-BREF Quality Of Life Assessment. Psychological Medicine. 1998;28:551-558.

Criteria	+	?	-
Design	+ RCT or cohort study		- cross-sectional study
Patient selection	+ clear description setting and selection process, selection criteria mentioned and response ≥70%	? insufficient data to estimate risk of bias	- no clear description setting and selection process, selection criteria not mentioned and response <70%
Comparability	+ matched controls or comparable baseline for age, comorbidities, dialysis vintage	? insufficient data to estimate risk of bias	- non-matched or non- comparable groups
Outcome	+ self-reported HRQoL or trained interviewer	? insufficient data to estimate risk of bias	- no clear protocol for interview or administering questionnaire
Confounding	+ Adjusted analyses or stratified presentation in results	? insufficient data to estimate risk of bias	- confounding factors not mentioned or only as part of discussion
Follow-up	+ follow-up >6 months and <30% loss in the first year, with non- selective reasons	N/A not applicable	- follow-up <6 months and >30% loss in the first year
Selective reporting	+ all pre-defined HRQoL scores in protocol or methods section are reported	? insufficient data to estimate risk of bias	- not all pre-defined scores are reported
Overall (risk of bias)	low: ≥4 plus signs in above mentioned elements	moderate: 3 plus signs in above mentioned elements or 1-2 plus signs with ≥1 question mark	high: ≤2 plus signs in above mentioned elements
Conflict of interest	+ mentioned, non- conflicted	? not-mentioned	- mentioned and conflicted

Supplementary table S3. Criteria used in Risk of Bias assessment

Supplementary table S4. Cr	itical apprais:	al of 41 studie	Sč						
Author, year	Design	Patient selection	Compa- rability	Out- come	Con- founding	Follow-up	Selective reporting	Overall Risk of Bias	Conflict of interest
Al Wakeel, 2012	ı	+	۰.	+	+	N/A	+	Low	۰.
Alvares, 2012	ı	+	ı	+	+	N/A	+	Low	+
Atapour, 2016	ı	+	+	Λ.	+	N/A	+	Low	۵.
Barata, 2015	ı	ς.	<u>.</u>	.	ı	N/A	.	Moderate	۵.
Basok, 2009	ı	. .	c .	+	ı	N/A	+	Moderate	ο.
Baykan, 2012	ı	ο.	<u>م</u> .	ο.	ı	N/A	+	Moderate	+
Borowiak, 2009	ı	ς.	<u>.</u>	ı	<u>.</u> .	N/A	+	Moderate	۵.
Brown , 2010		+	+	+	+	N/A	+	Low	I
Bujang, 2015 and Liu, 2014	I	+	+	+	1	N/A	+	Low	+
Chen , 2017	ı	+	c.	+	+	N/A	+	Low	+
Chkhotua, 2011 and Maglakelidze, 2011	I	0.		+	I	N/A	ı	Moderate	0.
Czyzewski, 2014	ı	ο.	ο.	+	ı	N/A	ı	Moderate	۰.
Da Silva-Gane, 2012	+	+	ı	+	+	+	ı	Low	+
De Fijter, 2018	ı	+	ı	+	+	N/A	+	Low	+
Fructuoso, 2011	ı	ο.	ı	ı	+	N/A	+	High	A .
Garcia-Llana, 2013	ı	+	ı	+	~ •	N/A	+	Moderate	I
Ginieri-Coccossis, 2008	ı	ο.	+	+	ı	N/A	+	Moderate	+
Goncalves, 2015		~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	۰ م			N/A		High	

Supplementary table S4. Crit	tical appraisa	l of 41 studie	s (continued	[]					
Author, year	Design	Patient selection	Compa- rability	Out- come	Con- founding	Follow-up	Selective reporting	Overall Risk of Bias	Conflict of interest
Griva, 2014 and Yang, 2015	I	+	ı	+	+	N/A	+	Low	+
Günalay, 2018	ı	0.	ı	<u>.</u>	ı	N/A	+	High	۰.
Ibrahim, 2011	ı	+	۵.	ı	I	N/A	α.	High	۰.
Ikonomou, 2015	ı	+	<u>.</u> .	+	+	N/A	+	Low	+
Iyasere, 2016	ı	+	+	+	+	N/A	+	Low	I
Kang, 2017	ı	• •	+	<u>.</u>	+	N/A	+	Moderate	+
Kim , 2013	ı	+	۰.	+	0.	N/A		Moderate	۰.
Kontodimopoulos, 2008 and 2009	I	+	+	+	+	N/A	+	Low	0.
Nakayama, 2015	ı	+	+	+	ı	N/A	+	Low	ı
Neumann, 2018	+	ο.	+	+	+	+	+	Low	+
Okpechi, 2013	ı	ı	ı	ı	ı	N/A	+	High	۰.
Ören, 2013	ı	+	ı	ı	+	N/A	ı	High	۵.
Painter, 2012	+	+	+	+	ı	+	+	Low	+
Ramos, 2015	ı	+	+	+	+	N/A	+	Low	۵.
Ruiz de Alegria - Fernandez de Retana, 2013	+	+	+	+	+	+	1	Low	+
Tannor, 2017	ı	+	ı	+	ı	N/A	+	Moderate	+
Theofilou, 2011 and 2013	-	+	۰.	-	-	N/A	~	High	~

Chapter 7

Author, year	Design	Patient	Compa-	Out-	Con-	Follow-up	Selective	Overall	Conflict of
		selection	rability	come	founding		reporting	Risk of Bias	interest
Turkmen, 2012	I	+	+	۰.	ı	N/A	I	Moderate	+
Watanabe, 2014	ı	ο.	+	+	ı	N/A	+	Moderate	+
Wright, 2015	ı	+	. .	+		N/A	+	Moderate	+
Wu, 2013	I	ο.	+	+		N/A	+	Moderate	+
Ying, 2014	ı	+	.	ı		N/A	ı	High	۰.
Yongsiri, 2014	ı	۰.	+	۰.		N/A	+	Moderate	+

Supplementary table S4. Critical appraisal of 41 studies (continued)

Study ID	Panel A. Physical HRQoL measured with different questionnaires	SMD (95% Cl)	% Weigh
SF	1		
Al Wakeel		-0.19 (-0.47, 0.09)	2.74
Alvares	★ !	-0.20 (-0.29, -0.12)	3.37
Atapour		0.42 (0.01, 0.83)	2.19
Basok		-0.42 (-1.01. 0.17)	1.58
Bavkan		0.73 (0.29, 1.18)	2.07
Brown		0.16 (-0.18, 0.49)	2.51
Chen		0 73 (0 00 0 46)	2 03
Thebotus and Maglakolidzo	The second se	0.02 (0.27 0.22)	2.45
Children and Maglakendze		-0.02 (-0.37, 0.33)	1.05
Da Silva, Cano		0.51 (-0.10, 0.79)	2 22
Galitar		0.01 (0.23, 0.96)	2.33
Fijter		0.39 (=0.07, 0.85)	2.01
-ructuoso	1	1.09 (0.44, 1.74)	1.42
Garcia–Llana	1 1	0.59 (0.08, 1.10)	1.82
soncaives		-0.26 (-0.49, -0.04)	2.94
ariva and Yang		-0.19 (-0.36, -0.01)	3.12
brahim		0.30 (0.04, 0.55)	2.84
konomou		0.16 (-0.22, 0.53)	2.33
yasere		0.13 (-0.12, 0.38)	2.86
Kang		-0.16 (-0.27, -0.04)	3.30
(im		-0.06 (-0.35, 0.22)	2.70
Kontodimopoulos	_	0.00 (-0.26, 0.26)	2.83
Vakayama		-0.29 (-0.58, 0.01)	2.66
Neumann		-0.22 (-0.43, -0.01)	3.00
Okpechi		0.05 (-0.42, 0.51)	1.99
Painter		0.15 (-0.68, 0.97)	1.04
Ramos		-0.07(-0.36, 0.21)	2.72
Ruiz de Alegria – Fernandez de Retana		0.41 (0.01, 0.82)	2.23
Tannor		0.04 (-0.34, 0.42)	2.31
Turkmen		-0.60(-0.92,-0.27)	2.53
Watanahe		1 06 (0 50 1 54)	1 06
Wright a	I I I I I I I I I I I I I I I I I	0.22 (0.27 0.24)	1.60
Wright b		-0.22 (-0.77, 0.34)	1.09
Whight D		0.04 (-0.43, 0.57)	3.70
Wu	1 1001 1000	0.28 (=0.02, 0.33)	2.70
ring	and a second sec	0.51 (0.23, 0.79)	2./1
Uren Subbabal di anno di Telesci a le sesti		0.37 (0.14, 0.60)	2.92
Subtotal (I-squared = 79.8%, p = 0.000)	•	0.11 (0.01, 0.22)	84.51
WHOQOL-BREF	1	1 33 /0 99 1 76	3.00
alata Naissa sed Liv		- 1.32 (0.88, 1./0)	2.09
sujang and Liu		-0.07 (-0.18, 0.04)	3.32
Inieri–Coccossis		0.48 (0.06, 0.90)	2.15
neoniou		0.46 (0.13, 0.80)	2.50
rongsiri	•	0.12 (-0.39, 0.63)	1.83
Subtotal (I–squared = 91.4%, p = 0.000)		0.45 (-0.04, 0.94)	11.88
EQ-5D	i		
Borowiak		0.11 (-0.28, 0.50)	2.27
Günalay		-0.41 (-1.09, 0.28)	1.34
Subtotal (I–squared = 39.5%, p = 0.199)		-0.07 (-0.55, 0.41)	3.61
Overall (I–squared = 81.3%, p = 0.000)		0.14 (0.04, 0.24)	100.00
NOTE: Weights are from random effects a	nalvsis		
	-15 -1 -5 0 5 1 15		

Legend: SMD, standardized mean difference; 95% CI, 95% confidence Interval; SF, Short Form (including KDQOL); WHOQOL–BREF, World Health Organization Quality of Life–BREF; EQ–5D, EuroQol–5D; HRQoL, Health–Related Quality of Life.

Supplementary figure S1A. Meta-analysis of Health-Related Quality of Life in different questionnaires.

Study ID	Panel B. Mental HRQoL measured with different questionnaires	SMD (95% CI)	% Weigh
SF			
Al Wakeel		0.80 (0.52, 1.09)	2.71
Alvares		0.01 (-0.08, 0.09)	3.54
Atapour		0.09 (-0.32, 0.50)	2.16
Basok		-0.48 (-1.07, 0.12)	1.48
Baykan	· · · · ·	0.45 (0.01, 0.88)	2.04
Brown	<u> </u>	0.34 (0.01, 0.67)	2.50
Chen		-0.07 (-0.30, 0.16)	3.00
Chkhotua and Maglakelidze		-0.05 (-0.40, 0.30)	2.43
Czyzewski		0.65 (0.16, 1.13)	1.85
Da Silva–Gane		-0.16 (-0.53, 0.21)	2.33
Fiiter		0.32 (-0.14, 0.78)	1.95
Fructuoso		0.30 (-0.32, 0.92)	1.41
Garcia-Llana		0.24 (-0.26, 0.75)	1 78
Goncalves		-0.63 (-0.86 -0.40)	2.00
Goncaives		0.03 (-0.15, 0.30)	2.77
landa and rang		0.02 (-0.15, 0.20)	3.23
loranim		0.31 (0.06, 0.37)	2.00
konomou		0.30 (-0.08, 0.88)	2.29
ryasere		-0.12 (-0.36, 0.13)	2.91
Kang		-0.19 (-0.31, -0.08)	3.45
Kim		0.03 (-0.26, 0.31)	2.73
Kontodimopoulos		-0.09 (-0.35, 0.16)	2.87
Nakayama		-0.04 (-0.34, 0.25)	2.68
Neumann		-0.03 (-0.24, 0.18)	3.08
Okpechi		0.02 (-0.45, 0.48)	1.92
Painter	*	-0.25 (-1.08, 0.57)	0.95
Ramos		0.14 (-0.15, 0.42)	2.75
Ruiz de Alegria – Fernandez de Retana		0.08 (-0.32, 0.48)	2.20
Tannor		-0.31 (-0.70, 0.07)	2.26
Turkmen -		-1.15 (-1.50, -0.81)	2.44
Watanabe		0.20 (-0.24, 0.65)	2.01
Wright a		0.02 (-0.53, 0.57)	1.61
Wright h		0.08(-0.45.0.61)	1.69
Win		0.25 (-0.03, 0.54)	2.72
Vine		0.45 (0.17, 0.74)	2.73
ring		0.45 (0.17, 0.74)	2.13
Oren		0.32 (0.09, 0.55)	2.99
Subtotal (I–squared = 80.0%, p = 0.000)	e	0.05 (-0.06, 0.16)	84.56
WHOQOL-BREF			
Barata	Li — • —	0.84 (0.42, 1.26)	2.11
Bujang and Liu		0.04 (-0.07, 0.15)	3.47
Ginieri–Coccossis	1.0	0.13 (-0.28, 0.55)	2.12
Theofilou		0.05 (-0.28, 0.38)	2.51
Yongsiri		-0.08 (-0.59, 0.43)	1.75
Subtotal (I–squared = 70.9%, p = 0.008)	\sim	0.18 (-0.09, 0.45)	11.96
		0 1 1 / 0 00 0 000	0.00
Borowiak		0.11 (-0.28, 0.50)	2.23
Gunalay		-0.41 (-1.09, 0.28)	1.24
Subtotal (I–squared = 39.5%, p = 0.199)		-0.07 (-0.55, 0.41)	3.47
Overall (I–squared = 78.1%, p = 0.000)	•	0.06 (-0.03, 0.15)	100.0
NOTE: Weights are from random effects and	lysis		
1			
-1.	5 –1 –.5 0 .5 1	1.5	

Legend: SMD, standardized mean difference; 95% CJ, 95% confidence interval; SF, Short Form (including KDQOL); WHOQOL–BREF, World Health Organization Quality of Life–BREF; EQ–5D, EuroQol–5D; HRQoL, Health–Related Quality of Life.

Supplementary figure S1B. Meta-analysis of Health-Related Quality of Life in different questionnaires.

ID Panel A: Physical HRQoL me	easured in three age categories (<45 years, 45–60 years, >60 years)	SMD (95% CI)	Weigh
<45 years			
Baykan		0.73 (0.29, 1.18)	2.45
Fructuoso		1.09 (0.44, 1.74)	1.61
Okpechi		0.05 (-0.42, 0.51)	2.35
Painter	*	0.15 (-0.68, 0.97)	1.16
Tannor		0.04 (-0.34, 0.42)	2.78
Subtotal (I–squared = 67.3%, p = 0.01	6)	0.39 (-0.01, 0.80)	10.35
45–60 years			
AlWakeel		-0.19 (-0.47, 0.09)	3.38
Alvares	* I I	-0.20 (-0.29, -0.12)	4.34
Atapour		0.42 (0.01, 0.83)	2.61
Basok	• • • • •	-0.42 (-1.01, 0.17)	1.82
Borowiak		0.11 (-0.28, 0.50)	2.72
Bujang and Liu		-0.07 (-0.18, 0.04)	4.26
Da Silva–Gane	· · · ·	0.61 (0.23, 0.98)	2.81
Garcia–Llana		0.59 (0.08, 1.10)	2.13
Goncalves	-	-0.26 (-0.49, -0.04)	3.69
Griva and Yang		-0.19 (-0.36, -0.01)	3.96
Günalay	· · · · ·	-0.41 (-1.09, 0.28)	1.52
lkonomou	·	0.16 (-0.22, 0.53)	2.81
Kang		-0.16 (-0.27, -0.04)	4.23
Kontodimopoulos		0.00 (-0.26, 0.26)	3.52
Neumann		-0.22 (-0.43, -0.01)	3.77
Ramos		-0.07 (-0.36, 0.21)	3.36
Ruiz de Alegria – Fernandez de Retana		0.41 (0.01, 0.82)	2.67
Turkmen		-0.60 (-0.92, -0.27)	3.09
Watanabe	· · · · · · · · · · · · · · · · · · ·	1.06 (0.59, 1.54)	2.31
Wu	· · •	0.26 (-0.02, 0.55)	3.33
Yongsiri		0.12 (-0.39, 0.63)	2.14
Ören		0.37 (0.14, 0.60)	3.65
Subtotal (I–squared = 79.9%, p = 0.00	0) 🗢	0.03 (-0.09, 0.15)	68.12
>60 years	1		
Brown		0.16 (-0.18, 0.49)	3.06
Chen		0.23 (0.00, 0.46)	3.66
Fijter		0.39 (-0.07, 0.85)	2.37
Ginieri–Coccossis		0.48 (0.06, 0.90)	2.56
lyasere		0.13 (-0.12, 0.38)	3.56
Nakayama		-0.29 (-0.58, 0.01)	3.27
Theofilou		0.46 (0.13, 0.80)	3.04
Subtotal (I–squared = 60.7%, p = 0.01	8)	0.20 (0.01, 0.39)	21.53
Overall (I–squared = 79.6%, p = 0.000)	•	0.11 (0.00, 0.21)	100.0
NOTE: Weights are from random effect	ts analysis		
		5	

Legend: SMD, standardized mean difference; 95% CI, 95% confidence interval; HRQoL, Health–Related Quality of Life.

Supplementary figure S2A. Meta-analysis of Health-Related Quality of Life in different age categories.

Study ID Panel R. Mental HROol, measured in three are categories (<45 years: 45_60 years: >60	SMD (95% CI)	% Weigh
<45 years	yearsy	
Pavkan	0.45 (0.01 0.88)	2.45
Fructuoso	0.30 (-0.32 0.02)	1.67
Okpechi	0.02 (-0.45, 0.48)	2 30
Paintar	-0.25 (-1.08.0.57)	1 11
Tappor	-0.23 (-1.06, 0.37)	2 72
Subtotal (I-squared = 48.7%, p = 0.100)	0.05 (-0.27, 0.37)	10.26
45_60 years		
Al Wakeel	0.80 (0.52, 1.09)	3.32
Alvares	0.01 (-0.08, 0.09)	4 44
Atapour	0.00 (-0.32, 0.50)	2.60
Basok	-0.48 (-1.07, 0.12)	1.75
Borowiak	0.11 (-0.28 0.50)	2.69
Buiang and Liu	0.04 (_0.07 0.15)	435
Da Silva, Cano	0.16 (0.53 0.31)	2.82
Garcia_Uana	0.10 (-0.35, 0.21)	2.05
Concelhos	0.62 (0.86 0.40)	2.12
Grive and Vang	0.03 (0.15 0.20)	4.02
Ginalay	0.02 (-0.13, 0.20)	1.02
Ikonomou	0.30 (0.08 0.68)	3.77
Kono	0.10 (0.31 0.09)	4.22
Kantadimonaular	-0.19 (-0.31, -0.06)	4.32
Noncoalmopoulos	-0.09 (-0.35, 0.16)	3.33
Personal International Interna	-0.03 (-0.24, 0.16)	3.01
Ramos	0.14 (-0.13, 0.42)	3.3/
Kulz de Alegria – Fernandez de Ketana	0.08 (-0.32, 0.48)	2.66
Turkmen	-1.15 (-1.50, -0.61)	2.90
Watanabe	0.20 (-0.24, 0.65)	2.41
wu	0.25 (-0.03, 0.54)	3.34
Yongsin	-0.08 (-0.59, 0.43)	2.09
Oren	0.32 (0.09, 0.55)	3.69
Subtotal (I–squared = 84.3%, p = 0.000)	-0.02 (-0.15, 0.11)	68.22
>60 years		
Brown	0.34 (0.01, 0.67)	3.04
Chen — • — • —	-0.07 (-0.30, 0.16)	3.70
Fijter	- 0.32 (-0.14, 0.78)	2.33
Ginieri–Coccossis –	0.13 (-0.28, 0.55)	2.55
lyasere	-0.12 (-0.36, 0.13)	3.58
Nakayama	-0.04 (-0.34, 0.25)	3.27
Theofilou	0.05 (-0.28, 0.38)	3.05
Subtotal (I–squared = 19.2%, p = 0.283)	0.04 (-0.09, 0.17)	21.52
Overall (I–squared = 77.9%, p = 0.000)	0.01 (-0.09, 0.11)	100.00
NOTE: Weights are from random effects analysis		
	I I	
_15 _1 _5 0 5	1 1.5	

Legend: SMD, standardized mean difference; 95% CI, 95% confidence interval; HRQoL, Health-Related Quality of Life.

Supplementary figure S2B. Meta-analysis of Health-Related Quality of Life in different age categories.



Chapter 8

The impact of COVID-19 on the mental health of dialysis patients

Anna A. Bonenkamp, Theresia A. Druiventak, Anita van Eck van der Sluijs, Frans J. van Ittersum, Brigit C. van Jaarsveld and Alferso C. Abrahams on behalf of the DOMESTICO study group.

Journal of Nephrology. 2021 Apr;34(2):337-344.

Abstract

Background Studies have shown increased anxiety, depression, and stress levels among different populations during the coronavirus disease 2019 (COVID-19) pandemic. However, the impact of the pandemic on the mental health of dialysis patients remains unknown. The aim of this study was to investigate the mental health of dialysis patients during the COVID-19 pandemic compared to the period preceding the pandemic.

Methods Data originate from the ongoing multicentre observational Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes. Patients who filled in a Health-Related Quality of Life (HRQoL) questionnaire during the pandemic and six to three months prior were included. The mean difference in Mental Component Summary (MCS) score of the Short Form 12 was analysed with multilevel linear regression. A McNemar test was used to compare presence of mental health-related symptoms during and prior to the COVID-19 pandemic.

Results A total of 177 patients were included. The mean MCS score prior to COVID-19 was 48.08 ± 10.15 , and 49.00 ± 10.04 during the COVID-19 pandemic. The adjusted mean MCS score was 0.93 point (95% CI -0.57 to 2.42) higher during the COVID-19 pandemic than during the period prior to the pandemic. Furthermore, no difference in the presence of the following mental health-related symptoms was found during the COVID-19 pandemic: feeling anxious, feeling sad, worrying, feeling nervous, trouble falling asleep, and trouble staying asleep.

Conclusions The mental health of dialysis patients appears to be unaffected by the COVID-19 pandemic. Dialysis patients may be better able to cope with the pandemic, since they have high resilience and are less impacted by social distancing measures.

Introduction

The coronavirus disease 2019 (COVID-19) outbreak that started in China rapidly spread across the globe, with major consequences for health and the healthcare system. Currently, the estimated number of infections worldwide is 66 million and the estimated number of deaths 1.5 million.¹ In the Netherlands, the first COVID-19 patient was diagnosed on February 27th, 2020.² In response, the Dutch government announced drastic measures; they obliged social distancing including working from home and closing all educational institutions, restaurants, cultural and sporting facilities, to limit further spread of the virus.

The current COVID-19 outbreak has been shown to increase levels of anxiety, depression, and stress among the general population.³⁻⁵ In patients with Alzheimer's disease and immunodeficiency, COVID-19 also resulted in higher anxiety levels and a higher risk of developing depression.^{6,7} Moreover, patients with chronic conditions had an increased risk of developing sleeping disorders.⁸ Patients with end-stage kidney disease (ESKD) who are treated with dialysis have a higher risk of a severe clinical course of COVID-19 and worse outcome.⁹ The knowledge that they have a higher risk of infection, can become more seriously ill and have a higher mortality risk might result in symptoms like feeling anxious, feeling sad, worrying, feeling nervous and sleeping problems. Moreover, the psychological well-being of dialysis patients may also be affected by fear among fellow patients and healthcare professionals. However, data regarding the impact of the COVID-19 pandemic on the mental health of dialysis patients during the COVID-19 pandemic compared to the period preceding the pandemic.

Methods

Study population and design

To compare the mental health of dialysis patients prior to the COVID-19 pandemic with a period during the COVID-19 pandemic, data were used from the ongoing Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO, Netherlands Trial Register identifier: NL6519).¹⁰ In this nationwide, prospective, observational study Health-Related Quality of Life (HRQoL) of home dialysis, i.e.

peritoneal dialysis and home haemodialysis, patients is compared with HRQoL of in-centre haemodialysis patients. All adult patients that started chronic dialysis were potentially eligible and all included patients provided written informed consent. The first patient was recruited in December 2017 and the end of the inclusion period is expected in 2021. HRQoL, and other patient-reported outcomes such as symptom burden, were measured with questionnaires.

For the present study, patients were included if they had completed a questionnaire during the COVID-19 pandemic, defined as the period between February 27th and July 1st, 2020, and a questionnaire 6 months prior to the COVID-19 pandemic. When the questionnaire administered 6 months prior to the COVID-19 pandemic was not available, the questionnaire administered 3 months prior to the COVID-19 pandemic was used.

Outcome parameters

The primary outcome parameter was mental health, assessed with the Mental Component Summary (MCS) score of the 12-item Short Form (SF-12) health survey. The MCS was calculated using standard algorithms, meaning that a healthy individual scores 50 points on a scale of 0-100 with a standard deviation of 10 points.^{11,12} Higher scores of the MCS reflect better HRQoL.¹¹ The secondary outcome parameters were the Physical Component Summary (PCS) score of the SF-12, and the presence and severity of mental health-related symptoms assessed with the Dialysis Symptom Index (DSI).^{11,13} These symptoms included feeling anxious, feeling sad, worrying, feeling nervous, trouble falling asleep, and trouble staying asleep. A 5-point Likert scale, ranging from 'not at all bothersome' to 'very bothersome', was used to evaluate the severity of these 6 symptoms.¹³

Data collection

The following sociodemographic and clinical data were collected at study baseline: sex, age, primary kidney disease, living situation (alone, with partner, or in a nursing home), level of education, work status, history of comorbidities, recent start, dialysis modality (in-centre haemodialysis, peritoneal dialysis, or home haemodialysis), and acute start at dialysis initiation. Primary kidney disease was classified according to the codes of the ERA-EDTA. A higher level of education includes university colleges and university of applied sciences. Comorbidity was scored according to the Charlson comorbidity index.¹⁴ Recent start of dialysis was defined as start of dialysis 6 months

prior to the COVID-19 pandemic. Acute start of dialysis was defined as an unplanned start of dialysis with no previous consultation of a nephrologist.

In addition, the questionnaires were reviewed to check whether participants had written comments related to COVID-19.

Statistical analysis

All normally distributed continuous variables are presented as mean with standard deviation (SD), non-normally distributed variables as median with interquartile range (IQR), and categorical variables as proportion.

Multilevel linear regression was used to assess the overall association between the COVID-19 pandemic and MCS or PCS score. The multilevel model was used to adjust for correlation of repeated observations within a patient. Both crude and adjusted analyses were performed. Adjusted models were corrected for sex, age, Charlson comorbidity index, higher educational level, dialysis modality, and recent start of dialysis.

A McNemar test was used to compare the presence of mental health-related symptoms prior to the COVID-19 pandemic with the period during the COVID-19 pandemic. A Wilcoxon signed-rank test was used to compare the severity of mental health-related symptoms prior to the COVID-19 pandemic with the period during the COVID-19 pandemic. In addition, the severity scores of the 6 mental health-related symptoms were added up to an overall symptom severity score ranging from 0 to 30, in which a severity score of 30 meant that in all mental health-related symptoms the maximum severity score was reported.^{13,15}

Missing values of SF-12 items and confounders were imputed with standard multiple imputation techniques using 10 repetitions and predictive mean matching (SPSS).¹⁶ A difference of 3 points on the MCS and PCS was considered clinically relevant and a p-value of <0.05 was considered statistically significant.^{17,18} All analyses were performed using SPSS Statistics version 26 (IBM) or STATA 14.

Results

A total of 177 patients were included, of whom 125 patients had filled in a questionnaire 6 months prior to the COVID-19 pandemic and 52 had filled in a questionnaire 3 months prior to the COVID-19 pandemic. The majority of patients (87%) had filled in their questionnaires completely. Patient characteristics are depicted in Table 1. The majority (63%) was male, the mean age of the study population was 64.9 ± 11.5 years and 61% started dialysis 3 to 6 months prior to the COVID-19 pandemic. Only 1% of the study population was infected with SARS-CoV-2.

Characteristics	Patients (n=177)
Sex, male, n (%)	112 (63)
Age, mean (SD), years	64.9 ± 11.5
Primary kidney disease, n (%)	
Glomerulonephritis/pyelonephritis	27 (21)
Cystic kidney disease	12 (9)
Renovascular kidney disease	28 (21)
Diabetes mellitus	24 (18)
Other/unknown	41 (31)
Living situation, n (%)	
Alone	49 (31)
With partner	95 (60)
In nursing home	4 (3)
Higher education, n (%)	34 (21)
Employed, n (%)	27 (16)
Charlson comorbidity index, median [IQR]	4 [2-5]
Recent start of dialysis, n (%)	107 (61)
Dialysis modality at dialysis initiation, n (%)	
In-centre haemodialysis	132 (75)
Peritoneal dialysis	43 (25)
Home haemodialysis	2 (1)
Acute start of dialysis, n (%)	25 (14)
Infected with SARS-CoV-2, n (%)	2 (1)

Table 1. Patient Characteristics

The MCS score was 48.08 ± 10.15 prior to the COVID-19 pandemic and 49.00 ± 10.04 during the COVID-19 pandemic (Figure 1A). The mean MCS score was 0.91 point (95% CI -0.59 to 2.41, *p*-value 0.2) higher during the COVID-19 pandemic than prior to the COVID-19 pandemic (Table 2). Adjustment for multiple confounders did not change this result.



Figure 1. Mental Component Summary score (A) and Physical Component Summary score (B) prior and during COVID-19 pandemic

MCS, Mental Component Summary; PCS, Physical Component Summary.

The PCS score was 35.92 ± 9.99 prior to the COVID-19 pandemic and 37.52 ± 10.38 during the COVID-19 pandemic (Figure 1B). The mean PCS score was 1.63 point (95% CI 0.28 to 2.99, *p*-value 0.02) higher during the COVID-19 pandemic than prior to the COVID-19 pandemic (Table 2). Adjustment for multiple confounders did not change this result.

Table 2. Linear regression of Health-Related Quality of Life score during COVID-19 pandemic

	Regression Coefficie	nt (95% CI)	
	Crude	Adjusted*	Adjusted**
MCS change during COVID-19	0.91 (-0.59 to 2.41)	0.91 (-0.59 to 2.41)	0.93 (-0.57 to 2.42)
PCS change during COVID-19	1.63 (0.28 to 2.99)	1.63 (0.28 to 2.98)	1.64 (0.28 to 2.99)

MCS, Mental Component Summary; PCS, Physical Component Summary.

* Adjusted for age and sex

** Adjusted for age, sex, Charlson comorbidity index, higher educational level, dialysis modality, and recent start of dialysis

As depicted in figure 2, patients on dialysis reported frequently that they were feeling sad (33% vs 35%), were worrying (35% vs 36%), had trouble falling asleep (37% vs 39%) and had trouble staying asleep (53% vs 51%). For all mental health-related symptoms, there was no significant difference in presence prior to the COVID-19 pandemic compared to the period during the COVID-19 pandemic.



Figure 2. Presence of mental health-related symptoms prior and during COVID-19 pandemic

Also, no difference was found regarding the total number of mental health-related symptoms: 74% of patients reported at least 1 symptom prior to the COVID-19 pandemic compared to 72% of patients during the COVID-19 pandemic while 7% of patients reported all 6 symptoms prior the COVID-19 pandemic compared to 6% of patients during the COVID-19 pandemic.

The severity of mental health-related symptoms was not significantly higher during the COVID-19 pandemic (Supplementary Figure 1). In addition, the total symptom severity score (ranging from 0-30) was not different between the two time periods (4 [IQR 0 - 8] prior to the pandemic vs 4 [IQR 0 - 9] during the COVID-19 pandemic).

Finally, a few patients wrote comments on the questionnaire concerning the COVID-19 pandemic. Patients commented that the COVID-19 pandemic had a huge impact on everyday life and that more help by informal caregivers was needed. For example, a patient wrote 'this corona period also affects our daily life. Due to my health condition, we have tried to avoid all threats. The domestic help is no longer coming and grocery shopping has been done by our children.' Another patient wrote 'Daily life has changed quite a bit due to corona. I stay indoors as much as possible'. Few patients also noted that they felt isolated, 'Loneliness because of corona. I am unable to receive visitors and all other activities have been discontinued.'

Discussion

This study showed that the COVID-19 pandemic did not affect the self-reported mental health of dialysis patients, as measured with HRQoL and symptom questionnaires. Dialysis patients did not report a higher burden or a higher severity of mental health-related symptoms such as feeling anxious, feeling sad, worrying, feeling nervous, trouble falling asleep, and trouble staying asleep.

A possible explanation for our findings could be that dialysis patients already suffer greatly from their kidney disease and treatment, which could limit the impact of the COVID-19 pandemic. Dialysis has a major impact on the mental health of dialysis patients resulting in a lower HRQoL than patients with other chronic illnesses such as malignancies.^{19,20} Mittal et al. found that patients with kidney disease had a 2.68 point lower MCS score compared to the general population, whereas patients with malignancies had a 0.31 lower MCS score compared to the general population.²⁰ In addition, dialysis patients have to deal with fluid restrictions, polypharmacy, and frequent hospital visits. As a result, dialysis patients have to adjust their everyday life for they encounter all these difficulties and adversities. As such, they have developed coping mechanisms in order to maintain satisfactory mental health. This ability to adapt is called resilience in literature and is often described as 'a measure of successful stress-coping ability'.²¹ Resilience includes having a positive perception, accepting a burdensome situation, and being motivated to overcome various difficulties.²² In a Spanish study, a higher level of resilience was associated with higher HRQoL scores.²³ The importance of resilience for both haemodialysis and peritoneal dialysis patients to overcome the burden of dialysis has been emphasized in multiple studies.²⁴⁻²⁶ In one of these studies the resilience of dialysis patients was quantified with a frequently used resilience scale. They found a score of 82.4 in dialysis patients, comparable to the general population (80.4) and reasonably higher than among patients visiting a general practitioner (71.8).^{21,24} Dialysis patients may have a high level of resilience compared to primary care patients, as they have learned to adapt over time to bear the burden of dialysis and their disease in general, which could explain their ability to deal better with different stressors such as the COVID-19 pandemic.

The large amount of unemployed dialysis patients in our population may also explain why the COVID-19 pandemic did not seem to affect mental health. A study showed that people who are unemployed had higher mental distress in general, but did not Chapter 8

experience an increase of mental distress during the COVID-19 pandemic as assessed with the 12-item General Health Questionnaire (change score –0.48 (95% CI –1.55 to 0.60).²⁷ Whereas people who are employed during the COVID-19 pandemic experienced an increase in mental distress compared to the period before COVID-19 (change score 0.63 (95% CI 0.20 to 1.06).^{27,28} In our population only 16% was employed, which is consistent with clinical practice as many dialysis patients are unemployed.

The third possible explanation for our results could be that 75% of our study population received in-centre haemodialysis, which might diminish mental problems that could have developed as a result of the national social isolation. Support from fellow patients, nurses, and health care professionals can contribute to a reduced sense of loneliness. Moreover, dialysis patients usually participate less in everyday activities than age-matched healthy individuals or even kidney transplant patients due to the nature of the dialysis treatment.²⁹ The regular visits to the hospital for dialysis treatments consumes an important part of the patient's time, with less time for social activities, work or travelling. Dialysis patients will be affected less by national policy measures such as social distancing since they experience fewer major changes in everyday life. In addition, in-centre haemodialysis patients might experience a sense of safety during their hospital visits that further limits the effect of the COVID-19 pandemic on mental health. In the Netherlands, many precautionary measures were taken at dialysis centres, such as screening for fever/complaints at entry for all patients, distance of 1.5 m whenever possible between people and wearing of face masks for dialysis patients, dialysis nurses and physicians early in the course of the pandemic. Also, dialysis patients that attended the hospital for haemodialysis sessions were able to obtain adequate information concerning COVID-19 directly from their health care professionals. In a Chinese study it was found that more information about the disease contributed to less anxiety levels.4

It should be noted that some dialysis patients did express feelings of loneliness due to social isolation in the additional comments of the questionnaire. Because of their vulnerability they were being extra careful to protect themselves; informal caregivers took over many tasks for the patients so that they could avoid contact with others as much as possible. In a national survey among the general Dutch population, more than half of the participants indicated moderate or severe feelings of loneliness from April to June 2020. Nonetheless, they found that concerns among the general Dutch population began to subside around the end of March 2020.³⁰ At this point the number of newly reported corona cases also began to decline. Compared to other countries in Europe including France, the United Kingdom and Italy, the number of newly reported COVID-19 patients and deaths was lower in the Netherlands, which could be a final explanation of the results in our study.³¹

The results of our study are in line with a recent study in the United Kingdom, which showed that the COVID-19 pandemic did not affect the mental health of patients with chronic illnesses as assessed with a generic HRQoL questionnaire (change score in the GHQ-12 0.40 (95% CI –0.30 to 1.09).²⁷ Contradictory, an online survey among 1.210 Chinese people found higher levels of stress, depression, and anxiety among those with a history of chronic illnesses.⁴ Another study conducted in Northern Spain also showed higher levels of stress, depression, and anxiety among those with a history of chronic illnesses.⁵ Unfortunately, none of all these studies have specified the participants' diseases, making a good comparison with our study population difficult. Also, in two studies no comparison with a historic control group or a pre-COVID-19 assessment of mental health was performed.^{4,5}

To our knowledge, this is the first study investigating whether the COVID-19 pandemic affects the mental health of dialysis patients. Strengths of this study include the use of validated self-reported HRQoL and symptom questionnaires and the use of an existing prospective and nationwide cohort of dialysis patients (DOMESTICO) for analysis. Moreover, the number of patients in our study would have been sufficient to detect a significant difference in SF-12 composite scores between time periods as small of 2.17, whereas a difference of 3 is defined clinically relevant in literature.^{17,18} We calculated in our sample size that a total of 123 patients was sufficient to detect a 3 point difference between time points (α =0.05, β =0.10). In our study, we had a 97% power to detect such a clinically relevant difference. A limitation of our study might be that the MCS score of the SF-12 questionnaire is not sensitive enough to detect differences over time in individuals, i.e. that the MCS score has limited responsiveness.³² To overcome this issue, we also used the DSI which provides more detailed information about the mental health of the patients. Another limitation might be that the chosen period of the COVID-19 pandemic was too short to demonstrate an association with mental health. The COVID-19 virus is still spreading and its effect on the economy is currently unclear. Therefore, if the pandemic lasts longer, a negative impact on mental health may still be revealed.

In conclusion, the mental health of dialysis patients assessed with SF-12 and DSI appears to be unaffected during the first wave of the COVID-19 pandemic. This could be explained by higher resilience, more unemployment among dialysis patients, less impact of social distancing on the dialysis population, strict precautionary measures and perceived support from health care professionals, which may all contribute to better coping with the COVID-19 pandemic. However, a second peak of COVID-19 is expected and the economic burden of the pandemic has yet to be discovered. Therefore, it is important to continue paying attention to the concerns and needs of our dialysis population.

References

- Coronavirus disease (COVID-19). | WHO https://covid19.who.int/ (accessed December 7, 2020).
- Ontwikkeling COVID-19 in grafieken | RIVM, https://www.rivm.nl/coronavirus-covid-19/ grafieken (accessed November 1, 2020).
- McGinty EE, Presskreischer R, Han H, et al. Psychological Distress and Loneliness Reported by US Adults in 2018 and April 2020. JAMA - Journal of the American Medical Association 2020; 324: 93–94.
- 4. Wang C, Pan R, Wan X, et al. Immediate Psychological Responses and Associated Factors during the Initial Stage of the 2019 Coronavirus Disease (COVID-19) Epidemic among the General Population in China. Int J Environ Res Public Health 2020; 17: 1729.
- 5. Ozamiz-etxebarria N, Dosil-santamaria M, Picaza-gorrochategui M, et al. Stress, anxiety , and depression levels in the initial stage of the COVID-19 outbreak in a population sample in the northern Spain Niveles de estrés, ansiedad y depresión en la primera fase del brote del COVID-19 en una muestra recogida en el norte de E. Cad Saude Publica 2020; 36: 1–9.
- 6. Beatriz Lara B, Carnes A, Dakterzada F, et al. Neuropsychiatric symptoms and quality of life in Spanish Alzheimer's disease patients during COVID-19 lockdown. Eur J Neurol 2020; 25: 1744-1747.
- Pulvirenti F, Cinetto F, Milito C, et al. Health-Related Quality of Life in Common Variable Immunodeficiency Italian Patients Switched to Remote Assistance During the COVID-19 Pandemic. J Allergy Clin Immunol Pract 2020; 8(6):1894-1899.e2. DOI: 10.1016/j. jaip.2020.04.003.
- 8. Gualano MR, Lo Moro G, Voglino G, et al. Effects of COVID-19 lockdown on mental health and sleep disturbances in Italy. Int J Environ Res Public Health 2020; 17: 1–13.
- 9. Williamson EJ, Walker AJ, Bhaskaran K, et al. Factors associated with COVID-19-related death using OpenSAFELY. Nature 2020; 584: 430–436.
- van Eck van der Sluijs A, Bonenkamp AA, Dekker FW, et al. Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO): rationale and design. BMC Nephrol 20, 361 (2019) DOI: 10.1186/s12882-019-1526-4.
- 11. Ware JE, Kosinski M, Keller SD. A 12-Item Short-Form Health Survey: Construction of Scales and Preliminary Tests of Reliability and Validity. Med Care 1996; 34: 220–233.
- 12. Gandek B, Ware JE, Aaronson NK, et al. Cross-validation of item selection and scoring for the SF-12 Health Survey in nine countries: Results from the IQOLA Project. J Clin Epidemiol 1998; 51: 1171–1178.
- 13. Weisbord SD, Fried LF, Arnold RM, et al. Development of a symptom assessment instrument for chronic hemodialysis patients: The dialysis symptom index. J Pain Symptom Manage 2004; 27: 226–240.
- Charlson ME, Pompei P, Ales KL, et al. A new method of classifying prognostic comorbidity in longitudinal studies: Development and validation. J Chronic Dis 1987; 40: 373-383.
- 15. Abdel-Kader K, Unruh ML, Weisbord SD. Symptom burden, depression, and quality of life in chronic and end-stage kidney disease. Clin J Am Soc Nephrol 2009; 4: 1057–1064.

- Eekhout I, De Vet HCW, Twisk JWR, et al. Missing data in a multi-item instrument were best handled by multiple imputation at the item score level. J Clin Epidemiol 2014; 67: 335–342.
- 17. Lacson E, Xu J, Lin SF, et al. A comparison of SF-36 and SF-12 composite scores and subsequent hospitalization and mortality risks in long-term dialysis patients. Clin J Am Soc Nephrol 2010; 5: 252–260.
- 18. Kosinski M. User's manual for the SF-12v2 health survey : with a supplement documenting the SF-12® health survey. Lincoln RI: QualityMetric Incorporated, 2007.
- van Sandwijk MS, Arashi D Al, van de Hare FM, et al. Fatigue, anxiety, depression and quality of life in kidney transplant recipients, haemodialysis patients, patients with a haematological malignancy and healthy controls. Nephrol Dial Transplant 2019; 34: 833–838.
- 20. Mittal SK, Ahern L, Flaster E, et al. Self-assessed physical and mental function of haemodialysis patients. Nephrol Dial Transplant 2001; 16: 1387–1394.
- 21. Connor KM, Davidson JRT. Development of a new Resilience scale: The Connor-Davidson Resilience scale (CD-RISC). Depress Anxiety 2003; 18: 76–82.
- 22. Kim EY, Lee YN, Chang SO. How Do Patients on Hemodialysis Perceive and Overcome Hemodialysis?: Concept Development of the Resilience of Patients on Hemodialysis. Nephrol Nurs J 2019; 46(5):521-530.
- 23. García-Martínez P, Temprado-Albalat MD, Ballester-Arnal R, et al. Predictive model of variables associated with health-related quality of life in patients with advanced chronic kidney disease receiving hemodialysis. Qual Life Res 2020; 29: 1817–1827.
- 24. Kukihara H, Yamawaki N, Ando M, et al. The mediating effect of resilience between family functioning and mental well-being in hemodialysis patients in Japan: A cross-sectional design. Health Qual Life Outcomes 2020; 18(1):233 DOI: 10.1186/s12955-020-01486-x.
- 25. Kim EY, Lee YN, Chang SO. Exploring Subjective Frames of Patients on Hemodialysis on Acquiring Resilience: A Q Methodology Study. Nephrol Nurs J 2018; 45(4):357-368
- 26. Chon MY, Yeun EJ, Jung KH, et al. Perceptions of resilience in patients undergoing peritoneal dialysis: A Q-methodology study. Nurs Heal Sci 2020; 22: 108–117.
- 27. Pierce M, Hope H, Ford T, et al. Mental health before and during the COVID-19 pandemic: a longitudinal probability sample survey of the UK population. The Lancet Psychiatry, 2020; 7: 883-892. DOI: 10.1016/S2215-0366(20)30308-4.
- Mazza C, Ricci E, Biondi S, et al. A Nationwide Survey of Psychological Distress among Italian People during the COVID-19 Pandemic: Immediate Psychological Responses and Associated Factors. Int J Environ Res Public Health; 17(9):3165 DOI: 10.3390/ ijerph17093165.
- 29. Purnell TS, Auguste P, Crews DC, et al. Comparison of life participation activities among adults treated by hemodialysis, peritoneal dialysis, and kidney transplantation: A systematic review. Am J Kidney Dis 2013; 62: 953–973.
- 30. Welbevinden en leefstijl | RIVM, https://www.rivm.nl/gedragsonderzoek/maatregelenwelbevinden/welbevinden-en-leefstijl (accessed November 1, 2020).
- 31. COVID-19 Map Johns Hopkins Coronavirus Resource Center, https://coronavirus.jhu.edu/map.html (accessed December 7, 2020).
- 32. Loosman WL, Hoekstra T, Van Dijk S, et al. Short-Form 12 or Short-Form 36 to measure quality-of-life changes in dialysis patients? Nephrol Dial Transplant 2015; 30: 1170–1176.









Chapter 9

Health-Related Quality of Life compared between kidney transplantation and nocturnal hemodialysis

Thijs T. Jansz*, Anna A. Bonenkamp*, Franciscus T. J. Boereboom, Franka E. van Reekum, Marianne C. Verhaar and Brigit C. van Jaarsveld. * These authors contributed equally to this work.

PLoS One. 2018 Sep 20;13(9):e0204405.
Abstract

Background Health-Related Quality of Life (HRQoL) is an important outcome measure in patients with end-stage renal disease. HRQoL is assumed to improve with kidney transplantation and also with nocturnal hemodialysis compared to conventional hemodialysis. However, there is no evidence regarding HRQoL to support the optimal treatment choice for patients on nocturnal hemodialysis who hesitate opting for transplantation. We therefore compared HRQoL between patients who were treated with kidney transplantation or nocturnal hemodialysis for one year.

Methods We assessed HQRoL using the Kidney Disease Quality of Life–Short Form questionnaire in a cross-sectional sample of patients who were treated with kidney transplantation (n=41) or nocturnal hemodialysis (n=31) for one year. All patients on nocturnal hemodialysis were transplantation candidates. Using linear regression, we compared HRQoL between kidney transplantation and nocturnal hemodialysis, and adjusted for age, sex, dialysis duration, cardiovascular disease, and presence of residual urine production.

Results At one year follow-up, mean age of the study population was 54 ± 13 years, and median dialysis duration was 3.2 (IQR 2.1–5.0) years. Kidney transplantation was associated with significantly higher HRQoL on the domain *effects* compared to nocturnal hemodialysis (adjusted difference 12.0 points, 95% CI 3.9; 20.1). There were potentially clinically relevant differences between kidney transplantation and nocturnal hemodialysis on the domains *burden* (adjusted difference 11.1 points, 95% CI -2.6; 24.8), *social support* (adjusted difference 6.2, 95% CI -6.6; 19.1), and the physical composite score (adjusted difference 3.0, 95% CI -2.0; 8.1), but these were not significant.

Conclusion After kidney transplantation, HRQoL is especially higher on the domain "effects of kidney disease" compared to nocturnal hemodialysis. This can be useful when counseling patients on nocturnal hemodialysis who may opt for transplantation.

Introduction

Health-Related Quality of Life (HRQoL) is an important indicator of well-being in patients with end-stage renal disease and is associated with survival and clinical outcomes.¹⁻⁴ Compared to the general population, patients with end-stage renal disease have severely diminished HRQoL, by some deemed even lower than in diseases such as congestive heart failure, chronic lung disease or cancer.⁵

The preferred treatment for end-stage renal disease is kidney transplantation, which is associated with improved HRQoL and survival.⁶ However, because of the limited availability of donor kidneys and because of transplant failure, many patients have to remain on dialysis.

An alternative to conventional dialysis modalities is frequent nocturnal hemodialysis. With this treatment, patients dialyze almost daily and twice as long (7–8 hours), generally at home. Thus, this treatment removes fluid more slowly and clears more solutes such as urea and phosphate.⁷ Nocturnal hemodialysis may hence improve intermediate outcomes^{8, 9} and possibly even survival, although mortality data remain inconsistent.^{10, 11} By dialyzing at night, patients save time during the day, and nocturnal hemodialysis has thus been reported to improve HRQoL¹²⁻¹⁴ to such an extent that some patients may even choose to forgo transplantation.¹⁵

How clinicians should deal with this reluctance toward transplantation is unclear. Currently, there is no evidence to support the optimal treatment choice for these patients, particularly not regarding patient-reported outcome measures. To fill this gap, we compared HRQoL measured with the Kidney Disease Quality of Life—Short Form (KDQOL-SF) between kidney transplant recipients and transplantation-eligible patients treated with nocturnal hemodialysis.

Methods

Study population

We analyzed a cross-sectional cohort from the ongoing NOCTx study (NCT00950573), a prospective cohort study designed to compare progression of coronary artery calcification between kidney transplant recipients, patients on frequent nocturnal

home hemodialysis, and patients on chronic peritoneal dialysis or conventional hemodialysis. Patients were eligible when aged between 18 and 75 years and were candidates for transplantation when on dialysis. All study participants gave written informed consent. NOCTx excluded patients with a life expectancy <3 months, preemptive transplantation, or non-adherence to dialysis regimens. NOCTx has been approved by the Medical Ethics Committee of the University Medical Center Utrecht and is conducted in accordance with the Declaration of Helsinki.

Between December 2009 and February 2016, NOCTx included 54 kidney transplant recipients and 39 patients on nocturnal hemodialysis who were referred for study participation to the University Medical Center of Utrecht, the Netherlands. For the present analyses, we included all kidney transplant recipients (n=41) and patients on nocturnal hemodialysis (n=31) who had one-year follow-up data. Most patients with a kidney transplant and on nocturnal hemodialysis entered NOCTx 2–3 months after switching to their respective treatment; thus, data from before switching were not available in these patients. We therefore analyzed data cross-sectionally after one year of treatment.

Treatment characteristics

Patients received treatment according to guidelines by the attending nephrologists. Kidney transplant recipients were treated in two tertiary centers, where standard immunosuppressant regimens consisted of a calcineurin inhibitor (tacrolimus), mycophenolate mofetil, and prednisone in tapering doses. Patients on nocturnal hemodialysis were trained and monitored in two dialysis centers that offered specialized training programs for nocturnal home hemodialysis. Patients dialyzed $\geq 4 \times 8$ hours per week at home, on a single needle, with a lower effective blood flow (150–220 mL/min), lower dialysate flow (300 mL/min), and a somewhat lower bicarbonate concentration compared to conventional hemodialysis, which was adjusted depending on laboratory results. Unfractionated heparin was used as anticoagulation.

Health-Related Quality of Life

We assessed HRQoL with the validated KDQOL-SF version 1.2.¹⁶ The KDQOL-SF consists of a general part and a disease-specific part. The general part, the Short Form with 36 questions (SF-36) version 1¹⁷, consists of eight domains that can be summarized in two scores. These summary scores are designed to reflect the general population in the United States when the means are 50 with a standard deviation of

10 points for physical functioning (physical composite score) and mental functioning (mental composite score).¹⁸ The composite scores were obtained from 12 questions in the SF-36 (PCS-12 and MCS-12).¹ The disease-specific part of the KDQOL-SF consists of 44 kidney disease-targeted questions, grouped in 12 domains. We focused on the domains *symptoms of kidney disease, effects of kidney disease, burden of kidney disease, cognitive function, quality of social interaction, sexual function, sleep, social support and overall health.* We did not evaluate the domains *work status, patient satisfaction* and *dialysis staff encouragement* in this study. The domains are scored from 0 to 100, with higher scores indicating better quality of life. Explanations of the disease-specific domains are available as Table 1 (adapted from Carmichael *et al.*¹⁹).

Domains	Interpretation			
	Low score	High score		
Symptoms of kidney disease	Extremely bothered by dialysis-related symptoms such as muscle cramps, pruritus, anorexia, and/or access problems	Not at all bothered		
Effect of kidney disease on daily life	Extremely bothered by fluid and dietary restriction, by an inability to travel, and dependency on doctors	Not at all bothered		
Burden of kidney disease	Extremely bothered by the time consumed by dialysis, its intrusiveness, and degree burden on family	Not at all bothered		
Cognitive function	Affected all of the time by inability to concentrate, confused, with poor reaction time	Not at all affected		
Quality of social interaction	Continual irritation and failure to get along with people with virtual isolation	No problems, socially interactive		
Sexual function	Experiencing severe problems with enjoyment and arousal	No problems		
Sleep	Very poor sleep with daytime somnolence	No problems with sleep		
Social support	Very dissatisfied	Satisfied with level of social support		
Overall health	Rates health as worst possible	Rates health as best possible		

 Table 1. Explanation of the Kidney Disease Quality of Life-Short Form (KDQOL-SF) kidney disease-specific domains.

Adapted from Carmichael et al.19

Other variables

At time of questionnaire completion, study personnel recorded demographical and clinical parameters (pre-dialysis blood pressure and post-dialysis weight averaged from routine measurements during 3 hemodialysis sessions or 2 outpatient visits for kidney transplant recipients) and laboratory parameters (total calcium, phosphate, parathyroid hormone, total cholesterol, albumin, hemoglobin, and C-reactive protein) routinely measured at local treatment facilities. Study personnel assessed presence of comorbidities by chart review, and assessed residual urine production with the most recent 24h-urine collection, which we classified as present (≥100mL/24u) or absent. Smoking status, oral anticoagulant use, and educational level were self-reported.

We defined diabetes mellitus as use of oral anti-diabetic medication or insulin therapy, and cardiovascular disease as any history of angina, myocardial infarction, percutaneous coronary intervention, coronary artery bypass grafting, stroke, intermittent claudication, peripheral artery angioplasty or bypass grafting. We defined higher education as any tertiary education. We estimated glomerular filtration rate with the Chronic Kidney Disease-Epidemiology Collaboration equation 2009 for kidney transplant recipients.

Statistical Analyses

We reported data as number (proportion) for categorical data, mean ±standard deviation for normally distributed variables, and median (interquartile range [IQR]) for non-normally distributed variables. We presented patient characteristics and HRQoL by renal replacement therapy. We compared categorical data with chi-squared tests, normally distributed variables with t-tests, and non-normally distributed variables with Mann-Whitney-U tests.

We used multiple linear regression analyses to examine the associations between renal replacement therapy and HRQoL. We regarded 5-point differences clinically relevant in the disease-specific domains, and 3-point differences clinically relevant in the composite scores.^{17, 18} We adjusted stepwise for potential confounders age (years), sex, educational level (high/low), dialysis duration (years), presence of diabetes mellitus, cardiovascular disease, and presence of residual urine production (≥100mL/24u or absent), and kept them in the model when coefficients changed >10%. In the final model, we adjusted for age, sex, dialysis duration, cardiovascular disease, and presence of residual urine production.

We reported regression coefficients with 95% confidence intervals (CI). We considered P-values ≤ 0.05 (two-tailed) statistically significant, did not attempt imputation for missing values, and performed all analyses with R 3.4.1.²⁰

Results

Study population

The mean age of the study population (n=72) was 54 ±13 years, 50 (69%) were male, median dialysis duration was 38 (IQR 25–60) months, and 17 (24%) had a history of cardiovascular disease. There were no significant differences in demographics or medical history between the kidney transplant recipients (n=41) and patients on nocturnal hemodialysis (n=31), but kidney transplant recipients had significantly lower phosphate levels and higher hemoglobin levels (Table 2). Kidney transplant recipients had an estimated glomerular filtration rate of 54.8 ±15.7 mL/min, while patients on nocturnal hemodialysis had median 0 (IQR 0–250) mL/day residual urine production. Patients on nocturnal hemodialysis dialyzed 38.3 ±7.2 hours per week in 4.8 ±0.8 sessions per week.

	Kidney transplantation (n = 41)	Nocturnal hemodialysis (n = 31)	P-value
Demographics			
Age (yr)	54.0 ±13.8	53.9 ±12.5	0.97
Male (%)	31 (75)	19 (62)	0.29
Body mass index (kg/m2)	25.5 ± 4.2	26.5 ±5.2	0.37
Systolic blood pressure (mmHg)	132 ±14	139 ±20	0.11
Diastolic blood pressure (mmHg)	80 ±10	75 ±12	0.12
Current smoker (%)	6 (15)	6 (19)	0.83
Oral anticoagulant use (%)	5 (13)	2 (7)	0.66
Higher education (%)	11 (28)	8 (26)	0.99
Medical history			
Dialysis duration (mo)	28 (24-58)	39 (28–66)	0.12
End-stage renal disease duration (mo)	28 (25-62)	39 (28–94)	0.15

Table 2. Characteristics of the 72 kidney transplant recipients and patients on nocturnal hemodialysis at one year of follow-up.

	Kidney transplantation (n = 41)	Nocturnal hemodialysis (n = 31)	P-value
Cause of end-stage renal disease (%)			0.23
Glomerulonephritis	9 (22)	11 (36)	
Interstitial nephritis	1 (2)	0 (0)	
Cystic kidney disease	14 (34)	5 (16)	
Renovascular	9 (22)	3 (10)	
Diabetes mellitus	1 (2)	2 (7)	
Other	3 (7)	5 (16)	
Unknown	4 (10)	5 (16)	
Comorbidities (%)			
Diabetes mellitus	3 (7)	4 (13)	0.70
Prior cardiovascular disease	7 (17)	10 (32)	0.22
Laboratory parameters			
Calcium (mmol/L)	2.41 ± 0.10	2.37 ± 0.20	0.30
Phosphate (mmol/L)	0.88 ± 0.21	1.42 ±0.39	< 0.001
Parathyroid hormone (pmol/L)	8.5 (6.4-12.0)	13.8 (7.6-22.8)	0.14
Cholesterol (mmol/L)	5.0 ±1.1	4.6 ±1.0	0.26
Albumin (g/L)	42.4 ±3.1	42.4 ±3.1	0.95
Hemoglobin (mmol/L)	8.9 ±1.0	7.0 ±0.8	< 0.001
C-reactive protein (mg/L)	3.0 (2.0-8.3)	5.0 (3.0-10.0)	0.29

Table 2. Characteristics of the 72 kidney transplant recipients and patients on nocturnalhemodialysis at one year of follow-up. (continued)

Results are presented as mean ±standard deviation, median (interquartile range), or number (proportion).

The current sample comprised 77% of all kidney transplant recipients and patients on nocturnal hemodialysis who entered NOCTx (n=93). Seven kidney transplant recipients (3 were lost to follow-up, 2 withdrew consent, 2 died) and 7 patients on nocturnal hemodialysis (3 received a transplant, 2 withdrew consent, 1 was lost to follow-up, 1 died) did not complete follow-up at one year, while 6 kidney transplant recipients and 1 patient on nocturnal hemodialysis did not complete quality of life questionnaires at the one-year follow-up. Their mean age (n=21) was 49 ±14 years (P=0.15 versus study population), 12 (57%) were male (P=0.43 versus study population), median dialysis duration was 65 (IQR 42–84) months (P=0.03 versus study population), and 4 (19%) had a history of cardiovascular disease (P=0.89 versus study population). Kidney transplant recipients were not more likely to complete follow-up than patients on nocturnal hemodialysis (P=0.88).

Health-Related Quality of Life at one year of treatment

The quality of life questionnaires were generally well-completed. In the following scales, one or more questionnaire items were missing resulting in a missing score: *sexual function* (5 respondents, 7%), SF-12 items (physical and mental composite scores; 2 respondents, 3%), *symptoms of kidney disease*, *effects of kidney disease*, *burden of kidney disease*, and *overall health* (1 respondent each, 1%).

Overall, kidney transplant recipients had numerically higher scores on the kidney disease-specific domains of HRQoL and the physical composite score compared to patients on nocturnal hemodialysis (Figure 1). Kidney transplant recipients scored significantly higher on the domain "effects of kidney disease" compared to patients on nocturnal hemodialysis, both in crude and adjusted analyses (Table 3). There were no significant differences on the other kidney disease-specific domains or the composite scores in both crude and adjusted analyses. When adjusted for age, sex, dialysis duration, cardiovascular disease, and residual urine production, kidney transplant recipients had potentially clinically relevant higher scores on the domains *burden of kidney disease, social support*, and the physical composite score compared to nocturnal hemodialysis, but these differences were not significant.





Figure 1. Disease-specific Health-Related Quality of Life scores and physical composite scores in the 72 kidney transplant recipients and patients on nocturnal hemodialysis. Mean HRQoL scores on the disease-specific domains *symptoms*, effects, *burden of kidney disease*, and the physical composite scores as bar charts in the 72 kidney transplant recipients and patients on nocturnal hemodialysis. We presented 95% confidence intervals alongside the bars. Mean scores for kidney transplantation and nocturnal hemodialysis: "symptoms" 86 and 81; "effects" 86 and 76; "burden" 75 and 67; physical composite score 47 and 43 points, respectively.

	Kidney transplantation (n = 41)	Nocturnal hemodialysis (n = 31)	Crude difference (95% CI)	Adjusted* difference (95% CI)				
Kidney disease-related quality of life								
Symptoms of kidney disease	86 ±11	81 ±10	-5.7 (-10.7; -0.7)	-4.6 (-10.6; 1.3)				
Effects of kidney disease	86 ±14	76 ±17	-9.8 (-16.9; -2.6)	-12.0 (-20.1; -3.9)				
Burden of kidney disease	75 ±27	67 ±24	-8.0 (-20.1; 4.1)	-11.1 (-24.8; 2.6)				
Cognitive function	81 ±19	78 ±18	-2.5 (-11.3; 6.3)	-4.3 (-14.2; 5.6)				
Quality of social interaction	79 ±15	77 ±14	-1.3 (-8.3; 5.8)	1.4 (-6.7; 9.5)				
Sexual function	72 ±30	64 ±33	-7.8 (-23.1; 7.5)	-2.0 (-19.1; 15.0)				
Sleep	66 ±23	63 ±16	-2.8 (-12.3; 6.8)	-3.3 (-14.5; 8.0)				
Social support	87 ±21	82 ±25	-4.7 (-15.5; 6.0)	-6.2 (-19.1; 6.6)				
Overall health	70 ±16	65 ±17	-4.3 (-12.3; 3.6)	-4.9 (-14.1; 4.3)				
SF-12 composite s	cores							
Physical composite score	47 ±10	43 ±8	-3.4 (-7.7; 0.9)	-3.0 (-8.1; 2.0)				
Mental composite score	51 ±10	52 ±11	0.6 (-4.2; 5.5)	1.2 (-4.4; 6.8)				

Table 3. Health-Related Quality of Life scores and differences in scores between the 72 kidney transplant recipients and patients on nocturnal hemodialysis at one year of follow-up.

Abbreviations: SF-12: short form-12 items. Scores are presented as mean ±standard deviation, and differences with 95% confidence intervals.

*Adjusted for age (years), sex (male/female), dialysis duration (years), cardiovascular disease, and presence of residual urine production (≥100mL/24u or absent).

Discussion

To our knowledge, this is the first study to compare HRQoL between kidney transplantation and nocturnal hemodialysis, demonstrating that kidney transplantation is associated with significantly higher quality of life on the domain *effects of kidney disease* compared to nocturnal hemodialysis. In addition, kidney transplant recipients

have potentially clinically relevant higher quality of life on the domains *burden* of kidney disease, social support, and the physical composite score, although not significantly higher in this study. Together, these findings suggest that HRQoL is generally better after kidney transplantation than on treatment with nocturnal hemodialysis.

The differences in HRQoL are the most evident on the domain *effects of kidney disease*. As this domain involves the restraints patients experience regarding their diet, ability to travel, and dependency on doctors, it is explainable that kidney transplant recipients score higher on this domain. After all, kidney transplant recipients are freer in terms of diet and travel than any patient on dialysis. Besides this domain, kidney transplant recipients have numerically higher adjusted scores on the domains *burden of kidney disease, social support*, and the physical composite score. Although not *statistically significant*, these differences may be *clinically relevant*.²¹ The original KDQOL-SF manual reads that 5-point differences regarding the composite scores^{17, 18}, which has been adopted by others.^{22, 23} Notably, a 3-point difference in the composite scores is associated with a mortality risk of approximately 6.0%.^{2, 3, 24} Given the size and consistent direction of these differences, we consider them relevant, even though they do not reach statistical significance with the current sample size.

In our experience, some patients treated with nocturnal hemodialysis decline kidney transplantation and prefer to stay on treatment with nocturnal hemodialysis. The current findings suggest that kidney transplantation - in which quality of life is known to improve²⁵⁻²⁷ - is a more favorable treatment option regarding HRQoL for transplantation-eligible patients on nocturnal hemodialysis, although individual outcomes may differ importantly.

For patients that are unlikely to receive a kidney transplant (e.g. HLA-sensitized patients), potential benefits of nocturnal hemodialysis remain relevant, such as an improvement of quality of life. Importantly, HRQoL has been shown to improve after conversion to nocturnal hemodialysis from conventional hemodialysis in several observational studies^{12, 14} and on selected domains in a randomized trial.²⁸ This is despite the fact that nocturnal hemodialysis is performed almost daily and requires considerable patient involvement. Notably, patients on nocturnal hemodialysis in our cohort have somewhat higher HRQoL scores compared to North-American

cohorts^{12, 13, 29}, which may be because all patients were transplantation candidates in our study. Remarkably, nocturnal hemodialysis does not seem to deteriorate sleep quality: patients on nocturnal hemodialysis have similar scores on the domain *sleep* to kidney transplant recipients in our study.

The results of this study should be interpreted within the context of some limitations. First, our study is not powered to demonstrate significance of all potentially relevant differences in kidney disease-specific HRQoL domains. For example, we would have needed 161 patients per group to show significance of an 8-point difference (as currently found) in the disease-specific domain *burden of kidney disease*. Second, the current data are cross-sectional after one year of treatment with kidney transplantation or nocturnal hemodialysis. A before/after comparison of HRQoL was not possible as patients were included in this study shortly after they had started treatment with either kidney transplantation or nocturnal hemodialysis. Third, we do not know the reasons why individual patients converted to nocturnal hemodialysis – there may have been patient selection. As noted in previous studies, healthier and more motivated patients may have been preferentially selected for nocturnal hemodialysis³⁰, which could influence HRQoL. Also, the current data are observational, although it should be noted that randomization to kidney transplantation would be unethical.

Our study has several strengths. First, questionnaire response rate in this study is high (91%) as compared to large studies on patients on hemodialysis.^{1, 2} The responders' demographic characteristics are largely similar to that of non-responders; therefore, we consider our findings generalizable to patients on nocturnal hemodialysis who may opt for kidney transplantation. Second, we focus on kidney disease-specific domains of HRQoL alongside the physical and mental composite scores, which increases the ability to detect more specific differences in patients' well-being. Finally, this study has only included patients on nocturnal hemodialysis who were transplantation candidates; simultaneously, no kidney transplant recipients had been transplanted pre-emptively, i.e. all recipients had a history of dialysis treatment. Both of these inclusion criteria enable valid comparisons between the treatment groups.

In conclusion, HRQoL is higher after kidney transplantation especially on the domain *effects of kidney disease* compared to nocturnal hemodialysis. This can be useful when counseling patients on nocturnal hemodialysis who may opt for transplantation.

References

- Lacson E, Jr., Xu J, Lin SF, Dean SG, Lazarus JM, Hakim RM. A comparison of SF-36 and SF-12 composite scores and subsequent hospitalization and mortality risks in long-term dialysis patients. Clin J Am Soc Nephrol. 2010;5(2):252-60. doi: 10.2215/CJN.07231009. PubMed PMID: 20019120; PubMed Central PMCID: PMCPMC2827595.
- 2. Mapes DL, Lopes AA, Satayathum S, McCullough KP, Goodkin DA, Locatelli F, *et al.* Health-related quality of life as a predictor of mortality and hospitalization: the Dialysis Outcomes and Practice Patterns Study (DOPPS). Kidney Int. 2003;64(1):339-49. doi: 10.1046/j.1523-1755.2003.00072.x. PubMed PMID: 12787427.
- Lowrie EG, Curtin RB, LePain N, Schatell D. Medical outcomes study short form-36: a consistent and powerful predictor of morbidity and mortality in dialysis patients. Am J Kidney Dis. 2003;41(6):1286-92. PubMed PMID: 12776282.
- 4. Kalantar-Zadeh K, Kopple JD, Block G, Humphreys MH. Association among SF36 quality of life measures and nutrition, hospitalization, and mortality in hemodialysis. J Am Soc Nephrol. 2001;12(12):2797-806. PubMed PMID: 11729250.
- Mittal SK, Ahern L, Flaster E, Maesaka JK, Fishbane S. Self-assessed physical and mental function of haemodialysis patients. Nephrol Dial Transplant. 2001;16(7):1387-94. PubMed PMID: 11427630.
- Tonelli M, Wiebe N, Knoll G, Bello A, Browne S, Jadhav D, *et al.* Systematic review: kidney transplantation compared with dialysis in clinically relevant outcomes. Am J Transplant. 2011;11(10):2093-109. doi: 10.1111/j.1600-6143.2011.03686.x. PubMed PMID: 21883901.
- 7. Pierratos A, Ouwendyk M, Francoeur R, Vas S, Raj DS, Ecclestone AM, *et al.* Nocturnal hemodialysis: three-year experience. J Am Soc Nephrol. 1998;9(5):859-68. PubMed PMID: 9596084.
- Jansz TT, Ozyilmaz A, Grooteman MPC, Hoekstra T, Romijn M, Blankestijn PJ, et al. Long-term clinical parameters after switching to nocturnal haemodialysis: a Dutch propensity-score-matched cohort study comparing patients on nocturnal haemodialysis with patients on three-times-a-week haemodialysis/haemodiafiltration. BMJ Open. 2018;8(3):e019900. doi: 10.1136/bmjopen-2017-019900. PubMed PMID: 29523566; PubMed Central PMCID: PMCPMC5855195.
- 9. Walsh M, Culleton B, Tonelli M, Manns B. A systematic review of the effect of nocturnal hemodialysis on blood pressure, left ventricular hypertrophy, anemia, mineral metabolism, and health-related quality of life. Kidney Int. 2005;67(4):1500-8. doi: 10.1111/j.1523-1755.2005.00228.x. PubMed PMID: 15780103.
- Pauly RP, Gill JS, Rose CL, Asad RA, Chery A, Pierratos A, *et al.* Survival among nocturnal home haemodialysis patients compared to kidney transplant recipients. Nephrol Dial Transplant. 2009;24(9):2915-9. doi: 10.1093/ndt/gfp295. PubMed PMID: 19584107.
- Rocco MV, Daugirdas JT, Greene T, Lockridge RS, Chan C, Pierratos A, *et al.* Long-term Effects of Frequent Nocturnal Hemodialysis on Mortality: The Frequent Hemodialysis Network (FHN) Nocturnal Trial. Am J Kidney Dis. 2015;66(3):459-68. doi: 10.1053/j.ajkd.2015.02.331. PubMed PMID: 25863828; PubMed Central PMCID: PMCPMC4549208.

- Lockridge RS, Jr., Spencer M, Craft V, Pipkin M, Campbell D, McPhatter L, *et al.* Nightly home hemodialysis: five and one-half years of experience in Lynchburg, Virginia. Hemodialysis international International Symposium on Home Hemodialysis. 2004;8(1):61-9. doi: 10.1111/j.1492-7535.2004.00076.x. PubMed PMID: 19379403.
- Garg AX, Suri RS, Eggers P, Finkelstein FO, Greene T, Kimmel PL, et al. Patients receiving frequent hemodialysis have better health-related quality of life compared to patients receiving conventional hemodialysis. Kidney Int. 2017;91(3):746-54. doi: 10.1016/j.kint.2016.10.033. PubMed PMID: 28094031; PubMed Central PMCID: PMCPMC5333984.
- Van Eps CL, Jeffries JK, Johnson DW, Campbell SB, Isbel NM, Mudge DW, et al. Quality of life and alternate nightly nocturnal home hemodialysis. Hemodialysis international International Symposium on Home Hemodialysis. 2010;14(1):29-38. doi: 10.1111/j.1542-4758.2009.00419.x. PubMed PMID: 20377650.
- Rosenthal MM, Molzahn AE, Chan CT, Cockfield SL, Kim SJ, Pauly RP. Why take the chance? A qualitative grounded theory study of nocturnal haemodialysis recipients who decline kidney transplantation. BMJ Open. 2016;6(5):e011951. doi: 10.1136/ bmjopen-2016-011951. PubMed PMID: 27194322; PubMed Central PMCID: PMCPMC4874163.
- Korevaar JC, Merkus MP, Jansen MA, Dekker FW, Boeschoten EW, Krediet RT, et al. Validation of the KDQOL-SF: a dialysis-targeted health measure. Qual Life Res. 2002;11(5):437-47. PubMed PMID: 12113391.
- 17. Ware JE, Snow KK, Kosinski M, Gandek B. SF-36 Health Survey-Manual and Interpretation Guide. Boston: The Health Institute, New England Medical Center; 1993.
- Ware JE, Kosinski M, Keller SD. SF-36 Physical and Mental Health Summary Scales: A User's Manual, 2nd Ed. Boston: The Health Institute, New England Medical Center; 1994.
- 19. Carmichael P, Popoola J, John I, Stevens PE, Carmichael AR. Assessment of quality of life in a single centre dialysis population using the KDQOL-SF questionnaire. Qual Life Res. 2000;9(2):195-205. PubMed PMID: 10983483.
- 20. R Core Team. R: A Language and Environment for Statistical Computing. Vienna, Austria: R Foundation for Statistical Computing; 2016.
- van Rijn MHC, Bech A, Bouyer J, van den Brand J. Statistical significance versus clinical relevance. Nephrol Dial Transplant. 2017;32(suppl_2):ii6-ii12. doi: 10.1093/ndt/gfw385. PubMed PMID: 28064161.
- Mazairac AH, Grooteman MP, Blankestijn PJ, Penne EL, van der Weerd NC, den Hoedt CH, *et al.* Differences in quality of life of hemodialysis patients between dialysis centers. Qual Life Res. 2012;21(2):299-307. doi: 10.1007/s11136-011-9942-3. PubMed PMID: 21633878; PubMed Central PMCID: PMCPMC3276757.
- 23. Hall YN, Larive B, Painter P, Kaysen GA, Lindsay RM, Nissenson AR, et al. Effects of six versus three times per week hemodialysis on physical performance, health, and functioning: Frequent Hemodialysis Network (FHN) randomized trials. Clin J Am Soc Nephrol. 2012;7(5):782-94. doi: 10.2215/CJN.10601011. PubMed PMID: 22422538; PubMed Central PMCID: PMCPMC3338281.
- 24. DeOreo PB. Hemodialysis patient-assessed functional health status predicts continued survival, hospitalization, and dialysis-attendance compliance. Am J Kidney Dis. 1997;30(2):204-12. PubMed PMID: 9261030.

- 25. Laupacis A, Keown P, Pus N, Krueger H, Ferguson B, Wong C, *et al.* A study of the quality of life and cost-utility of renal transplantation. Kidney Int. 1996;50(1):235-42. PubMed PMID: 8807593.
- Molnar-Varga M, Molnar MZ, Szeifert L, Kovacs AZ, Kelemen A, Becze A, *et al.* Healthrelated quality of life and clinical outcomes in kidney transplant recipients. Am J Kidney Dis. 2011;58(3):444-52. doi: 10.1053/j.ajkd.2011.03.028. PubMed PMID: 21658828.
- 27. Overbeck I, Bartels M, Decker O, Harms J, Hauss J, Fangmann J. Changes in quality of life after renal transplantation. Transplant Proc. 2005;37(3):1618-21. doi: 10.1016/j. transproceed.2004.09.019. PubMed PMID: 15866689.
- Culleton BF, Walsh M, Klarenbach SW, Mortis G, Scott-Douglas N, Quinn RR, et al. Effect of frequent nocturnal hemodialysis vs conventional hemodialysis on left ventricular mass and quality of life: a randomized controlled trial. JAMA. 2007;298(11):1291-9. doi: 10.1001/jama.298.11.1291. PubMed PMID: 17878421.
- 29. Manns BJ, Walsh MW, Culleton BF, Hemmelgarn B, Tonelli M, Schorr M, *et al.* Nocturnal hemodialysis does not improve overall measures of quality of life compared to conventional hemodialysis. Kidney Int. 2009;75(5):542-9. doi: 10.1038/ki.2008.639. PubMed PMID: 19109588.
- Tennankore KK, Na Y, Wald R, Chan CT, Perl J. Short daily-, nocturnal- and conventional-home hemodialysis have similar patient and treatment survival. Kidney Int. 2018;93(1):188-94. doi: 10.1016/j.kint.2017.06.014. PubMed PMID: 28844317.

Supplemental material

Extra Supplemental table 1. Crude differences in Health-Related Quality of Life between kidney transplantation and dialysis at one year of follow-up.

	KT (n = 41)	NHD) (n = 31)		PD (n = 18)		CHD (n = 25)	
		B (95% CI)	p-value	B (95% CI)	p-value	B (95% CI)	p-value
Kidney disease-related quality of life							
Symptoms of kidney disease	4	-5.7 (-12.1; 0.7)	0.08	-8.6 (-16.2; -1.1)	0.03	-5.7 (-12.6; 1.2)	0.10
Effects of kidney disease	*	-9.8 (-17.5; -2.1)	0.01	-13.6 (-22.7; -4.5)	<0.01	-16.3 (-24.5; -8.1)	<0.001
Burden of kidney disease	*	-8.0 (-20.4; 4.4)	0.20	-16.5 (-31.2; -1.8)	0.03	-21.8 (-35.0; -8.6)	0.001
Work status	*	-11.3 (-31.8; 9.3)	0.28	-6.2 (-30.6; 18.1)	0.61	-7.1 (-29.0; 14.8)	0.52
Cognitive function	*	-2.5 (-10.9; 5.8)	0.55	5.1 (-4.8; 15.1)	0.31	-0.3 (-9.2; 8.7)	0.95
Quality of social interaction	*	-1.3 (-8.3; 5.7)	0.72	2.0 (-6.2; 10.3)	0.63	1.6 (-5.9; 9.0)	0.68
Sexual function	*	-7.8 (-24.6; 9.0)	0.36	-12.7 (-32.9; 7.5)	0.22	-13.4 (-33.6; 6.7)	0.19
Sleep	*	-2.8 (-12.1; 6.6)	0.56	-1.2 (-12.3; 9.9)	0.83	-1.7 (-11.6; 8.3)	0.74
Social support	*	-4.7 (-16.2; 6.8)	0.41	-7.4 (-21.0; 6.3)	0.29	-19.0 (-31.3; -6.7)	<0.01
Overall health	*	-4.3 (-12.1; 3.4)	0.27	-10.1 (-19.2; -0.9)	0.03	-7.1 (-15.3; 1.1)	0.09
SF-12 composite	scores						
Physical composite score	4	-3.4 (-7.9; 1.1)	0.14	-5.9 (-11.4; -0.3)	0.04	-3.5 (-8.8; 1.7)	0.19
Mental composite score	*	0.6 (-3.7; 5.0)	0.77	2.7 (-2.7; 8.1)	0.32	-0.3 (-5.4; 4.8)	0.92

*Reference group.

Abbreviations: KT: kidney transplantation; NHD: nocturnal hemodialysis; PD: peritoneal dialysis; CHD: conventional hemodialysis; 95% CI: 95% confidence interval; SF-12: short form-12 items.

	KT	NHD	PD	CHD
	(n = 41)	(n = 31)	(n = 18)	(n = 25)
-		B (95% CI)	B (95% CI)	B (95% CI)
Kidney disease-related quality of	of life			
Symptoms of kidney disease	*	-6.1 (-13.0; 0.8)	-10.9 (-20.1; -1.8)	-6.8 (-15.3; 1.6)
Effects of kidney disease	×-	-10.7 (-18.9; -2.6)	-14.4 (-25.3; -3.5)	-17.5 (-27.4; -7.7)
Burden of kidney disease	×-	-10.7 (-23.9; 2.4)	-22.2 (-39.7; -4.7)	-27.7 (-43.6; -11.8)
Cognitive function	×-	-3.5 (-12.3; 5.3)	1.4 (-10.4; 13.2)	-1.9 (-12.6; 8.9)
Quality of social interaction	26-	0.2 (-7.1; 7.4)	2.9 (-6.9; 12.6)	5.0 (-3.8; 13.9)
Sexual function	*	-7.2 (-25.0; 10.6)	-13.0 (-37.2; 11.2)	-10.7 (-34.2; 12.7)
Sleep	26-	-2.9 (-12.8; 7.1)	-2.1 (-15.4; 11.2)	-1.8 (-13.8; 10.2)
Social support	×-	-6.2 (-18.4; 6.0)	-8.2 (-24.6; 8.1)	-21.9 (-36.7; -7.1)
Overall health	26-	-4.5 (-12.7; 3.8)	-10.5 (-21.5; 0.5)	-6.8 (-16.8; 3.2)
SF-12 composite scores				
Physical composite score	*	-4.7 (-9.6; 0.1)	-9.3 (-16.0; -2.6)	-6.2 (-12.5; 0.2)
Mental composite score	*	0.1 (-4.6; 4.8)	2.1 (-4.4; 8.7)	-1.5 (-7.7; 4.7)

Extra Supplemental table 2. Adjusted* differences in Health-Related Quality of Life between kidney transplant recipients (n=41) and patients on nocturnal hemodialysis (n=31), peritoneal dialysis (n=18) and conventional hemodialysis (n=25).

Regression coefficients (B) are to be interpreted as the absolute differences in quality of life scores. Reference: kidney transplantation.

SF-12: short form-12 items.

*Adjusted for age (years), sex (male/female), dialysis duration (years), and presence of residual urine production (≥100mL/24u or absent)



Extra Supplemental Figure S1. Bar charts to present Health-Related Quality of Life on the Physical Component score in kidney transplant, nocturnal hemodialysis, peritoneal dialysis and hemodialysis patients. 95% confidence intervals are presented alongside bars.



Extra Supplemental Figure S2. Bar charts to present Health-Related Quality of Life on the disease-specific domains *symptoms*, *effects*, *burden of kidney disease*, and *overall health* in kidney transplant, nocturnal hemodialysis, peritoneal dialysis and hemodialysis patients. 95% confidence intervals are presented alongside bars.



Part IV

Discussion and future perspectives



Chapter 10 Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO): rationale and design

Anita van Eck van der Sluijs*, Anna A. Bonenkamp*, Friedo W. Dekker, Alferso C. Abrahams and Brigit C. van Jaarsveld on behalf of the DOMESTICO study group * These authors contributed equally to this work.

BMC Nephrology 2019 Sep 18;20(1):361.

Abstract

Background More than 6200 End Stage Kidney Disease patients in the Netherlands are dependent on dialysis, either performed at home or in a dialysis centre. Visiting a dialysis centre three times a week is considered a large burden by many patients. However, recent data regarding the effects of dialysis at home on quality of life, clinical outcomes, and costs compared with in-centre haemodialysis are lacking.

Methods The Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO) is a nationwide, prospective, observational cohort study that will include adult patients starting with a form of dialysis. Health-Related Quality of Life, as the primary outcome, clinical outcomes and costs, as secondary outcomes, will be measured every 3-6 months in patients on home dialysis, and compared with a control group consisting of in-centre haemodialysis patients. During a 3-year period 800 home dialysis patients (600 peritoneal dialysis and 200 home haemodialysis patients) and a comparison group of 800 in-centre haemodialysis patients will be included from 53 Dutch dialysis centres (covering 96% of Dutch centres) and 1 Belgian dialysis centre (covering 4% of Flemish centres).

Discussion DOMESTICO will prospectively investigate the effect of home dialysis therapies on Health-Related Quality of Life, clinical outcomes and costs, in comparison with in-centre haemodialysis. The findings of this study are expected to ameliorate the shared decision-making process and give more guidance to healthcare professionals, in particular to assess which type of patients may benefit most from home dialysis.

Trial registration The DOMESTICO study is registered with the National Trial Register on https://www.trialregister.nl/trial/6519 (number: NL6519, date of registration: 22 August 2017) and the Central Committee on Research Involving Human Subjects (CCMO) (number: NL63277.029.17).

Background

In the Netherlands, over 6200 patients with End Stage Kidney Disease (ESKD) are dependent on dialysis, and over the past 15 years, the number of dialysis patients has increased by more than 20%.¹⁻³ The burden of dialysis is high and the Health-Related Quality of Life (HRQoL), which is presently considered to be the most important outcome parameter in dialysis patients, is much worse than that of healthy people.⁴ As patient survival is poor, with a median five-year survival rate of only 45%, optimising HRQoL is of great importance for this growing group of patients.^{5,6}

Besides its impact on HRQoL, dialysis is also an expensive treatment. In the Netherlands, the estimated costs are approximately 570 million euro per year (639 million US dollars) and are still increasing.[Personal communications, G.A. De Wit, National Institute for Public Health and the Environment, 2019] This makes dialysis by far the highest cost-consuming treatment in internal medicine, not only calculated per individual patient, but also if total treatment costs are taken into account.⁷

Home dialysis has a potential positive effect on HRQoL because it offers flexibility to patients and greater freedom.⁸ Moreover, home dialysis is possibly a more costeffective therapy if less nursing staff is needed, when patients perform their treatment autonomously or with help of an informal caregiver. Despite these potential advantages, currently more than 80% of dialysis patients are treated with in-centre haemodialysis (ICHD). Furthermore, the percentage of patients treated with home dialysis is steadily decreasing in the Netherlands, from 32% in 2002 to 18% in 2018. This decline is mainly attributable to a reduction in the number of patients performing peritoneal dialysis (PD), the main home based therapy, with 1,519 PD patients (30% of total dialysis patients) in 2002 versus 894 PD patients (14% of total dialysis patients) in 2018.¹

Available evidence regarding the effects of home dialysis compared with ICHD on HRQoL, a Patient Reported Outcome (PRO), is limited. Most studies have a cross-sectional design and lack adequate correction for confounding among dialysis groups.⁹⁻³⁸ Also, the characteristics of patients starting with some kind of home dialysis treatment have changed remarkably over the past years. Previously, those patients were typically young, working people with little comorbidities, whereas during the last years the general home dialysis population is older and often suffers

from multiple comorbidities.² This could influence clinical outcomes such as mortality and hospitalisation rate. Finally, there are limited data available regarding the cost-effectiveness of home dialysis.

To investigate the effect of home dialysis on HRQoL, clinical outcomes, and costs, the Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO) has been initiated. The aim of this study is to compare HRQoL, clinical outcomes, and cost-effectiveness of home dialysis with ICHD. The hypothesis is that home dialysis is associated with better HRQoL, at least comparable clinical outcomes and lower costs, compared to ICHD.

Methods

Study design

DOMESTICO is a nationwide, prospective, observational cohort study comparing home dialysis with ICHD. The maximum follow-up period of the study is 48 months. At present, 53 Dutch dialysis centres (covering 96% of Dutch centres) and 1 Belgian dialysis centre have agreed to recruit patients (Figure 1). The study is conducted according to the principles of the Declaration of Helsinki and in accordance with the Medical Research Involving Human Subjects Act (WMO).

Study population

All patients, aged 18 years and older, with ESKD that start with a form of dialysis in the participating centres, between December 2017 and December 2020, are eligible for this study. These patients are allowed to have a history of renal replacement therapy (RRT), however they have to (re)start dialysis during the study period for example due to kidney transplant failure (with or without previous dialysis). All these patients are defined as 'incident patients'. Prevalent dialysis patients, and patients with a life expectancy shorter than 3 months or an expected kidney transplantation within 3 months, are excluded. Patients have to provide written informed consent before participating in the study.



Figure 1. Participating centres

The red dots indicate the participating centres: 53 Dutch dialysis centres (covering 96% of Dutch centres) and 1 Belgian dialysis centre.

Inclusion

Patients are included in the period within four weeks before to four weeks after start of dialysis. If patients are missed for inclusion within this timeframe (for example, due to acute start of dialysis), they can be included at 3 months (\pm 2 weeks) after start of dialysis. Start of dialysis is defined as the first PD session performed at (a nursing) home (excluding PD-training) or, in case of ICHD, the first haemodialysis session performed in a centre (excluding continuous RRT).

The first patient was included in December 2017 and the study has currently started in 45 centres with 338 participating patients (Figure 2).



Figure 2. Participating patients

(Early) termination

For each participating patient, the study ends on 20 December 2021. Early study termination occurs if the patient withdraws from the study or stops dialysis treatment. Reasons to stop dialysis include kidney transplantation, recovery of kidney function, the wish to stop dialysis, or death.

Outcomes

Primary outcome parameter

The primary outcome parameter is the patient's HRQoL, a PRO, determined with the 12-item Short Form (SF-12) health survey.³⁹ The Dialysis Symptom Index (DSI) was added to the SF-12, to also assess symptom burden – also a PRO. Both questionnaires were carefully selected as recommended Patient Reported Outcome Measures (PROMs) in nephrological care by the Dutch Kidney Patients Association, the Dutch Federation for Nephrology, Nefrovisie (the Dutch Quality Institute for Nephrology), and Leiden University Medical Center.^{41,42}

The SF-12 is the shorter version of the Short Form-36 (SF-36), one of the most widely used surveys to assess HRQoL.^{43,44} The SF-36 consists of eight domains: Physical

functioning, Role-physical, Bodily pain, General health, Vitality, Social function, Role-emotional and Mental health. These domains are summarised in the Physical Component Summary (PCS) score and the Mental Component Summary (MCS) score. In the SF-12 these summary scores are calculated from the 12 most important questions (explaining ~90% variance) of the SF-36 questionnaire.^{39,45} As the average difference in summary scores between SF-36 and SF-12 is quite small, for time-efficiency reasons, the SF-12 can be used reliably in cohort studies.⁴⁶

The DSI consists of 30 questions evaluating the severity of symptoms relevant to dialysis and ESKD patients (Table 1). Patients report the level of burden of specific symptoms on a 5-point Likert scale, options range from 'not at all bothersome' to 'very bothersome'.⁴⁰

1. Constipation	16. Chest pain
2. Nausea	17. Headache
3. Vomiting	18. Muscle soreness
4. Diarrhoea	19. Difficulty concentrating
5. Decreased appetite	20. Dry skin
6. Muscle cramps	21. Itching
7. Swelling in legs	22. Worrying
8. Shortness of breath	23. Feeling nervous
9. Lightheadedness or dizziness	24. Trouble falling asleep
10. Restless legs or difficulty keeping legs still	25. Trouble staying asleep
11. Numbness or tingling in feet	26. Feeling irritable
12. Feeling tired or lack of energy	27. Feeling sad
13. Cough	28. Feeling anxious
14. Dry mouth	29. Decreased interest in sex
15. Bone or joint pain	30. Difficulty becoming sexually aroused

Table 1. Items Dialysis Symptom Index

Secondary outcome parameters

Secondary outcome parameters are hospitalisation, mortality, other clinical parameters, costs, and technique failure.

The cause of each hospitalisation episode will be categorised into the following categories (using ICD-10 codes)⁴⁷:

Cardiac (including myocardial ischaemia/infarction, cardiac arrest/arrhythmia, cardiac failure, fluid overload/pulmonary edema, haemorrhagic pericarditis);

- Vascular disease (including pulmonary embolus, stroke, cerebrovascular haemorrhage, ruptured vascular aneurysm, mesenteric infarction, peripheral arterial disease);
- Infection, non-dialysis related (including bacteraemia/sepsis, cardiac infection, HIV, osteomyelitis, respiratory infection, urinary tract infection);
- Dialysis related (including dialysis access infection, peritonitis, PD catheter leakage/exchange/removal, fistula operation, renal fluid overload, bleeding);
- Malignancy;
- Bleeding, non-dialysis related (including intracranial bleeding, gastro-intestinal bleeding, other causes of bleeding);
- Other causes.

Mortality will be categorised into the following categories (using ERA-EDTA codes)⁴⁸:

- Sudden death 'with unknown cause';
- Cardiac (including myocardial ischaemia/infarction, cardiac arrest/arrhythmia, cardiac failure, fluid overload/pulmonary edema, haemorrhagic pericarditis);
- Vascular (including pulmonary embolus, stroke, cerebrovascular haemorrhage, ruptured vascular aneurysm, mesenteric infarction, peripheral arterial disease);
- Infectious, dialysis related (including dialysis access infection, peritonitis);
- Infectious, non-dialysis related (including bacteraemia/sepsis, cardiac infection, HIV, osteomyelitis, respiratory infection, urinary tract infection);
- Malignancy;
- Bleeding (including dialysis related bleeding, intracranial bleeding, gastrointestinal bleeding, other causes of bleeding);
- Overall deterioration in clinical condition/stopping dialysis;
- Other causes.

Besides hospitalisation and mortality, several clinical parameters will be recorded including blood pressure and use of antihypertensive drugs, haemoglobin and use of erythropoiesis-stimulating agents, phosphate levels and use of phosphate binders, vascular access parameters, and nutritional status. Direct healthcare costs, patient costs, and costs with regard to productivity losses will be assessed with a subset of questions from the Institute for Medical Technology Assessment (iMTA) Productivity Cost Questionnaire (iPCQ) and the iMTA Medical Cost Questionnaire (iMCQ).^{49,50} To capture all health care costs for the population under research a small number of disease specific services are added to the standard iMCQ, e.g. home dialysis. Given the fact that many patients need substantial help from close relatives, also use of informal care by patients will be assessed. The costs related to the healthcare consumption, the dialysis procedures, the diagnostic tests and (over-the-counter) medication will be derived from the patient's medical chart during the study. Unit costs will be derived from the Dutch manual for costing studies.⁵¹

To further examine cost-effectiveness, the EuroQol-5D-5L (EQ-5D-5L) questionnaire will be used. The EQ-5D-5L measures HRQoL on the following 5 domains: mobility, self-care, usual activities, pain/discomfort, and anxiety/depression. Each domain has 5 levels of functioning, ranging from 'no problems' to 'extreme problems'. The EQ-5D-5L also contains a visual analogue scale on which the current health state can be indicated. The EQ-5D scores can be used to calculate utilities, which describe HRQoL on a scale from 0 (dead) to 1 (perfect health). Utilities can be combined with survival to calculate quality adjusted life years (QALYs). As outcome measure for cost-effectiveness, the costs per additional QALY will be analysed.^{52,53}

All participating patients will also receive a self-management screening questionnaire (SeMaS) at baseline, in order to investigate whether self-management can predict a successful home dialysis treatment. This questionnaire shows the abilities and possible barriers for self-management by asking questions about the burden of disease, locus of control, self-efficacy, social support, coping style, anxiety, depression, and skills.^{54,55}

Table 2 provides an overview of the moments when participating patients will fill in the aforementioned questionnaires.

Finally, technique failure rate of home dialysis, defined by a composite outcome of death or transfer to ICHD, will be assessed. Both a 30-days and a 180-days definition of technique failure will be used according to the minimum number of days the patient received ICHD after cessation of home dialysis.⁵⁶ Permanent technique failure is defined by death or transfer to ICHD (using the 180-days definition), or cessation of dialysis. Death-censored technique failure will be reported separately. Transfer

to kidney transplantation is not considered to be technique failure and will also be reported separately.⁵⁶

Visit	SF-12 and DSI	iPCQ and iMCQ	EQ-5D-5L	SeMaS
Baseline	Х	Х	Х	Х
At 3 and 6 months	Х	Х	Х	
At 9 months and every 6 months thereafter		Х		
At 12 months and every 6 months thereafter	Х	Х	Х	

 Table 2. Overview questionnaires

SF-12: Short Form-12; DSI: Dialysis Symptom Index; iPCQ: Institute for Medical Technology Assessment (iMTA) Productivity Cost Questionnaire; iMCQ: iMTA Medical Cost Questionnaire; SeMaS: self-management screening questionnaire

Data collection

All study outcomes, except the SeMaS, will be assessed at baseline, after 3 months, 6 months, and thereafter every 6 months until end of follow-up or end of the study (Table 2).

Data will be registered in case report forms (CRF). IBM Data Collection will be used as CRF. The database is developed by Nefrovisie and follows the principles of Good Clinical Practice (i.e. it has an audit trail, possibility for electronic signing, direct validation of inserted data, authorisation per form and user). Nefrovisie will also host the database for the duration of the study. The database will be archived for future research during 15 years after termination of the study.

Statistical analysis

All statistical analyses will be performed using statistical software such as SPSS and Stata.

Univariable and multivariable regression analysis will be conducted. In case of repeated measures, multilevel analysis or generalised estimating equations will be applied. Possible confounders determined a priori are age, gender, marital status, level of education, work status, cause of renal failure, prior RRT with dialysis vintage, comorbidities, albumin, body mass index, and protein energy wasting. Cumulative incidence of hospitalisation, mortality, and technique failure will be reported in

Kaplan Meier curves. In case of missing data, multiple imputation techniques will be used to impute the missing values where appropriate.

Overall costs will be compared across the treatment groups and 95% confidence intervals will be estimated using bootstrapping techniques. The cost-effectiveness of different dialysis modalities will be determined using a state transition model. This model captures the changes in treatment modality, including transplantation, over time. The results of the DOMESTICO study will be used as input parameters for this model.

Sample size calculation

For the primary outcome HRQoL, obtained with the SF-12, a sample size of 350 patients is required. To obtain a clinically relevant difference between groups of 3 points in the SF-12 summary scores, after a median of 12 months follow-up, 175 patients per group are needed (assumed standard deviation = 10 points, $\alpha = 0.05$, $\beta = 0.20$).^{46,57-59}

However, for the EQ-5D-5L, an important component for the secondary outcome cost-effectiveness, a sample size of 1400 patients (700 patients per group) is needed. A difference of 0.03 - 0.07 points between groups after a mean follow-up of 12 months is considered clinically relevant.^{44,60,61} The standard deviation in dialysis groups ranges from 0.1 to 0.22.^{62,63} Assuming a common standard deviation of 0.20 and the lowest, still clinically relevant score, a total of 1400 patients (700 patients per group) will be sufficient to detect a difference of 0.03 points in the EQ-5D-5L score between groups ($\alpha = 0.05$, $\beta = 0.20$).

When approximately 10% loss to follow up is taken into account, a group of 800 home dialysis patients and a comparison group of 800 ICHD patients has to be included in order to have sufficient power to analyse both outcomes. Since the ratio between PD patients and home haemodialysis (HHD) patients in the Netherlands is expected to be 3:1 in future years, the home dialysis group will consist of 600 PD and 200 HHD patients.

Discussion

Dialysis has a great impact on the HRQoL of ESKD patients and dialysis is a very expensive treatment. More than 80% of Dutch dialysis patients are treated with ICHD although home dialysis could result in a better HRQoL and could be more cost effective. Therefore, we initiated the DOMESTICO study, which will investigate the effects of home dialysis on HRQoL in relation to clinical outcomes and costs, in comparison with ICHD. This nationwide cohort study will include 1600 incident dialysis patients over a period of 3 years. At time of submission of this manuscript, 338 patients have been included.

Although a randomised controlled trial (RCT) would yield the ultimate answer to our research question, this is not in accordance with the concept of shared decision making. A patient's choice between home dialysis and ICHD is considered too fundamental, to let it be determined by chance. Indeed, an RCT in the Netherlands comparing PD with ICHD conducted in the past, stopped early due to poor patient recruitment; only 38 patients consented to be randomly assigned to either PD or ICHD.⁶⁴ Hence, DOMESTICO is designed as a prospective, observational cohort study collecting extensive parameters to correct for confounding by indication.

The results of this study will be of great importance for future ESKD patients when choosing a treatment, as HRQoL is increasingly acknowledged by clinicians and patients as an important aspect in the decision-making process. In addition, the results with respect to clinical outcomes will ameliorate the shared decision-making process. Finally, the data could give more guidance to healthcare professionals, in particular to assess which type of patients may benefit most from home dialysis.

List of abbreviations

CCMO: Central Committee on Research Involving Human Subjects CRF: case report forms DOMESTICO: Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes DSI: Dialysis Symptom Index EQ-5D-5L: EuroQol-5D-5L ESKD: End Stage Kidney Disease HHD: home haemodialysis HRQoL: Health-Related Quality of Life ICHD: in-centre haemodialysis iMTA: Institute for Medical Technology Assessment iMCQ: iMTA Medical Cost Questionnaire iPCQ: iMTA Productivity Cost Questionnaire MCS: Mental Component Summary PCS: Physical Component Summary PD: peritoneal dialysis PRO: Patient Reported Outcome PROM: Patient Reported Outcome Measure QALY: quality adjusted life year RCT: randomised controlled trial RRT: renal replacement therapy SeMaS: self-management screening questionnaire SF: Short Form

References

- 1. Nefrovisie. http://www.nefrovisie.nl/nefrodata/. Accessed 20th June 2019.
- Hemke AC, Dekker FW, Bos WJW, et al. Oorzaken voor verminderd aandeel peritoneale dialyse als nierfunctievervangende behandeling in Nederland. Ned Tijdschr Geneeskd. 2012;156:1-8.
- Volksgezondheidenzorg.info. https://www.volksgezondheidenzorg.info/. Accessed 2 May 2019.
- 4. Gorodetskaya I, Zenios S, McCulloch CE, et al. Health-related quality of life and estimates of utility in chronic kidney disease. Kidney Int. 2005;68:2801-2808.
- 5. van de Luijtgaarden MWM, Jager KJ, Segelmark M, et al. Trends in dialysis modality choice and related patient survival in the ERA-EDTA Registry over a 20-year period. Nephrol Dial Transplant. 2016;31:120-128.
- 6. Kramer A, Pippias M, Noordzij M, et al. The European Renal Association European Dialysis and Transplant Association (ERA-EDTA) Registry Annual Report 2015: a summary. Clin Kidney J. 2018;11(1):108-122.
- 7. Nederlandse Zorgautoriteit. http://opendisdata.nl/. Accessed 2 May 2019.
- 8. Ageborg M, Allenius BL, Cederfjall C. Quality of life, self-care ability, and sense of coherence in hemodialysis patients: a comparative study. Hemodial Int. 2005;9:S8-14.
- 9. Al Wakeel J, Al Harbi A, Bayoumi M, et al. Quality of life in hemodialysis and peritoneal dialysis patients in Saudi Arabia. Ann Saudi Med. 2012;32(6):570-574.
- 10. Atapour A, Nasr S, Boroujeni AM, et al. A comparison of the quality of life of the patients undergoing hemodialysis versus peritoneal dialysis and its correlation to the quality of dialysis. Saudi J Kidney Dis Transpl. 2016;27(2):270-280.
- 11. Barata NE. Dyadic Relationship and Quality of Life Patients with Chronic Kidney Disease. J Bras Nefrol. 2015;37(3):315-322.
- 12. Basok EK, Atsu N, Rifaioglu MM, et al. Assessment of female sexual function and quality of life in predialysis, peritoneal dialysis, hemodialysis, and renal transplant patients. Int Urol Nephrol. 2009;41(3):473-481.
- 13. Baykan H, Yargic I. Depression, anxiety disorders, quality of life and stress coping strategies in hemodialysis and continuous ambulatory peritoneal dialysis patients. Klinik Psikofarmakoloji Bulteni. 2012;22(2):167-176.
- 14. Borowiak E, Braksator E, Nowicki M, et al. Quality of life of chronic hemodialysis and peritoneal dialysis patients. Clinical and Experimental Medical Letters. 2009;50(1):37-42.
- 15. Brown EA, Johansson L, Farrington K, et al. Broadening Options for Long-term Dialysis in the Elderly (BOLDE): differences in quality of life on peritoneal dialysis compared to haemodialysis for older patients. Nephrol Dial Transplant. 2010;25(11):3755-3763.
- 16. Chen JY, Wan EYF, Choi EPH, et al. The Health-Related Quality of Life of Chinese Patients on Hemodialysis and Peritoneal Dialysis. Patient. 2017:1-10.
- 17. Czyzewski L, Sanko-Resmer J, Wyzgal J, et al. Assessment of health-related quality of life of patients after kidney transplantation in comparison with hemodialysis and peritoneal dialysis. Ann Transplant. 2014;19:576-585.
- Fructuoso M, Castro R, Oliveira L, et al. Quality of life in chronic kidney disease. Nefrologia. 2011;31(1):91-96.

- Goncalves FA, Dalosso IF, Borba JM, et al. Quality of life in chronic renal patients on hemodialysis or peritoneal dialysis: a comparative study in a referral service of Curitiba - PR. Jornal brasileiro de nefrologia. 2015;37(4):467-474.
- Ibrahim N, Chiew-Tong NK, Desa A. Symptoms and health-related quality of life in patients with heamodialysis and continuous ambulatory peritoneal dialysis. Research Journal of Medical Sciences. 2011;5(5):252-256.
- 21. Ikonomou M, Skapinakis P, Balafa O, et al. The impact of socioeconomic factors on quality of life of patients with chronic kidney disease in Greece. J Ren Care. 2015;41(4):239-246.
- 22. Kim JY, Kim B, Park KS, et al. Health-related quality of life with KDQOL-36 and its association with self-efficacy and treatment satisfaction in Korean dialysis patients. Qual Life Res. 2013;22(4):753-758.
- Kontodimopoulos N, Pappa E, Niakas D. Gender- and age-related benefit of renal replacement therapy on health-related quality of life. Scand J Caring Sci. 2009;23(4):721-729.
- 24. Liu WJ, Musa R, Chew TF, et al. Quality of life in dialysis: A Malaysian perspective. Hemodial Int. 2014;18(2):495-506.
- Maglakelidze N, Pantsulaia T, Tchokhonelidze I, et al. Assessment of health-related quality of life in renal transplant recipients and dialysis patients. Transplant Proc. 2011;43(1):376-379.
- Nakayama M, Ishida M, Ogihara M, et al. Social functioning and socioeconomic changes after introduction of regular dialysis treatment and impact of dialysis modality: a multicentre survey of Japanese patients. Nephrology (Carlton). 2015;20(8):523-530.
- 27. Okpechi IG, Nthite T, Swanepoel CR. Health-related quality of life in patients on hemodialysis and peritoneal dialysis. Saudi J Kidney Dis Transpl. 2013;24(3):519-526.
- Ören B, Enc N. Quality of life in chronic haemodialysis and peritoneal dialysis patients in Turkey and related factors. International journal of nursing practice. 2013;19(6):547-556.
- 29. Ramos EC, Santos I, Zanini R, et al. Quality of life of chronic renal patients in peritoneal dialysis and hemodialysis. Jornal brasileiro de nefrologia. 2015;37(3):297-305.
- 30. Tannor EK, Archer E, Kapembwa K, et al. Quality of life in patients on chronic dialysis in South Africa: a comparative mixed methods study. BMC Nephrol. 2017;18(1):4.
- 31. Theofilou P. Quality of life in patients undergoing hemodialysis or peritoneal dialysis treatment. J Clin Med Res. 2011;3(3):132-138.
- Turkmen K, Yazici R, Solak Y, et al. Health-related quality of life, sleep quality, and depression in peritoneal dialysis and hemodialysis patients. Hemodial Int. 2012;16(2):198-206.
- Watanabe Y, Ohno Y, Inoue T, et al. Home hemodialysis and conventional in-center hemodialysis in Japan: a comparison of health-related quality of life. Hemodial Int. 2014;18 Suppl 1:S32-S38.
- Wright LS, Wilson L. Quality of Life and Self-Efficacy in Three Dialysis Modalities: Incenter Hemodialysis, Home Hemodialysis, and Home Peritoneal Dialysis. Nephrol Nurs J. 2015;42(5):463-476.
- Wu F, Cui L, Gao X, et al. Quality of life in peritoneal and hemodialysis patients in China. Ren Fail. 2013;35(4):456-459.
- 36. Yang F, Griva K, Lau T, et al. Health-related quality of life of Asian patients with endstage renal disease (ESRD) in Singapore. Qual Life Res. 2015;24(9):2163-2171.
- 37. Ying SC, Krishnan M. Interpretation of quality of life outcomes amongst end stage renal disease patients in selected hospitals of Malaysia. International Journal of Pharmaceutical Sciences and Research. 2014;5(1):60-69.
- 38. Yongsiri S, Thammakumpee J, Prongnamchai S, et al. The association between bioimpedance analysis and quality of life in pre-dialysis stage 5 chronic kidney disease, hemodialysis and peritoneal dialysis patients. J Med Assoc Thai. 2014;97(3):293-299.
- 39. Ware JE, Kosinski MM, Keller SD. A 12-Item Short-Form Health Survey: Construction of Scales and Preliminary Tests of Reliability and Validity. Medical Care. 1996;34:220-233.
- 40. Weisbord SD, Fried LF, Arnold RM, et al. Development of a Symptom Assessment Instrument for Chronic Hemodialysis Patients: The Dialysis Symptom Index. Journal of Pain and Symptom Management. 2004;27(3):226-240.
- 41. Van der Willik EM, Leegte M, van Ittersum F, et al. First Dutch registry of patientreported outcome measures (PROMS) has a low response in dialysis patients [abstract]. Nephrol Dial Transplant. 2018;33 Suppl 1:i262.
- 42. Van der Willik EM, Meuleman Y, Prantl K, et al. Patient-reported outcome measures: selection of a valid questionnaire for routine symptom assessment in patients with advanced chronic kidney disease a four-phase mixed methods study. BMC Nephrol. 2019;Accepted.
- 43. Ware JE. SF-36 Health Survey. Manual and Interpretation Guide. 1997.
- 44. Wyld M, Morton RL, Hayen A, et al. A Systematic Review and Meta-Analysis of Utility-Based Quality of Life in Chronic Kidney Disease Treatments. PLOS. 2012;9(9):1-10.
- 45. Gandek B, Ware JE, Aaronson NK, et al. Cross-validation of item selection and scoring for the SF-12 Health Survey in nine countries: Results from the IQOLA Project. J Clin Epidemiol. 1998;51:1171-1178.
- 46. Loosman WL, Hoekstra T, van Dijk S, et al. Short-Form 12 or Short-Form 36 to measure quality-of-life changes in dialysis patients? Nephrol Dial Transplant. 2015;30:1170-1176.
- 47. World Health Organization. ICD-10. Internationale statistische classificatie van ziekten en met gezondheid verband houdende problemen, tiende revisie. 2014.
- 48. ERA-EDTA codes mortality. https://www.nefrovisie.nl/wp-content/uploads/2016/08/ renine_codes_doodsoorzaak.pdf. Accessed 2 May 2019.
- 49. Bouwmans C, Krol M, Severens H, et al. The iMTA Productivity Cost Questionnaire A Standardized Instrument for Measuring and Valuing Health-Related Productivity Losses. Value in Health. 2015:753-758.
- 50. Institute for Medical Technology Assessment, Productivity and Health Research Group. Handleiding iMTA Medical Cost Questionnaire (iMCQ). . iMTA, Erasmus University Rotterdam, 2018. Questonnaire can be retrieved from: https://www.imta.nl/ questionnaires/. Accessed 7 June 2016.
- 51. Hakkaart-van Roijen L, Van der Linden N, Bouwmans CAM, et al. Kostenhandleiding: Methodologie van kostenonderzoek en referentieprijzen voor economische evaluaties in de gezondheidszorg. Zorginstituut Nederland, Diemen, 2015.
- 52. Versteegh MM, Vermeulen KM, Evers SMAA, et al. Dutch Tariff for the Five-Level Version of EQ-5D. Value in Health. 2016;19:343-352.
- 53. Brooks R, Group E. EuroQol: the current state of play. Health Policy. 1996;37:53-72.
- 54. Eikelenboom N, van Lieshout J, Wensing M, et al. Implementation of personalized selfmanagement support using the self-management screening questionnaire SeMaS; a study protocol for a cluster randomized trial. Trials. 2013;14(336):1-9.

- 55. Eikelenboom N, Smeele I, Faber M, et al. Validation of Self-Management Screening (SeMaS), a tool to facilitate personalised counselling and support of patients with chronic diseases. BMC Family Practice. 2015;16(165):1-12.
- 56. Lan PG, Clayton PA, Johnson DW, et al. Duration of hemodialysis following peritoneal dialysis cessation in Australia and New Zealand: proposal for a standardized definition of technique failure. Peritoneal Dialysis International. 2016;36:623-630.
- 57. Hall YN, Larive B, Painter P, et al. Effects of Six versus Three Times per Week Hemodialysis on Physical Performance, Health, and Functioning: Frequent Hemodialysis Network (FHN) Randomized Trials. Clin J Am Soc Nephrol. 2012;7:782-794.
- Lowrie EG, Curtin RB, LePain N, et al. Medical outcomes study short form-36: a consistent and powerful predictor of morbidity and mortality in dialysis patients. Am J Kidney Dis. 2003;41(6):1286-1292.
- 59. Korevaar JC, Jansen MAM, Merkus MP, et al. Quality of life in predialysis end-stage renal disease patients at the initiation of dialysis therapy. Peritoneal Dialysis International. 2000;20:69-75.
- 60. Dolan P. Modeling valuations for EuroQol health states. Medical Care. 1997;35(11):1095-1108.
- 61. Walters SJ, Brazier JE. Comparison of the minimally important difference for two health state utility measures: EQ-5D and SF-6D. Quality of Life Research. 2005;14:1523-1532.
- Culleton BF, Walsh M, Klarenbach SW, et al. Effect of Frequent Nocturnal Hemodialysis vs Conventional Hemodialysis on Left Ventricular Mass and Quality of Life. JAMA. 2007;297(11):1291-1299.
- 63. Jardine MJ, Gray NA, De Zoysa J, et al. Design and participant baseline characteristics of 'A Clinical Trial of IntensiVE Dialysis': The ACTIVE Dialysis Study. Nephrology. 2015;20:257-265.
- 64. Korevaar JC, Feith GW, Dekker FW, et al. Effect of starting with hemodialysis compared with peritoneal dialysis in patients new on dialysis treatment: A randomized controlled trial. Kidney Int. 2003;64:2222-2228.



Chapter 11

Summary and general discussion

Worldwide, home dialysis is utilized significantly less often than in-centre haemodialysis (CHD) with few exceptions.^{1, 2} This is remarkable, as home dialysis - both peritoneal dialysis (PD) and home haemodialysis (HHD) - is associated with similar survival as CHD and with more independence and greater scheduling flexibility than CHD.³ Previously, home dialysis was mostly performed by younger patients. However, more elderly patients are currently receiving home dialysis, because the Netherlands and other developed countries are facing an ageing population (**Chapter 2**).⁴⁻⁶ This development urges the need to organize care for the current elderly dialysis population and to prioritize Health-Related Quality of Life (HRQoL) – rather than survival - as an important outcome. This thesis aimed to re-evaluate home dialysis eligibility, technique survival and HRQoL in the context of this demographic shift. In this final chapter, the main findings of this thesis will be placed in a broader perspective and recommendations for future research will be made.

Part I – Eligibility for home dialysis in the current dialysis population

More patients in the current patient population might be eligible for home dialysis, than receive this dialysis modality at present (20% of the dialysis population is treated with either PD or HHD). In previous studies, it was estimated that >80% of patients are eligible for home dialysis.⁷⁻⁹ When patients are given comprehensive education, over 60% of patients chose a home dialysis modality including those who are older.⁹, ¹⁰ Using Dutch registry data, we found that the current home dialysis population is ageing due to both an increase in kidney transplantation in younger patients and an increase in the absolute number of elderly patients (**Chapter 2**). The relative cumulative incidence of home dialysis, however, remained more or less the same for patients aged 65-74 (ranging from 29 to 25% over 20 years' time) and above 75 (ranging from 17 to 19%).

The high absolute numbers of elderly patients facing dialysis require adaptions in the organization of pre-dialysis education and dialysis care, while the constant relative numbers require further consideration. Since over 50% of elderly patients opt for PD⁹, more elderly people may initiate home dialysis if they are offered home dialysis as treatment option. Compared to the Netherlands, Canada has a comparable proportion of elderly patients on PD (19%).¹¹ Yet in the greater Toronto area, the uptake of PD was statistically different in regions with and without home care support (47% to 37% resp.), indicating that sufficient home care support is a key factor in implementing home dialysis to elderly patients.⁹ In other countries, such as in Australia and New

Zealand, the proportion of elderly people on home dialysis (both PD and HHD) is much higher than in the Netherlands (resp. 24 and 47%).¹² In these countries, home dialysis might be favoured due to long travel distances. As a result mobile buses providing HD services were quite popular.¹³ Nevertheless, these data suggest that a higher proportion of older patients might be eligible to receive home dialysis.

An important reason to favor home dialysis in the elderly is the high flexibility in dialysis schedules of home dialysis, which can be a major advantage for elderly and frail patients.¹³ Other important advantages of home dialysis that may be important to the elderly patient are related to haemodynamics. PD is associated with better cardiovascular stability during dialysis - due to the slow removal of molecules, while intensive HD is associated with better blood pressure control.¹⁴⁻¹⁶ Assisted PD and HHD can be safely offered to elderly patients. Previous studies indicated that among elderly patients on assisted PD compared to self-care PD mortality and technique failure rates are similar.^{17, 18} A French study found that assisted PD patients had worse survival, but this finding was possibly due to differences in frailty between the groups.¹⁹ Home care assistance in elderly patients who cannot perform self-care due to functional limitations, is the lack of home support.^{9, 13} Other challenges in the elderly population are the development of PD leakage due to weak abdominal muscles in PD and the occurrence of falls due to intradialytic hypotension in HHD.¹³

Both age and comorbidity are frequent reported barriers to home dialysis.²⁰ Due to improved treatments in cardiac disease and diabetes, patients with severe comorbidities are currently surviving long enough to reach End Stage Kidney Disease (ESKD) and the start of dialysis. Patients initiating dialysis appear to have more comorbidities than patients with several solid tumours.²¹ The results in this regard are striking: dialysis patients have a high prevalence of hypertension (81%), diabetes mellitus (54%), cardiovascular disease (41%) and heart failure (37%), compared to respectively a maximum of 54%, 29%, 21% and 8% in patients with solid tumours.²¹ Many nephrologists may perceive severe comorbidity as a barrier to initiate home dialysis, despite options to provide support. Yet, using data from the retrospective DOMESTICO study we found that patients with high comorbidity scores were as likely to start home dialysis as CHD, if corrected for confounding factors including BMI and age (**Chapter 3**). Obese patients with high comorbidity scores were however significantly less likely to receive home dialysis. The presented results suggest that BMI

and age play a more important role in dialysis modality initiation than comorbidity, even though both factors are no absolute contraindications for initiating home dialysis.¹⁴ Of note, we found no difference in average BMI and age among centres, while the proportion of home dialysis varies considerably in the Netherlands. These insights might imply that dialysis modality initiation depends not only on patient characteristics, but also on the weighting of these characteristics by the dialysis centre, i.e. by practice policies. In addition, other centre-specific characteristics unrelated to patient characteristics, such as staff shortage, insufficient training in home dialysis for young nephrologists or reimbursement, might play a role in the decision not to start home dialysis. Indeed, in previous studies it was suggested that factors such as regional health policies, small centre-size and offering limited dialysis treatment options determine the variance in home dialysis use.^{22,23} Although centres should ideally offer both home and in-centre dialysis, both night and daytime dialysis, extended hours dialysis and home dialysis assistance, providing all these options is often not feasible within one centre. More regional collaboration among dialysis centres would facilitate providing all different dialysis treatment options to patients.

It is difficult to determine upfront which patients will benefit from receiving home dialysis. A home visit prior to initiating dialysis modality and pre-dialysis education, helps to identify patients with suitable housing (Chapter 4). In this chapter, we describe the results of an implementation project on patient selection. We found that suitable housing had a significant association with long-term home dialysis treatment. In our opinion, this determinant is best assessed during a home visit. Moreover, the health personnel involved in this project acknowledged that the home visit was the most important addition to the pre-dialysis programme. The social worker, who performed the home visit, had a valuable role in patient selection. It should be recognized that the specificity of the social worker's assessment was significantly higher than the nephrologist's assessment, indicating that the social worker was particularly capable of selecting patients *unsuitable* for home dialysis. In addition to suitable housing, we found that a strong social support system, an active lifestyle and being able to balance burden and capacity were other key determinants of whether patients received home dialysis. In previous research, lack of social support has indeed been identified as an important barrier to home dialysis.²⁴ We believe that involving family members in decision making, may help to improve family support for home dialysis. A home visit might also be the perfect opportunity for such treatment education. In a study on treatment education at home, family members present during education sessions demonstrated improved understanding of dialysis and experienced fewer concerns and distress.²⁵ We would recommend considering a home visit as part of a pre-dialysis programme, for the various reasons presented above. Another important feature of the presented implementation project, was the success of the 'home dialysis first' policy. It has been reported that such a policy, results in more patients receiving home dialysis.^{26,27}

Although in this thesis home dialysis consistently refers to both PD and HHD, as is concomitant with literature, it is important to consider that both modalities are completely different therapies. The main similarity between these therapies is the advantages of scheduling flexibility and autonomy that are associated with a treatment at home. The dialysis tasks of both therapies are very different and require different degrees of patient empowerment. Therefore, PD patients and HHD patients are not likely equal to each other. Throughout this thesis, sensitivity analyses were performed to acknowledge the important difference in patient populations. In **chapter 2**, for example, we found that the home haemodialysis use increased over time as compared to a decline of the overall home dialysis incidence. However in most chapters, including in the systematic review of **chapter 7**, we had to conclude that the sample size of HHD patients was too small to draw conclusions on this patient group. Thus, most data provided in this thesis concerns PD patients.

Part II - Enhancing technique survival of peritoneal dialysis

Among patients on dialysis, five-year survival rates are below 50%.²⁸ For comparison, the survival rate of patients with solid tumours including breast and colorectal cancer is higher than that of patients treated with dialysis.²¹ As previously stated, being treated with dialysis is associated with a high prevalence of comorbidity.²¹ As a result, dialysis patients are once to twice annually admitted to the hospital.⁴ Using data of the retrospective DOMESTICO study, we demonstrated that the risk for hospital admission is higher in PD patients compared to haemodialysis patients (**Chapter 5**). This may indicate a difference between treatments, but could also be explained by circumstantial factors. For example, patients treated at home may be more likely to be admitted to the hospital to receive treatment for infections and fluid overload, than patients seen thrice weekly on the dialysis ward where treatment can be intensively monitored on an outpatient basis. Another important finding of this study was that PD patients were most frequently hospitalized for peritonitis, which is in keeping

with findings of previous studies.²⁹⁻³² More attention to prevention of PD-associated infections is necessary for reducing the number of hospitalizations in the future.

PD-related infections consisting of PD peritonitis and exit-site infections were a major cause of PD technique failure (**Chapter 6**). In this chapter, we present the causes and risk factors of PD technique failure, the primary outcome, in the retrospective DOMESTICO study. In accordance with previous research, we showed that the highest risk of PD technique failure is noted in the first year of PD treatment.³³ Few studies however studied the reasons of technique failure, especially over an extended period.³⁴⁻ ³⁶ We found that infections were the most important cause of early technique failure. Other important causes for specifically early technique failure in DOMESTICO were PD fluid leakage and catheter problems. Infections and PD catheter problems can be considered modifiable causes of technique failure, and major barriers to long-term PD treatment if insufficiently addressed. We also found that ultrafiltration failure was the cause for technique failure (early and late combined) in less than 10%. This is in line with previous reports^{34, 37, 38}, further indicating the possible reversibility of technique failure. Successful long-term treatment with PD thus depends on the prevention of technique failure, i.e. the prevention of PD infections and catheter problems.

There is at present no consensus on antibiotic regimens in PD peritonitis.³⁹ In fact, many recommendations from PD guidelines are based on level C evidence.³⁹⁻⁴¹ Current study groups, including PDOPPS, are evaluating evidence for antibiotic regimens and PD catheter access.⁴² Yet, also experiences from single centres are contributing to this evidence, for example from our study group.^{43, 44} Leakage and catheter problems are thus other modifiable causes of technique failure. For example, incorrect implementation technique or too early PD initiation are known causes for early leakage.⁴⁰ The recommended period between PD catheter implementation and PD initiation is two weeks.⁴⁰ On the other hand, in urgent PD, the catheter is used within 3 days with small dialysate volumes.⁴⁵ These examples indicate that the success of the therapy depends on knowledge and experience of the physician.

Another important finding in the DOMESTICO study that should be emphasized is that death was the leading cause of both early and late technique failure. The mortality rate of PD patients is identical to the rate in CHD patients, as already stated in the first paragraph.³ Thus high mortality is an expected result of the disease and *not* the result of the dialysis modality. This is important to consider when comparing research on CHD and PD patients, since technique failure is an outcome solely reported in research on home dialysis patients. CHD patients do not usually transit from CHD to PD, thus technique failure on CHD results in death. Hence, mortality is the reported outcome in research on CHD patients, while in studies on home dialysis patients both mortality and technique failure (including death) are reported outcomes. Accordingly, to optimize transparency of home dialysis studies it is important to report technique failure according to the standardized definition, death-censored technique failure *and* mortality.

A large number of studies have investigated risk factors for technique failure. Both demographic and patient characteristics, such as diabetes mellitus, cardiovascular disease and residual GFR, and centre characteristics, such as small centre size, are associated with early technique failure.^{34, 37, 38} In DOMESTICO however, we demonstrated that centre size was not associated with technique failure rate as shown in a funnel plot (Chapter 6). Funnel plots are a graphical method to compare performance indicators among health care providers. The important feature in funnel plots is the use of control limits, that are based upon the sample sizes of individual hospitals. Funnel plots are superior to league tables ranking health-care providers, because they are easy to interpret and identify outliers instead of implying the existence of ranking.^{46, 47} In addition, funnel plots can be adjusted for important confounders. Our funnel plot was adjusted for both age and sex, but not for other case-mix variables such as comorbidity. Another important limitation of our study is that some centres only provided few patients, resulting in broad control limits. Despite these limitations, we believe that the funnel plot indicates that centre size is not causally related to technique failure and that size is a mere proxy for a successful home dialysis program. We suggest that dedication and organization play the most important role in technique survival.

Part III - Shift towards Health-Related Quality of Life

Progress in technology notwithstanding, traditionally important outcomes are disappointing in dialysis patients: survival is low, hospital admissions are frequent and the technique failure rate is high. HRQoL is becoming an increasingly important outcome in ESKD patients, who consider HRQoL a core outcome alongside more traditional outcomes.⁴⁸⁻⁵¹ This is not surprising, as the burden of living with ESKD negatively affects HRQoL. The low HRQoL of dialysis patients as compared with HRQoL of the general population and patients with other chronic diseases such as

malignancies and diabetes mellitus^{52, 53}, implies the need to identify interventions to improve HRQoL. Lower levels of HRQoL or mental health-related factors are also associated with traditional outcomes, both mortality and frequent hospitalization.⁵⁴⁻⁵⁶ The postulated mechanism behind this association might be that lower HRQoL leads to non-adherence to therapy. Non-adherence likely results in hospital admissions and eventually could lead to death.^{57, 58} Elaborating on this, HRQoL may also be linked to technique failure by the same mechanism. Sufficient coping is crucial for homebased therapies, because home dialysis patients need a certain ability to perform self-care. Insufficient coping, for example through feelings of anxiety, may contribute to technique failure.⁵⁹ These factors are more modifiable than risk factors such as comorbidity and may therefore have significant impact on improvement of technique survival. Although both the emerging interest in HRQoL and its association with traditional outcomes emphasize the importance of HRQoL as an important outcome measure, few studies on home dialysis report HRQoL as outcome. Less than 4% of randomized trials on PD patients described results from HRQoL questionnaires.⁶⁰

In our systematic review and meta-analysis of forty-six articles reporting on HRQoL in 41 study populations, we found marginally higher HRQoL scores for home dialysis patients across the world (**Chapter 7**). Since the overall quality of included studies was poor and few longitudinal studies exist, the results should be interpreted with caution. The few recent longitudinal studies that have been performed report conflicting results.⁶¹⁻⁶⁴ For example one study showed a decline in HRQoL in both incident CHD and PD patients over time⁶², while another study found that HRQoL improved after a transfer from CHD to home haemodialysis.⁶³ No randomized controlled trials (RCTs) were included in this meta-analysis, since no recent RCTs on HRQoL of home and CHD patients have been performed. To date, there have been few RCTs on home dialysis and CHD, because recruitment in these studies is challenging due to patients preferring one of the modalities.^{65, 66} This may be illustrated by the experiences from the NECOSAD study group: while 773 incident dialysis patients were eligible, only 38 patients could be randomized in this trial.⁶⁶

Another important limitation of the presented meta-analysis is that most studies, even half of the included longitudinal studies, included prevalent patients. Using prevalent patients makes comparability among study populations challenging, since HRQoL tends to decline over time and dialysis vintage thus is an important confounder.⁶⁷ An important bias introduced in studies with prevalent patients is survival bias, since patients with higher HRQoL scores may survive and then be able to transit to home dialysis. In other words, these difficulties in observational studies highlight the importance of longitudinal studies on HRQoL of incident home dialysis.

The following example from a nocturnal HD trial further stresses the importance of longitudinal studies: in a study randomising patients between nocturnal home haemodialysis and CHD, no difference was found between the change in HRQoL scores of both groups after randomization.⁶⁸ However, when pre-randomisation scores were used as baseline, i.e. when the patient was still unaware of treatment choice, the general HRQoL of nocturnal home haemodialysis patients did improve over time.⁶⁸ This example shows that treatment choice alone, even prior to dialysis initiation, can effect HRQoL. Another important finding in this trial was that the kidney disease specific domains 'effects of kidney disease' and 'burden of kidney disease' were significantly higher in nocturnal HHD patients, in which a higher score indicates a more favourable HRQoL. This example also highlights the additional value of disease-specific HRQoL domains, i.e. 'burden of kidney disease' or a questionnaire addressing symptom burden, compared to general HRQoL measures.

In the NOCTx study, one-year follow-up HRQoL data were used for the reason presented in the previous paragraph. Baseline in this study was 2 to 3 months after transfer to nocturnal HD or obtaining a kidney transplant. The primary aim in the NOCTx study was to prospectively measure progression of arterial calcification in nocturnal haemodialysis patients and in kidney transplant patients, hence baseline results were collected 2-3 months after treatment initiation to allow for stabilization of laboratory data and treatment modality. In chapter 9, we present the results of the secondary outcome, HRQoL as measured with the Kidney Disease Quality of Life Instrument (KDQOL-SF), of 75 patients with 1-year follow-up data. General HRQoL did not demonstrate a statistically significant difference between nocturnal haemodialysis and kidney transplantation, although clinically relevant changes in the physical component score and kidney-disease specific domains were observed in favor of kidney transplantation. Only the domain 'effects of kidney disease' was associated with a significantly lower score in incident nocturnal HD patients compared to incident kidney transplant patients, implicating less favorable HRQoL for nocturnal HD patients. A kidney transplant for transplantation-eligible patients should always be the preferred therapy, both in terms of traditional outcomes and HRQoL, as our study suggests. Yet, for patients ineligible for kidney transplantation, nocturnal HD

may have potential advantages compared to other dialysis modalities. Nocturnal HD patients may be more free in diet restrictions and in the ability to work or travel due to nightly HD regimens. Intensive HD regimens, such as nocturnal HHD, are associated with higher survival rates compared to CHD.⁶⁹⁻⁷¹ The hypothesized mechanism of improved survival from intensive HD regimens is that these regimens lead to better solute and fluid removal¹⁵, resulting in better blood pressure, phosphate control and nutritional status and thus in turn reducing morbidity and mortality.¹⁶ These mechanisms likely also contribute to improved HRQoL when patients switch to nocturnal HD.⁷²⁻⁷⁴ The extra supplemental tables of the manuscript indeed showed that some kidney-disease specific domains of nocturnal patients were higher than in CHD patients, consistent with previous reports on nocturnal HD and CHD.^{72, 73} The CHD group in the NOCTx study were all transplantation-eligible patients, making them fairly more comparable to the group of incident kidney transplant and nocturnal HD patients then CHD patients in general. These tables were however omitted from the final article on request of the editor, because the CHD group consisted of prevalent patients. Nevertheless, these results might be useful when counselling on dialysis modality choice in patients who are ineligible for kidney transplantation.

This thesis was written during a worldwide and memorable pandemic, that of COVID-19. Therefore, it was indispensable to investigate the effect of this major pandemic on the main outcome parameter of this thesis within the studied population. In chapter 8, data from the ongoing prospective DOMESTICO study were used; eligible patients had HRQoL scores both before and during the COVID-19 pandemic. Mental health was assessed using both the Short Form-12, i.e. general HRQoL, and mental healthrelated symptoms of the Dialysis Symptom Index. We found that the mental health of dialysis patients was unaffected by the major pandemic; no difference was found in either the Mental Component Summary and in the presence of specific symptoms during the COVID-19 pandemic. We hypothesize that major life events do not effect dialysis patients due to their high resilience. This is striking, as the COVID-19 pandemic negatively impacted the mental health of patients with many other chronic diseases, such as malignancies and Alzheimer's disease.75,76 Another explanation for the presented results, also stated in the original article, is that the majority of the studied patients were treated with CHD and thus had frequent social contact with healthcare professionals despite social distancing. Social distancing negatively affects important mental health-related outcomes such as depression and social anxiety, as stated in a recent review.⁷⁷ This is important to consider, since this is unlikely to be the last outbreak in human history, and the current COVID-19 pandemic resulted in an increased interest in home dialysis.⁷⁸ Sufficient support from the health care professionals, for example provided by eHealth, to maintain social connection and quality of care is then needed.

HRQoL as a core outcome fits better in the paradigm shift towards patient-centred care then traditional outcomes such as survival. Patient-centred care focusses on finding the treatment that suits the patient best, instead of finding the treatment with the best clinical outcomes. Since there are few major differences in clinical outcomes between dialysis modalities, a focus on patient-centred care is especially valuable in ESKD. With the shift towards patient-centred care, understanding patient-reported outcomes such as HRQoL is pivotal for an in-depth conversation with the patient about dialysis modality choice. There are several examples in the context of kidney disease showing the focus on patient-centred care. Examples include renewed interest in abandoning traditional thrice-weekly regimens by also offering nocturnal dialysis or incremental dialysis, and thus individualizing treatment according to patient needs.^{79, 80} As another example, several centres are providing extra support in the home to safely conduct dialysis, such as assisted PD or providing home haemodialysis by a dialysis nurse. However, probably the intriguing examples are the initiatives of single centres. To illustrate, an example from Bravis hospital, a centre that also offers haemodialysis at the general practitioner's practice, the so-called *dialysis hub*, thus bringing dialysis to patients without burdening them with long travel.⁸¹ Another example: offering nocturnal haemodialysis within the safe surrounding of a dialysis centre but with private sleeping rooms (Diapriva Dialysis Centre). The latter examples, including the example in chapter 4, may be exemplified as 'good practices', i.e. as practices initiated by health care professionals with high success within their centre but lacking an evidence-based foundation.⁸²

One of the principles of patient-centred care is shared decision making. In my opinion, shared decision making should be achieved by listening carefully to the patient's beliefs, providing information by a multidisciplinary team and offering all treatment options, not necessarily the available or feasible options. Only then one can make an informed decision. In European surveys, 25-39% of patients reported that they were not informed about any treatment modality options.^{83, 84} In higher income countries, a substantial proportion of patients were not informed about PD (16%) and HHD (27%).⁸³ The presented examples of incomplete treatment modality education are obviously are barrier to home dialysis. While the nephrologist reported that he never

made the decision without consulting the patient⁸⁵, 10% of CHD patients felt the doctor decided for them compared to 1,5% of patients on home dialysis.⁸³ This might be the result of patients simply being unable to remember the decision on treatment modality (recall bias). However, due to the difference between CHD and home dialysis patients, this phenomenon might also be the result of implicit persuasion, i.e. unintentionally steering patients towards the treatment that seems to suit best. Steering behaviour is frequently observed in oncology⁸⁶, but has recently also been recognized in treatment modality consultations in nephrology.⁸⁷ Selectively presenting treatment options, advantages and disadvantages is common, and could lead to less involvement of the patient in the treatment modality decision.⁸⁷

Discussing treatment modality options in ESKD, also means fully considering the patient's social system. As mentioned earlier in this thesis, life-long dialysis dependency requires a certain structure to everyday-life not only to the patient but also to the patient's social system, especially spouses and children. Since family members are frequently also carers to the patient, especially elderly patients, it is important to include informal caregivers in the treatment modality decision. The treatment modality decision is thus never a decision by one person, but should be team-made. An informed treatment modality decision will eventually lead to treatment satisfaction and thus higher HRQoL scores. The decision may sometimes be an unexpected choice for the clinician (personal experience). During this research project, I met a young employed man with small children. Although he started on PD, he eventually opted for CHD because he did not want his home to turn into a hospital.

Part IV - Future perspectives and conclusions

Although home dialysis has been associated with similar survival, its impact on HRQoL and cost-effectiveness in especially elderly patients require further elucidation. The prospective DOMESTICO study will investigate the effect of home dialysis therapies on HRQoL, cost-effectiveness and clinical outcomes, compared to CHD in the current dialysis population (Chapter 10). Several patient-reported outcome measures are being used, including the Short-Form 12 to measure generic HRQoL and the Dialysis Symptom Index to measure symptom burden. Symptom burden is related to the concept of HRQoL, according to the model of Wilson & Cleary: a high symptom burden causes a decline in functional status and in turn lower HRQoL.⁸⁸ The DOMESTICO study will also include some interesting secondary outcomes, such as caregiver burden and productivity losses, that are still knowledge gaps.

As previously stated, HRQoL is an important outcome to dialysis patients.⁴⁸⁻⁵¹ HRQoL is however still a non-traditional outcome in clinical practice, because it only recently gained its role as a core outcome in research.^{48, 49} Even though establishing a core outcome set is intended to enhance shared decision making and quality of care, patient-reported outcome measures are not frequently used in daily practice. Dialysis patients have low HRQoL and high symptom burden, but due to underreporting not all symptoms are adequately treated.⁸⁹ It is therefore pivotal, that health care workers integrate HRQoL and symptom questionnaires into everyday care. One way to achieve this is to routinely apply those questionnaires in clinical practice.⁹⁰ Introducing patient-reported outcomes within the dialysis registry in the Netherlands, i.e. by the means of the PROMS-NL (both Short Form-12 and Dialysis Symptom Index) of Nefrovisie, is a good example of this.⁹¹ Subsequently, data on HRQoL and especially symptoms should be used to relieve symptom burden, which is best achieved if all health care workers perceive benefits from collecting data on patient-reported outcomes. It is important that the entire multi-disciplinary team values patientreported outcomes and understands their interpretation, to relieve the workload associated with collecting questionnaires and avoid survey fatigue among patients through good motivation. In a study from the UK concerning routinely measuring HRQoL and symptom questionnaires, difficulty in understanding questionnaire results by health professionals was a barrier to integrating questionnaires in everyday care. This resulted in the study team providing workshops to discuss the interpretation of results.⁸⁹ Likewise in our own prospective study, the inclusion rate and questionnaire collection seem to be associated with the feedback provided by the nephrologist to the patient and the endeavor of the entire health care team involved. Future studies should therefore focus on 'good practices' of integrating questionnaires into everyday care. Another important direction for future studies is whether management of symptoms has an effect on HRQoL.

Cost-effectiveness is the other important outcome in the prospective DOMESTICO study. Most economic evaluation studies to date, have used health insurance claims to compare costs between various dialysis modalities.⁹²⁻⁹⁴ These studies suggest that home dialysis is associated with lower costs. Also a recent Dutch study performed with Vektis data concluded that CAPD was associated with the lowest costs, including also indirect non-medical costs such as transportation.⁹² Still this study did not include all costs to estimate cost-effectiveness, most importantly indirect non-medical costs related to productivity losses were unknown while these costs can be rather important

in dialysis patients.⁹⁵ Productivity costs are best described as 'costs associated with production loss and replacement costs due to illness, disability and death of productive persons'.⁹⁶ A large proportion of dialysis patients are unemployed. This may be partly due to disability of the disease, but also by difficulty to combine the thrice-weekly regimen of CHD with a paid job schedule. As a result, employed patients frequently opt for home dialysis. If more patients receive home dialysis, more patients might be able to go back to work and thus contribute to reducing production losses. In the current aging population this may seem of less importance, but it is important to consider that elderly patients frequently need informal caregivers to support them with dialysis-related tasks. Productivity losses of these informal caregivers might result in overall higher costs. Overall, including all medical and non-medical costs would yield the ultimate answer to the cost-effectiveness of home dialysis.

As mentioned earlier in this chapter, more patients might be eligible for home dialysis treatment. Important barriers that were presented in this discussion are insufficient education of patients and operational barriers: patients can receive home dialysis only if the treatment option is offered and if the treatment is available. This requires a robust healthcare organization and infrastructure, such as sufficient access to home care support, and policies that promote home dialysis. After all, providing home dialysis to incident patients is more challenging than starting patients on CHD: home dialysis requires infrastructure and a dedicated team. An increase in the number of patients receiving home dialysis would enable health-care resources for home dialysis to be used more efficiently. Combined with potentially lower costs, home dialysis might be a sustainable option for the growing population of patients with ESKD. We, as physicians, are responsible not only for the health of our individual patients, but also for future patients with ESKD.

This thesis aimed to identify opportunities to offer home dialysis to a larger patient population and to improve the technique survival of peritoneal dialysis. The prevention of PD infections and catheter problems might contribute to a long-term treatment with PD. Furthermore, the presented results might help to improve treatment decision making and overall HRQoL of all dialysis patients. Access to home dialysis should be improved, to provide more patients an opportunity to choose home dialysis as treatment modality.

References

- 1. Li PK-T, Chow KM, Van de Luijtgaarden MWM, Johnson DW, Jager KJ, Mehrotra R, et al. Changes in the worldwide epidemiology of peritoneal dialysis. Nature Reviews Nephrology. 2017;13(2):90-103.
- Robinson BM, Akizawa T, Jager KJ, Kerr PG, Saran R, Pisoni RL. Factors affecting outcomes in patients reaching end-stage kidney disease worldwide: differences in access to renal replacement therapy, modality use, and haemodialysis practices. Lancet. 2016;388(10041):294-306.
- 3. van de Luijtgaarden MW, Jager KJ, Segelmark M, Pascual J, Collart F, Hemke AC, et al. Trends in dialysis modality choice and related patient survival in the ERA-EDTA Registry over a 20-year period. Nephrol Dial Transplant. 2016;31(1):120-8.
- System USRD. United States Renal Data System. 2019 USRDS Annual Data Report: Epidemiology of kidney disease in the United States. National Institutes of Health, National Institute of Diabetes and Digestive and Kidney Diseases, Bethesda, MD, 2019 [Internet]. Available from: https://www.usrds.org/annual-data-report/ Accessed: 7-8-2021. . 2019.
- ANZDATA Registry. 43rd Report, Chapter 1: Incidence of Renal Replacement Therapy for End Stage Kidney Disease. Australia and New Zealand Dialysis and Transplant Registry, Adelaide, Australia. 2020. Available from: http://www.anzdata.org.au Accessed: 13-07-2021. 2020.
- ERA-EDTA Registry: ERA-EDTA Registry Annual Report 2018. Amsterdam UMC, location AMC, Department of Medical Informatics, Amsterdam, the Netherlands, 2020. 2018.
- 7. Mendelssohn DC, Mujais SK, Soroka SD, Brouillette J, Takano T, Barre PE, et al. A prospective evaluation of renal replacement therapy modality eligibility. Nephrol Dial Transplant. 2009;24(2):555-61.
- 8. Mehrotra R, Marsh D, Vonesh E, Peters V, Nissenson A. Patient education and access of ESRD patients to renal replacement therapies beyond in-center hemodialysis. Kidney International. 2005;68(1):378-90.
- 9. Oliver MJ, Quinn RR, Richardson EP, Kiss AJ, Lamping DL, Manns BJ. Home care assistance and the utilization of peritoneal dialysis. Kidney Int. 2007;71(7):673-8.
- Shukla AM, Easom A, Singh M, Pandey R, Rotaru D, Wen X, et al. Effects of a Comprehensive Predialysis Education Program on the Home Dialysis Therapies: A Retrospective Cohort Study. Perit Dial Int. 2017;37(5):542-7.
- Canadian Institute for Health Information. Treatment of End-Stage Organ Failure in Canada, Canadian Organ Replacement Register, 2010 to 2019: End-Stage Kidney Disease and Kidney Transplants — Data Tables. Ottawa, ON: CIHI; 2020. Available from: https:// www.cihi.ca/en/organ-replacement-in-canada-corr-annual-statistics-2020 Accessed: 1-10-2021.
- 12. Brown EA, Johansson L. Dialysis options for end-stage renal disease in older people. Nephron Clin Pract. 2011;119 Suppl 1:c10-3.
- 13. Auguste BL, Chan CT. Home Dialysis Among Elderly Patients: Outcomes and Future Directions. Canadian Journal of Kidney Health and Disease. 2019;6:2054358119871031.

- 14. Eroglu E, Heimburger O, Lindholm B. Peritoneal dialysis patient selection from a comorbidity perspective. Semin Dial. 2020.
- 15. Leypoldt JK. Kinetics of beta2-microglobulin and phosphate during hemodialysis: effects of treatment frequency and duration. Semin Dial. 2005;18(5):401-8.
- Rocco MV, Lockridge RS, Jr., Beck GJ, Eggers PW, Gassman JJ, Greene T, et al. The effects of frequent nocturnal home hemodialysis: the Frequent Hemodialysis Network Nocturnal Trial. Kidney Int. 2011;80(10):1080-91.
- 17. Brown EA, Wilkie M. Assisted Peritoneal Dialysis as an Alternative to In-Center Hemodialysis. Clin J Am Soc Nephrol. 2016;11(9):1522-4.
- Li PK, Law MC, Chow KM, Leung CB, Kwan BC, Chung KY, et al. Good patient and technique survival in elderly patients on continuous ambulatory peritoneal dialysis. Perit Dial Int. 2007;27 Suppl 2:S196-201.
- Castrale C, Evans D, Verger C, Fabre E, Aguilera D, Ryckelynck JP, et al. Peritoneal dialysis in elderly patients: report from the French Peritoneal Dialysis Registry (RDPLF). Nephrol Dial Transplant. 2010;25(1):255-62.
- 20. van de Luijtgaarden MW, Noordzij M, Stel VS, Ravani P, Jarraya F, Collart F, et al. Effects of comorbid and demographic factors on dialysis modality choice and related patient survival in Europe. Nephrol Dial Transplant. 2011;26(9):2940-7.
- Naylor KL, Kim SJ, McArthur E, Garg AX, McCallum MK, Knoll GA. Mortality in Incident Maintenance Dialysis Patients Versus Incident Solid Organ Cancer Patients: A Population-Based Cohort. American Journal of Kidney Diseases. 2019;73(6):765-76.
- 22. Couchoud C, Savoye E, Frimat L, Ryckelynck JP, Chalem Y, Verger C. Variability in case mix and peritoneal dialysis selection in fifty-nine French districts. Perit Dial Int. 2008;28(5):509-17.
- 23. Ethier I, Cho Y, Hawley C, Pascoe EM, Roberts MA, Semple D, et al. Effect of patient- and center-level characteristics on uptake of home dialysis in Australia and New Zealand: a multicenter registry analysis. Nephrol Dial Transplant. 2020.
- 24. Walker RC, Hanson CS, Palmer SC, Howard K, Morton RL, Marshall MR, et al. Patient and caregiver perspectives on home hemodialysis: a systematic review. Am J Kidney Dis. 2015;65(3):451-63.
- 25. Ismail SY, Luchtenburg AE, Timman R, Zuidema WC, Boonstra C, Weimar W, et al. Home-based family intervention increases knowledge, communication and living donation rates: a randomized controlled trial. Am J Transplant. 2014;14(8):1862-9.
- de Maar JS, de Groot MA, Luik PT, Mui KW, Hagen EC. GUIDE, a structured predialysis programme that increases the use of home dialysis. Clin Kidney J. 2016;9(6):826-32.
- 27. Honkanen EOaR, V.M. . What happened in Finland to increase home hemodialysis? Hemodialysis International 2008;12.
- Kramer A, Boenink R, Stel VS, Santiuste de Pablos C, Tomović F, Golan E, et al. The ERA-EDTA Registry Annual Report 2018: a summary. Clinical Kidney Journal. 2021;14(1):107-23.
- 29. Oliver MJ, Al-Jaishi AA, Dixon SN, Perl J, Jain AK, Lavoie SD, et al. Hospitalization Rates for Patients on Assisted Peritoneal Dialysis Compared with In-Center Hemodialysis. Clinical journal of the American Society of Nephrology : CJASN. 2016;11(9):1606-14.

- Lafrance JP, Rahme E, Iqbal S, Elftouh N, Vallée M, Laurin LP, et al. Association of dialysis modality with risk for infection-related hospitalization: a propensity scorematched cohort analysis. Clin J Am Soc Nephrol. 2012;7(10):1598-605.
- Jeon Y, Kim HD, Hong YA, Kim HW, Yang CW, Chang YK, et al. Clinical outcomes of infection-related hospitalization in incident peritoneal dialysis patients. Kidney Res Clin Pract. 2020;39(4):460-8.
- 32. Banshodani M, Kawanishi H, Moriishi M, Shintaku S, Tsuchiya S. Association between Dialysis Modality and Infectious Diseases: Peritoneal Dialysis versus Hemodialysis. Blood Purif. 2021;50(3):370-9.
- 33. Cho Y, See EJ, Htay H, Hawley CM, Johnson DW. Early Peritoneal Dialysis Technique Failure: Review. Perit Dial Int. 2018;38(5):319-27.
- 34. Kolesnyk I, Dekker FW, Boeschoten EW, Krediet RT. Time-dependent reasons for peritoneal dialysis technique failure and mortality. Perit Dial Int. 2010;30(2):170-7.
- 35. Guo A, Mujais S. Patient and technique survival on peritoneal dialysis in the United States: evaluation in large incident cohorts. Kidney Int Suppl. 2003(88):S3-12.
- 36. Jaar BG, Plantinga LC, Crews DC, Fink NE, Hebah N, Coresh J, et al. Timing, causes, predictors and prognosis of switching from peritoneal dialysis to hemodialysis: a prospective study. BMC Nephrology. 2009;10(1):3.
- 37. See EJ, Johnson DW, Hawley CM, Pascoe EM, Badve SV, Boudville N, et al. Risk Predictors and Causes of Technique Failure Within the First Year of Peritoneal Dialysis: An Australia and New Zealand Dialysis and Transplant Registry (ANZDATA) Study. Am J Kidney Dis. 2018;72(2):188-97.
- Béchade C, Guittet L, Evans D, Verger C, Ryckelynck JP, Lobbedez T. Early failure in patients starting peritoneal dialysis: a competing risks approach. Nephrol Dial Transplant. 2014;29(11):2127-35.
- 39. Li PK, Szeto CC, Piraino B, de Arteaga J, Fan S, Figueiredo AE, et al. ISPD Peritonitis Recommendations: 2016 Update on Prevention and Treatment. Perit Dial Int. 2016;36(5):481-508.
- Crabtree JH, Shrestha BM, Chow KM, Figueiredo AE, Povlsen JV, Wilkie M, et al. Creating and Maintaining Optimal Peritoneal Dialysis Access in the Adult Patient: 2019 Update. Perit Dial Int. 2019;39(5):414-36.
- 41. Szeto CC, Li PK, Johnson DW, Bernardini J, Dong J, Figueiredo AE, et al. ISPD Catheter-Related Infection Recommendations: 2017 Update. Perit Dial Int. 2017;37(2):141-54.
- 42. Perl J, Davies SJ, Lambie M, Pisoni RL, McCullough K, Johnson DW, et al. The Peritoneal Dialysis Outcomes and Practice Patterns Study (PDOPPS): Unifying Efforts to Inform Practice and Improve Global Outcomes in Peritoneal Dialysis. Perit Dial Int. 2016;36(3):297-307.
- 43. Abrahams AC, Ruger W, Ter Wee PM, van Ittersum FJ, Boer WH. Improved Outcome of Enteric Peritonitis in Peritoneal Dialysis Patients Aged 50 Years and Older with Temporary Discontinuation of Peritoneal Dialysis and Intravenous Meropenem. Perit Dial Int. 2017.
- 44. van der Sluijs AVE, Eekelschot KZ, Frakking FN, Haas PA, Boer WH, Abrahams AC. Salvage of the peritoneal dialysis catheter in Candida peritonitis using amphotericin B catheter lock. Perit Dial Int. 2021;41(1):110-4.
- 45. Htay H, Johnson DW, Craig JC, Teixeira-Pinto A, Hawley CM, Cho Y. Urgent-start peritoneal dialysis versus conventional-start peritoneal dialysis for people with chronic kidney disease. Cochrane Database Syst Rev. 2020;12(12):Cd012913.

- 46. van der Willik EM, van Zwet EW, Hoekstra T, van Ittersum FJ, Hemmelder MH, Zoccali C, et al. Funnel plots of patient-reported outcomes (PROs) to evaluate healthcare quality: basic principles, pitfalls and considerations. Nephrology (Carlton). 2020.
- 47. Spiegelhalter DJ. Funnel plots for comparing institutional performance. Stat Med. 2005;24(8):1185-202.
- Manera KE, Johnson DW, Craig JC, Shen JI, Gutman T, Cho Y, et al. Establishing a Core Outcome Set for Peritoneal Dialysis: Report of the SONG-PD (Standardized Outcomes in Nephrology-Peritoneal Dialysis) Consensus Workshop. Am J Kidney Dis. 2020;75(3):404-12.
- 49. Evangelidis N, Tong A, Manns B, Hemmelgarn B, Wheeler DC, Tugwell P, et al. Developing a Set of Core Outcomes for Trials in Hemodialysis: An International Delphi Survey. American Journal of Kidney Diseases. 2017;70(4):464-75.
- 50. Verberne WR, Das-Gupta Z, Allegretti AS, Bart HAJ, van Biesen W, García-García G, et al. Development of an International Standard Set of Value-Based Outcome Measures for Patients With Chronic Kidney Disease: A Report of the International Consortium for Health Outcomes Measurement (ICHOM) CKD Working Group. Am J Kidney Dis. 2019;73(3):372-84.
- 51. Manns B, Hemmelgarn B, Lillie E, Dip SC, Cyr A, Gladish M, et al. Setting research priorities for patients on or nearing dialysis. Clin J Am Soc Nephrol. 2014;9(10):1813-21.
- 52. Mittal SK, Ahern L, Flaster E, Maesaka JK, Fishbane S. Self-assessed physical and mental function of haemodialysis patients. Nephrol Dial Transplant. 2001;16(7):1387-94.
- 53. van Sandwijk MS, Al Arashi D, van de Hare FM, van der Torren JMR, Kersten MJ, Bijlsma JA, et al. Fatigue, anxiety, depression and quality of life in kidney transplant recipients, haemodialysis patients, patients with a haematological malignancy and healthy controls. Nephrol Dial Transplant. 2019;34(5):833-8.
- 54. Lacson E, Jr., Xu J, Lin SF, Dean SG, Lazarus JM, Hakim RM. A comparison of SF-36 and SF-12 composite scores and subsequent hospitalization and mortality risks in long-term dialysis patients 2010 [updated FebPMC2827595]. 2009/12/19:[252-60].
- 55. van Dijk S, van den Beukel TO, Dekker FW, le Cessie S, Kaptein AA, Honig A, et al. Short-term versus long-term effects of depressive symptoms on mortality in patients on dialysis. Psychosom Med. 2012;74(8):854-60.
- Bossola M, Di Stasio E, Antocicco M, Panico L, Pepe G, Tazza L. Fatigue Is Associated with Increased Risk of Mortality in Patients on Chronic Hemodialysis. Nephron. 2015;130(2):113-8.
- 57. Brown EA, Finkelstein FO, Iyasere OU, Kliger AS. Peritoneal or hemodialysis for the frail elderly patient, the choice of 2 evils? Kidney Int. 2017;91(2):294-303.
- Iyasere OU, Brown EA, Johansson L, Huson L, Smee J, Maxwell AP, et al. Quality of Life and Physical Function in Older Patients on Dialysis: A Comparison of Assisted Peritoneal Dialysis with Hemodialysis. Clin J Am Soc Nephrol. 2016;11(3):423-30.
- 59. Griva K, Kang AW, Yu ZL, Lee VY, Zarogianis S, Chan MC, et al. Predicting technique and patient survival over 12 months in peritoneal dialysis: the role of anxiety and depression. Int Urol Nephrol. 2016;48(5):791-6.
- 60. Manera KE, Johnson DW, Cho Y, Sautenet B, Shen J, Kelly A, et al. Scope and heterogeneity of outcomes reported in randomized trials in patients receiving peritoneal dialysis. Clinical Kidney Journal. 2020;14(7):1817-25.

- 61. Ruiz de Alegria-Fernandez de Retana B, Basabe-Baranano N, Saracho-Rotaeche R. Coping mechanisms as a predictor for quality of life in patients on dialysis: a longitudinal and multi-centre study. Nefrologia. 2013;33(3):342-54.
- 62. Neumann D, Lamprecht J, Robinski M, Mau W, Girndt M. Social relationships and their impact on health-related outcomes in peritoneal versus haemodialysis patients: a prospective cohort study. Nephrology Dialysis Transplantation. 2018;33(7):1235-44.
- 63. Painter P, Krasnoff JB, Kuskowski M, Frassetto L, Johansen K. Effects of modality change on health-related quality of life. Hemodial Int. 2012;16(3):377-86.
- 64. Da Silva-Gane M, Wellsted D, Greenshields H, Norton S, Chandna SM, Farrington K. Quality of life and survival in patients with advanced kidney failure managed conservatively or by dialysis. Clinical journal of the American Society of Nephrology : CJASN. 2012;7(12):2002-9.
- Suri RS, Garg AX, Chertow GM, Levin NW, Rocco MV, Greene T, et al. Frequent Hemodialysis Network (FHN) randomized trials: study design. Kidney Int. 2007;71(4):349-59.
- 66. Korevaar JC, Feith GW, Dekker FW, van Manen JG, Boeschoten EW, Bossuyt PM, et al. Effect of starting with hemodialysis compared with peritoneal dialysis in patients new on dialysis treatment: a randomized controlled trial. Kidney Int. 2003;64(6):2222-8.
- 67. Merkus MP, Jager KJ, Dekker FW, De Haan RJ, Boeschoten EW, Krediet RT. Quality of life over time in dialysis: the Netherlands Cooperative Study on the Adequacy of Dialysis. NECOSAD Study Group. Kidney Int. 1999;56(2):720-8.
- 68. Manns BJ, Walsh MW, Culleton BF, Hemmelgarn B, Tonelli M, Schorr M, et al. Nocturnal hemodialysis does not improve overall measures of quality of life compared to conventional hemodialysis. Kidney Int. 2009;75(5):542-9.
- 69. Weinhandl ED, Liu J, Gilbertson DT, Arneson TJ, Collins AJ. Survival in daily home hemodialysis and matched thrice-weekly in-center hemodialysis patients. J Am Soc Nephrol. 2012;23(5):895-904.
- 70. Chertow GM, Levin NW, Beck GJ, Depner TA, Eggers PW, Gassman JJ, et al. Incenter hemodialysis six times per week versus three times per week. N Engl J Med. 2010;363(24):2287-300.
- 71. Mathew A, McLeggon J-A, Mehta N, Leung S, Barta V, McGinn T, et al. Mortality and Hospitalizations in Intensive Dialysis: A Systematic Review and Meta-Analysis. Canadian journal of kidney health and disease. 2018;5:2054358117749531-.
- 72. Culleton BF, Walsh M, Klarenbach SW, Mortis G, Scott-Douglas N, Quinn RR, et al. Effect of frequent nocturnal hemodialysis vs conventional hemodialysis on left ventricular mass and quality of life: a randomized controlled trial. JAMA. 2007;298(11):1291-9.
- Van Eps CL, Jeffries JK, Johnson DW, Campbell SB, Isbel NM, Mudge DW, et al. Quality of life and alternate nightly nocturnal home hemodialysis. Hemodial Int. 2010;14(1):29-38.
- Chiu YW, Teitelbaum I, Misra M, de Leon EM, Adzize T, Mehrotra R. Pill burden, adherence, hyperphosphatemia, and quality of life in maintenance dialysis patients. Clin J Am Soc Nephrol. 2009;4(6):1089-96.
- Ayubi E, Bashirian S, Khazaei S. Depression and Anxiety Among Patients with Cancer During COVID-19 Pandemic: A Systematic Review and Meta-analysis. J Gastrointest Cancer. 2021;52(2):499-507.

- 76. Manca R, De Marco M, Venneri A. The Impact of COVID-19 Infection and Enforced Prolonged Social Isolation on Neuropsychiatric Symptoms in Older Adults With and Without Dementia: A Review. Front Psychiatry. 2020;11:585540.
- 77. Morina N, Kip A, Hoppen TH, Priebe S, Meyer T. Potential impact of physical distancing on physical and mental health: a rapid narrative umbrella review of meta-analyses on the link between social connection and health. BMJ Open. 2021;11(3):e042335.
- 78. Mendu ML, Divino-Filho JC, Vanholder R, Mitra S, Davies SJ, Jha V, et al. Expanding Utilization of Home Dialysis: An Action Agenda From the First International Home Dialysis Roundtable. Kidney Medicine. 2021.
- 79. Kalantar-Zadeh K, Unruh M, Zager PG, Kovesdy CP, Bargman JM, Chen J, et al. Twiceweekly and incremental hemodialysis treatment for initiation of kidney replacement therapy. Am J Kidney Dis. 2014;64(2):181-6.
- Pierratos A, Ouwendyk M, Francoeur R, Vas S, Raj DS, Ecclestone AM, et al. Nocturnal hemodialysis: three-year experience. Journal of the American Society of Nephrology. 1998;9(5):859-68.
- Bravis start dialyse op huisartsenlocatie. Availlable from: https://www.bravisziekenhuis. nl/over-bravis/laatste-nieuws/bravis-start-dialyse-op-huisartsenlocatie Accessed: 18-11-2019.
- van Eck van der Sluijs A, Vonk S, van Jaarsveld BC, Bonenkamp AA, Abrahams AC. Good practices for dialysis education, treatment, and eHealth: A scoping review. PLoS One. 2021;16(8):e0255734.
- 83. de Jong RW, Stel VS, Rahmel A, Murphy M, Vanholder RC, Massy ZA, et al. Patientreported factors influencing the choice of their kidney replacement treatment modality. Nephrol Dial Transplant. 2021.
- 84. Van Biesen W, van der Veer SN, Murphey M, Loblova O, Davies S. Patients' perceptions of information and education for renal replacement therapy: an independent survey by the European Kidney Patients' Federation on information and support on renal replacement therapy. PLoS One. 2014;9(7):e103914.
- 85. de Jong RW, Jager KJ, Vanholder RC, Couchoud C, Murphy M, Rahmel A, et al. Results of the European EDITH nephrologist survey on factors influencing treatment modality choice for end-stage kidney disease. Nephrol Dial Transplant. 2021.
- 86. Engelhardt EG, Pieterse AH, van der Hout A, de Haes HJ, Kroep JR, Quarles van Ufford-Mannesse P, et al. Use of implicit persuasion in decision making about adjuvant cancer treatment: A potential barrier to shared decision making. Eur J Cancer. 2016;66:55-66.
- van Dulmen S, Peereboom E, Schulze L, Prantl K, Rookmaaker M, van Jaarsveld BC, et al. The use of implicit persuasion in decision-making about treatment for end-stage kidney disease. Perit Dial Int. 2021:8968608211027019.
- 88. Wilson IB, Cleary PD. Linking Clinical Variables With Health-Related Quality of Life: A Conceptual Model of Patient Outcomes. JAMA. 1995;273(1):59-65.
- 89. van der Veer SN, Aresi G, Gair R. Incorporating patient-reported symptom assessments into routine care for people with chronic kidney disease. Clin Kidney J. 2017;10(6):783-7.
- 90. Breckenridge K, Bekker HL, Gibbons E, van der Veer SN, Abbott D, Briancon S, et al. How to routinely collect data on patient-reported outcome and experience measures in renal registries in Europe: an expert consensus meeting. Nephrol Dial Transplant. 2015;30(10):1605-14.

- 91. van der Willik EM, Hemmelder MH, Bart HAJ, van Ittersum FJ, Hoogendijk-van den Akker JM, Bos WJW, et al. Routinely measuring symptom burden and health-related quality of life in dialysis patients: first results from the Dutch registry of patient-reported outcome measures. Clinical Kidney Journal. 2020;14(6):1535-44.
- 92. Mohnen SM, van Oosten MJM, Los J, Leegte MJH, Jager KJ, Hemmelder MH, et al. Healthcare costs of patients on different renal replacement modalities - Analysis of Dutch health insurance claims data. PLoS One. 2019;14(8):e0220800.
- Li B, Cairns JA, Fotheringham J, Tomson CR, Forsythe JL, Watson C, et al. Understanding cost of care for patients on renal replacement therapy: looking beyond fixed tariffs. Nephrol Dial Transplant. 2015;30(10):1726-34.
- 94. Couillerot-Peyrondet AL, Sambuc C, Sainsaulieu Y, Couchoud C, Bongiovanni-Delaroziere I. A comprehensive approach to assess the costs of renal replacement therapy for end-stage renal disease in France: the importance of age, diabetes status, and clinical events. Eur J Health Econ. 2017;18(4):459-69.
- 95. Bouwmans C, Krol M, Severens H, Koopmanschap M, Brouwer W, Roijen LH-v. The iMTA Productivity Cost Questionnaire: A Standardized Instrument for Measuring and Valuing Health-Related Productivity Losses. Value in Health. 2015;18(6):753-8.
- Brouwer WBF, Koopmanschap MA, Rutten FFH. Productivity costs in cost-effectiveness analysis: Numerator or denominator: A further discussion. Health Economics. 1997;6(5):511-4.



Appendices

Nederlandse samenvatting Appendix A Appendix B DOMESTICO Study Group members List of publications Dankwoord About the author

Nederlandse samenvatting

Wereldwijd neemt het aantal patiënten met chronische nierinsufficiëntie en eindstadium nierfalen toe.^{1, 2} Het groeiend aantal patiënten vormt een groot economisch probleem, omdat dialyse een relatief dure therapie is. Dialyse bestaat grofweg uit centrumhemodialyse (CHD) en thuisdialyse, waaronder zowel peritoneale dialyse (PD) als thuis hemodialyse (THD) vallen. In veruit de meeste landen is het aantal patiënten dat met thuisdialyse behandeld wordt veel lager dan het aantal patiënten dat behandeld wordt in een dialysecentrum.^{2, 3} Dit is opvallend, omdat thuisdialyse geassocieerd is met gelijke overleving als CHD maar ook met meer vrijheid en flexibiliteit dan CHD.⁴ Bovendien is thuisdialyse mogelijk goedkoper.⁵⁻⁹ Met een wereldwijd groeiend aantal patiënten met eindstadium nierfalen, is het dus interessant om uitkomsten van thuisdialyse in de huidige populatie verder te onderzoeken.

In Nederland hebben we te maken met een uitgesproken groei in het aantal oudere patiënten met nierfalen en dus met een verandering van de dialysepopulatie.¹⁰ Deze ontwikkeling vraagt om meer individualisering van de zorg en om meer aandacht voor kwaliteit van leven dan voor overleving. Het uiteindelijke doel in de spreekkamer is de therapie te vinden die het beste bij de patiënt past.

In dit proefschrift wordt in de huidige populatie gekeken naar de geschiktheid voor thuisdialyse, naar manieren om de techniekoverleving op thuisdialyse te verbeteren en naar kwaliteit van leven, als belangrijke patiënt-gerapporteerde uitkomst.

Deel I - geschiktheid voor thuisdialyse in de huidige populatie

Er zijn waarschijnlijk meer patiënten geschikt voor thuisdialyse dan er op dit moment behandeld worden met thuisdialyse. In Nederland wordt 20% van de dialysepatiënten behandeld met PD of THD, terwijl in studies geschat wordt dat meer dan 80 % van de patiënten medisch gezien in aanmerking komt voor thuisdialyse.¹¹⁻¹³ Uit eerdere studies bleek ook dat wanneer patiënten uitgebreide voorlichting krijgen, meer dan 60% voor thuisdialyse kiest.^{13, 14} In **hoofdstuk 2** laten we zien dat de Nederlandse thuisdialyse populatie de afgelopen jaren ouder wordt, door zowel de vergrijzing als doordat jongere patiënten vaker getransplanteerd worden. Hoewel het absolute aantal oudere thuisdialysepatiënten dus toenam, zagen we dat de relatieve incidentie op thuisdialyse over de jaren min of meer gelijk bleef. In patiënten in de leeftijd van 65-74, veranderde de incidentie thuisdialyse van 29 tot 25% in 20 jaar tijd, en bij patiënten boven de 75 jaar veranderde de incidentie tussen de 17 en 19%. In vergelijking tot andere landen waar het percentage oudere patiënten op thuisdialyse hoger ligt (Australië 24%, Nieuw Zeeland 47%)¹⁵, suggereren deze resultaten dat er mogelijk meer oudere patiënten in Nederland kunnen starten met thuisdialyse.

Naast leeftijd, wordt ook het hebben van veel comorbiditeiten vaak gezien als barrière voor het starten van thuisdialyse.16 In de retrospectieve DOMESTICO studie vonden wij echter geen relatie tussen het hebben van veel comorbiditeit en thuisdialyse, als er gecorrigeerd werd voor onder andere BMI en leeftijd (hoofdstuk 3). Patiënten met veel comorbiditeiten hadden een even grote kans te starten met CHD als thuisdialyse. Alleen patiënten met ernstig overgewicht én veel comorbiditeiten hadden een kleinere kans te starten met thuisdialyse. In dit hoofdstuk suggereren we dat er andere factoren, zoals BMI en leeftijd, allicht een grote rol spelen in de selectie van thuisdialyse dan comorbiditeit. De gemiddelde waardes van BMI en leeftijd verschilde tussen de centra onderling echter nauwelijks, terwijl er wel grote verschillen zijn tussen centra in het aantal patiënten dat thuis dialyseert. Juist daarom is het goed mogelijk dat centrum factoren, zoals verpleegkundige tekorten, een belangrijke rol spelen. Ook in studies uit het buitenland wordt geopperd dat de variatie in het aantal thuisdialyse patiënten tussen centra mogelijk deels verklaard kan worden door factoren als kleine centrumgrootte en het aanbieden van weinig opties voor meer flexibele dialyse (langere dialysesessies, dialyse gedurende de nacht etc).^{17, 18} Ook in andere landen om ons heen wordt eenzelfde variatie gezien in het percentage patiënten op thuisdialyse tussen verschillende centra. Vaak is het echter niet haalbaar om alle mogelijke keuzeopties (nachtdialyse, intensieve dialyse, thuisdialyse met een dialyse assistent etc.) aan te bieden. Derhalve is regionale samenwerkingen nodig om alle keuzeopties aan de patiënt te kunnen aanbieden.

Het is vanuit de spreekkamer lastig te beoordelen welke patiënt baat zal hebben van thuisdialyse. In **hoofdstuk 4** van dit proefschrift, beschrijven we de resultaten van een implementatieproject over voorlichting. Dit project bestond uit vragenlijsten over de geschiktheid voor thuisdialyse (voor zowel patiënt, maatschappelijk werker als nefroloog) en een huisbezoek. We vonden dat een maatschappelijk werker (tijdens een huisbezoek) de geschiktheid van een patiënt voor thuisdialyse goed kan beoordelen. Met name de specificiteit van het oordeel van de maatschappelijk werker was hoog, aangevende dat zij goed in staat was te onderscheiden welke patiënt waarschijnlijk minder geschikt was voor thuisdialyse. Belangrijke vragen die een hoge associatie met thuisdialysebehandeling bleken te hebben waren onder andere, 'is het huis geschikt voor thuisdialyse', 'is er een goede verhouding tussen draagkracht en draaglast' en 'heeft de patiënt een goed steunsysteem'. Het missen van steun vanuit de omgeving, is in eerder onderzoek genoemd als een belangrijke barrière voor het starten van thuisdialyse.¹⁹ Het betrekken van de familie bij de behandelkeuze, draagt waarschijnlijk bij aan de steun die later gegeven wordt. Gedurende het huisbezoek bestaat er ook de mogelijkheid de familie te betrekken bij de diagnose van eindstadium nierfalen. Daarom raden wij in dit hoofdstuk aan een huisbezoek op te nemen in het pre-dialyse programma.

Deel II - verbeteren van techniek overleving op peritoneale dialyse

Dialysepatiënten worden gemiddeld 1-2 keer per jaar opgenomen in het ziekenhuis.²⁰ Gebruikmakend van data uit de retrospectieve DOMESTICO studie, toonden wij in **hoofdstuk 5** dat PD patiënten twee keer vaker dan CHD patiënten opgenomen worden in het ziekenhuis. Het is goed mogelijk dat patiënten vaker opgenomen worden door de aard van de therapie, maar het kan ook het resultaat zijn van de locatie van de therapie. Men kan zich goed voorstellen dat patiënten die een thuisbehandeling krijgen laagdrempeliger opgenomen worden in het ziekenhuis, dan patiënten die driemaal per week gezien worden op de dialysezaal. PD patiënten werden het vaakste opgenomen voor een peritonitis. Daarom benadrukken wij in dit hoofdstuk dat het belangrijk is om infecties vanuit de PD catheter te voorkomen, om zo mogelijk het risico op ziekenhuisopnames te verlagen.

PD catheter gerelateerde infecties en peritonitiden waren ook een zeer belangrijke oorzaak van techniekfalen (hoodstuk 6). In dit hoofdstuk presenteren we de primaire uitkomstmaat van de retrospectieve DOMESTICO studie: techniekfalen. Het eerste jaar na start van PD had het hoogste risico op techniekfalen, zoals ook uit eerder onderzoek gebleken is.²¹ Weinig onderzoeken hebben echter gekeken naar de redenen van techniekfalen, met name de onderzoeken naar laat techniekfalen zijn zeer gering.²²⁻ ²⁴ Infecties waren een belangrijke oorzaak voor zowel laat als vroeg techniekfalen, catheter problemen en lekkages waren belangrijke oorzaken voor het vroeg falen van de techniek. Dit zijn allen oorzaken die in essentie te voorkomen zijn en dus een potentiele positieve invloed kunnen hebben op langdurige techniekoverleving op PD. Wij vonden bovendien dat ultrafiltratie-falen maar een oorzaak was voor 10% van de gevallen van techniekfalen (vroeg en laat gecombineerd), passend bij de hypothese die hierboven geschetst wordt. Langdurige therapie met PD is afhankelijk van het voorkomen van techniekfalen, en dus de preventie van PD infecties en catheter problemen.

Deel III - meer aandacht voor kwaliteit van leven

Kwaliteit van leven is een belangrijk onderwerp in studies geworden, sinds patiënten kwaliteit van leven zien als een belangrijke uitkomstenmaat.²⁵⁻²⁸ Kwaliteit van leven past beter bij een individualistische aanpak van zorg, dan een traditionele uitkomstmaat als overleving. Het doel moet zijn om de therapie te kiezen die het beste bij de patiënt past, en dat is niet per definitie de therapie met de beste uitkomstmaten. Bovendien zit er weinig verschil in uitkomstmaten tussen de verschillende dialysemodaliteiten, zodat kwaliteit van leven extra belangrijk kan zijn voor patiënten met nierfalen. In hoofdstuk 7 presenteren we de resultaten van een systematische review en metaanalyse naar kwaliteit van leven van thuisdialyse en CHD patiënten. Op basis van 46 studies uit verschillende landen, vonden wij dat thuisdialysepatiënten gemiddeld hogere kwaliteit van leven scores hadden. Echter, de kwaliteit van deze studies was laag en bovendien waren er maar weinig studies die over de tijd (longitudinaal) kwaliteit van leven gemeten hebben. Bovendien waren in de studies met name prevalente patiënten geïncludeerd. Algemeen neemt kwaliteit van leven van dialysepatiënten in de loop van de tijd af²⁹, terwijl in sommige studies een switch naar bijvoorbeeld thuishemodialyse gepaard lijkt te gaan met een verbetering van kwaliteit van leven.³⁰ Daarom moeten de resultaten van deze meta-analyse met de nodige voorzichtigheid worden geïnterpreteerd en concluderen wij dat longitudinaal vervolgonderzoek noodzakelijk is.

Ten tijde van dit promotietraject raakte de wereld in de ban van het SARS-CoV-2 virus. In hoofdstuk 8 onderzoeken we daarom het effect van de COVID-19 pandemie op de mentale gezondheid van dialysepatiënten. Hiervoor werden kwaliteit van leven metingen voor en tijdens de pandemie gebruikt van dialysepatiënten uit de prospectieve DOMESTICO studie. We vonden dat de pandemie geen invloed had op de algehele score van mentale kwaliteit van leven, noch op verschillende psychische symptomen zoals angst en somberheid. Wij suggereren dat dialysepatiënten gemiddeld veerkrachtig zijn, omdat zij in het dagelijks leven al geconfronteerd worden met vele uitdagingen. Een andere verklaring kan zijn dat dialysepatiënten, met name CHD patiënten, minder last hebben van de door het kabinet opgestelde maatregelen. Immers komen zij gemiddeld 3 keer per week naar het ziekenhuis. De NOCTx studie was een Nederlandse studie naar kransslagaderverkalking in patiënten met een niertransplantatie in vergelijking tot patiënten die behandeld werden met nachtelijke hemodialyse. Als secundaire uitkomstmaat werd ook kwaliteit van leven meegenomen. De algemene kwaliteit van leven was niet statistisch verschillend tussen de twee therapieën (hoofdstuk 9). Alleen het domein 'effecten van nierziekte' was geassocieerd met een significante lagere score in nachtelijke hemodialyse patiënten. In een aantal andere onderdelen van kwaliteit van leven werden wel nog klinisch relevante verschillen gevonden: namelijk de algehele score van fysieke kwaliteit van leven en een aantal nierziekte specifieke domeinen waren beter in patiënten met een niertransplantatie. Niertransplantatie gaat gepaard met de beste overleving en met vermoedelijk ook een betere kwaliteit van leven, zoals onze studie toonde. Daarom heeft een niertransplantatie ook altijd de eerste voorkeur, maar voor patiënten die niet geschikt zijn voor niertransplantatie, kan nachtelijke hemodialyse overwogen worden.

Deel IV - toekomstperspectief en conclusie

De kwaliteit van leven van thuisdialyse patiënten alsmede de kosteneffectiviteit van deze therapieën zullen onderzocht gaan worden in de prospectieve DOMESTICO studie. In **hoofdstuk 10** staat het doel en de opzet van deze studie beschreven. Veel artikelen in dit proefschrift hebben gediend als een aanzet of een onderbouwing voor dit multicenter onderzoek. Resultaten uit de prospectieve studie en dit proefschrift zouden kunnen helpen bij de besluitvorming rondom therapiekeuze voor patiënten met nierfalen, met als doel de therapie te vinden die het beste bij de patiënt en diens omgeving past.

Concluderend vond dit proefschrift redenen om aan te nemen dat in de huidige populatie mogelijk meer patiënten geschikt zijn voor thuisdialyse. Daarnaast zouden preventieve interventies op het gebied van PD infecties en catheters de levensduur op PD kunnen verlengen. De gepresenteerde resultaten over kwaliteit van leven kunnen helpen in de besluitvorming rondom therapiekeuze voor alle dialysepatiënten. Wij zijn als onderzoeksgroep van mening dat er meer aandacht mag zijn voor thuisdialyse, zodat thuisdialyse vaker als behandelmogelijkheid overwogen wordt.

Referenties

- 1. Xie Y, Bowe B, Mokdad AH, Xian H, Yan Y, Li T, *et al.* Analysis of the Global Burden of Disease study highlights the global, regional, and national trends of chronic kidney disease epidemiology from 1990 to 2016. Kidney Int. 2018;94(3):567-81.
- Robinson BM, Akizawa T, Jager KJ, Kerr PG, Saran R, Pisoni RL. Factors affecting outcomes in patients reaching end-stage kidney disease worldwide: differences in access to renal replacement therapy, modality use, and haemodialysis practices. Lancet. 2016;388(10041):294-306.
- Li PK-T, Chow KM, Van de Luijtgaarden MWM, Johnson DW, Jager KJ, Mehrotra R, et al. Changes in the worldwide epidemiology of peritoneal dialysis. Nature Reviews Nephrology. 2017;13(2):90-103.
- 4. van de Luijtgaarden MW, Jager KJ, Segelmark M, Pascual J, Collart F, Hemke AC, *et al.* Trends in dialysis modality choice and related patient survival in the ERA-EDTA Registry over a 20-year period. Nephrol Dial Transplant. 2016;31(1):120-8.
- 5. Mohnen SM, van Oosten MJM, Los J, Leegte MJH, Jager KJ, Hemmelder MH, *et al.* Healthcare costs of patients on different renal replacement modalities - Analysis of Dutch health insurance claims data. PLoS One. 2019;14(8):e0220800.
- Klarenbach S, Tonelli M, Pauly R, Walsh M, Culleton B, So H, *et al.* Economic evaluation of frequent home nocturnal hemodialysis based on a randomized controlled trial. J Am Soc Nephrol. 2014;25(3):587-94.
- Couillerot-Peyrondet AL, Sambuc C, Sainsaulieu Y, Couchoud C, Bongiovanni-Delaroziere I. A comprehensive approach to assess the costs of renal replacement therapy for end-stage renal disease in France: the importance of age, diabetes status, and clinical events. Eur J Health Econ. 2017;18(4):459-69.
- 8. Klarenbach SW, Tonelli M, Chui B, Manns BJ. Economic evaluation of dialysis therapies. Nat Rev Nephrol. 2014;10(11):644-52.
- Li B, Cairns JA, Fotheringham J, Tomson CR, Forsythe JL, Watson C, *et al.* Understanding cost of care for patients on renal replacement therapy: looking beyond fixed tariffs. Nephrol Dial Transplant. 2015;30(10):1726-34.
- 10. Hoekstra T, Dekker FW, Cransberg K, Bos WJ, van Buren M, Hemmelder MH. RENINE annual report 2018. Available at: https://www.nefrovisie.nl/jaarrapportages/ Accessed: 04-01-2021. 2018.
- 11. Mendelssohn DC, Mujais SK, Soroka SD, Brouillette J, Takano T, Barre PE, *et al*. A prospective evaluation of renal replacement therapy modality eligibility. Nephrol Dial Transplant. 2009;24(2):555-61.
- 12. Mehrotra R, Marsh D, Vonesh E, Peters V, Nissenson A. Patient education and access of ESRD patients to renal replacement therapies beyond in-center hemodialysis. Kidney International. 2005;68(1):378-90.
- 13. Oliver MJ, Quinn RR, Richardson EP, Kiss AJ, Lamping DL, Manns BJ. Home care assistance and the utilization of peritoneal dialysis. Kidney Int. 2007;71(7):673-8.
- Shukla AM, Easom A, Singh M, Pandey R, Rotaru D, Wen X, et al. Effects of a Comprehensive Predialysis Education Program on the Home Dialysis Therapies: A Retrospective Cohort Study. Perit Dial Int. 2017;37(5):542-7.

- 15. Brown EA, Johansson L. Dialysis options for end-stage renal disease in older people. Nephron Clin Pract. 2011;119 Suppl 1:c10-3.
- 16. van de Luijtgaarden MW, Noordzij M, Stel VS, Ravani P, Jarraya F, Collart F, *et al.* Effects of comorbid and demographic factors on dialysis modality choice and related patient survival in Europe. Nephrol Dial Transplant. 2011;26(9):2940-7.
- 17. Couchoud C, Savoye E, Frimat L, Ryckelynck JP, Chalem Y, Verger C. Variability in case mix and peritoneal dialysis selection in fifty-nine French districts. Perit Dial Int. 2008;28(5):509-17.
- 18. Ethier I, Cho Y, Hawley C, Pascoe EM, Roberts MA, Semple D, *et al.* Effect of patientand center-level characteristics on uptake of home dialysis in Australia and New Zealand: a multicenter registry analysis. Nephrol Dial Transplant. 2020.
- 19. Walker RC, Hanson CS, Palmer SC, Howard K, Morton RL, Marshall MR, *et al.* Patient and caregiver perspectives on home hemodialysis: a systematic review. Am J Kidney Dis. 2015;65(3):451-63.
- 20. System USRD. United States Renal Data System. 2019 USRDS Annual Data Report: Epidemiology of kidney disease in the United States. National Institutes of Health, National Institute of Diabetes and Digestive and Kidney Diseases, Bethesda, MD, 2019 [Internet]. Available from: https://www.usrds.org/annual-data-report/ Accessed: 7-8-2021. 2019.
- 21. Cho Y, See EJ, Htay H, Hawley CM, Johnson DW. Early Peritoneal Dialysis Technique Failure: Review. Perit Dial Int. 2018;38(5):319-27.
- 22. Jaar BG, Plantinga LC, Crews DC, Fink NE, Hebah N, Coresh J, *et al.* Timing, causes, predictors and prognosis of switching from peritoneal dialysis to hemodialysis: a prospective study. BMC Nephrology. 2009;10(1):3.
- 23. Guo A, Mujais S. Patient and technique survival on peritoneal dialysis in the United States: evaluation in large incident cohorts. Kidney Int Suppl. 2003(88):S3-12.
- 24. Kolesnyk I, Dekker FW, Boeschoten EW, Krediet RT. Time-dependent reasons for peritoneal dialysis technique failure and mortality. Perit Dial Int. 2010;30(2):170-7.
- Manera KE, Johnson DW, Craig JC, Shen JI, Gutman T, Cho Y, *et al.* Establishing a Core Outcome Set for Peritoneal Dialysis: Report of the SONG-PD (Standardized Outcomes in Nephrology-Peritoneal Dialysis) Consensus Workshop. Am J Kidney Dis. 2020;75(3):404-12.
- 26. Evangelidis N, Tong A, Manns B, Hemmelgarn B, Wheeler DC, Tugwell P, *et al.* Developing a Set of Core Outcomes for Trials in Hemodialysis: An International Delphi Survey. American Journal of Kidney Diseases. 2017;70(4):464-75.
- Verberne WR, Das-Gupta Z, Allegretti AS, Bart HAJ, van Biesen W, García-García G, et al. Development of an International Standard Set of Value-Based Outcome Measures for Patients With Chronic Kidney Disease: A Report of the International Consortium for Health Outcomes Measurement (ICHOM) CKD Working Group. Am J Kidney Dis. 2019;73(3):372-84.
- 28. Manns B, Hemmelgarn B, Lillie E, Dip SC, Cyr A, Gladish M, *et al.* Setting research priorities for patients on or nearing dialysis. Clin J Am Soc Nephrol. 2014;9(10):1813-21.
- 29. Merkus MP, Jager KJ, Dekker FW, De Haan RJ, Boeschoten EW, Krediet RT. Quality of life over time in dialysis: the Netherlands Cooperative Study on the Adequacy of Dialysis. NECOSAD Study Group. Kidney Int. 1999;56(2):720-8.
- 30. Painter P, Krasnoff JB, Kuskowski M, Frassetto L, Johansen K. Effects of modality change on health-related quality of life. Hemodial Int. 2012;16(3):377-86.

Appendix A

Treatment distribution ir	1 percentage of incident	patients (treatmer	nt modality at
day 91)			

	ERA- EDTA 2003	ERA- EDTA 2012	ERA- EDTA 2018	ERA- EDTA 2003	ERA- EDTA 2012	ERA- EDTA 2018
	HD/HDF	HD/HDF	HD/HDF	PD	PD	PD
Austria	88	83	83	10	12	11
Belgium, Dutch- speaking	86	88	87	14	10	11
Denmark	64	65	60	32	28	30
Finland	69	72	71	29	27	25
Greece	89	92	93	10	7	7
Iceland	81	47	48	14	32	33
Norway	68	64	59	19	20	29
Spain, Catalonia	90	73	77	6	17	14
Spain, Valencian region	85	79	75	15	18	20
Sweden	65	60	59	31	32	33
The Netherlands	67	71	68	28	16	17
United Kingdom, England/Wales	66	70	72	30	21	20
United Kingdom, Scotland	75	78	72	22	14	19

Adapted from ERA-EDTA annual report: 2003, Table A.3.9(1); 2012, Table A.3.9 (2); 2018, Table B.3.10 (3)

	ERA- EDTA 2003	ERA- EDTA 2012	ERA- EDTA 2018	ERA- EDTA 2003	ERA- EDTA 2012	ERA- EDTA 2018
	HD/HDF	HD/HDF	HD/HDF	PD	PD	PD
Austria	47	45	45	4	5	4
Belgium, Dutch- speaking	52	53	52	6	5	4
Denmark	45	42	38	16	10	10
Finland	32	33	33	9	7	7
Greece	74	74	77	8	5	5
Iceland	34	26	22	16	10	7
Norway	22	24	25	4	4	7
Spain, Catalonia	51	41	38	3	4	4
Spain, Valencian region	57	54	50	5	5	6
Sweden	36	34	33	11	9	8
The Netherlands	34	35	31	13	6	5
United Kingdom, England/Wales	39	42	39	15	7	6
United Kingdom, Scotland	40	42	37	11	5	4

Treatment distribution in percentage prevalent patients (including kidney transplants)

Adapted from ERA-EDTA annual report: 2003, Table A.4.9 (1); 2012, Table A.4.9 (2); 2018, Table B.4.10 (3)

1. ERA-EDTA Registry: ERA-EDTA Registry 2003 Annual Report. Academic Medical Center, Amsterdam, The Netherlands, May 2005.

2. ERA-EDTA Registry: ERA-EDTA Registry Annual Report 2012. Academic Medical Center, Department of Medical Informatics, Amsterdam, The Netherlands, 2014.

3. ERA-EDTA Registry: ERA-EDTA Registry Annual Report 2018. Amsterdam UMC, location AMC, Department of Medical Informatics, Amsterdam, the Netherlands, 2020. 2018.

Appendix B

Short-Form 12 (generic)

This survey asks for your views about your health. This information will help keep track of how you feel and how well you are able to do your usual activities.

Answer each question by choosing just one answer. If you are unsure how to answer a question, please give the best answer you can.

1. In general, would you say your health is

- O Excellent
- O Very good
- O Good
- O Fair
- O Poor

The following questions are about activities you might do during a typical day. Does your health now limit you in these activities? If so, how much?

	Yes, limited a lot	Yes, limited a little	No, not limited at all
2. Moderate activities such as moving a table, pushing a vacuum cleaner, bowling, or playing golf.	Ο	Ο	Ο
3. Climbing several flights of stairs.	0	0	0

During the <u>past 4 weeks</u>, have you had any of the following problems with your work or other regular daily activities <u>as a result of your physical health</u>?

	Yes	No
4. Accomplished less than you would like.	Ο	0
5. Were limited in the kind of work or other activities.	Ο	0
During the past 4 weeks, have you had any of the following problems with your work or other regular daily activities as a result of any emotional problems (such as feeling depressed or anxious)?

	Yes	No
6. Accomplished less than you would like.	Ο	0
7. Did work or activities less carefully than usual.	Ο	0

8. During the past 4 weeks, how much did pain interfere with your normal work (including work outside the home and housework)?

- O Not at all
- O A little bit
- O Moderately
- O Quite a bit
- O Extremely

These questions are about how you have been feeling during the past 4 weeks<. For each question, please give the one answer that comes closest to the way you have been feeling.

How much of the time during the past 4 weeks...

	All of the time	Most of the time	A good bite of the time	Some of the time	A little of the time	None of the time
9. Have you felt calm & peaceful?	0	0	0	0	Ο	0
10. Did you have a lot of energy?	0	0	0	0	Ο	0
11. Have you felt down-hearted and blue?	0	0	0	0	0	0

12. During the past 4 weeks, how much of the time has your physical health or emotional problems interfered with your social activities (like visiting friends, relatives, etc.)?

O All of the time

O Most of the time

O Some of the time

O A little of the time

O None of the time

Dialysis Symptom Index (kidney disease specific HRQoL questionnaire)

Following is a list of physical and emotional symptoms that people on dialysis may have. For each symptom, please indicate if you had the symptom <u>during the past week</u> by clicking 'yes' or 'no'. If 'yes', please indicate how much that symptom bothered you.

During the past week: Did you expe this symptom?	rience	If ye	es, how much did it bother you?			
		Not at all	A little bit	Some- what	Quite a bit	Very much
Constipation	No/Yes					
Nausea	No/Yes					
Vomiting	No/Yes					
Diarrhea	No/Yes					
Decreased appetite	No/Yes					
Muscle cramps	No/Yes					
Swelling in legs	No/Yes					
Lightheadedness or dizziness	No/Yes					
Shortness of breath	No/Yes					
Restless legs or difficulty keeping legs still	No/Yes					
Numbness or tingling in feet	No/Yes					
Feeling tired or lack of energy	No/Yes					
Cough	No/Yes					
Dry mouth	No/Yes					
Bone or joint pain	No/Yes					
Chest pain	No/Yes					

During the past week: Did you experience this symptom?		If yes, how much did it bother you?				
		Not at all	A little bit	Some- what	Quite a bit	Very much
Headache	No/Yes					
Muscle soreness	No/Yes					
Difficulty concentrating	No/Yes					
Dry skin	No/Yes					
Itching	No/Yes					
Worrying	No/Yes					
Feeling nervous	No/Yes					
Trouble falling asleep	No/Yes					
Trouble staying asleep	No/Yes					
Feeling irritable	No/Yes					
Feeling sad	No/Yes					
Feeling anxious	No/Yes					
Decreased interest in sex	No/Yes					
Difficulty becoming sexually aroused	No/Yes					

Under each heading, please tick the ONE box that best describes your health TODAY.				
MOBILITY				
I have no problems in walking about				
I have slight problems in walking about				
I have moderate problems in walking about				
I have severe problems in walking about				
I am unable to walk about				
SELF-CARE				
I have no problems washing or dressing myself				
I have slight problems washing or dressing myself				
I have moderate problems washing or dressing myself				
I have severe problems washing or dressing myself				
I am unable to wash or dress myself				
USUAL ACTIVITIES (e.g. work, study, housework, family or leisure	activities)			
I have no problems doing my usual activities				
I have slight problems doing my usual activities				
I have moderate problems doing my usual activities				
I have severe problems doing my usual activities				
I am unable to do my usual activities				
PAIN / DISCOMFORT				
I have no pain or discomfort				
I have slight pain or discomfort				
I have moderate pain or discomfort				
I have severe pain or discomfort				
I have extreme pain or discomfort				
ANXIETY / DEPRESSION				
I am not anxious or depressed				
I am slightly anxious or depressed				
I am moderately anxious or depressed				
I am severely anxious or depressed				
I am extremely anxious or depressed				

EQ-5D-5L (generic HRQoL questionnaire for economic evaluation)

UK (English) © 2009 EuroQol Group EQ-5D™ is a trade mark of the EuroQol Group

Appendices



UK (English) © 2009 EuroQol Group EQ-5D™ is a trade mark of the EuroQol Group

DOMESTICO Study Group members

Steering committee members: AC Abrahams, University Medical Centre Utrecht; BC van Jaarsveld, Amsterdam University Medical Centres (VU University, Amsterdam) and Diapriva Dialysis Centre Amsterdam; FW Dekker, Leiden University Medical Centre; FJ van Ittersum, Amsterdam University Medical Centres (VU University, Amsterdam); H Bart/W Konijn, Dutch Kidney Patients Association (NVN); MH Hemmelder, Maastricht UMC+; MAGJ ten Dam, Nefrovisie and Canisius-Wilhelmina Hospital Nijmegen.

Junior investigators: A van Eck van der Sluijs, *University Medical Centre Utrecht*; AA Bonenkamp, *Amsterdam University Medical Centres (VU University, Amsterdam)*; B van Lieshout, *Amsterdam University Medical Centres (VU University, Amsterdam)*; S Vonk, *University Medical Centre Utrecht*; MC Verhaar, *University Medical Centre Utrecht*;

DOMESTICO retrospective study committee members: : FTJ Boereboom, Dianet Utrecht; FW Dekker, Leiden University Medical Centre; CWH de Fijter, OLVG Amsterdam; DG Struijk, Dianet Amsterdam; YM Vermeeren, Gelre Hospitals Apeldoorn.

DOMESTICO prospective 'Quality of life and clinical outcomes' committee: Frans van Ittersum, Amsterdam University Medical Centers (VU University, Amsterdam); Lars Penne, Northwest Clinics Alkmaar; Aegida Neradova, Dianet Dialysis Center Amsterdam and Amsterdam University Medical Centers (University of Amsterdam); Dick Struijk, Dianet Amsterdam and Amsterdam University Medical Centers (University of Amsterdam); Akin Özyilmaz, University Medical Center Groningen and Dialysis Center Groningen.

DOMESTICO prospective 'Costs' committee: MM Versteegh, Institute for Medical Technology Assessment; L Hakkaart-van Roijen, Institute of Health Policy & Management and Institute for Medical Technology Assessment; GA de Wit, Julius Center for Health Sciences and Primary Care; FT Boereboom, Dianet Utrecht and Diakonessenhuis; MH Hemmelder, Nefrovisie and Medical Center Leeuwarden/ Maastricht UMC+; TA Kanters, Institute for Medical Technology Assessment; G de Graaf, Institute for Medical Technology Assessment.

Appendices

DOMESTICO 'Good practices and shared decision making' committee: P.W.G. du Buf-Vereijken, Amphia Hospital Breda; K Prantl, Dutch Kidney Patients Association (NVN); MH Hemmelder, Nefrovisie and Medical Center Leeuwarden; JA Bijlsma, Dianet Dialysis Center Amsterdam and Amsterdam University Medical Centers (University of Amsterdam); EC Hagen, Niercentrum Midden Nederland Amersfoort; Anton Luik, VieCuri Medical Center Venlo;

Investigators of the retrospective DOMESTICO study: : MR Korte, Albert Schweitzer Hospital Dordrecht; TT Cnossen, Amphia Hospital Breda; BC van Jaarsveld, Amsterdam University Medical Centres (VU University, Amsterdam) and Diapriva Dialysis Centre Amsterdam; HP Krepel, Bravis Hospital Roosendaal; MAGJ ten Dam, Canisius-Wilhelmina Hospital Nijmegen; CJAM Konings, Catharina Hospital Eindhoven; CJ Doorenbos, Deventer Hospital; A Lips, Dialysiscentre Beverwijk; A Özyilmaz, Dialysis Centre Groningen; DG Struijk, Dianet Amsterdam; FTJ Boereboom, Dianet Utrecht; S van Esch, Elisabeth-TweeSteden Hospital Tilburg; GF van Breda, Elyse Clinics; EJ Hoorn and D Severs, Erasmus Medical Centre Rotterdam; AH Boonstra, Flevohospital Almere; RW Nette, Franciscus Gasthuis & Vlietland Rotterdam; YM Vermeeren, Gelre Hospitals Apeldoorn; HD Thang and NH Hommes, Haaglanden Medical Centre The Hague; M van Buren, HagaHospital The Hague; JM Hofstra, Hospital Gelderse Vallei Ede; SHA Diepeveen, Isala Zwolle; S Boorsma, Laurentius Hospital Roermond; JI Rotmans, Leiden University Medical Centre; F van der Sande and EJR Litjens, Maastricht UMC+; HS Brink and R Wijering, Medical Spectrum Twente Enschede; EC Hagen, Niercentrum Midden Nederland Amersfoort; EL Penne, Northwest Clinics Alkmaar; CWH de Fijter and HFH Brulez, OLVG Amsterdam; HW van Hamersvelt, Radboudumc Nijmegen; SJ Huisman, Reinier de Graaf Gasthuis Delft; CE Douma, Spaarne Gasthuis Hoofddorp; AC Abrahams, University Medical Centre Utrecht; AJ Luik, VieCuri Medical Centre Venlo; RJL Klaassen, Zaans Medical Centre Zaandam; AG Weenink, ZorgSaam Hospital Terneuzen; MME Krekels, Zuyderland Sittard.

Investigators of the prospective DOMESTICO study: PB Leurs, Admiraal de Ruyter Hospital Goes; MR Korte, Albert Schweitzer Hospital Dordrecht; AM Schrander, Alrijne Hospital; TT Cnossen, Amphia Hospital Breda; BC van Jaarsveld, Amsterdam University Medical Centers (VU University, Amsterdam) and Diapriva Dialysis Center Amsterdam; A de Vriese, AZ St-Jan Brugge (Belgium); J Lips, Bernhoven Uden; HP Krepel, Bravis Hospital Roosendaal; MAGJ ten Dam, Canisius-Wilhelmina

Hospital Nijmegen; CJAM Konings, Catharina Hospital Eindhoven; A van Eck van der Sluijs, Deventer Hospital; A Lips, Dialysis center Beverwijk; A Özyilmaz, Dialysis Center Groningen; A Neradova, Dianet Amsterdam; FTJ Boereboom, Dianet Utrecht; S van Esch, Elisabeth-TweeSteden Hospital Tilburg; CR Susanto, Elkerliek Hospital; GF van Breda, Elyse Clinics; EJ Hoorn and D Severs, Erasmus Medical Center Rotterdam; AH Boonstra, Flevohospital Almere; RW Nette and MAM Verhoeven, Franciscus Gasthuis & Vlietland Rotterdam; YM Vermeeren, Gelre Hospitals Apeldoorn; DHT Ijpelaar, Groene Hart Hospital Gouda; NH Hommes, Haaglanden Medical Center The Hague; M van Buren, HagaHospital The Hague; JM Hofstra, Hospital Gelderse Vallei Ede; SHA Diepeveen, Isala Zwolle; EK Hoogeveen, Jeroen Bosch Hospital 's-Hertogenbosch; T Cornelis, Jessa Hospital Hasselt (Belgium); S Boorsma, Laurentius Hospital Roermond; II Rotmans, Leiden University Medical Center; AM van Alphen, Maasstad Hospital Rotterdam; EJR Litjens and B Zomer, Maastricht UMC+; WMT Janssen, Martini Hospital Groningen; A Kuijper and CH Beerenhout, Máxima Medical Center Veldhoven; J Broekroelofs and L Bierma, Medical Center Leeuwarden; HS Brink and RMJ Wijering, Medical Spectrum Twente Enschede; RJ Bosma, Niercentrum Midden Nederland Amersfoort; EL Penne, Northwest Clinics Alkmaar; CWH de Fijter and HFH Brulez, OLVG Amsterdam; HW van Hamersvelt, Radboudumc Nijmegen; SJ Huisman, Reinier de Graaf Gasthuis Delft; MP Kooistra and JC Verhave, Rijnstate Arnhem; G van Kempen, Saxenburgh Group; HHTI Klein, Slingeland Hospital Doetinchem; CE Douma, Spaarne Gasthuis Hoofddorp; WJW Bos, St. Antonius Hospital Nieuwegein; ID Snoep, Tergooi Hilversum; I Mulder, Treant Zorggroep Emmen; CFM Franssen, University Medical Center Groningen; AC Abrahams, University Medical Center Utrecht; K Francois, UZ Brussel (Belgium); AJ Luik, VieCuri Medical Center Venlo; RJL Klaassen and A van Tellingen, Zaans Medical Center Zaandam; MMG Dekker, Ziekenhuisgroep Twente; AG Weenink, ZorgSaam Hospital Terneuzen; MME Krekels, Zuyderland Sittard.

List of publications

This thesis

- Bonenkamp AA, Hoekstra T, Hemmelder MH, van Eck van der Sluijs A, Abrahams AC, van Ittersum FJ, van Jaarsveld BC. Trends in home dialysis use differ among age categories in past two decades: a Dutch registry study. *European Journal of Clinical Investigation*. 2021 Jul 22:e13656. Epub ahead of print.
- Bonenkamp AA, Vonk S, Abrahams AC, Vermeeren YM, van Eck van der Sluijs A, Hoekstra T, van Ittersum FJ, van Jaarsveld BC on behalf of the DOMESTICO study group, Comorbidity is not associated with dialysis modality choice in patients with end-stage kidney disease. *Nephrology*. 2022. *in print*
- Bonenkamp AA, Reijnders TDY, van Eck van der Sluijs A, Hagen EC, Abrahams AC, van Ittersum FJ, van Jaarsveld BC. Key elements in selection of pre-dialysis patients for home dialysis. *Peritoneal Dialysis International*. 2021 Sep;41(5):494-501.
- van Eck van der Sluijs A, Bonenkamp AA, van Wallene VA, Hoekstra T, Lissenberg-Witte BI, Dekker FW, van Ittersum FJ, Verhaar MC, van Jaarsveld BC, Abrahams AC; DOMESTICO study group. Differences in hospitalisation between peritoneal dialysis and haemodialysis patients. *European Journal of Clinical Investigation*. 2022 Feb 7:e13758. Epub ahead of print.
- Bonenkamp AA, van Eck van der Sluijs A, Dekker FW, Struijk DG, de Fijter CW, Vermeeren YM, van Ittersum FJ, Verhaar MC, van Jaarsveld BC, Abrahams AC. Technique failure in peritoneal dialysis: Modifiable causes and patient-specific risk factors. *Peritoneal Dialysis International.* 2022. *In print*
- Bonenkamp AA, van Eck van der Sluijs A, Hoekstra T, Verhaar MC, van Ittersum FJ, Abrahams AC, van Jaarsveld BC. Health-Related Quality of Life in Home Dialysis Patients Compared to In-Center Hemodialysis Patients: A Systematic Review and Meta-analysis. *Kidney Medicine*. 2020 Feb 11;2(2):139-154.
- Jansz TT, Bonenkamp AA, Boereboom FTJ, van Reekum FE, Verhaar MC, van Jaarsveld BC. Health-related quality of life compared between kidney transplantation and nocturnal hemodialysis. *PLoS One. 2018 Sep* 20;13(9):e0204405.
- Bonenkamp AA, Druiventak TA, van Eck van der Sluijs A, van Ittersum FJ, van Jaarsveld BC, Abrahams AC; DOMESTICO study group. The Impact of COVID-19 on the mental health of dialysis patients. *Journal of Nephrology*. 2021 Apr;34(2):337-344.

 van Eck van der Sluijs A, Bonenkamp AA, Dekker FW, Abrahams AC, van Jaarsveld BC; DOMESTICO study group. Dutch nOcturnal and hoME dialysis Study To Improve Clinical Outcomes (DOMESTICO): rationale and design. BMC Nephrology 2019 Sep 18;20(1):361

Other publications

- Bonenkamp AA, van Gelder MK, Abrahams AC, Boereboom FTJ, Cornelis T, Luik AJ, Özyilmaz A, van der Sande FM, van Eck van der Sluijs A, Gerritsen KGF, van Jaarsveld BC. Home haemodialysis in the Netherlands: State of the art. *Netherlands Journal of Medicine*. 2018 May;76(4):144-157
- Colombijn JMT, Bonenkamp AA, van Eck van der Sluijs A, Bijlsma JA, Boonstra AH, Özyilmaz A, Abrahams AC, van Jaarsveld BC; DOMESTICO study group. Impact of Polypharmacy on Health-Related Quality of Life in Dialysis Patients. *American Journal of Nephrology.* 2021;52(9):735-744
- van Eck van der Sluijs A, Vonk S, van Jaarsveld BC, **Bonenkamp AA**, Abrahams AC. Good practices for dialysis education, treatment, and eHealth: A scoping review. *PLoS One*. 2021 Aug 11;16(8):e0255734.
- van Eck van der Sluijs A, van Jaarsveld BC, Allen J, Altabas K, Béchade C, Bonenkamp AA, Burkhalter F, Clause AL, Corbett RW, Dekker FW, Eden G, François K, Gudmundsdottir H, Lundström UH, de Laforcade L, Lambie M, Martin H, Pajek J, Panuccio V, Ros-Ruiz S, Steubl D, Vega A, Wojtaszek E, Davies SJ, Van Biesen W, Abrahams AC. Assisted peritoneal dialysis across Europe: Practice variation and factors associated with availability. *Peritoneal Dialysis International.* 2021 Nov;41(6):533-541.

